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Preface

This book covers the fundamental mechanics of fluids as they are treated at
the senior level and at the introductory graduate level. Many excellent books
exist that treat special areas of fluid mechanics such as ideal-fluid flow or
boundary-layer theory. However, there are very few books at this level that
sacrifice an in-depth study of one of these special areas of fluid mechanics for
a briefer treatment of a broader area of the fundamentals of fluid mechanics.
This situation exists despite the fact that many institutions of higher learn-
ing offer a broad, fundamental course to a wide spectrum of their students
before offering more advanced specialized courses to those who are special-
izing in fluid mechanics. This book is intended to remedy this situation.

The book is divided into five parts. Part I, “Governing Equations,” deals
with the derivation of the basic conservation laws, flow kinematics, and
some basic theorems of fluid mechanics. Part II is titled “Ideal-Fluid Flow,”
and it covers two-dimensional potential flows, three-dimensional potential
flows, and surface waves. Part III, “Viscous Flows of Incompressible Fluids,”
contains chapters on exact solutions, low-Reynolds-number approximations,
boundary-layer theory, and buoyancy-driven flows. Part IV of the book is
titled “Compressible Flow of Inviscid Fluids,” and this part contains chapters
that deal with shock waves, one-dimensional flows, and multidimensional
flows. Finally, Part V, which is titled “Methods of Mathematical Analysis,”
presents a summary of some of the commonly used methods of analysis as
used in this book as well as many others. Appendixes are also included that
summarize vectors, tensors, the governing equations in the common coordi-
nate systems, Fourier series, and thermodynamics.

The treatment of the material is such as to emphasize the phenomena asso-
ciated with the various properties of fluids while providing techniques for
solving specific classes of fluid-flow problems. The treatment is not geared to
any one discipline, and it may readily be studied by physicists and chemists
as well as by engineers from various branches. Since the book is intended
for teaching purposes, phrases such as “it can be shown that” and similar
clichés that cause many hours of effort for many students have been avoided.
In order to aid the teaching process, several problems are included at the end
of each of the 14 chapters. These problems serve to illustrate points brought
out in the text and to extend the material covered in the text.

Most of the material contained in this book can be covered in about 50
lecture hours. For more extensive courses, the material contained here may
be completely covered and even augmented. Parts II, 11, and IV are essen-
tially independent so that they may be interchanged or any one or more of
them may be omitted. This permits a high degree of teaching flexibility and
allows the instructor to include or substitute material that is not covered in

xxi



xxii Preface

the text. Such additional material may include free convection, density strati-
fication, hydrodynamic stability, and turbulence with applications to pollu-
tion, meteorology, etc. These topics are not included here, not because they
do not involve fundamentals, but rather because I set up a priority of what I
consider to be the basic fundamentals.

For the fourth edition, Chapter 14, which is titled “Some Useful Methods
of Analysis,” has been added. The problems in all chapters have also been
reviewed, and some of them have been revised in order to clarify and/or
extend the questions. Some new problems have also been included, bringing
the total number to 140.

Many people are to be thanked for their direct or indirect contributions
to this text. I had the privilege of taking lectures from F. E. Marble, C. B.
Millikan, and P. G. Saffman, and some of the style and methods of these
great scholars are probably evident in the following pages. Thanks are also
due to the many instructors and students who have used the book as a text
and who have pointed out errors and/or ambiguities in the material in the
earlier versions of this text.

I. G. Currie



Part 1

Governing Equations

In this first part of the book, a sufficient set of equations will be derived,
based on physical laws and postulates, governing the dependent variables
of a fluid that is moving. The dependent variables are the fluid-velocity com-
ponents, pressure, density, temperature, and internal energy or some simi-
lar set of variables. The equations governing these variables will be derived
from the principles of mass, momentum, and energy conservation and from
equations of state. Having established a sufficient set of governing equations,
some purely kinematical aspects of fluid flow are discussed, at which time
the concept of vorticity is introduced. The final section of this part of the book
introduces certain relationships that can be derived from the governing equa-
tions under certain simplifying conditions. These relationships may be used
in conjunction with the basic governing equations or as alternatives to them.

Taken as a whole, this part of the book establishes the mathematical equa-
tions that result from invoking certain physical laws postulated to be valid
for a moving fluid. These equations may assume different forms, depending
upon which variables are chosen and upon which simplifying assumptions
are made. The specific chapters contained in Part I are as follows:

Chapter 1: Basic Conservation Laws
Chapter 2: Flow Kinematics
Chapter 3: Special Forms of the Governing Equations

Parts II, III, and IV of the book are devoted to solving the governing
equations established in Part I for different categories of fluid flow, thereby
explaining quantitatively some of the phenomena that are observed in a flow-
ing fluid. In addition, they establish the methodologies that are employed in
the various branches of fluid mechanics.






1

Basic Conservation Laws

The essential purpose of this chapter is to derive the set of equations that
results from invoking the physical laws of conservation of mass, momen-
tum, and energy. In order to realize this objective, it is necessary to discuss
certain preliminary topics. The first topic of discussion is the two basic ways
in which the conservation equations may be derived: the statistical method
and the continuum method. Having selected the basic method to be used in
deriving the equations, one is then faced with the choice of reference frame
to be employed, Eulerian or Lagrangian. Next, a general theorem, called
Reynolds’ transport theorem, is derived, since this theorem relates deriva-
tives in the Lagrangian framework to derivatives in the Eulerian framework.

Having established the basic method to be employed and the tools to be
used, the basic conservation laws are then derived. The conservation of mass
yields the so-called continuity equation. The conservation of momentum
leads ultimately to the Navier—Stokes equations, while the conservation of
thermal energy leads to the energy equation. The derivation is followed by a
discussion of the set of equations so obtained, and finally a summary of the
basic conservation laws is given.

1.1 Statistical and Continuum Methods

There are basically two ways of deriving the equations that govern the
motion of a fluid. One of these methods approaches the question from the
molecular point of view. That is, this method treats the fluid as consisting
of molecules whose motion is governed by the laws of dynamics. The mac-
roscopic phenomena are assumed to arise from the molecular motion of the
molecules, and the theory attempts to predict the macroscopic behavior of
the fluid from the laws of mechanics and probability theory. For a fluid that
is in a state not too far removed from equilibrium, this approach yields the
equations of mass, momentum, and energy conservation. The molecular
approach also yields expressions for the transport coefficients, such as the
coefficient of viscosity and the thermal conductivity, in terms of molecular
quantities such as the forces acting between molecules or molecular diam-
eters. The theory is well developed for light gases, but it is incomplete for
polyatomic gas molecules and for liquids.
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The alternative method used to derive the equations governing the motion
of a fluid uses the continuum concept. In the continuum approach, individ-
ual molecules are ignored, and it is assumed that the fluid consists of con-
tinuous matter. At each point of this continuous fluid, there is supposed to
be a unique value of the velocity, pressure, density, and other so-called field
variables. The continuous matter is then required to obey the conservation
laws of mass, momentum, and energy, which give rise to a set of differen-
tial equations governing the field variables. The solution to these differential
equations then defines the variation of each field variable with space and
time, which corresponds to the mean value of the molecular magnitude of
that field variable at each corresponding position and time.

The statistical method is rather elegant, and it may be used to treat gas
flows in situations where the continuum concept is no longer valid. However,
as was mentioned before, the theory is incomplete for dense gases and for
liquids. The continuum approach requires that the mean free path of the
molecules be very small compared with the smallest physical-length scale of
the flow field (such as the diameter of a cylinder or other body about which
the fluid is flowing). Only in this way can meaningful averages over the
molecules at a “point” be made and the molecular structure of the fluid be
ignored. However, if this condition is satisfied, there is no distinction among
light gases, dense gases, or even liquids—the results apply equally to all.
Since the vast majority of phenomena encountered in fluid mechanics fall
well within the continuum domain and may involve liquids as well as gases,
the continuum method will be used in this book. With this background, the
meaning and validity of the continuum concept will now be explored in
some detail. The field variables such as the density p and the velocity vector
u will in general be functions of the spatial coordinates and time. In symbolic
form, this is written as p = p(x, t) and u = u(x, t), where x is the position vector
whose Cartesian coordinates are x, y, and z. At any particular point in space,
these continuum variables are defined in terms of the properties of the vari-
ous molecules that occupy a small volume in the neighborhood of that point.

Consider a small volume of fluid AV containing a large number of mol-
ecules. Let Am and v be the mass and velocity, respectively, of any individual
molecule contained within the volume AV, as indicated in Figure 1.1. The
density and the velocity at a point in the continuum are then defined by the

following limits:
S

Am
AV—¢ AV

E VAm
u=lim

AV—e Am
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AV

FIGURE 1.1
Individual molecule in small volume AV having mass Am and velocity v.

where ¢ is a volume that is sufficiently small that €'/3 is small compared with
the smallest significant length scale in the flow field but is sufficiently large
that it contains a large number of molecules. The summations in the above
expressions are taken over all the molecules contained within the volume
AV. The other field variables may be defined in terms of the molecular prop-
erties in an analogous way.

A sufficient condition, though not a necessary condition, for the contin-
uum approach to be valid is

1
—xex]’®
n

where 7 is the number of molecules per unit volume, and L is the smallest
significant length scale in the flow field, which is usually called the macro-
scopic length scale. The characteristic microscopic length scale is the mean free
path between collisions of the molecules. Then, the above condition states
that the continuum concept will certainly be valid if some volume ¢ that is
much larger than the volume occupied by a single molecule of the fluid but
much smaller than the cube of the smallest macroscopic length scale (such as
cylinder diameter) can be found. Since a cube of gas, at normal temperature
and pressure, whose side is 2 um contains about 2 x 108 molecules and the
corresponding figure for a liquid is about 2 x 10" molecules, the continuum
condition is readily met in the vast majority of flow situations encountered
in physics and engineering. It may be expected to break down in situations
where the smallest macroscopic length scale approaches microscopic dimen-
sions, such as in the structure of a shock wave, and where the microscopic
length scale approaches macroscopic dimensions, and such as when a rocket
passes through the edge of the atmosphere.
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1.2 Eulerian and Lagrangian Coordinates

Having selected the continuum approach as the method that will be used
to derive the basic conservation laws, one is next faced with a choice of ref-
erence frames in which to formulate the conservation laws. There are two
basic coordinate systems that may be employed, these being Eulerian and
Lagrangian coordinates.

In the Eulerian framework, the independent variables are the spatial coor-
dinates x, y, and z and time ¢t. This is the familiar framework in which most
problems are solved. In order to derive the basic conservation equations in
this framework, attention is focused on the fluid that passes through a con-
trol volume that is fixed in space. The fluid inside the control volume at any
instant in time will consist of different fluid particles from that which was
there at some previous instant in time. If the principles of conservation of
mass, momentum, and energy are applied to the fluid passing through the
control volume, the basic conservation equations are obtained in Eulerian
coordinates.

In the Lagrangian approach, attention is fixed on a particular mass of
fluid as it flows. Suppose we could color a small portion of the fluid without
changing its density. Then, in the Lagrangian framework, we follow this
colored portion as it flows and changes its shape, but we always consider
the same particles of fluid. The principles of mass, momentum, and energy
conservation are then applied to this particular element of fluid as it flows,
resulting in a set of conservation equations in Lagrangian coordinates. In
this reference frame, x, y, z, and t are no longer independent variables since
if it is known that our colored portion of fluid passed through the coor-
dinates x,, y,, and z, at some time f,, then its position at some later time
may be calculated if the velocity components u, v, and w are known. That
is, as soon as a time interval (f — t;) is specified, the velocity components
uniquely determine the coordinate changes (x — x,), (v — ), and (z — z;) so
that x, y, z, and t are no longer independent. The independent variables in
the Lagrangian system are x,, v, zo, and t, where x,, y,, and z, are the coor-
dinates that a specified fluid element passed through at time ¢;. That is, the
coordinates x,, y,, and z, identify which fluid element is being considered,
and the time t identifies its instantaneous location.

The choice of which coordinate system to employ is largely a matter of
taste. It is probably more convincing to apply the conservation laws to a con-
trol volume that always consists of the same fluid particles rather than one
through which different fluid particles pass. This is particularly true when
invoking the law of conservation of energy, which consists of applying the
tirst law of thermodynamics, since the same fluid particles are more read-
ily justified as a thermodynamic system. For this reason, the Lagrangian
coordinate system will be used to derive the basic conservation equations.
Although the Lagrangian system will be used to derive the basic equations,
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the Eulerian system is the preferred one for solving the majority of prob-
lems. In the next section, the relation between the different derivatives will
be established.

1.3 Material Derivative

Let o be any field variable such as the density or temperature of the fluid.
From the Eulerian viewpoint, a may be considered to be a function of the
independent variables x, y, z, and t. However, if a specific fluid element is
observed for a short period of time &t as it flows, its position will change
by amounts &x, 8y, and 6z while its value of a will change by an amount
da. That is, if the fluid element is observed in the Lagrangian framework,
the independent variables are x,, y,, z,, and t, where x,, y,, and z, are initial
coordinates for the fluid element. Thus, x, y, and z are no longer independent
variables but are functions of ¢ as defined by the trajectory of the element.
During the time 8¢, the change in @ may be calculated from differential cal-
culus to be

a—aét + a—a6x+a—a6y + a—Otf)z.
ot ox ay 0z

Equating the preceding change in a to the observed change Sa in the
Lagrangian framework and dividing throughout by 5t gives

do  da dx da  dy o Oz da
b ——

S ot ot ax Ot gy Ot 9z

The left-hand side of this expression represents the total change in « as
observed in the Lagrangian framework during the time &8¢, and in the limit,
it represents the time derivative of a in the Lagrangian system, which will be
denoted by Do/Dt. It may be also noted that in the limit as 8 — 0, the ratio
dx/dt becomes the velocity component in the x direction, namely, u. Similarly,
dy/dt — v and 8z/8t — w as &t — 0, so that in the limit, the expression for the
change in a becomes

Da da oo oo oo
— =t U—+V—+W—.
Dt ot ax oy 0z

In vector form, this equation may be written as follows:
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Do da
—=—+(u )o.
Dt ot
Alternatively, using the Einstein summation convention where repeated
subscripts are summed, the tensor form may be written as

@=aﬁ+ukaﬁ‘ (1.1)
Dt ot *ox,

The term Da/Dt in Equation 1.1 is the so-called material derivative. It rep-
resents the total change in the quantity a as seen by an observer who is fol-
lowing the fluid and is watching a particular mass of the fluid. The entire
right-hand side of Equation 1.1 represents the total change in o expressed in
Eulerian coordinates. The term u,(da/0x,) expresses the fact that in a time-
independent flow field in which the fluid properties depend upon the spatial
coordinates only, there is a change in a due to the fact that a given fluid
element changes its position with time and therefore assumes different
values of a as it flows. The term da/dt is the familiar Eulerian time deriva-
tive and expresses the fact that at any point in space, the fluid properties
may change with time. Then, Equation 1.1 expresses the Lagrangian rate of
change Do/Dt of a for a given fluid element in terms of the Eulerian deriva-
tives da/dt and do/dx,.

1.4 Control Volumes

The concept of a control volume, as required to derive the basic conservation
equations, has been mentioned in connection with both the Lagrangian and
Eulerian approaches. Irrespective of which coordinate system is used, there
are two principal control volumes from which to choose. One of these is a
parallelepiped of sides &x, 8y, and 6z. Each fluid property, such as the veloc-
ity or pressure, is expanded in a Taylor series about the center of the control
volume to give expressions for that property at each face of the control vol-
ume. The conservation principle is then invoked, and when 8x, 6y, and 6z
are permitted to become vanishingly small, the differential equation for that
conservation principle is obtained. Frequently, shortcuts are taken, and the
control volume is taken to have sides of length dx, dy, and dz with only the
first term of the Taylor series being carried out.

The second type of control volume is arbitrary in shape, and each con-
servation principle is applied to an integral over the control volume. For

example, the mass within the control volume is | pdV, where p is the fluid

v
density and the integration is carried out over the entire volume V of the
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fluid contained within the control volume. The result of applying each con-
servation principle will be an integrodifferential equation of the type

f LodV =0
1%

where L is some differential operator and o is some property of the fluid. But
since the control volume V was arbitrarily chosen, the only way this equation
can be satisfied is by setting La = 0, which gives the differential equation of
the conservation law. If the integrand in the above equation was not equal to
zero, it would be possible to redefine the control volume V in such a way that
the integral of La was not equal to zero, contradicting the integrodifferential
equation above.

Each of these two types of control volumes has some merit, and in this
book, each will be used at some point, depending upon which gives the bet-
ter insight into the physics of the situation under discussion. The arbitrary
control volume will be used in the derivation of the basic conservation laws
since it seems to detract less from the principles being imposed. Needless to
say, the results obtained by the two methods are identical.

1.5 Reynolds’ Transport Theorem

The method that has been selected to derive the basic equations from the con-
servation laws is to use the continuum concept and to follow an arbitrarily
shaped control volume in a Lagrangian frame of reference. The combination of
the arbitrary control volume and the Lagrangian coordinate system means that
material derivatives of volume integrals will be encountered. As was mentioned
in the previous section, it is necessary to transform such terms into equivalent
expressions involving volume integrals of Eulerian derivatives. The theorem
that permits such a transformation is called Reynolds’ transport theorem.

Consider a specific mass of fluid and follow it for a short period of time &t
as it flows. Let a be any property of the fluid such as its mass, momentum in
some direction, or energy. Since a specific mass of fluid is being considered
and since x, Yy, Z, and t are the independent variables in the Lagrangian
framework, the quantity a will be a function of ¢ only as the control volume
moves with the fluid. That is, @ = a(t) only and the rate of change of the inte-
gral of a will be defined by the following limit:

D o®dv-tim

DtJve 30 Ot Jv(tsst)

a(t+8t) dV - f a(t) dv
V(t)

where V(t) is the control volume containing the specified mass of fluid
and which may change its size and shape as it flows. The quantity a(t + &f)
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integrated over V(f) will now be subtracted, and then added inside the above
limit.

Do avotim a(t+0t) AV - [ aft+ot) dV

DtJve 30 Ot Jv(t+st) 140

+$ at+5) AV [ a() dV

V(t) V()

The first two integrals inside this limit correspond to holding the inte-
grand fixed and permitting the control volume V to vary, whereas the second
two integrals correspond to holding V fixed and permitting the integrand o
to vary. The latter component of the change is, by definition, the integral of
the familiar Eulerian derivative with respect to time. Then, the expression
for the Lagrangian derivative of the integral of a may be written in the fol-
lowing form:

Droapaveim L alt+5t) dV +f 9% gy
DtJve 8=0 Ot Jv(t+sn-vit) vy ot

The remaining limit, corresponding to the volume V changing while o
remains fixed, may be evaluated from geometric considerations.

Figure 1.2a shows the control volume V(f) that encloses the mass of fluid
being considered both at time t and at time ¢ + 6¢. During this time interval, the

(b)

FIGURE 1.2
(a) Arbitrarily shaped control volume at times f and ¢ + 8¢ and (b) superposition of the control
volumes at these times showing element 8V of the volume change.
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control volume has moved downstream and has changed its size and shape.
The surface that encloses V(t) is denoted by S(f), and at any point on this sur-
face, the velocity may be denoted by u and the unit outward normal by n.

Figure 1.2b shows the control volume V(¢ + &t) superimposed on V(f), and
an element of the difference in volumes is detailed. The perpendicular dis-
tance from any point on the inner surface to the outer surface is u - n 8t, so
that an element of surface area 8S will correspond to an element of volume
change 8V in which 8V = u - n &t 8S. Then, the volume integral inside the
limit in the foregoing equation may be transformed into a surface integral in
which dV is replaced by u - n 5t dS.

D oapydvetim [ a(t+dHu-nds “f 9% 4y
V()

V() 5t—0 S(t) ot

Ja
= Hu-ndS + —dV.
J;(t)a( u J:/u) ot

Having completed the limiting process, the Lagrangian derivative of a vol-
ume integral has been converted into a surface integral and a volume integral
in which the integrands contain only Eulerian derivatives. As was mentioned
in the previous section, it is necessary to obtain each term in the conserva-
tion equations as the volume integral of something. The foregoing form of
Reynolds’ transport theorem may be put in this desired form by converting
the surface integral to a volume integral by use of Gauss’ theorem, which is
formulated in Appendix A. In this way, the surface-integral term becomes

fs(t)cx(t)u-nd5= ) -(au)dV.

Substituting this result into the foregoing expression and combining the
two volume integrals give the preferred form of Reynolds’ transport theorem:

D oo
— dv = — . dv
thv “ fv ot + low

or, in tensor notation,
Bf a dV =f a—Ot+i(ocuk) dv. (1.2)
DtJv v ot ax,

Equation 1.2 relates the Lagrangian derivative of a volume integral of a
given mass to a volume integral in which the integrand has Eulerian deriva-
tives only.
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Having established the method to be used to derive the basic conserva-
tion equations and having established the necessary background material,
it remains to invoke the various conservation principles. The first such prin-
ciple to be treated will be the conservation of mass.

1.6 Conservation of Mass

Consider a specific mass of fluid whose volume V is arbitrarily chosen. If this
given fluid mass is followed as it flows, its size and shape will be observed
to change but its mass will remain unchanged. This is the principle of mass
conservation that applies to fluids in which no nuclear reactions are taking
place. The mathematical equivalence of the statement of mass conservation
is to set the Lagrangian derivative D/Dt of the mass of fluid contained in V,

which is f pdV, equal to zero. That is, the equation that expresses conserva-
14

tion of mass is

prdV=0.
DtJv

This equation may be converted to a volume integral in which the integrand
contains only Eulerian derivatives by use of Reynolds” transport theorem
(Equation 1.2), in which the fluid property a is, in this case, the mass density p:

p d
—+—(pu) dV=0.
ot ox, (puy)

Since the volume V was arbitrarily chosen, the only way in which the above
equation can be satisfied for all possible choices of V is for the integrand to be
zero. Then, the equation expressing conservation of mass becomes

p a
o + o, (pu,) = 0. (1.3a)

Equation 1.3a expresses more than the fact that mass is conserved. Since it
is a partial differential equation, the implication is that the velocity is continu-
ous. For this reason, Equation 1.3a is usually called the continuity equation.
The derivation that has been given here is for a single-phase fluid in which no
change of phase is taking place. If two phases were present, such as water and
steam, the starting statement would be that the rate at which the mass of fluid
1 is increasing is equal to the rate at which the mass of fluid 2 is decreasing.
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The generalization to cases of multiphase fluids and to cases of nuclear reac-
tions is obvious. Since such cases cause no changes in the basic ideas or prin-
ciples, they will not be included in this treatment of the fundamentals.

In many practical cases of fluid flow, the variation of density of the fluid
may be ignored, as for most cases of the flow of liquids. In such cases, the
fluid is said to be incompressible, which means that as a given mass of fluid
is followed, not only will its mass be observed to remain constant but its
volume, and hence its density, will be observed to remain constant as well.
Mathematically, this statement may be written as follows:

Dp_.
Dt

In order to use this special simplification, the continuity equation is first
expanded by use of a vector identity given in Appendix A:

) 9 ou
By, P oM,
ot Xy, X,

The first and second terms in this form of the continuity equation will be
recognized as being the Eulerian form of the material derivative as given by
Equation 1.1. That is, an alternative form of Equation 1.3a is

Do o0 . (1.3b)
Dt ax;

This mixed form of the continuity equation in which one term is given as
a Lagrangian derivative and the other as an Eulerian derivative is not useful
for actually solving fluid-flow problems. However, it is frequently used in
the manipulations that reduce the governing equations to alternative forms,
and for this reason, it has been identified for future reference. An immediate
example of such a case is the incompressible fluid under discussion. Since
Dp/Dt = 0 for such a fluid, Equation 1.3b shows that the continuity equation
assumes the simpler form p(du,/dx;) = 0. Since p cannot be zero in general, the
continuity equation for an incompressible fluid becomes

Ot _ 0 (incompressible). (1.30)
0x

It should be noted that Equation 1.3c is valid not only for the special case of
Dp/Dt = 0 in which p is constant everywhere but also for stratified fluid flows
of the type depicted in Figure 1.3. A fluid particle that follows the lines p = p,
or p = p, will have its density remain fixed at p = p, or p = p, so that Dp/Dt =0.
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FIGURE 1.3
Flow of density-stratified fluid in which Dp/Dt = 0 but for which dp/dx # 0 and dp/dy # 0.

However, p is not constant everywhere, so that dp/dx = 0 and dp/dy # 0. Such
density stratifications may occur in the ocean (owing to salinity variations)
or in the atmosphere (owing to temperature variations). However, in the
majority of cases in which the fluid may be considered to be incompressible,
the density is constant everywhere.

Equation 1.3, in either the general form (Equation 1.3a) or the incompress-
ible form (Equation 1.3c¢), is the first condition that has to be satisfied by
velocity and density. No dynamical relations have been used to this point,
but the conservation-of-momentum principle will utilize dynamics.

1.7 Conservation of Momentum

The principle of conservation of momentum is, in effect, an application of
Newton’s second law of motion to an element of the fluid. That is, when con-
sidering a given mass of fluid in a Lagrangian frame of reference, it is stated
that the rate at which the momentum of the fluid mass is changing is equal to
the net external force acting on the mass. Some individuals prefer to think of
forces only and restate this law in the form that the inertia force (due to accel-
eration of the element) is equal to the net external force acting on the element.

The external forces that may act on a mass of the fluid may be classed as
either body forces, such as gravitational or electromagnetic forces, or surface
forces, such as pressure forces or viscous stresses. Then, if f is a vector that
represents the resultant of the body forces per unit mass, the net external

body force acting on a mass of volume V will be f pfdV. Also, if P is a sur-
v
face vector that represents the resultant surface force per unit area, the net

external surface force acting on the surface S containing V will be f Pds.
s
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According to the statement of the physical law that is being imposed in this
section, the sum of the resultant forces evaluated above is equal to the rate of
change of momentum (or inertia force). The mass per unit volume is p and its

momentum is pu, so that the momentum contained in the volume V is f pu
\%4

dV. Then, if the mass of the arbitrarily chosen volume V is observed in the
Lagrangian frame of reference, the rate of change of momentum of the mass

contained with V will be (D/Df) f pudV. Thus, the mathematical equation
14

that results from imposing the physical law of conservation of momentum is

DBthpu dV=fP dS+prde.

In general, there are nine components of stress at any given point, one nor-
mal component and two shear components on each coordinate plane. These
nine components of stress are most easily illustrated by use of a cubical ele-
ment in which the faces of the cube are orthogonal to the Cartesian coordi-
nates, as shown in Figure 1.4, and in which the stress components will act ata
point as the length of the cube tends to zero. In Figure 1.4, the Cartesian coor-
dinates x, y, and z have been denoted by x,, x,, and x;, respectively. This per-
mits the components of stress to be identified by a double-subscript notation.
In this notation, a particular component of the stress may be represented by

022
17ﬁ 091
023,
: 012
033 |
|
| t-i— 0'11
031
o
| 13
/I_ R
033 -
7
7
Xy 7
7
X1

X3

FIGURE 1.4
Representation of nine components of stress that may act at a point in a fluid.
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the quantity o;, in which the first subscript indicates that this stress compo-
nent acts on the plane x; = constant and the second subscript indicates that it
acts in the x; direction.

The fact that the stress may be represented by the quantity c;, in which i
and j may be 1, 2, or 3, means that the stress at a point may be represented
by a tensor of rank 2. However, on the surface of our control volume, it was
observed that there would be a vector force at each point, and this force was
represented by P. The surface force vector P may be related to the stress ten-
sor c; as follows: The three stress components acting on the plane x; = con-
stant are 6y, 615, and 6,;. Since the unit normal vector acting on this surface
is n,, the resulting force acting in the x, direction is P, = 6,,n,. Likewise, the
forces acting in the x, direction and the x; direction are, respectively, P, =
o1, and P; = 6,311;. Then, for an arbitrarily oriented surface whose unit nor-
mal has components 7, 11,, and n;, the surface force will be given by P, = 6,
in which i is summed from 1 to 3. That is, in tensor notation, the equation
expressing conservation of momentum becomes

D
aﬂ/puj dV=fsoijni dS+prfj dv.

The left-hand side of this equation may be converted to a volume integral in
which the integrand contains only Eulerian derivatives by use of Reynolds’
transport theorem, Equation 1.2, in which the fluid property o here is the
momentum per unit volume pu; in the X; direction. At the same time, the sur-
face integral on the right-hand side may be converted into a volume integral
by use of Gauss’ theorem as given in Appendix B. In this way, the equation
that is evolved from Newton’s second law becomes

i(u)+i(uu)dV— ﬂdV+ f; dV
y oot U gy, Pt fvax,. fvp’ '

All these volume integrals may be collected to express this equation in the
form f {}dV =0, where the integrand is a differential equation in Eulerian
\4

coordinates. As before, the arbitrariness of the choice of the control volume V
is now used to show that the integrand of the above integrodifferential equa-
tion must be zero. This gives the following differential equation to be satisfied
by the field variables in order that the basic law of dynamics may be satisfied:

i( u)+i( uu)—%+ f -
o PTG PHIME = R

The left-hand side of this equation may be further simplified if the two
terms involved are expanded in which the quantity pu;u, is considered to be
the product of pu; and u,.
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auj p % Jo

p—+u-a—+u-i(pu)+pu =—ij+pf-
at ot Tox, o “ox, ox, 1

The second and third terms on the left-hand side of this equation are now
seen to sum to zero since they amount to the continuity equation (Equation
1.3a) multiplied by the velocity u;. With this simplification, the equation that
expresses conservation of momentum becomes

ou;

; du; 9oy f (1.4)
—+PQU, — = —+ P B
Pt TP oy T, T

It is useful to recall that this equation came from an application of Newton’s
second law to an element of the fluid. The left side of Equation 1.4 represents
the rate of change of momentum of a unit volume of the fluid (or the inertia
force per unit volume). The first term is the familiar temporal acceleration
term, while the second term is a convective acceleration and accounts for local
accelerations (such as when flowing around obstacles) even when the flow is
steady. Note also that this second term is nonlinear since the velocity appears
quadratically. On the right-hand side of Equation 1.4 are the forces causing the
acceleration. The first of these is due to the gradient of surface shear stresses,
while the second is due to body forces, such as gravity, which act on the mass
of the fluid. A clear understanding of the physical significance of each of the
terms in Equation 14 is essential when approximations to the full govern-
ing equations must be made. The surface-stress tensor 6; has not been fully
explained up to this point, but it will be investigated in detail in a later section.

1.8 Conservation of Energy

The principle of conservation of energy amounts to an application of the first
law of thermodynamics to a fluid element as it flows. The first law of thermo-
dynamics applies to a thermodynamic system that is originally at rest and,
after some event, is finally at rest again. Under these conditions, it is stated
that the change in internal energy, due to the event, is equal to the sum of the
total work done on the system during the course of the event and any heat
that was added. Although a specified mass of fluid in a Lagrangian frame
of reference may be considered a thermodynamic system, it is, in general,
never at rest and therefore never in equilibrium. However, in the thermody-
namic sense, a flowing fluid is seldom far from a state of equilibrium, and
the apparent difficulty may be overcome by considering the instantaneous
energy of the fluid to consist of two parts: intrinsic or internal energy and
kinetic energy. That is, when applying the first law of thermodynamics, the
energy referred to is considered to be the sum of the internal energy per unit
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- . 1 . o
mass ¢ and the kinetic energy per unit mass Suu In this way, the modified

form of the first law of thermodynamics that will be applied to an element
of the fluid states that the rate of change of the total energy (intrinsic plus
kinetic) of the fluid as it flows is equal to the sum of the rate at which work is
being done on the fluid by external forces and the rate at which heat is being
added by conduction.

With this basic law in mind, we again consider any arbitrary mass of fluid
of volume V and follow it in a Lagrangian frame of reference as it flows. The

total energy of this mass per unit volume is pe + Epu-u, so that the total

energy contained in V will be f pe + %pu‘u dV. As was established in
\4

the previous section, there are two types of external forces that may act on
the fluid mass under consideration. The work done on the fluid by these
forces is given by the product of the velocity and the component of each force
that is colinear with the velocity. That is, the work done is the scalar product
of the velocity vector and the force vector. One type of force that may act on
the fluid is a surface stress whose magnitude per unit area is represented by

the vector P. Then, the total work done owing to such forces will be f u-Pds,
s

where S is the surface area enclosing V. The other type of force that may act
on the fluid is a body force whose magnitude per unit mass is denoted by
the vector f. Then, the total work done on the fluid due to such forces will be

f u-pfdV. Finally, an expression for the heat added to the fluid is required.
\4

Let the vector q denote the conductive heat flux leaving the control volume.
Then, the quantity of heat leaving the fluid mass per unit time per unit sur-
face area will be q - n, where n is the unit outward normal, so that the net

amount of heat leaving the fluid per unit time will be | q-ndS.
s

Having evaluated each of the terms appearing in the physical law that is to
be imposed, the statement may now be written in analytic form. In doing so,
it must be borne in mind that the physical law is being applied to a specific,
though arbitrarily chosen, mass of fluid so that Lagrangian derivatives must
be employed. In this way, the expression of the statement that the rate of
change of total energy is equal to the rate at which work is being done plus
the rate at which heat is being added becomes

Bf pe+lpu-u dV=fu-P dS+fu-pf dV—fq-n ds.
DtJv 2 s v s

This equation may be converted to one involving Eulerian derivatives
only by use of Reynolds’ transport theorem, Equation 1.2, in which the fluid
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property a is here the total energy per unit volume pe+lpu'u . The
resulting integrodifferential equation is 2

9 e+1uu +i e+l u-u uy dV
fv ot PP o, 2P k

=fsu‘PdS+fVu-pde—fsq'n ds.

The next step is to convert the two surface integrals into volume integrals
so that the arbitrariness of VV may be exploited to obtain a differential equa-
tion only. Using the fact that the force vector P is related to the stress tensor
o; by the equation P; = 6,1, as was shown in the previous section, the first
surface integral may be converted to a volume integral as follows:

fsu'PdS=fu]0qnz ds = f u0;)dV.

Here, use has been made of Gauss’ theorem as documented in Appendix B.
Gauss’ theorem may be applied directly to the heat-flux term to give

fands= [qmds= [ q’ av.

Since the stress tensor o;; has been brought into the energy equation, it is
necessary to use the tensor notation from this point on. Then, the expression
for conservation of energy becomes

d 1 ] 1
— Pe+_pUl; +—— pe+ _puUl; U dv
fv ot PP ox, TPt

=fva%(u] DAV [ upfdv- [ 7l % av.

Having converted each term to a volume integral, the conservation equa-

tion may be considered to be of the form f {}dV =0, where the choice of V
\4

is arbitrary. Then, the quantity inside the brackets in the integrand must be
zero, which results in the following differential equation:

d 1 )
— pe+— puu +a pe+5pujuj U =§(u] i)+ upf; - —
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This equation may be made considerably simpler by using the equations
that have already been derived, as will now be demonstrated. The first term

on the left-hand side may be expanded by considering pe and u;u; to be

the products (p)(e) and (p) — u , respectively. Then,

9 e+lu = —e ap —luu +luu@
atp P pat ot pat 2 2 7 ot

Similarly, the second term on the left-hand side of the basic equation may
be expanded by considering peu; to be the product (e)(pu;) and %pujujuk to
be the product %ujuj (pty). Thus,

9 pe+lpuu U ei(pu)+pu ge
o, 20 TR T gy R “ox,

+1uui(u)+ uiluu
2]’axkpk pkaxk 2

In this last equation, the quantity (d/0x,)(pu), which appears in the first and
third terms on the right-hand side, may be replaced by —dp/dt in view of the
continuity equation (Equation 1.3a). Hence, it follows that

9 e+luuu —e@+u£luu@+uiluu
ox, Dot Pt ot TP gn T Mty TP G MM

Now, when the two components constituting the left-hand side of the basic
conservation equation are added, the two terms with minus signs above are
canceled by corresponding terms with plus signs to give

i e+1u +i (i’+l uu. u
o POT P F o PeT P T

= %+ uﬁ+iluu +uiluu
Pt TP o, TP oMM TP ot

de au] au
=p—+puU u +pu;ul
p Pkak pjat Y kak
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Then, noting that

9 d0;; ou;
Tox, 7 o,

the equation that expresses the conservation of energy becomes

p%+pu de u%+ uu %—u aOij+0 %+u f i)
or TP G TPy TPt T T T i gy, TP a;

Now, it can be seen that the third and fourth terms on the left-hand side are
canceled by the first and third terms on the right-hand side since these terms
collectively amount to the product of u; with the momentum equation (Equation
1.4). Thus, the equation expressing conservation of thermal energy becomes

de d ou; 9q;
0% wpu, 2o, T (1.5)

ot ax, ' ox, oy

The terms that were dropped in the last simplification were the mechanical-
energy terms. The equation of conservation of momentum, Equation 14,
may be regarded as an equation of balancing forces with j as the free sub-
script. Therefore, the scalar product of each force with the velocity vector,
or the multiplication by u,, gives the rate of doing work by the mechanical
forces, which is the mechanical energy. On the other hand, Equation 1.5 is a
balance of thermal energy, which is what is left when the mechanical energy
is subtracted from the balance of total energy. The result is usually referred
to as simply the energy equation.

As with the equation of momentum conservation, it is instructive to interpret
each of the terms appearing in Equation 1.5 physically. The entire left-hand
side represents the rate of change of internal energy, with the first term being
the temporal change while the second being due to local convective changes
caused by the fluid flowing from one area to another. The entire right-hand
side represents the cause of the change in internal energy. The first of these
terms represents the conversion of mechanical energy into thermal energy
due to the action of the surface stresses. As will be seen later, part of this con-
version is reversible and part is irreversible. The final term in the equation
represents the rate at which heat is being added by conduction from outside.

1.9 Discussion of Conservation Equations

The basic conservation laws, Equations 1.3a, 14, and 1.5, represent five scalar
equations that the fluid properties must satisfy as the fluid flows. The continuity
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and the energy equations are scalar equations, while the momentum equation
is a vector equation that represents three scalar equations. Two equations of
state may be added to bring the number of equations up to seven, but our basic
conservation laws have introduced 17 unknowns. These unknowns are the sca-
lars p and ¢, the density and the internal energy, respectively; the vectors #; and
q; the velocity and heat flux, respectively, each vector having three components;
and the stress tensor o which has, in general, nine independent components.

In order to obtain a complete set of equations, the stress tensor o, and the
heat-flux vector g, must be further specified. This leads to the so-called consti-
tutive equations in which the stress tensor is related to the deformation-rate
tensor and the heat-flux vector is related to temperature gradients. Although
the latter relation is very simple, the former is quite complicated and requires
either an intimate knowledge of tensor analysis or a clear understanding of
the physical interpretation of certain tensor quantities. For this reason, prior
to establishing the constitutive relations, the tensor equivalents of rotation
and rate of shear will be established.

1.10 Rotation and Rate of Shear

It is the purpose of this section to consider the rotation of a fluid element
about its own axis and the shearing of a fluid element and to identify the
tensor quantities that represent these physical quantities. This is most easily
done by considering an infinitesimal fluid element of rectangular cross sec-
tion and observing its change in shape and orientation as it flows.

Figure 1.5 shows a two-dimensional element of fluid (or the projection of
a three-dimensional element) whose dimensions at time ¢t = 0 are 5x and dy.
The fluid element is rectangular at time t = 0, and its centroid coincides with
the origin of a fixed coordinate system. For purposes of identification, the
corners of the fluid element have been labeled A, B, C, and D.

After a short time interval 8¢, the centroid of the fluid element will have
moved downstream to some new location as shown in Figure 1.5. The dis-
tance the centroid will have moved in the x direction will be given by

3t
Ax = fo ulx(t), y(t)] dt.

Since the values of x and y must be close to zero for short times such as 8¢,
the velocity component # may be expanded in a Taylor series about the point
(0,0) to give

St
we u(0,0)+x(f)ZZ(O,O)+y(t)3:(O,0)+-~ dt
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FIGURE 1.5
Infinitesimal element of fluid at time ¢ = 0 (indicated by ABCD) and at time ¢ = 8¢ (indicated by
ABCD)).

where the dots represent terms that are smaller than those presented and
that will eventually vanish as the limit of 8t — 0 is taken. Integrating the
leading term explicitly gives

Ot

A = (0, )3t + | X(0) 22 (0,0)+ y(t)g;‘(o, 04 di

= (0, 0)3t + .

Similarly,
Ay =v(0,0) 6t + ---.

As well as moving bodily, the fluid element will rotate and will be dis-
torted as indicated by the corners, which are labeled A; B;, C, and D’ to rep-
resent the element at time t = &t. The rotation of the side CD to its new
position C'D’ is indicated by the angle 6o, where a is positive when mea-
sured counterclockwise. Similarly, the rotation of the side BC to its new
position B'C’is indicated by the angle 8, where f is positive when measured
clockwise. Expressions for da and 8f in terms of the velocity components
may be obtained as follows.
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From the geometry of the element as it appears at time t = &t,

5o tan-! ¥ component of D C

x component of D C

1% l6x,—lf>y o+ — v —léx,—léy Ot +---
! 2 2 2 2
= tan

Ox +---

. . . . 1 1
where v is evaluated first at point D, whose coordinates are Eéx,—if)y ,

and secondly at point C, whose coordinates are - %6x, —%6y . The x com-

ponent of the side D'C’ will be only slightly different from dx, and it turns out
that the precise departure from this value need not be evaluated explicitly.

Expanding the velocity component v in a Taylor series about the point (0,0)
results in the following expression for da:

v(0,0) + %Evc(av/ dx)(0,0) - %6}/(81] /oy)(0,0)+--- ot

da = tan™!
ox(1+--)
- 0(0,0) - %Evc(av/ dx)(0,0) - %63/(37; /ay)(0,0)+--- ot
ox(1+--)
— tan! [6x(dv/0x)(0,0) + ---15¢

ox(L+--)

4 [(8v/0x)(0,0) +---15¢t
(1+--)

= tan

=tan”’ ?(0,0)+~-- ot .
X

Since the argument of the arctangent is small, the entire right-hand side
may be expanded to give
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o= 20(0,0) 4 Ot +---
Jx

oo v
—=—(0,0
ot ox 0,0+

This expression represents the change in the angle a per unit time so that
in the limit as 8x, 8y, and 6t all tend to zero, this expression becomes

. 0o
=—1(0,0
a ax( )

where a is the time derivative of the angle a. By an identical procedure, it
follows that the time derivative of the angle f is given by

c U
p= @(0,0)-

Recall that a is measured counterclockwise and f is measured clockwise.
Thus, the rate of clockwise rotation of the fluid element about its centroid is
given by

1 ou ov

*([3— )‘E @_8735 .

Likewise, the shearing action is measured by the rate at which the sides B'C’
and D'C’ are approaching each other and is therefore given by the quantity

The foregoing analysis was carried out in two dimensions, which may be
considered as the projection of a three-dimensional element on the xy plane.
If the analysis is carried out in the other planes, it may be verified that the
rate of rotation of the element about its own axes and the rate of shearing are
given by the following quantities in the yz, xz, and xy planes, respectively:

ow o 1w g 1w

rate of rotation: 1w oo ; 1 ; 1 (1.6a)
2 gy 9z 2 o0z dx 2 dy ox

rate of shearing: 1 a—w+a—v ; 1 a—u+a—w ; 1 %4_671) . (L6b)
2 9y 9z 2 9z dx 2 dy ox
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Both of these quantities may be represented by a tensor of rank 2, as we
will now show. Define the deformation-rate tensor e; and split it into its sym-
metric and antisymmetric parts as follows:

ou,

1

1 oy auj 1 o auj
e =—"= -—L 4= +—L
Toox; 2 ox; ox, 2 ox;  ox;

The antisymmetric part of the deformation-rate tensor represents the rate
of rotation of a fluid element about its own axes, while the symmetric part
of the deformation-rate tensor represents the rate of shearing of the fluid
element. The antisymmetric part has three independent components, corre-
sponding to Equation 1.6a, and these components correspond to those of the
curl of the velocity vector. The symmetric part has six independent compo-
nents, including the components given in Equation 1.6b, and it corresponds
to the six independent components of stress at a point. This conclusion will
be verified a little later in this chapter. That is, there are a total of nine com-
ponents of the deformation-rate tensor since it is a tensor of rank 2, and these
components collectively represent the rotation and the rate of shear of the
fluid element.

1.11 Constitutive Equations

In this section, the nine elements of the stress tensor o;; will be related to the
nine elements of the deformation-rate tensor ¢, by a set of parameters. All
these parameters except two will be evaluated analytically, and the remain-
ing two, which are the viscosity coefficients, must be determined empiri-
cally. In order to achieve this end, the postulates for a Newtonian fluid will
be introduced directly. Water and air are by far the most abundant fluids on
earth, and they behave like Newtonian fluids, as do many other common
fluids. It should be pointed out, however, that some fluids do not behave in a
Newtonian manner, and their special characteristics are among the topics of
current research. One example is the class of fluids called viscoelastic fluids,
whose properties may be used to reduce the drag of a body. Since this book is
concerned with the classical fundamentals only, the Newtonian fluid will be
treated directly. If the various steps are clearly understood, there should be
no conceptual difficulty in following the details of some of the more complex
fluids such as viscoelastic fluids.

Certain observations and postulates will now be made concerning the
stress tensor. The precise manner in which the postulates are made is largely
a matter of taste, but when the Newtonian fluid is being treated, the resulting
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equations are always the same. The following are the four conditions the
stress tensor is postulated to satisfy:

1. When the fluid is at rest, the stress is hydrostatic and the pressure
exerted by the fluid is the thermodynamic pressure.

2. The stress tensor o;; is linearly related to the deformation-rate tensor

e, and depends only on that tensor.
3. Since there is no shearing action in a solid-body rotation of the fluid,
no shear stresses will act during such a motion.

4. There are no preferred directions in the fluid, so that the fluid prop-
erties are point functions.

Condition 1 requires that the stress tensor o; be of the form
o= —pd; + T;

where t; depends upon the motion of the fluid only and is called the shear-stress
tensor. The quantity p is the thermodynamic pressure and §;; is the Kronecker
delta. The pressure term is negative since the sign convention being used here
is that normal stresses are positive when they are tensile in nature.

The remaining unknown in the constitutive equation for stress is the shear-
stress tensor t;. Condition 2 postulates that the stress tensor, and hence the
shear-stress tensor, is linearly related to the deformation-rate tensor. This is
the distinguishing feature of Newtonian fluids. In general, the shear-stress
tensor could depend upon some power of the velocity gradients other than
unity, and it could depend upon the velocity itself as well as the velocity
gradient. The condition postulated here can be verified experimentally in
simple flow fields in most common fluids, and the results predicted for more
complex flow fields yield results that agree with physical observations. This
is the sole justification for condition 2 above.

There are nine elements in the shear-stress tensor t;, and each of these ele-
ments may be expressed as a linear combination of the nine elements in the
deformation-rate tensor ey, (just as a vector may be represented as a linear com-
bination of components of the base vectors). That is, each of the nine elements
of 7; will, in general, be a linear combination of the nine elements of ¢, so that
81 parameters are needed to relate t; to ¢;. This means that a tensor of rank 4 is
required so that the general form of t; will be, according to condition 2,

o oy,
o= Ol
7 1) axl

It was shown in the previous section that the tensor du,/dx, like any
other tensor of rank 2, could be broken down into an antisymmetric part
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and a symmetric part. Here, the antisymmetric part corresponds to the
rate of rotation of a fluid element, and the symmetric part corresponds to
the shearing rate. According to condition 3, if the flow field is executing a
simple solid-body rotation, there should be no shear stresses in the fluid.
However, for a solid-body rotation, the antisymmetric part of du,/dx;, namely,

1 . . o\
E(auk /9x; - du;/ dx, ), will not be zero. Hence, in order that condition 3 may

be satisfied, the coefficients of this part of the deformation-rate tensor must
be zero. That is, the constitutive relation for stress must be of the form

1 o, Ay
T.=—0B., —+—=
if ) Bl]kl axl axk

The 81 elements of the fourth-rank tensor B, are still undetermined, but
condition 4 has yet to be imposed. This condition is the so-called condition of
isotropy, which guarantees that the results obtained should be independent
of the orientation of the coordinate system chosen. In Appendix B (a sum-
mary of some useful tensor relations), it is pointed out that the most general
isotropic tensor of rank 4 is of the form

Bijr = A8;By + 1(8:5; + 88;) + v(8:8; — 8

where A, p, and y are scalars. The proof of this is straightforward but tedious.
The general tensor is subjected to a series of coordinate rotations and inflec-
tions, and the condition of invariance is applied. In this way, the 81 quantities
contained in the general tensor are reduced to three independent quantities
in the isotropic case. In the case of the fourth-rank tensor relating the shear-
stress tensor to the deformation-rate tensor, namely, B;;;, not only must it be
isotropic but it must be symmetric in view of condition 3 as well. That is,
the coefficient y must be zero in this case so that the expression for the shear
stress becomes

1 . ou
Ty = 5[7&5176“ + (80 +0;0;)] aixl,(-l- @

Using the fact that §;; = 0 unless [ = k shows that

1
*7\61761(1 s + 9y - >“6ij o
2 ax;,  0x, 0x;,

in which / has been replaced by k. Likewise, replacing k by i and I by j shows
that
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1 d 9 1 o 0u;
- 6ik6jl ﬂ_,_ﬂ =_ %_Fi]
2 ax,  0xy 2 0x; dx;

and replacing I by i and k by j shows that

1 1
g, M, om 1004
2 ax;,  9x; 2 ox; o

1

au]- ou:
+

Hence, the expression for the shear-stress tensor becomes

ou ou; ou;
Ty =M+ 4L
X, ox;  ox;

Thus, the constitutive relation for stress in a Newtonian fluid becomes

1

+
Xy ox;  ox;

w9y

O = —pd; + A, (17)

which shows that the stress is represented by a second-order symmetric
tensor.

The nine elements of the stress tensor 6, have now been expressed in terms
of the pressure and the velocity gradients, which have all been previously
introduced, and two coefficients A and p. These coefficients cannot be deter-
mined analytically and must be determined empirically. Up to this point,
both A and p are just coefficients, but their nature and physical significance
will be discussed in the next section.

The second constitutive relation involves the heat-flux vector g;, which rep-
resents conduction alone. Fourier’s law of heat conduction states that the heat
flux by conduction is proportional to the negative temperature gradient so
that

o

Lk
9 ax;

(1.8)

This is the constitutive equation for the heat flux, where the proportional-
ity factor k in Fourier’s law is the thermal conductivity of the fluid. In using
Equation 1.8, it is implicitly assumed that the concept of temperature, as
employed in equilibrium thermodynamics, also applies to a moving fluid.
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1.12 Viscosity Coefficients

It was pointed out in the previous section that the parameters A and p, which
appear in the constitutive equations for stress, must be determined experi-
mentally. It is the purpose of this section to establish a physical interpreta-
tion of these two parameters and thus show the manner in which they may
be evaluated.

Consider a simple shear flow of an incompressible fluid in which the veloc-
ity components are defined by

u =u(y)
v=w=0

That is, only the x component of velocity is nonzero, and that component is
a function of y only. From the definition of this flow field, the components of
the stress tensor may be evaluated from Equation 1.7 to give

du

Opp =0y = ———
dy

O3 =0pn =033 =-p

O3 =03 = Oy = O3 = 0.

That is, the normal components of the stress are defined by the thermody-
namic pressure, and the nonzero shear components of the stress are propor-
tional to the velocity gradient with the parameter u as the proportionality
factor. However, from Newton’s law of viscosity, the proportionality factor
between the shear stress and the velocity gradient in a simple shear flow is
the dynamic viscosity. Hence, the quantity p that appears in the constitutive
equation for stress is the dynamic viscosity of the fluid. Frequently, the kine-
matic viscosity, defined by v = p/p, is used instead of the dynamic viscosity.

The parameter A in Equation 1.7 is usually referred to as the second viscos-
ity coefficient. In order to establish its significance, the average normal stress
component p will be calculated.

_ 1
-p= 5(011 + 0y +03) .

This average normal stress is the mechanical pressure in the fluid, and it is
equal to one-third of the trace of the stress tensor. Since the mechanical pres-
sure is either purely hydrostatic or hydrostatic plus a component induced by
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the stresses that result from the motion of the fluid, it will, in general, be dif-
ferent from the thermodynamic pressure p. Using Equation 1.7, the mechani-
cal pressure p may be evaluated as follows:

—f)=l —p+7\%+2 ou + —p+}\%+2 L
3 0x; dx 0x; ay
+ —p+k%+2 9w
X, 0z
cpen e 2 O
ax, 3 ox
=-p+ )\+g %
3 Ix

That is, the difference between the thermodynamic pressure and the
mechanical pressure is proportional to the divergence of the velocity vector.
The proportionality factor is usually referred to as the bulk viscosity and is
denoted by K. That is,

_ ou
—p=K=—2k
Por =Ko

where K = A + % . Of the three viscosity coefficients p, A, and K, only two are

independent and the third is defined by the equation that defines K. For pur-
poses of physical interpretation of these viscosity coefficients, it is preferred
to discuss p (which has already been done) and K, leaving A to be defined by
A-K-2 .
3

In order to identify the physical significance of the bulk viscosity, some of
the results of the kinetic theory of gases will be used. The mechanical pres-
sure is a measure of the translational energy of the molecules only, whereas
the thermodynamic pressure is a measure of the total energy, which includes
vibrational and rotational modes of energy as well as the translational mode.
For liquids, other forms of energy are also included such as intermolecular
attraction. These different modes of molecular energy have different relax-
ation times, so that in any given flow field, it is possible to have energy trans-
ferred from one mode to another. The bulk viscosity is a measure of this
transfer of energy from the translational mode to the other modes, as may be
seen from the relation p - p = K(du; /dx;). For example, during the passage
through a shock wave, the vibrational modes of energy are excited at the



32 Fundamental Mechanics of Fluids

expense of the translational modes, so that the bulk viscosity will be nonzero
in this case.

The above discussion has been for a polyatomic molecule of a liquid or a
gas. If the fluid is a monatomic gas, the only mode of molecular energy is the
translational mode. Then, for such a gas, the mechanical pressure and the ther-
modynamic pressure are the same, so that the bulk viscosity is zero. That is,

L2
3

which is called Stokes’ relation, so that there is only one independent viscosity
coefficient in the case of monatomic gases. For polyatomic gases and for lig-
uids, the departure from K = 0 is frequently small, and many authors incor-
porate Stokes’ relation in the constitutive relation (Equation 1.7) for stress. In
any case, for incompressible fluids, Equation 1.7 shows that it is immaterial

2 . . . .
whether A =- 3 or not, for then the term involving A is zero by virtue of

the continuity equation.

1.13 Navier—Stokes Equations

The equation of momentum conservation (Equation 14), together with the
constitutive relation for a Newtonian fluid (Equation 1.7), yields the famous
Navier—Stokes equations, which are the principal conditions to be satisfied by
a fluid as it flows. Having obtained an expression for the stress tensor, the term
do;/dx; that appears in Equation 1.4 may be evaluated explicitly as follows:

00 ; 9 9 ou.  ou;
7 _péij_,_;\aijﬂ*_ o 9
0X;, dax;  0x;

ox;  0x; ] ;

0 om0 ow o

ox;  ox; 0x;, 0x; ox; o,

where, in the first two terms, i has been replaced by j since it is only wheni =
that these terms are nonzero. Substituting this result into Equation 1.4 gives

ou, ou; ) a .9 . o,
oty L =P 0y e PR +0f;. (19a)
ot 00X ox;  dx; 0x; x; ax; o,
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Equation 1.9a is known as the Navier—Stokes equations, representing three
scalar equations corresponding to the three possible values of the free sub-
script j. In the most frequently encountered situations, the fluid may be
assumed to be incompressible and the dynamic viscosity may be assumed to
be constant. Under these conditions, the second term on the right-hand side
of Equation 1.9a is identically zero and the viscous-shear term becomes

2 2
au]- _ au]-

dx;0x; dx;0x;

1 1

d ou; au]' 0 o
+ [

ox,  ox; ox; ox; ox;

That is, the viscous-shear term is proportional to the Laplacian of the
velocity vector, and the constant of proportionality is the dynamic viscosity.
Then, the Navier-Stokes equations for an incompressible fluid of constant
density become

o, ou; 9 0%u;
Lypuy L =Py !

P ot X, 9x; 0x;0x;

+pf;- (1.9b)

In the special case of negligible viscous effects, Equation 1.9a becomes

p%+puk%=—a—p+pf». (19¢)
ot o, ox; )

Equation 19c¢ is known as the Euler equations and the fluid is said to be
inviscid.

1.14 Energy Equation

The term o,(duy/dx) that appears in the equation of energy conservation
(Equation 1.5) may now be evaluated explicitly by use of Equation 1.7.

au; u ou,  ou,  du;
O = —pdy+ My 4 4L L
0x; 0X; ox; ox;  ox

Using the fact that the first two terms on the right-hand side of this equa-
tion are zero unless i = j, this expression becomes
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1

T o, ax; X, ax;  ox;  0x;

2
Ouj _ e o O O O O

It will be recalled that the term o,(du/dx,) represents the work done by the
surface forces. The first term in the expression for this work done, namely,
—p(01,/dxy), represents the reversible transfer of energy due to compression.
The remaining two terms are collectively called the dissipation function and
are denoted by @. That is,

2
e w0 0y

9x; ax;  dx; dx;

O =) (1.10)

The reason @ is called the dissipation function is that it is a measure of the
rate at which mechanical energy is being converted into thermal energy. This
may be readily verified by considering an incompressible fluid in a Cartesian
coordinate system. Then,

oo w ow o
ox; 0x; Ox;
_oaw 0w 1y w1 o
ox; ox; 2 dx; dx; 2 dx;  Ox;
2
_1 w9
2 ox; ox

which is a positive-definite quantity. This shows that the dissipation func-
tion always works to increase irreversibly the internal energy of an incom-
pressible fluid.

In terms of the dissipation function, the total work done by the surface
stresses is given by

Using this result and the constitutive relation for the heat flux (Equation
1.8) in the equation of conservation of energy, Equation 1.5, yields the energy
equation for a Newtonian fluid:
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de de ou, d aT
P P —=mp ok ko
ot ax; ox,  ox;  0x;

+@ (1.11)

where @ is defined by Equation 1.10.

1.15 Governing Equations for Newtonian Fluids

The equations that govern the motion of a Newtonian fluid are the continu-
ity equation (Equation 1.3a), the Navier-Stokes equations (Equation 1.9a), the
energy equation (Equation 1.11), and equations of state. For purposes of sum-
mary and discussion, these equations will be repeated here.

ap d
—+—(pu)=0 1.3a
ot ox, (puy) (1.3a)
ou; ou; ap 0 ou 9 ou, U,
R S Lv—L 4pfi (19

Pt TP o T Ty Tax, Tax, Tax, o, T an, o (19)

d ) d d oT d : d du; Ju;
0% you, e Ly, 8 G OT Ay O

ot X, ox,  dx;  0x; 0X; ox;  9x; ox;
p=ppT) (1.12)
e=e(p, 7). (1.13)

The last two equations are general representations of the thermal and
caloric equations of state, respectively. The most frequently encountered
form of the thermal equation of state is the ideal-gas law p = pRT, whereas
the most frequently encountered form of the caloric equation of state is e =
CyI, where Cy is the specific heat at constant volume.

The preceding set of equations represents seven equations that are to be
satisfied by seven unknowns. Each of the continuity, energy, and state equa-
tions supplies one scalar equation, whereas the Navier-Stokes equations
supply three scalar equations. The seven unknowns are the pressure, den-
sity, internal energy, temperature, and velocity components, that is, p, p, ¢, T,
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and u;. The parameters A, p, and k are assumed to be known from experimen-
tal data, and they may be constants or specified functions of the temperature
and pressure.

It is not always necessary to solve the complete set of equations in order
to define the flow field analytically. For example, if compressible effects are
thought to be unimportant in the flow field being considered, the incom-
pressible form of the governing equations may be used. The continuity
equation and the Navier-Stokes equations are then simpler, as indicated
by Equations 1.3c and 1.9b, respectively, but the greatest simplification
comes from the fact that the energy equation is mathematically uncoupled
from these two equations. The continuity and Navier-Stokes equations
offer four scalar equations involving only p and u;. That is, the pressure
and velocity fields may be established without reference to the energy
equation. After doing this, the temperature field may be established sepa-
rately, and the temperature may have the trivial solution T = constant. In
cases of forced convection heat transfer in which the flow is turbulent,
the continuity and Navier—Stokes equations are frequently replaced by an
empirical velocity distribution, and the energy equation is solved to yield
the temperature distribution. More frequently, however, thermal effects
are unimportant, and the continuity and Navier-Stokes equations alone
must be solved.

The most common type of body force acting on a fluid is due to gravity, so
that the body force f; that appears in the Navier-Stokes equations is defined
in magnitude and direction by the acceleration due to gravity. Sometimes,
however, electromagnetic effects are important, and in such cases, f = (p.E +
J x B), which is the Lorentz force. Here, p, is the charge density, E is the
electric field vector, J is the electric current density, and B is the magnetic
field vector. The electric and magnetic fields themselves must obey a set of
physical laws that are expressed by Maxwell’s equations. The solution to
such problems requires the simultaneous solution of the equations of fluid
mechanics and of electromagnetism. One special case of this type of cou-
pling is the field known as magnetohydrodynamics.

It may also be pointed out that the governing equations summarized here
contain the equations of hydrostatics and heat conduction as special cases.
If the fluid is at rest, the velocity components will all be zero, so that the
Navier—Stokes equations (Equation 1.9a) become

o

+pf..
axj pf]

If the body force f; is now set equal to the gravitational force, the equa-
tion of hydrostatics is obtained. For example, if gravity acts in the negative z
direction, f] = - ge,, where e, is the unit vector in the z direction. Then,
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p
P _ e
o, pge:

which shows that dp/dx = dp/dy = 0 and dp/0dz = — pg. In the case of zero veloc-
ity, the energy equation becomes

de_ o T

p =
ot ax]- ax]-

Introducing the enthalpy i = ¢ + p/p and using the fact that p and p are
constant in a stationary fluid give

oo ar

p =
o ox; - ox;

If the fluid is thermally perfect, i will be a function of T only, so that

oh  oh T aT

where C, is the specific heat at constant pressure, which is the appropriate
process for this case. Then, the energy equation becomes

oc, T -2 9T
ot ax]- ax]-

which is the equation of heat conduction.

1.16 Boundary Conditions

The Navier-Stokes equations are, mathematically, a set of three elliptic,
second-order partial differential equations. The appropriate type of bound-
ary conditions is therefore Dirichlet or Neumann conditions on a closed
boundary. Physically, this usually amounts to specifying the velocity on all
solid boundaries. Within the continuum approximation, the experimentally
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determined boundary condition is that there is no slip between the fluid and
a solid boundary at the interface. On the molecular scale, slippage is possible,
but it is confined within a layer whose dimensions are of the same order as
the mean free path between the molecules. Then, if U represents the velocity
of a solid boundary, the boundary condition that should be imposed on our
continuum velocity is

u=U on solid boundaries. (1.14)

In the case of an infinite expanse of fluid, one common form of Equation
114 is thatu — 0 as x — oo.

If thermal effects are included, a boundary condition on the tempera-
ture is also required. As in the case of heat-conduction problems, this
may take the form of specifying the temperature or the heat flux on some
boundary.

PROBLEMS

1.1 Derive the continuity equation from first principles using an infini-
tesimal control volume of rectangular shape and having dimensions
(x, 8y, 82). Identify the net mass flow rate through each surface of
this element as well as the rate at which the mass of the element is
increasing. The resulting equation should be expressed in terms
of the Cartesian coordinates (x,y,z,f), the Cartesian velocity compo-
nents (1,0,w), and the fluid density p.

1.2 Derive the continuity equation from first principles using an infini-
tesimal control volume of cylindrical shape and having dimensions
(OR, R86, 8z). Identify the net mass flow rate through each surface of
this element as well as the rate at which the mass of the element is
increasing. The resulting equation should be expressed in terms of
the cylindrical coordinates (R,6,z,t), the cylindrical velocity compo-
nents (ug,Uyu.), and the fluid density p.

1.3 Derive the continuity equation from first principles using an infini-
tesimal control volume of spherical shape and having dimensions (&7,
156, rsind dw). Identify the net mass flow rate through each surface
of this element as well as the rate at which the mass of the element
is increasing. The resulting equation should be expressed in terms
of the spherical coordinates (1,6,0,t), the spherical velocity compo-
nents (u,ug1,), and the fluid density p.

14 Obtain the continuity equation in cylindrical coordinates by
expanding the vector form in cylindrical coordinates. To do this,
make use of the following relationships connecting the coordi-
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nates and the velocity components in Cartesian and cylindrical

coordinates:
x=RcosB
y=Rsin0
z=2z

U=1upcosO—1u,sind

U = Ug Sin 0 + u, cos 0

1.5 Obtain the continuity equation in spherical coordinates by expand-
ing the vector form in spherical coordinates. Make use of the vector
relationships outlined in Appendix A and follow the procedures
used in Problem 1.4.

1.6 Evaluate the radial component of the inertia term (u - V)u in cylin-
drical coordinates using the following identities:

x=Rcos0
y=Rsin0
ue, +ve, = iy eg + Uy €

and any other vector identities from Appendix A as required. Here,
R and 0 are cylindrical coordinates, 1 and 1, are the correspond-
ing velocity components, and e, and e, are the unit base vectors.

1.7 Evaluate the radial component of the inertia term (u - V)u in spheri-
cal coordinates by use of the vector identities given in Appendix A.

1.8 Start with the shear stress tensor t;. Write out the independent
components of this tensor in Cartesian coordinates (x,y,z) using the
Cartesian representation (u,u,w) for the velocity vector. Specialize
these expressions for the case of a monatomic gas for which the
Stokes relation applies.
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1.9 Write out the expression for the dissipation function, ®, for the same
conditions and using the same notation as defined in Problem 1.8.

1.10 Write down the equations governing the velocity and pressure in
steady, two-dimensional flow of an inviscid, incompressible fluid,
but for which the density is not a constant, in which the effects of
gravity may be neglected. If the fluid is stratified, the density p will
depend, in general, on both x and y. Show that the transformation

u* = }ﬁu
Po

in which p, is a constant reference density, transforms the govern-
ing equations into those of a constant-density fluid whose velocity
components are u* and v*.
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Flow Kinematics

This chapter explores some of the results that may be deduced about the
nature of a flowing continuum without reference to the dynamics of the
continuum.

The first topic, flow lines, introduces the notions of streamlines, pathlines,
and streaklines. These concepts not only are useful for flow-visualization
experiments but also supply the means by which solutions to the governing
equations may be interpreted physically.

The concepts of circulation and vorticity are then introduced. Although
these quantities are treated only in a kinematic sense at this stage, their full
usefulness will become apparent in the later chapters when they are used in
the dynamic equations of motion.

The concept of the streamline leads to the concept of a stream tube or a
stream filament. Likewise, the introduction of the vorticity vector permits
the topic of vortex tubes and vortex filaments to be discussed. Finally, this
chapter ends with a discussion of the kinematics of vortex filaments or
vortex lines. In this treatment, a useful analogy with the flow of an incom-
pressible fluid is used. The results of this study form part of the so-called
Helmholtz equations, with the remaining parts being taken up in the next
chapter, which deals with, among other things, the dynamics of vorticity.

2.1 Flow Lines

Three types of flow lines are used frequently for flow-visualization pur-
poses. These flow lines are called streamlines, pathlines, and streaklines, and in
a general flow field, they are all different. The definitions and equations of
these various flow lines will be obtained separately below.

2.1.1 Streamlines

Streamlines are lines whose tangents are everywhere parallel to the veloc-
ity vector. Since, in unsteady flow, the velocity vector at a given point will
change both its magnitude and its direction with time, it is meaningful to
consider only the instantaneous streamlines in the case of unsteady flows.

41
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In order to establish the equations of the streamlines in a given flow field,
consider first a two-dimensional flow field in which the velocity vector u has
components 1 and v in the x and y directions, respectively. Then, by virtue of
the definition of a streamline, its slope in the xy plane, namely, dy/dx, must
be equal to that of the velocity vector, namely, v/u. That is, the equation of the
streamline in the xy plane is

dy _
dx

= |Q

where, in general, both u and v will be functions of x and y. Integration of
this equation with respect to x and y, holding t fixed, will then yield the
equation of the streamline in the xy plane at that instant in time.

In the case of a three-dimensional flow field, the foregoing analysis is valid
for the projection of the velocity vector on the xy plane. By similarly treating
the projections on the xz plane and on the yz plane, the slopes of the stream-
lines are found to be

dz _w
dx u
dz _w
dy o

on the xz and yz planes, respectively. These three equations defining the
streamline may be written in the form

dy dx dz_dx dz _dy
v U woou w v

Written in this form, it is clear that these three equations may be expressed
in the following more compact form:

Integration of these equations for fixed ¢ will yield, for that instant in time,
an equation of the form z = z(x, y), which is the required streamline. The
easiest way of carrying out the required integration is to try to obtain the
parametric equations of the curve z = z(x, y) in the form x = x(s), ¥ = y(s), and
z = z(s). Elimination of the parameter s among these equations will then yield
the equation of the streamline in the form z = z(x, y).
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Thus, a parameter s is introduced whose value is zero at some reference
point in space and whose value increases along the streamline. In terms of
this parameter, the equations of the streamline become

de _dy dz_
M_ U_ZU_

ds.

These three equations may be combined in tensor notation to give

dx; _ u(x;, t)  tfixed 2.1)
ds

in which it is noted that if the velocity components depend upon time, the
instantaneous streamline for any fixed value of t is considered. If the stream-
line that passes through the point (x,, v, zo) is required, Equation 2.1 is inte-
grated and the initial conditions that when s = 0, x = x,, ¥ = y,, and z = z, are
applied. This will result in a set of equations of the form

X;=x; (X, Yo, Zo, 1, 5)

which, as s takes on all real values, traces out the required streamline.
As an illustration of the determination of streamline patterns for a given
flow field, consider the two-dimensional flow field defined by

u=ux(1+2t)
v=y
w=0.

From Equation 2.1, the equations to be satisfied by the streamlines in the
xy plane are

% =x(1+2t)
dy
as v

Integration of these equations yields
X = Cle(1+2t)s

y=Cye’
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which are the parametric equations of the streamlines in the xy plane. In par-
ticular, suppose the streamlines passing through the point (1, 1) are required.
Using the initial conditions that when s = 0, x = 1 and y = 1 shows that C, =

C, = L. Then, the parametric equations of the streamlines passing through
the point (1, 1) are

X = e(1+2t)s
y =e°.

The fact that the streamlines change with time is evident from the preced-

ing equations. Suppose the streamline passing through the point (1, 1) at
time t = 0 is required; then,

Hence, the equation of the streamline is

x=y.

This streamline is shown in Figure 2.1 together with other flow lines,
which are discussed below.

Bl
1
d
4 <>
Gé\ @q’&
Yy 3 %\ S) [ —
) . |l —
3\ gaghling ——]
2
L

FIGURE 2.1

Comparison of the streamline through the point (1, 1) at t = 0 with the pathline of a particle that

passed through the point (1, 1) at f = 0 and the streakline through the point (1, 1) at f = 0 for the
flow field u = x(1 + 2t), v =y, w =0.
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2.1.2 Pathlines

A pathline is a line traced out in time by a given fluid particle as it flows.
Since the particle under consideration is moving with the fluid at its local
velocity, pathlines must satisfy the equation

dx
dt

L= u(x;, ). (2.2

The equation of the pathline that passes through the point (x,, y,, z,) at time
t = 0 will then be the solution to Equation 2.2, which satisfies the initial condi-
tion that when t = 0, x = x,, y =y, and z = z,. The solution will therefore yield
a set of equations of the form

x; = X{(Xo, Yo, Zo 1)

which, as t takes on all values greater than zero, will trace out the required
pathline.

As an illustration of the manner in which the equation of a pathline is
obtained, consider again the flow field defined by

u=x(1+21)
v=y
w = 0.

From Equation 2.2, the differential equations to be satisfied by the path-
lines are

% =x(1+2t)

dy
L _y
dt
Integration of these equations gives
X = Clet(m)

y =Cyel.

These are the parametric equations of all the pathlines in the xy plane for
this particular flow field. In particular, if the pathline of the particle that
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passed through the point (1, 1) at ¢ = 0 is required, these parametric equations
become

x = et0+)

y=e.

Eliminating t from these equations shows that the equation of the required
pathline is

X = y1+log v,

This pathline is shown in Figure 2.1, from which it will be seen that the
streamline that passes through (1, 1) at t = 0 does not coincide with the path-
line for the particle that passed through (1, 1) at t = 0.

2.1.3 Streaklines

A streakline is a line traced out by a neutrally buoyant marker fluid that is
continuously injected into a flow field at a fixed point in space. The marker
fluid may be smoke (if the main flow involves air or some other gas) or a dye
(if the main flow involves water or some other liquid).

A particle of the marker fluid that is at the location (, y, z) at time t must
have passed through the injection point (x,, y,, z,) at some earlier time ¢ = 7.
Then, the time history of this particle may be obtained by solving the equa-
tion for the pathline (Equation 2.2) subject to the initial conditions that x =
X0, Y = Yo and z = z; when t = 7. Then, as 7 takes on all possible values in the
range — < 1 < t, all fluid particles on the streakline will be obtained. That
is, the equation of the streakline through the point (x,, y,, z,) is obtained by
solving Equation 2.2 subject to the initial conditions that when t = 7, x = x,,
Y =Yy and z = z,,. This will yield an expression of the form

x; = X; (Yo, Yo, Zo/ 1, D)

Then, as 7 takes on the values 7t < t, these equations will define the instan-
taneous location of that streakline.

As an illustrative example, consider the flow field that was used to illus-
trate the streamline and the pathline. Then, the equations to be solved for
the streakline are

% =x(1+2t)

dy _

ar Y
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which integrate to give
X = Clet(m)
y=Cyel.
Using the initial conditions that x = y = 1 when t = 7, these equations become
X = et(1+t)—r (1+7)
y=eT
These are the parametric equations of the streakline that pass through the
point (1, 1), and they are valid for all times t. In particular, at ¢ = 0, these equa-
tions become
x = ¢~t(1+7)
y=e".

Eliminating T from these parametric equations shows that the equation of
the streakline that passes through the point (1, 1) is, at time ¢ = 0,

X = yl—log v,

This streakline is shown in Figure 2.1 along with the streamline and the
pathline that were obtained for the same flow field. It will be noticed that
none of the three flow lines coincide.

2.2 Circulation and Vorticity

The circulation contained within a closed contour in a body of fluid is defined
as the integral around the contour of the component of the velocity vector
that is locally tangent to the contour. That is, the circulation I' is defined as

r= ff,u-cu 2.3)

where d1 represents an element of the contour. The integration is taken
counterclockwise around the contour, and the circulation is positive if this
integral is positive.
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The vorticity of an element of fluid is defined as the curl of its velocity vec-
tor. That is, the vorticity o is defined by

o=V xu 2.4

In tensor notation, Equation 2.4 may be written in the form

ou;
w; = _Sijk 5 .
k

From this definition, it is evident that, by comparison with Equation 1.6a,
the vorticity vector is numerically twice the angular speed of rotation of the
fluid element about its own axes. That is, the vorticity is equal to twice the
antisymmetric part of the deformation-rate tensor e;. It should be noted that
a fluid element may travel on a circular streamline while having zero vortic-
ity. Vorticity is proportional to the angular velocity of a fluid element about
its principal axes, not the angular velocity of the center of gravity of the ele-
ment about some reference point. Thus, a particle traveling on a circular
streamline will have no vorticity, provided that it does not revolve about its
own center of gravity as it moves.

The vorticity contained in a fluid element is related to the circulation
around the element. This relationship may be obtained from an application of
Stokes’ theorem to the definition of circulation as follows. From Equation 2.3

F=fu-d1

=fA( xu)-ndA

where the contour integral has been converted to a surface integral by use
of Stokes” theorem, in which A is the area defined by the closed contour
around which the circulation is calculated and n is the unit normal to the
surface. Finally, invoking the definition of the vorticity vector, this relation-
ship becomes

r= f o-ndA. 2.5)

Equation 2.5 shows that, for arbitrary choices of contours and enclosing
areas A, if ® = 0, then I" = 0 and vice versa. Flows for which @ = 0 are called
irrotational, and flows for which this is not so are called rotational. The distinc-
tion between rotational and irrotational flow fields is an important one from
an analytic point of view, as will be seen in later chapters.
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2.3 Stream Tubes and Vortex Tubes

The concept of a streamline, which was introduced in an earlier section, may
be used to define a stream tube, which is a region whose sidewalls are made
up of streamlines. For any closed contour in a flow field, each point on the
contour will have a streamline passing through it. Then, by considering all
points on the contour, a series of streamlines that form a surface are obtained,
and this surface is called a stream tube. Figure 2.2a shows a length of stream
tube defined by a contour whose area is A;. The corresponding area at some
other section is shown as A,, where in general A, will be different from A,
and the shapes of the two cross sections of the stream tube will be different.
If the cross section of a stream tube is infinitesimally small, the stream tube
is usually referred to as a stream filament.

By analogy with streamlines and stream tubes, the useful concepts of
vortex lines and vortex tubes may be introduced. A vortex line is a line
whose tangents are everywhere parallel to the vorticity vector. Then, for
any closed contour in a flow field, each point on the contour will have a
vortex line passing through it, and the series of vortex lines defined by the
closed contour form a vortex tube. Figure 2.2b shows a length of vortex tube
defined by a contour whose area is A;. The cross-sectional area and shape
at any other section of the vortex tube will, in general, be different. A vortex
tube whose area is infinitesimally small is usually referred to as a vortex
filament.

FIGURE 2.2
(a) Stream tube and (b) vortex tube subtended by a contour of area A, in a flow field.
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2.4 Kinematics of Vortex Lines

Certain properties of flow lines may be established by studying the kine-
matics of vortex lines. The results so obtained form part of what is some-
times referred to as the Helmholtz theorems of vorticity. The other parts of the
Helmholtz theorems involve the dynamics of vorticity, which will be taken
up in the next chapter.

Equation 2.4 defines the vorticity vector as the curl of the velocity vector.
Since the divergence of the curl of any vector is identically zero, it follows
that

V-o=0.

Since the vorticity vector is divergence-free, it follows that there can be no
sources or sinks of vorticity in the fluid itself. That is, vortex lines must either
form closed loops or terminate on the boundaries of the fluid. The boundar-
ies of the fluid may be either a solid surface or a free surface.

The fact that the vorticity vector is divergence-free leads to an analogy
with the flow of an incompressible fluid. In this analogy, the counterpart of
the velocity vector is the vorticity vector, and the counterpart of the volume
tflow rate is the circulation. To establish this analogy, a sequence of opera-
tions will be performed first on the velocity vector for an incompressible
flow field and then on the vorticity vector.

The continuity equation for an incompressible fluid is

V-u=0.

Integrating this expression over some volume V gives

fv udV =0.

By use of Gauss’ theorem, this volume integral may be converted to the
equivalent surface integral
f u-nds=0
S

where the surface s encloses the volume V. Now, consider the surface s to be
the entire outer surface of an element of a stream tube or stream filament, as
shown in Figure 2.2a, including the ends. Then, since u - n = 0 on the walls of
the stream tube by definition, it follows that

f u-nds+f u-nds=0.
A Ay
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Since n is defined as the outward unit normal,

f u-nds =-Q,
A

and

fAzu-nds=Q2

where Q; is the volume flow rate crossing the area A,, and Q, is the volume
flow rate crossing the area A,. That is, the fact that the vector u is divergence-
free leads to the result

Qi=0Q,

which states that the volume of fluid crossing the area A; per unit time is
equal to that crossing the area A, per unit time. Since the fluid was assumed
to be incompressible, this result appears intuitively obvious.

Turning now to the vorticity vector, it was shown that

V-o=0

so that

fv ‘odV =0

fm-nds=0

where the surface s enclosed the volume V. Now, consider the surface s to be
the entire outer surface of an element of a vortex tube or a vortex filament
as shown in Figure 2.2b, including the ends. By definition of the vortex lines
that make up the surface of the vortex tube, ® - n = 0 on the walls of the vor-
tex tube. Then,

and

o-nds + o-nds = 0.
Ay Ay

However, from Equation 2.5

®-nds=-T,
A
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and

o-nds=T,.
Ay

Hence, the fact that @ is divergence-free results in the condition
Ir,=r,.

That is, the circulation around the limiting contour of the area A, is equal
to that around A,. Alternatively, this result may be stated in the form that
the circulation at each cross section of a vortex tube is the same. This means
that if the cross-sectional area of the vortex tube increases, the average value
of the vorticity across that section must decrease, just as the average velocity
would decrease to satisfy continuity. In fact, the result I') = I', may be put in
the form of the simple, one-dimensional continuity equation. If ®; denotes
the average vorticity across the area A; and o, denotes that across A,, the
result

o-nds + w-nds=0
Ay Ay

becomes
0,4, = 0,A,. (2.6)

The fact that the vorticity vector o is divergence-free means that vortex
tubes must terminate on themselves, at a solid boundary or at a free surface.
Smoke rings terminate on themselves, while a vortex tube in a free surface
flow over a solid boundary may have one end at the solid boundary forming
the bottom and the other end at the free surface.

PROBLEMS

2.1 Itis required to reproduce the graph shown in Figure 2.1 of the text.
The curves should be drawn from a sufficiently large number of cal-
culated points that a smooth curve is obtained in each case. Print
your graph in landscape mode and adjust its size so that the size of
one unit in the x direction corresponds to the size of one unit in the y
direction. Also, make the range and the grid lines in both the x and
y directions correspond to those presented in Figure 2.1 of the text
and make the diagram “fill” a regular-size sheet of paper.

2.2 Consider the two-dimensional flow field defined by the following
velocity components:
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u=—— v=1 w=0.

For this flow field, find the equation of
(@) The streamline through the point (1,1) at t =0
(b) The pathline for a particle released at the point (1,1) at t =0
(c) The streakline at t = 0, which passes through the point (1,1)
2.3 Atwo-dimensional flow field has the following velocity components:

u=x1+1t v=1 w =0.

Determine the following quantities for this flow field:

(@) The equation of the streamline that passes through the point (1,1)
asseen at t = 0.

(b) The equation of the pathline for a particle released at the point
(1,1) at time t = 0.

() The equation of the streakline that passes through the point (1,1)
asseenatt=0.

(d) The density at time ¢ = 0 on a particular streamline in this flow
has the value p,, which is a constant. Find an expression for the
density p at any subsequent time ¢ on the same streamline.

2.4 Show that the streamlines and particle paths coincide for the follow-
ing flow field for values of i = 1, 2, and 3:

u;=x; /(1 +1).
2.5 The velocity components for a particular flow field are as follows:
u=16x*+y v=10 w=yz2

(@) Determine the circulation, I, for this flow field around the
following contour by integrating the velocity around it

counterclockwise:
0<x<10 y=0
0<y<5 x=10
102x>0 y=5

52y20 x=0.
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(b) Calculate the vorticity vector, @, for the given flow field and

hence evaluate:
f o-ndA
A

where A is the area of the rectangle defined in (a) and n is the unit
normal to that area. Compare the result obtained in (b) with that
obtained in (a).

2.6 Consider the two-dimensional velocity distribution defined as follows:

X
U=- U= L

x2+y2 x2+y2'

Determine the circulation for this flow field around the following
contour by integrating around it counterclockwise:

-1<x<+1 y=-1
-1<y<+l x=+1
+1>x>-1 y=+1
+H12y>-1 x=-1.

2.7 A particular three-dimensional flow field has the following velocity
components:

u=9x2+2y v=10x w = -2yz2.

(@) Using the same contour as defined in Problem 2.6 on the plane
z =5, determine the circulation for the given flow field.

(b) Calculate the vorticity vector for the given flow field at any point
(xy) in the plane z = 5.

(o) Using the value obtained in (b) for the vorticity vector ® in the
plane z = 5, evaluate the following integral:

wa-ndA.

In the above, A is the area of the rectangle defined in (a), and n is
the unit normal to that area in the positive z direction. Compare the
result obtained in (c) with that obtained in (a).
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2.8 The velocity components for a particular two-dimensional flow field

29

210

are defined as follows:

Yy X

U=- v= .
X +y x* +y?

(@) Using the same contour as defined in Problem 2.6, determine the
circulation for the given flow field.

(b) Calculate the vorticity vector for the given flow field at any point
(y)-

(o) Calculate the divergence of the velocity vector for the given flow
field at any point (x,y).

Consider the two-dimensional velocity distribution of a fluid defined
as follows:

u=oy v = fx.

In the above, a and p are both constants.

(@) Determine the circulation I for this flow field around the same
contour as in Problem 2.6 by integrating in the counterclockwise
direction.

(b) Evaluate the following integral for this flow field:

fAm-ndA.

(o) Find the equations of the streamlines for the given flow field in
terms of an unspecified constant of integration.

(d) For @ = -1 and p = +1, determine the equation of the streamline
that passes through the point (1,0).

() For a = p = 1, determine the equations of the streamlines that
pass through the origin, that is, the point (0,0).

Calculate the vorticity at any point (R,0) for each of the following
two-dimensional flow fields:

@ ug=0, uy=wR
(b) ug =0, uy=1/27R

In the above, R and 0 are cylindrical coordinates while ® and I" are
constants.
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Special Forms of the Governing Equations

Some alternative forms of the governing equations, as derived in Chapter 1,
will be discussed here. The results are all obtained from the governing equa-
tions under various degrees of approximation such as negligible viscous
effects. Some of the results are frequently referred to as theorems. They are
used either as alternatives to the general equations derived in Chapter 1, under
the specified restrictions, or as supplementary information to these equations.

The first result established is Kelvin's theorem. This theorem establishes
the conditions under which irrotational motion remains irrotational and
so justifies the simplifying methods of analysis that are utilized for irrota-
tional flows. Then, the Bernoulli equations are derived. These equations are
integrals of the Euler equations under certain conditions. They are used to
relate the pressure and velocity fields when the velocity is established sepa-
rately from, for example, the condition of irrotationality. Crocco’s equation
is derived next. This equation relates the entropy of the fluid to the vorticity
and shows that, under certain conditions, isentropic flows are irrotational,
and vice versa. Finally, the vorticity equation is derived for a fluid of constant
density and viscosity. This equation is useful in the study of rotational flows.

3.1 Kelvin’s Theorem

This theorem states that for an inviscid fluid in which the density is constant,
or in which the pressure depends on the density alone, and for which any
body forces that exist are conservative, the vorticity of each fluid particle will
be preserved. Kelvin’s theorem covers the remainder of the Helmholtz theo-
rems of vorticity that were not treated in Section 2.4 during the discussion of
the kinematics of vortex lines. Although Kelvin’s theorem appears to be kine-
matic in nature, the dynamic equations of motion are required in the proof.

Suppose that a body force f; per unit mass that may act on the fluid is con-
servative, such as gravity for example. Then, f; may be written as the gradient
of some scalar function G, giving

_ G
axj'

fi

57
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Then, from Equation 1.9¢, the equations of motion for an inviscid fluid sub-
jected to only conservative body forces are

ou; ou;
P pou, o 06

P ot X, 0x; 0x;

or, in terms of the material derivative,

Duj 1o oG

s + )
Dt p ox; ox;

It is this form of the momentum equation, which is valid for an inviscid
fluid subjected to only conservative body forces, that will be used to prove
Kelvin’s theorem.

In order to determine the rate of change of vorticity associated with a given
fluid element, the material derivative of the circulation I" will be calculated.
From Equation 2.3

DI D
—=—Qu;dx
Dt DtJ '
Du; D(dx
=§ —Ldx; +u (dx;)
Dt /7 Dt

The quantity D(dx;)/Dt is the material derivative of an element dx; of the
contour around which the circulation is to be calculated. Its value may be
established as follows:

D(dxj)= % =d %+u % =du,.
Dt Dt at - Fox, /

Here, the material derivative has been converted into its Eulerian equiva-
lent, using Equation 1.1, in which ¢ and the spatial coordinates are indepen-
dent. Thus, ax,./ ot = 0 and ax,./ ox, = 8, which is zero unless k = j, at which
time its value is unity. This shows that the value of Dx;/Dt is u; and hence the
value of D(dx))/Dt is du,. In this way, the expression for the rate of change of
circulation becomes

Dr_ o Dy,

b ) o
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The quantity Du;/Dt will now be eliminated from this expression by using
the momentum equations that were derived above for an inviscid fluid in
which any body forces were conservative. Thus, the rate of change of circula-
tion becomes

br =§ —la—pdxj +de]- +u;du,
Dt p ox; ax;

dp 1
=56 —?+dG+§d(ujuj)

where it has been observed that (dp/dx)dx; = dp, which is the total spatial
variation of p, and likewise (BG/dx]-)dxj = dG. It is now observed that, since
the integration is to be carried out around a closed contour, the integral of
dG and that of d(uu) are both zero since the body force and the velocity are
both assumed to be single valued. Then,

Now, if p is a constant, the remaining integral is zero for the same reason
that the other integrals were zero. However, this integral is zero under less
restrictive conditions also. Suppose the pressure p may be considered to be a
function of the density p only as, for example, in isentropic flows. Then, for
some function g,

p=g(p)

so that

dp = g'(p)dp.

The expression for DI'/Df now becomes
Dbr _ _95&@_
Dt p

That is, this integral falls into the same category as the two previous inte-
grals, and its value around any closed contour is zero. This gives the result
known as Kelvin’s theorem:

—=0. 3.1
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Equation 3.1 says that if we follow a given contour as it flows, the total
vorticity inside that contour will not change. Recall that the right-hand side
of Equation 3.1 could be proved to be zero by considering the fluid to be
inviscid, the body forces to be conservative, and either the density to be con-
stant or the pressure to be a function of the density only. Relaxing any of
these conditions leads to, in general, a nonzero term on the right-hand side of
Equation 3.1. Thus, it may be deduced that the total vorticity may be changed
by the action of viscosity, the application of nonconservative body forces, or
density variations that are not simply related to the pressure variation.

It should be noted that Equation 3.1 applies to a simply connected region.
That is, for any closed contour in the fluid that contains only fluid, there will
be some definite value of the circulation I Equation 3.1 asserts that under the
conditions specified in the derivation, the value of I will not change around
that contour even though the contour itself may be deformed by the flow. A
closed contour that originally does not include a body cannot at any subse-
quent time contain a body such as a two-dimensional airfoil. There is there-
fore no conflict in the fact that such an airfoil may have a circulation around
it while immersed in an irrotational flow.

From Kelvin’s theorem and the results established in Section 2.4, it is evi-
dent that the total vorticity associated with a vortex filament is fixed and
will not change as the vortex filament flows with the fluid. Distortion of the
vortex filament may take place, but the total vorticity associated with it will
remain the same. The vortex filament will always consist of the same fluid
particles as it flows, and if the vortex filament is elongated, the vorticity at
any section of the filament will increase and the total vorticity associated
with the filament will remain fixed.

The principal use of Kelvin's theorem is in the study of incompressible,
inviscid fluid flows. If a body is moving through such a fluid, or if a uniform
flow of such a fluid passes around a body, then the vorticity far from the
body will be zero. Then, according to Kelvin's theorem, the vorticity in the
fluid will everywhere be zero, even adjacent to the body. Then, the condition
V x u =0 may be used to replace the Euler equations so that the condition of
irrotationality becomes the alternative form of the equations of motion for
the fluid. Again, it is emphasized that this kinematic equivalent is valid only
because of Kelvin’s theorem, and in turn, the Euler equations were used to
prove Kelvin’s theorem.

3.2 Bernoulli Equation

For an inviscid fluid in which all body forces are conservative and either the
flow is steady or itis irrotational, the equations of momentum conservation may
be integrated to yield a single scalar equation called the Bernoulli equation.



Special Forms of the Governing Equations 61

In the previous section, it was pointed out that the equations of motion for
an inviscid fluid in which all body forces are conservative could be written
in the form

ou; ou; d oG
pj+puk4=—i+pf-

Using a vector identity given in Appendix A, the second term on the left-
hand side of these equations may be rewritten as follows:

auj 1
——=(u* )u= —u'u -ux( xu)

u
k ax, 2

= —uu-uxo.
2

In this way, the Euler equations may be written in the following vector form:

du 1 1
—+ —uu -uxw=-— p+ G
ot 2 P

It is now proposed to show that the term (1/p)Vp, which appears on the

right-hand side of this equation, may be written as V f dp/p . To do this, we

form the scalar product of an element of a space curve d/, such as an element
of a streamline, with the vector quantity (1/p) Vp:

de-

p =—deé- p=1dp.
p

|-
D | =

Here, the result d¢ - V = dx(d/dx) + dy(d/9y) + dz(9/9z) = d has been used,
where the scalar operator d is the total spatial derivative. Then, using d and
its inverse integral operation, it follows that

where, again, the equivalence of d and d{ - V has been used. The vectors that
form the scalar product with d?f in this last equation must be equal since df
was arbitrarily chosen; hence, it follows that

-

1
p p
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Using this result, the Euler equations become

a—u+ f%+lu-u—G =uxm. (3.2a)
ot p 2

The vector Equation 3.2a may be integrated for steady flow and for unsteady
or steady irrotational flow.

Considering first steady flow, Equation 3.2a becomes

%+lu-u—G =uxm.
p 2

Forming the scalar product of the velocity vector u with this equation gives
u- fdp+1u-u—G =u-(uxw).
p 2

However, the vector product of u with @ will yield a vector that is perpen-
dicular to u; hence, the quantity u - (u x ®) is zero. Furthermore, the operator
u - Vis the steady-state form of the material derivative. Thus, the preceding
equation states that as we flow along a streamline in steady flow, the quan-

tity f dp/p+u-u/2 - G remains constant. That is,
dp 1 .
f —+ Eu -u - G = constant along each streamline. (3.2b)
p

This result is referred to as the Bernoulli integral or the Bernoulli equation. It
should be recalled that it is valid for the steady flow of a fluid in which viscous
effects are negligible and in which all body forces are conservative. In many
cases, the flow around some body originates in a uniform flow, and in such
cases, and in some other cases, the constant on the right-hand side of Equation

3.2b is the same for each streamline. Then, the quantity [ dp/p+u-u/2-Gis

constant everywhere. The constant is usually referred to as the Bernoulli constant.

Equation 3.2a may also be integrated under slightly different circum-
stances from those that led to Equation 3.2b. Rather than considering steady
flows, consider irrotational flows. Then, the vorticity @ will be zero so that
Equation 3.2a becomes

6—u+ f%+lu-u—G =0.
ot p 2

Now, since ® =V x u =0, it follows that the velocity vector u may be written as
the gradient of some scalar, for example, ¢, since V x V¢ = 0 for any function ¢.
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The quantity ¢ is known as the velocity potential, and it will be used extensively
in Chapter 4. Then, replacing u by V¢ in the preceding equation gives

) dp 1
—+[—=+= ¢ ¢-G =0.
ot +fp +2 L

Forming the scalar product of this vector equation with an element of a
space curve df gives

W, o, 1 o
d8t+fp+2¢¢c_0

where again the fact that df - V = d has been used, where d is the total spatial
derivative. Thus, integration yields

L e
8t+fp+2 ¢ ¢-G=F(t) (3.20)

where F(f) is some function of time that may be added after integrating over
the space coordinates. F(f) is usually referred to as the unsteady Bernoulli con-
stant, even though it is not strictly a constant. Recall that Equation 3.2c is
valid for irrotational motion of a fluid in which viscous effects are negli-
gible and in which all body forces are conservative. Kelvin’s theorem usually
helps to verify the validity of the condition of irrotationality by relating the
flow under consideration to a simpler form of the flow far upstream.

3.3 Crocco’s Equation

This equation relates the vorticity of a flow field to the entropy of the fluid.
Under certain conditions, it will be shown that isentropic flows are irrota-
tional, and vice versa. Then, if it is known that a flow field is essentially isen-
tropic, the mathematical simplifications associated with irrotational motion
may be employed. This simplification will be employed in the chapters deal-
ing with compressible fluid flow, and it is justified by Crocco’s equation.

In order to establish Crocco’s equation, consider the flow of an inviscid
fluid in which there are no body forces. Then, from Equation 19¢, the Euler
equations that guarantee dynamic equilibrium, become

E+(u' )u=—l p.
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The nonlinear term may be expanded as follows using a vector identity
given in Appendix A:

(u Jus= %u-u —ux( xu).

Hence, the Euler equation becomes

ou 1
— 4+ —u'u -uUxX®=-—

ot 2 P

° |-

It is this form of the Euler equation that is the starting point for the deriva-
tion of Crocco’s equation. In order to relate the dynamics of the flow to its
thermodynamics, it is proposed to eliminate the pressure p and the density
p, which appear in the term on the right-hand side of the above equation,
in favor of the enthalpy / and the entropy s. To do this, we use the first law
of thermodynamics and the definition of the entropy. From Appendix E, a
change in internal energy de is caused by work done on the fluid —pd(1/p)
and by any heat that is added to the fluid dg. That is,

de=-pd +dg

+T ds

Tk T~

=-pd

where the last relation follows from the definition of the entropy. Now p and
p have been related to e and s. In order to eliminate e in favor of the enthalpy
h, we use the equation that defines the enthalpy, namely, e = & — p/p. Then, the
foregoing thermodynamic relation becomes

dh-d =-pd +T ds.

oI
D | =

Since d(p/p) = pd(1/p) + dp/p, this equation simplifies to
1
-—dp=T ds-dh.
p

Using again the result established in the previous section that df - V =d, it
follows that

1
-— p=T s- h.
0 p
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This result will now be used to eliminate the pressure and the density that
appear on the right-hand side of the Euler equations:

8—u+ lu-u —uxw=T s- h
ot 2

Rearranging this vector equation slightly yields the result known as
Crocco’s equation:

uxw+T s= h+lu-u +a—u. (3.3a)
2 ot

Equation 3.3a is valid for flows in which viscous effects are negligible and in
which there are no body forces.

Under conditions of steady, adiabatic flow, Equation 3.3a may be reduced to
a scalar equation. To show this, it will first be shown that for adiabatic flow
of an inviscid fluid in which there are no body forces, the quantity i, = h +
u - u/2 is constant along each streamline. The quantity h, is called the stagna-
tion enthalpy.

From Problem 3.1, the energy equation for adiabatic flow of an inviscid
fluid is

D Dp

"Dt "D
The Euler equations for a flow without body forces are

Du __
Por =~ F

Forming the scalar product of this equation with the velocity vector u gives

D1
Par 2 P

Adding this equation to the energy equation derived above yields

D 1 Dp
p— h+-uu =——-up
Dt 2 Dt
or
Dh, dp

Dt ot
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Then, for steady flow, the right-hand side of this equation will be zero. That
is, for steady, adiabatic flow of an inviscid fluid in which there are no body
forces, the quantity Dh,/Dt will be zero. Hence, the stagnation enthalpy h,
will be constant along each streamline.

Equation 3.3a was derived for an inviscid fluid that is not subjected to any
body forces. Then, if, in addition, the flow is steady and adiabatic, Equation
3.3a becomes

ux o+ T Vs=Vh,

where the quantity /4, is constant along each streamline. Hence, Vi, will be a
vector perpendicular to the streamlines. However, u x ® is also perpendicu-
lar to the streamlines, so that the remaining vector, namely, T Vs, must also
be perpendicular to the streamlines. Then, the above vector equation may be
written in the following scalar form:

u +T§=d—h°. (3.3b)
dn dn
Here, U and Q are, respectively, the magnitudes of the velocity vector u
and the vorticity vector . The coordinate 7 is perpendicular to the stream-
lines locally. Equation 3.3b is valid for steady, adiabatic flow of an inviscid
fluid in which there are no body forces.
Usually when the stagnation enthalpy is constant along each streamline, it
is constant everywhere. That is, the value of /1, along each streamline is the
same. Under these conditions, d/,/dn = 0, so that Equation 3.3b becomes

u +t¥_o (3.30)
dn

In this form, Crocco’s equation clearly shows that if s is constant, Q must
be zero. Likewise, if Q is zero, ds/dn must be zero so that s must be constant.
That is, isentropic flows are irrotational and irrotational flows are isentropic.
This result is true, in general, only for steady flows of inviscid fluids in which
there are no body forces and in which the stagnation enthalpy is constant.

3.4 Vorticity Equation

The equation to be satisfied by the vorticity vector ® for a fluid of constant
density and constant viscosity will be derived in this section. Such an equa-
tion is useful in the study of viscous flows in incompressible fluids, which
is the topic of Part III of this book. One reason that the vorticity equation
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is of interest is that it enables us to learn more about the physics of given
flow fields. Also, in the analysis of some flow fields, it is frequently possible
to make some statement about the vorticity distribution that facilitates the
analysis if the problem is posed in terms of the vorticity.

From Equation 1.9b, the Navier-Stokes equations for a fluid of constant
density and viscosity and for which relevant body forces are conservative
are as follows:

a—u+(u- Ju=- Py tus G

ot

The quantity v = p/p is the kinematic viscosity. Replacing the nonlinear term
by its equivalent form given by the vector identities in Appendix A, this vec-
tor equation becomes

ad 1
au —u'u —ux( xu)=- +v ‘u+ G.

+ P
ot 2 p

The vorticity equation is obtained by taking the curl of this equation and
noting that the curl of the gradient of any scalar is zero. Hence,

— - x(uxw)=v ‘o.
Using a vector identity given in Appendix A, the second term on the left-
hand side may be expanded to give
Vxuxo)=uV-0o)-o(V-u)-u-V)o + (®- V)u
However, V - ® = 0, since the divergence of the curl of any vector is zero and
V- u =0 from the continuity equation. Hence, the vorticity equation becomes

—+u Jo=(w Ju+v ‘o (34a)

For two-dimensional flows, the vorticity vector ® will be perpendicular to
the plane of the flow, so that (® - V)u will be zero. Then,

aa—(;)+(u‘ w=v ‘0. (3.4b)

The vorticity equation, in either the general form (Equation 3.4a) or the
two-dimensional form (Equation 3.4b), has another advantage over and
above those mentioned in the preliminary remarks. It will be noted from
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these equations that the pressure p does not appear explicitly. Thus, the
vorticity and velocity fields may be obtained without any knowledge of the
pressure field.

In order to determine the pressure distribution in terms of the vorticity, the
Navier-Stokes equations are again used in the form

a—u+(u' u=-— Piv 2us G

Taking the divergence of this equation and using the result of Problem 3.2
together with the continuity equation V - u = 0, it follows that the equation to
be satisfied by the pressure p is

2 P =0-o+u-( 2u)—% 2(wru)+ *G. (3.5)
p

If the body force represents only gravity, the Laplacian of G will be zero.
From the foregoing results, we see that the vorticity satisfies an advection—
diffusion equation, whereas the pressure satisfies a Poisson equation
(Figures 3.1 and 3.2).

FIGURE 3.1

Results from the large eddy numerical simulation of a liquid jet in a gaseous cross flow show-
ing the deformation and breakup of the liquid jet. The jet velocity is 40 m/s, its viscosity is 10+
Pa-s, and its density is 14.4 kg/m?. The cross flow bulk velocity is 40 m/s, the gas viscosity is
1075 Pa-s, and its density is 1.2 kg/m?. (Courtesy of Professor Nasser Ashgriz, University of
Toronto, Toronto, Canada.)
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FIGURE 3.2

Results from numerical calculations showing droplet dynamics and spray formation from an
annular nozzle with outer and inner co-flowing gas. The droplets break up into smaller ones
due to the high velocity gaseous flow, generating a spray of smaller droplets. The outer gas
velocity is 140 m/s; the inner gas velocity, as well as the velocity of the droplets, is 40 m/s.
(Courtesy of Professor Nasser Ashgriz, University of Toronto, Toronto, Canada.)

PROBLEMS
3.1 In vector form, the thermal energy equation is

pg—i=—p ‘u+ (b T)+ .

By using the definition of the enthalpy /;, show that an equivalent
form of this equation is

bn_Dp “(k T)+®.

3.2 Show that, for an incompressible fluid, the following identity holds
between the velocity vector u and the vorticity vector e:

[(u- >u1=% 2(u-w)-u-( u)-0-o.

3.3 In cylindrical coordinates, the velocity components for the uniform
flow of an incompressible and inviscid fluid around a circular cylin-
der are
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aZ
ug =U 1—P cos 0

2

Uy = -U 1+% sin 6.

Here, U is the constant magnitude of the velocity approaching the
cylinder, and a is the radius of the cylinder. Determine the pressure
p(R, 6) at any point in the fluid in the absence of any body forces.
Take the pressure far from the cylinder to be constant and equal
to p,.

Specialize the result obtained above to obtain an expression for
the pressure p(a, 6) on the surface of the cylinder.

Further Reading—Part I

Part I of this book has been concerned with the derivation of the equations
governing the motion of a fluid. The number of books dealing with fluid
mechanics in which these equations are derived is large. The following rep-
resents a sample of some of these books.

Aris, R.: Vectors, Tensors, and the Basic Equations of Fluid Mechanics, Prentice-Hall, Inc.,
Englewood, N.J., 1964.

Batchelor, G. K.: An Introduction to Fluid Dynamics, Cambridge University Press,
London, 1967.

Chorin, A. J. and Marsden, J. E.: A Mathematical Introduction to Fluid Mechanics, 3rd ed.,
Springer-Verlag, Berlin, 1993.

Lagerstrom, P. A.: Laminar Flow Theory, in F. K. Moore (ed.): Theory of Laminar Flows,
Princeton University Press, Princeton, N.J., 1964.

Lamb, H.: Hydrodynamics, 6th ed., Dover, New York, 1932.

Landau, L. D. and Lifschitz, E. M.: Fluid Mechanics, 2nd ed., Pergamon Press, London,
1987.

Panton, R. L.: Incompressible Flow, John Wiley & Sons, New York, 1984.

Serrin, J.: Mathematical principles of classical fluid mechanics, in S. Fliigge (ed.):
Handbuch der Physik, vol. VIII/1, Springer-Verlag OHG, Berlin, 1959.

Sherman, F. S.: Viscous Flow, McGraw-Hill Book Company, New York, 1990.

Yih, C.-S.: Fluid Mechanics, McGraw-Hill Book Company, New York, 1969.



Part 11

Ideal-Fluid Flow

This part of the book deals with the flow of ideal fluids, that is, fluids that
are inviscid and incompressible. The results are therefore limited to flow
fields in which viscous effects of the fluid are negligible and compressibility
of the fluid is unimportant. Then, any phenomena that are predicted by the
governing equations will be due to the inertia of the fluid. The mathematical
simplification that results from neglecting viscous and compressible effects
is great, and consequently, the topic of ideal-fluid flow is, mathematically, the
best understood.

Part II contains Chapters 4, 5, and 6. Chapter 4 deals with two-dimensional
potential flows. Apart from some fundamental flows, the flow around some
two-dimensional bodies such as cylinders, ellipses, and airfoils is cov-
ered. Chapter 5 treats three-dimensional potential flows including the flow
around submerged bodies such as spheres. Finally, Chapter 6 deals with
surface waves on liquids. This chapter includes traveling waves, standing
waves, and waves at the interface of two fluids.

II.1 Governing Equations and Boundary Conditions

Since the fluid is assumed to be incompressible, the equation of mass con-
servation is Equation 1.3c. The equations of momentum conservation for an
inviscid fluid are the Euler equations, which are expressed by Equation 1.9¢.
That is, the equations governing the velocity and pressure fields for an ideal
fluid are
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V-u=0 (IL1)
1
—+(u- )u=—5 p+f. (IL.2)

Equations II.1 and IL.2 are sufficient to establish the velocity and the pres-
sure in the flow independent of any temperature distribution that may exist.
It was pointed out in Chapter 1 that compressibility is the fluid property that
couples the equations of thermodynamics to those of dynamics so that in the
study of ideal fluids, the equations of thermodynamics need not be solved
concurrently with the equations of motion. The study of ideal-fluid flows is
frequently referred to as hydrodynamics, and Equations II.1 and IL.2 are fre-
quently called the equations of hydrodynamics.

Within macroscopic length scales, the proper boundary condition to be
satisfied by the velocity is the no-slip boundary condition expressed by
Equation 1.14. It is not possible to satisfy this boundary condition with the
Euler equations. The reason lies in the fact that the Euler equations are one
order lower than the Navier-Stokes equations because the viscous terms
are absent in the former equations. Thus, the true boundary condition must
be relaxed somehow under the approximation of negligible viscous effects.
Since it is primarily viscous effects that prohibit a fluid from slipping along
a solid boundary, the condition of no tangential slip at boundaries is relaxed.
That is, the condition of no normal velocity at a solid boundary is retained,
but the condition of no tangential velocity is dropped. Thus, the boundary
condition that should be used with the Euler equations is

u-n=U:-n on solid boundaries (IL.3)

where n is the unit normal to the surface of the body, and U is the velocity vec-
tor of the body. Comparison of Equation I.3 with Equation 1.14 shows that the
former constitutes one component of the true boundary condition and the two
tangential components are unspecified. Physically, this means that the condi-
tion of no slip on a solid boundary has become the condition that the surface
of the body must be a streamline. Any boundary condition that is to be satis-
fied far from the body, such as the flow becoming uniform, is unaffected by
the inviscid approximation.

II.2 Potential Flows

If the flow of an ideal fluid about a body originates in an irrotational flow,
such as a uniform flow, for example, then Kelvin's theorem (Equation 3.1)
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guarantees that the flow will remain irrotational even near the body. That is,
the vorticity vector @ will be zero everywhere in the fluid. Then, since V x
V¢ = 0 for any scalar function ¢, the condition of irrotationality will be satis-
fied identically by choosing

u = Vé. (IL4)

The function ¢ is called the velocity potential, and flow fields that are irrota-
tional, and so can be represented in the form of Equation 1.4, are frequently
referred to as potential flows. In order to find the equation that the velocity
potential ¢ satisfies, the expression for u given by Equation 114 is substituted
into the continuity equation (Equation I1.1) to give

V2 = 0. (IL5)

Thus, by solving Equation IL.5 and utilizing Equation 1.4, the velocity field
may be established without directly using the equations of motion (Equation
I1.2). This is so because the condition of irrotationality has been used, and
this condition is justified by Kelvin’s theorem, which uses Equation I1.2 in its
proof. However, the equations of motion must be used directly to obtain the
pressure distribution. Solving Equation I1.5 for the velocity potential ¢ deter-
mines the velocity distribution only, and in order to determine the pressure,
use must be made of the equations of dynamics. Rather than use Equation
I1.2, its integrated form, that is the Bernoulli equation, will be used. Using
Equation 3.2¢, the pressure may be determined from the following relation:

. r, 1o e_Ge
6t+p+2 ¢ ¢-G=F(1). (IL6)

Having determined the velocity potential ¢, this becomes a simple alge-
braic equation for the pressure.

From the foregoing, it is evident that a simpler form of the governing equa-
tions exists for potential flows. Rather than solving Equations II.1 and II.2
directly, Equation IL5, together with the appropriate boundary conditions,
may be solved to yield the velocity potential and hence the velocity field.
Having done this, Equation II.6 may be used to establish the pressure field.
This formulation has certain simplifying features. First, it will be noticed
that the differential equation to be solved, given by Equation IL5, is linear,
whereas Equation II.2 is nonlinear. Of course, the nonlinearity cannot be
completed circumvented, and indeed, it appears in the term V¢ - V¢ in the
Bernoulli equation. However, in this equation, it poses no difficulty in the
analysis. One of the most useful properties of linear differential equations is
that different solutions may be superimposed to yield other solutions. This
property will be used extensively in the following chapters.
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Two-Dimensional Potential Flows

It was pointed out in the introduction to Part II that potential flows may
be analyzed in a much simpler way than general fluid flows. Within the
category of potential flows, the two-dimensional subset lends itself to even
greater simplification. It will be shown in this chapter that the simplification
is so great that solutions to Equations IL.5 and I1.6 may be obtained without
actually solving any differential equations. This is achieved through use of
the powerful tool of complex variable theory.

The chapter begins by introducing the stream function, which, together
with the velocity potential, leads to the definition of a complex potential.
Through this complex potential, some elementary solutions corresponding
to sources, sinks, and vortices are examined. The superposition of such ele-
mentary solutions then leads to the solution for the flow around a circular
cylinder. The method of conformal transformations is then introduced, and
the Joukowski transformation is used to establish the solutions for the flow
around ellipses and airfoils. The Schwarz—Christoffel transformation is then
introduced and used to study the flow in regions involving sharp corners.
Included in this chapter are examples of free-surface configurations.

4.1 Stream Function

The velocity potential ¢ was defined in such a way that it automatically sat-
isfied the condition of irrotationality. The continuity equation then showed
that ¢ had to be a solution of Laplace’s equation. A second function may be
defined by a complementary procedure for two-dimensional incompressible
fluid flows. That is, a function may be defined in such a way that it automati-
cally satisfies the continuity equation, and the equation it must satisfy will be
determined by the condition of irrotationality.

The continuity equation, in Cartesian coordinates, for the flow field under
consideration is

%+a—v=0.
ox oy

75
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Now, introduce a function y that is defined as follows:

u=v (@1a)
ay

SR (4.1b)
ox

With this definition, the continuity equation is satisfied identically for all
functions y. y is called the stream function, and by virtue of its definition, it is
valid for all two-dimensional flows, both rotational and irrotational.

The equation that the stream function y must satisfy is obtained from the
condition of irrotationality. Denoting the components of the vorticity vector
o by (&, 1, §), it is first observed that, in two dimensions, the only nonzero
component of the vorticity vector is {, the component perpendicular to the
plane of the flow. Secondly, it is noted that { = dv/dx — du/dy. Thus, the condi-
tion of irrotationality is

dv 0
oo
ox dy

Substituting for u and v from Equation 4.1 shows that y must satisfy the
following equation:

2 2
27‘5 . aay“’ - 0. (4.2)

That is, the stream function v, like the velocity potential ¢, must satisfy
Laplace’s equation. The stream function y has some useful properties that
will now be derived.

The flow lines that correspond to y = constant are the streamlines of the
flow field. To show this, it is noted that y is a function of both x and y in
general so that the total variation in y associated with a change in x and a
change in y may be calculated from the expression

np o
dyp=""dx+d
v ox ¥ ay Y

=-vdx+udy
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where Equation 4.1 has been used. Then, the equation of the line y = constant
will be

0=-vdx+udy
or
dy 2
dv  u
v

where the subscript denotes that this expression for dy/dx is valid for y held
constant. However, it was shown in Chapter 2 that this is precisely the equa-
tion of the streamlines in the xy plane. Hence, the lines corresponding to
y = constant are the streamlines, and each value of the constant defines a
different streamline. It is this property of the function y that justifies the
name stream function.

Another property of the stream function  is that the difference of its
values between two streamlines gives the volume of fluid that is flowing
between these two streamlines. To show this, consider two streamlines cor-
responding to y =y, and y = y, as shown in Figure 4.1. A control surface AB
of arbitrary shape but positive slope is shown joining these two streamlines,
and an element of this surface shows the positive volumetric flow rates cross-
ing it in the x and y directions per unit depth perpendicular to the flow field.
Then, the total volume of fluid flowing between the streamlines per unit
time per unit depth of flow field will be

FIGURE 4.1
Two streamlines showing the components of the volumetric flow rate across an element of
control surface joining the streamlines.
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B B
Q =fAudy —fAvdx.

However, it was observed earlier that dy = —v dx + u dy, so that, integrating
this expression between the two points A and B, it follows that

VP, -, =—fjvdx+fjudy.

Comparing these two expressions confirms that y, — y; = Q.

Finally, it should be noted that the streamlines y = constant and the lines
¢ = constant, which are called equipotential lines, are orthogonal to each other.
This may be shown by noting that if ¢ depends upon both x and y, the total
change in ¢ associated with changes in both x and y will be

99 99
dp=—"dx+_—d
¢ ax x+ay Y

= udx + vdy

where Equation I1.4 has been used. Then, the lines corresponding to ¢ = con-
stant will be defined by the following relation:

0 =udx + vdy
or
dy __u
dx . v
That is,
dy 1

dv - (dy/dx),

In words, the slope of the line ¢ = constant is the negative reciprocal of the
slope of the line y = constant, so that these sets of lines must be orthogonal.
This property of the streamlines and the equipotential lines is the basis of a
numerical procedure for solving two-dimensional potential-flow problems.
The method is referred to as the flow net.
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4.2 Complex Potential and Complex Velocity

The velocity components # and v may be expressed in terms of either the
velocity potential or the stream function. From Equations 1.4 and 4.1, these
expressions are

)
ox  dy
0200
ay ax

That is, the functions ¢ and y are related by the expressions

% _
ox oy
% __
dy  ox

But these will be recognized as the Cauchy—Riemann equations for the
functions ¢(x, y) and y(x, y). Then, consider the complex potential F(z), which
is defined as follows:

F(z) = d(xy) + iw(xy) @.3)

where z = x + iy. Now, if F(z) is an analytic function, it follows that ¢ and y
will automatically satisfy the Cauchy—Riemann equations. That is, for every
analytic function F(z), the real part is automatically a valid velocity potential,
and the imaginary part is a valid stream function.

The foregoing result suggests a very simple way of establishing solutions
to the equations of two-dimensional potential flows. By equating the real
part of a given analytic function to ¢ and the imaginary part to y, the theory
of complex variables guarantees that V2¢ = 0 and V?y = 0 as required. The
flow field corresponding to that analytic function may be determined by
studying the streamlines y = constant. The corresponding velocity compo-
nents may be calculated from Equation I1.4 or Equation 4.1, and the pressure
may be obtained using Equation I1.6. This approach has the disadvantage
of being inverse in the sense that a problem is first solved and then exam-
ined to see what the physical problem was in the first place. However, for
teaching purposes, this is of no consequence. Another disadvantage is that
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the method cannot be generalized to three-dimensional potential flows. On
the other hand, this approach avails itself of the powerful results of com-
plex variable theory and avoids the difficulties of solving partial differential
equations. For these reasons, the complex-potential approach will be used in
this chapter.

Another quantity of prime interest, apart from the complex potential F(z),
is the derivative of F(z) with respect to z. Since F(z) is supposed to be analytic,
dF/dz will be a point function whose value is independent of the direction
in which it is calculated. Then, denoting this derivative by W, its value will
be given by

W(z)=£=£
dz ox
=@+ia—w
ax ox
that is,
dF
WE)=—=u-1i 44
(2) - Uu-1iv 4.4

where Equations 4.3, 11.4, and 4.1b have been used. In view of this result, the
quantity W(z) is called the complex velocity, although its imaginary part is —iv.
Equation 4.4 offers a convenient alternative to the use of Equations 1.4 and
4.1 for finding the velocity components that correspond to any particular
complex potential.

A useful property of the complex velocity is that, when multiplied by its
own complex conjugate, it gives the scalar product of the velocity vector with
itself. To show this, consider W(z) and its complex conjugate W(z). Then,

WW = (u - iv)(u + iv) @.5)
WW = u* + 07,

The significance of this result is that the quantity u - u = V¢ - Vi = 12 + 2
appears in the Bernoulli equation.

Frequently, it is advantageous to work in cylindrical coordinates rather
than Cartesian coordinates. An expression for the complex velocity may be
readily obtained in cylindrical coordinates by converting the Cartesian com-
ponents of the velocity vector (1, v) to cylindrical components (uy, 1). Figure
4.2 shows a velocity vector OP decomposed into its Cartesian components
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FIGURE 4.2
Decomposition of velocity vector OP into its Cartesian components (1, v) and its cylindrical
components (ig, U).

(shown solid) and also its cylindrical components (shown dotted). From this
figure, each of the Cartesian velocity components may be expressed in terms
of the two cylindrical components as follows:

U .
U = Uy cos O + u, cos 5—6 = Uy cosO —1u,sind
. . .
U = Uy SIN O + U, sin 5—6 = U Sin 0 + 1, cos 0.

Substituting these expressions into Equation 4.4 gives the expression for
the complex velocity W in terms of uy and

W = (ug cos0 — u, sin 0) — i(u sin 6 + 1, cos 0)
= Uy (cos O — isin B) — i1, (cos O — i sin O)
that is,
W = (ug — iuge™™. 4.6

The foregoing results (Equations 4.3 through 4.6) are sufficient to establish
the flow fields, which are represented by simple analytic functions.
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4.3 Uniform Flows

The simplest analytic function of z is proportional to z itself, and the corre-
sponding flow fields are uniform flows.

First, consider F(z) to be proportional to z where the constant of propor-
tionality is real. That is,

F@z)=cz
where c is real. Then, from Equation 4.4,
WiE)=u-iv=c.

Then, by equating real and imaginary parts of this equation, the velocity
components corresponding to this complex potential are

u=c
v=0.

But this is just the velocity field for a uniform rectilinear flow as shown in
Figure 4.3a. Thus, the complex potential for such a flow whose velocity mag-
nitude is U in the positive x direction will be

F(z) = Uz. 4.7a)

Next, consider the complex potential to be proportional to z with an imagi-
nary constant of proportionality. Then,

F(z) = —icz

where c is real. The minus sign has been included to make the velocity com-
ponent positive when c is positive. For this complex potential,

W(z) =u —iv = —ic

() (b) (©
y y y

- L

FIGURE 4.3
Uniform flow in (a) x direction, (b) y direction, and (c) angle o to x direction.
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so that the velocity components are

This is a uniform vertical flow as shown in Figure 4.3b. Then, the complex
potential for such a flow whose velocity magnitude is V in the positive y
direction will be

F(z) = -iVz. 4.7b)

Finally, consider a complex constant of proportionality so that

F(z) = ce*z
where ¢ and o are real. For this complex potential,
W(z) = u — iv = ¢ cos o — ic sin .
Hence, the velocity components of the flow field are

Uu=ccosuo

v=csina
This corresponds to a uniform flow inclined at an angle « to the x axis as
shown in Figure 4.3c. Hence, the complex potential for such a flow whose
velocity magnitude is V will be

F(z) = Ve iz, 4.70)

This last result, of course, contains the two previous results as special cases
corresponding to « = 0 and a = n/2.

4.4 Source, Sink, and Vortex Flows

Complex potentials that correspond to the flow fields generated by sources,
sinks, and vortices are obtained by considering F(z) to be proportional to log
z. When considering log z, we consider the principal part of this multivalued
function corresponding to 0 < 0 < 2.
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Consider, first, the constant of proportionality to be real. Then,
F(z)=clogz
= clog Re®
= clog R + ic0.

Hence, from Equation 4.3,

¢=clogR
Y = ch.

That is, the equipotential lines are the circles R = constant, and the stream-
lines are the radial lines 6 = constant. This gives a flow field as shown in
Figure 4.4a in which the streamlines are shown solid and the direction of
the flow is shown for ¢ > 0. The direction of the flow is readily confirmed
by evaluating the velocity components. In view of the geometry of the flow,
cylindrical coordinates are preferred, so that

-0

W(z) =

e

N o
e

Comparison with Equation 4.6 shows that the velocity components are

FIGURE 4.4
Streamlines (shown solid) and equipotential lines (shown dashed) for (a) source flow and
(b) vortex flow in the positive sense.
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The flow field indicated in Figure 4.4a is called a source. The velocity is
purely radial, and its magnitude decreases as the flow leaves the origin. In
fact, the origin is a singular point corresponding to infinite velocity, and as
the fluid flows radially outward, its velocity is decreased in such a way that
the volume of fluid crossing each circle is constant, as required by the conti-
nuity equation.

Sources are characterized by their strength, denoted by m, which is defined
as the volume of fluid leaving the source per unit time per unit depth of the
flow field. From this definition, it follows that

27
m =f uzRdo
0

2

),

Here, the result u = ¢/R has been used. Then, ¢ may be replaced by m/2x,
giving the following complex potential for a source of strength m:

cdO = 2mc.

F(z) = %1ogz.

The source corresponding to this complex potential is located at the origin,
the location of the singularity. Then, the complex potential for a source of
strength m located at the point z = z, will be

F(z) = %log(z _z). .8)

Clearly, the complex potential for a sink, which is a negative source, is
obtained by replacing m by —m in Equation 4.8.

Now, consider the constant of proportionality in the logarithmic complex
potential to be imaginary. That is, consider

F(z) = —iclog z

where c is real and the minus sign is included to give a positive vortex, as
will be seen. Then, using cylindrical coordinates,

F(z) = —iclog Re"

=c6-iclogR.
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Then, from Equation 4.3, the velocity potential and the stream function are

¢=cO
= -clogR.
That is, the equipotential lines are the radial lines 6 = constant, and the

streamlines are the circles R = constant as shown in Figure 4.4b. The velocity
components may be evaluated by use of the complex velocity:

c c
W(z)=-i—=-i—e™.
(2) . R

Comparison with Equation 4.6 shows that the velocity components are

Hence, the direction of the flow is positive (counterclockwise) for ¢ > 0, and
the resulting flow field is called a vortex.

A vortex is characterized by its strength, which may be measured by the
circulation I" associated with it. From Equation 2.3, the circulation I" associ-
ated with the singularity at the origin is

r=g>‘u-d1
27
=f uy,Rd6
0

27

=f cdO = 2mc.
0

Here, the result 1, = ¢/R has been used. Then, ¢ may be replaced by I'/2x,
giving the following complex potential for a positive (counterclockwise) vor-
tex of strength I

r
F = ‘71 .
(z2)=-i——logz

The singularity in this expression is located at z = 0. That is, the line vortex
is located at z = 0. Then, the complex potential for a positive vortex located
at z =z, will be
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F(z) - —i%log(z ~2). “9)

The complex potential for a negative vortex would be obtained by replac-
ing I" by -I" in Equation 4.9. Note, however, that the negative coefficient is
associated with the positive vortex.

The flow field represented by Equation 4.9, which is shown in Figure 4.4b
for z, = 0, corresponds to a so-called free vortex. That is, for any closed contour
that does not include the singularity, the circulation will be zero and the
flow will be irrotational. All the circulation and vorticity associated with this
type of vortex are concentrated at the singularity. This is in contrast with the
solid-body rotation vortex mentioned in Chapter 2.

The principal application of the source, the sink, and the vortex is in the
superposition with other flows to yield more practical flow fields.

4.5 Flow in Sector

The flows in sharp bends or sectors are represented by complex potentials
that are proportional to z", where n > 1. A special case of such complex poten-
tials would be n = 1, which represents a uniform rectilinear flow. Then, in
order that this special case will reduce to Equation 4.7a, consider the complex
potentials

F(z) = Uz".

Substituting z = Re®® and separating the real and imaginary parts of this
function give

F(z) = UR" cos nB + iUUR" sin n6.
Then, the velocity potential and the stream function are
¢ = UR" cos no
y = UR" sin n6.
From this, it is evident that when 0 = 0 and when 0 = nt/#n, the stream func-
tion  is zero. That is, the streamline y = 0 corresponds to the radial lines

0 =0 and 0 = n/n. Between these two lines, the streamlines are defined by R"
sin 116 = constant. This gives the flow field shown in Figure 4.5. The direction
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FIGURE 4.5
Streamlines (shown solid) and equipotential lines (shown dashed) for flow in a sector.

of the flow along the streamlines may be determined from the complex
velocity as follows:
W(z) = nUz"" = nUR"'e'"°

= (nUR"" cos n® + inlUR"™" sin n@)e™.

Thus, by comparison with Equation 4.6, the velocity components are

uz = nUR"™ cos nd

uy = —nUR"" sin n.

Then, for 0 < 0 < (n/2n), uy is positive while 1, is negative, and for (x/2n)
< 0 < (n/n), ug is negative and u, remains negative. This establishes the flow
directions as indicated in Figure 4.5.

From the foregoing, the complex potential for the flow in a corner or sector
of angle nn/n is

F(z) = Uz". (4.10)
For n = 1, Equation 4.10 gives the complex potential for a uniform rectilin-

ear flow, and for n = 2, it gives the complex potential for the flow in a right-
angled corner.
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4.6 Flow around Sharp Edge
The complex potential for the flow around a sharp edge, such as the edge of
a flat surface, is obtained from the function z!'/2. Then, consider the complex
potential

F(z) = cz'/?
where cis real and 0 < 0 < 2x. Then, in cylindrical coordinates,

F(z) = cR1/2¢0/2

so that the velocity potential and stream function are
0
¢ = cRY? cos —
2
. 0
Yy = cRY? sin._.

Thus, the lines 0 = 0 and 0 = 2x correspond to the streamline y = 0. The
other streamlines are defined by the equation R"2 sin 6/2 = constant, which
yields the flow pattern shown in Figure 4.6. The direction of the flow is
obtained from the complex velocity as follows:

c c _ig/2
W) = = Jrim

= c:os9 + ising e
- 2R1/2 2 °

Hence, the velocity components are

Then, for 0 <0 <, up >0 and u, < 0. Also, for 1 < 0 < 2x, uy <0 and u, < 0.
This gives the direction of flow as indicated in Figure 4.6.

The flow field shown in Figure 4.6 corresponds to the flow around a sharp
edge, and so the complex potential for such a flow is
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FIGURE 4.6
Streamlines (shown solid) and equipotential lines (shown dashed) for flow around a sharp
edge.

F(z) = cz'/2. 4.11)

An important feature of this result is that the corner itself is a singular point
at which the velocity components become infinite. Since both u; and u, vary
as the inverse of R'/?, it follows that the velocity is singular as the square root
of the distance from the edge. This result will be discussed in Section 4.15.

4.7 Flow due to Doublet

The function 1/z has a singularity at z = 0, and in the context of complex
potentials, this singularity is called a doublet. The quickest way of establish-
ing the flow field that corresponds to the complex potentials that are propor-
tional to 1/z would be to follow the methods used in the previous sections.
However, it turns out that the doublet may be considered to be the coalescing
of a source and a sink, and the required complex potential may be obtained
through a limiting procedure that uses this fact. This interpretation leads to
a better physical understanding of the doublet, and for this reason, it will be
followed here before studying the flow field.

Referring to the geometry indicated in Figure 4.7a, consider a source of
strength m and a sink of strength m, each of which is located on the real axis
a small distance € from the origin. The complex potential for such a configu-
ration is, from Equation 4.8,
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(@) (b)

L)

FIGURE 4.7
(a) Superposition of a source and a sink leading to (b) streamline pattern for the limit & — 0

with me = constant.

m m
F(z) = Elog(z +¢)— Elog(z -¢)

m Z+E€
=—-1Io

2% zZ—-¢
—ﬂo 1+¢/z

o 8 1-¢/z

If the nondimensional distance ¢/|z| is considered to be small, the argu-
ment of the logarithm may be expanded as follows:

82

m € 3
F(iz)=—log 1+— 1+—+0 —
® 2 8 z z z’

2

m € €
=—Ilog 1+2—+0 —
2 B z z?

where the designation O(e?/z%) means terms of order €2/z> or smaller. The
logarithm is now in the form log(1l + y), where y < 1, so that the equivalent
expansion v + O(y?) may be used. Then,
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Fo)=2 2%5:0 &
2z

It is now proposed to let € = 0 and m — o« in such a way that lim,_(me) =
np, where 1 is a constant. Then, the complex potential becomes

F(Z)—;.

Thus, the complex potential 1/z may be thought of as being the equivalent
of the superposition of a very strong source and a very strong sink that are
very close together.

In order to establish the flow field that the above complex potential repre-
sents, the stream function will be established as follows:

F(z)= —
X +1y

x-iy
x*+y?

Thus, the equation of the streamlines y = constant is
2 2
X+y +—y=0
P

or

2 2
X+ oyt = ——

2y 2y

But this is the equation of a circle of radius u/(2y) whose center is located at
y =-1/(2y). This gives the streamline pattern shown in Figure 4.7b. Although
the direction of the flow along the streamlines may be deduced from the
source and sink interpretation, it will be checked by evaluating the velocity
components. The complex velocity for this complex potential is
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W(Z) = —ZT = _?6—129

) -0
=-—>(cosO-isinB)e™.
Hence, the velocity components are
Ug = — Y cos 0
Uy = — E sin 6.

These expressions for u; and u, confirm the flow directions indicated in
Figure 4.7b.

The flow field illustrated in Figure 4.7b is called a doublet flow, and the
singularity that is at the heart of the flow field is called a doublet. Then, the
complex potential for a doublet of strength p that is located at z = z, is

F(z) =

4.12)

Zy

The principal use of the doublet is in the superposition of fundamental
flow fields to generate more complex and more practical flow fields. An
application of this will be illustrated in the next section.

4.8 Circular Cylinder without Circulation

The fundamental solutions to the foregoing flow situations provide the
basis for more general solutions through the principle of superposition.
Superposition is valid here since the governing equation, for either the veloc-
ity potential or the stream function, is linear. The first example of superposi-
tion of fundamental solutions will be the flow around a circular cylinder.

Consider the superposition of a uniform rectilinear flow and a doublet at
the origin. Then, from Equations 4.7a and 4.12, the complex potential for the
resulting flow field will be

F(z)=Uz+—.
Z
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It will now be shown that for a certain choice of the doublet strength, the
circle R = a becomes a streamline. On the circle R = g, the value of z is a¢®, so
that the complex potential on this circle is

F(z) = Uae® + —e™™
a
= Ua+— cosO+i Ua-— sinb.
a a
Thus, the value of the stream function on the circle R = a is
Y= Ua-— sinb.
a

For general values of p, y is clearly variable, but if we choose the strength of
the doublet to be p = Ua?, then y = 0 on R = a. The flow pattern for this doublet
strength is shown in Figure 4.8a. The flow field due to the doublet encounters
that due to the uniform flow and is bent downstream. For clarity, the flow
due to the doublet is shown as dashed lines in Figure 4.8a. It may be seen
that the doublet flow is entirely contained within the circle R = 4, whereas
the uniform flow is deflected by the doublet in such a way that it is entirely
outside the circle R = 4. The circle R = g itself is common to the two flow fields.

Under these conditions, a thin metal cylinder of radius a could be slid into
the flow field perpendicular to the uniform flow so that it coincides with the
streamline on R = a. Clearly, the flow due to the doublet and that due to the
free stream would be undisturbed by such a cylindrical shell. Having done
this, the flow due to the doublet could be removed, and the outer flow would
remain unchanged. Finally, the inside of the shell could be filled to yield a
solid cylinder. That is, for R > 4, the flow field due to the doublet of strength
Ua? and the uniform rectilinear flow of magnitude U give the same flow as

FIGURE 4.8
(a) Flow field represented by the complex potential F(z) = U(z + a%/z) and (b) flow around a
circular cylinder of radius a.
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that for a uniform flow of magnitude U past a circular cylinder of radius a.
The latter flow is shown in Figure 4.8b. Then, the complex potential for a
uniform flow of magnitude U past a circular cylinder of radius a is

2

Fz)=U z+ “? . 4.13)

This result is useful in its own right, but it will also be found useful in
later sections, through the technique of conformal transformations, to obtain
additional solutions.

The solution given by Equation 4.13 for the flow around a circular cylinder
predicts no hydrodynamic force acting on the cylinder. This statement will
be proved quantitatively in a later section, and in the meantime, it will be
proved qualitatively. Referring to Figure 4.8b, it can be seen that the flow is
symmetric about the x axis. That is, for each point on the upper surface, there
is a corresponding point on the lower surface, vertically below it, for which
the magnitude of the velocity is the same. Then, from the Bernoulli equation,
the magnitude of the pressure is the same at these two points. Hence, by
integrating p dx around the surface of the cylinder, the lift force acting on the
cylinder must be zero. Similarly, owing to the symmetry of the flow about
the y axis, the drag force acting on the cylinder is zero.

Although the foregoing result does not agree with our physical intuition,
the potential-flow solution for the circular cylinder, and indeed for other bod-
ies, is valuable. The absence of any hydrodynamic force on the cylinder is due
to the neglect of viscosity. It will be seen in Part III that viscous effects create
a thin boundary layer around the cylinder, and this boundary layer separates
from the surface at some point, creating a low-pressure wake. The resulting
pressure distribution creates a drag force. However, it will be pointed out
that the viscous boundary-layer solution is valid only in the thin boundary
layer around the cylinder, and the solution obtained from the boundary-layer
equations must be matched to that given by Equation 4.13 at the edge of the
boundary layer. That is, Equation 4.13 gives a valid solution outside the thin
boundary layer and upstream of the vicinity of the separation point. It also
indicates the idealized flow situation that would be approached if viscous
effects are minimized. For more streamlined bodies, such as airfoils, the
potential-flow solution is approached over the entire length of the body.

4.9 Circular Cylinder with Circulation

The flow field studied in the previous section not only was irrotational but
also produced no circulation around the cylinder itself. It was found that
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there was no hydrodynamic force acting on the cylinder under these condi-
tions. It will be shown in a later section that it is the circulation around a
body that produces any lift force that acts on it. It is therefore of interest to
study the flow around a circular cylinder that has a circulation around it.

It was established in a previous section that the streamlines for a vortex
flow form concentric circles. Therefore, if a vortex is added at the origin to
the flow around a circular cylinder, as described in the previous section, the
fact that the circle R = a is a streamline would be unchanged. Thus, from
Equations 4.13 and 4.9, z; being zero in the latter, the complex potential for the
flow around a circular cylinder with a negative bound vortex around it will be

a> il
Fz)=U z+— +—logz+c.
@) z 2 &

The negative vortex has been used since it will turn out that this leads to a
positive lift. A constant ¢ has been added to the complex potential for the fol-
lowing reason. For no circulation, it was found that not only was y constant on
R = abut also the value of the constant was zero. By adding the vortex, y will
no longer be zero on R = g, although it will have some other constant value.
Since it is frequently useful to have the streamline on R =a be y = 0, it is desir-
able to adjust things so that this condition is achieved. By adding a constant c
to the complex potential, we have the flexibility to choose c in such a way that
y = constant becomes y = 0. This adjustment has no effect on the velocity and
pressure distributions since the velocity components are defined by deriva-
tives of y, so that the absolute value of y at any point is of no significance.

In order to evaluate the constant ¢, the value of the stream function on the cir-
cle R =a will be computed. Then, putting z = ae®, the complex potential becomes

F(z) = U(ae® + ae™) + ;—rlog ae® +c
n

= ZUacose—L6+£loga+c.
2n 2m

Hence, on the circle R = g, the value of y is indeed constant, and by choos-
ing c = —(il'/2n) log a, the value of this constant will be zero. With this value
of ¢, the complex potential becomes

2 .
F(z)=U 2+ +£10g£. (4.14)
z 2% a

This describes a uniform rectilinear flow of magnitude U approaching a
circular cylinder of radius a that has a negative vortex of strength I' around
it. As required, this result agrees with Equation 4.13 when I" = 0.



Two-Dimensional Potential Flows 97

In order to visualize the flow field described by Equation 4.14, the corre-
sponding velocity components will be evaluated from the complex velocity.

2 o
a ir 1
WiEz)=U 1-— +_——
@) 72 2n z
2 o
U 1-% i +—lr e
R? 27R
2 o
-u eze_i ) +£ e
R? 2R
2 2
U 1-L cos+i U 1+% si116+L e ™,
R? R? 2R

Hence, by comparison with Equation 4.6, the velocity components are

2

g = U 1—% cos0 (4.152)
. r
Uy =-U 1+P sme—m. (4.15b)

On the surface of the cylinder, where R = a, Equation 4.15 becomes
ug =0

U, = —ZUSmG—L.
0 2ma

The fact that u; = 0 on R = a is to be expected since this is the boundary
condition (Equation I1.3). A significant point in the flow field is a point where
the velocity components all vanish—that is, a stagnation point. For this flow
field, the stagnation points are defined by

r

s = - 4nla

4.16)

where 65 is the value of 6 corresponding to the stagnation point. For I" = 0,
sin 65 = 0, so that 05 = 0 or n, which agrees with Figure 4.8b for the circular
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cylinder without circulation. For nonzero circulation, the value of 6 clearly
depends upon the magnitude of the parameter I'/(4rxlla), and it is convenient
to discuss Equation 4.16 for different ranges of this parameter.

First, consider the range 0 < I'/(4nla) < 1. Here, sin 05 < 0, so that 65 must
lie in the third and fourth quadrants. There are two stagnation points, and
clearly the one that was at 6 = & is now located in the third quadrant, while
the one that was located at 6 = 0 is now located in the fourth quadrant. The
two stagnation points will be symmetrically located about the y axis in order
that sin 65 = —constant may be satisfied. The resulting flow situation is shown
in Figure 4.9a.

Physically, the location of the stagnation points may be explained as fol-
lows: The flow due to the vortex and that due to the flow around the cylinder
without circulation reinforce each other in the first and second quadrants.
On the other hand, these two flow fields oppose each other in the third and
fourth quadrants, so that at some point in each of these regions, the net
velocity is zero. Thus, the effect of circulation around the cylinder is to make
the front and rear stagnation points approach each other, and for a negative
vortex, they do so along the lower surface of the cylinder.

Consider next the case when the nondimensional circulation is unity, that
is, when I'/(4=xUa) = 1. Here, sin 65 = -1, so that 05 = 3n/2. The corresponding
flow configuration is shown in Figure 4.9b. The two stagnation points have
been brought together by the action of the bound vortex such that they coin-
cide to form a single stagnation point at the bottom of the cylindrical surface.
It is evident that if the circulation is increased above this value, the single
stagnation point cannot remain on the surface of the cylinder. It will move
off into the fluid as either a single stagnation point or two stagnation points.

Finally, consider the case where I'/(4nlla) > 1. Since it seems likely that any
stagnation points in this case will not lie on the surface of the cylinder, the
velocity components must be evaluated from Equation 4.15. Then, if R, and
0 are the cylindrical coordinates of the stagnation points, it follows from
Equation 4.15 that R, and 05 must satisfy the equations

FIGURE 4.9

Flow of approach velocity U around a circular cylinder of radius a having a negative bound
circulation of magnitude I" for (a) 0 < I'/(4nUa) < 1, (b) I[/(4rnlUa) = 1, and (c) I'/(4xla) > 1.
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2

ui-2 cosf; =0

2
s

2
U 1+ sin Og =—L.
R? 2nR,

Since it is assumed that the stagnation points are not on the surface of the
cylinder, it follows that R, # 4, so that the first of these equations requires that
05 = n/2 or 3n/2. For these values of 0, the second of the above equations
becomes

2
u 1+a— =z r
R? 27R,

S

where the minus sign corresponds to 85 = n/2 and the plus sign to 85 = ©/2.
Since U > 0, the left-hand side of the above equation is positive, and since
I' > 0, the minus sign must be rejected on the right-hand side. This might
have been expected since for I'/(4nxla) = 1, the value of 65 was 3n/2, whereas
the minus sign corresponds to 85 = n/2, which would require a large jump in
0, for a small change in I. The equation for R, now becomes

2
Ui+t -
R?  2nR,
or
R-L R0
2nl
and hence
r r °
R=——x)]— -
4nl 4nl
or
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This result shows that as 4nla/T" — 0, R, — o for the plus sign, but the cor-
responding limit is indeterminate for the minus sign. This difficulty may be
overcome by expanding the square root for 4nla/T" < 1 as follows:

R, T 1 4dnlln °
— 1+ 1-— +

a  4nla 2 T

where the dots indicate terms of order (4na/I')* or smaller. In this form, it is
evident that as 4nlla/T'— 0, R, — 0 for the minus sign. Since this stagnation
point would be inside the surface of the cylinder, the minus sign may be
rejected, so that the coordinates of the stagnation point in the fluid outside
the cylinder are

3
65—7
R T 4nlla °
s _ 1+,1- 4
a 4xnlUa T

This gives a single stagnation point below the surface of the cylinder. The
corresponding flow configuration is shown in Figure 4.9¢, from which it
will be seen that there is a portion of the fluid that perpetually encircles the
cylinder.

The flow fields for the circular cylinder with circulation, as shown in
Figure 4.9, exhibit symmetry about the y axis. Then, following the argu-
ments used in the previous section, it may be concluded that there will be
no drag force acting on the cylinder. However, the existence of the circula-
tion around the cylinder has destroyed the symmetry about the x axis, so
there will be some force acting on the cylinder in the vertical direction.
For the negative circulation shown, the velocity on the top surface of the
cylinder will be higher than that for no circulation, whereas the velocity
on the bottom surface will be lower. Then, from Bernoulli’s equation, the
pressure on the top surface will be lower than that on the bottom surface,
so that the vertical force acting on the cylinder will be upward. That is, a
positive lift will exist. In order to determine the magnitude of this lift, a
quantitative analysis must be performed, and this will be done in the next
section.

The principal interest in the flow around a circular cylinder with circula-
tion is in the study of airfoil theory. By use of conformal transformations, the
flow around certain airfoil shapes may be transformed into that of the flow
around a circular cylinder with circulation.
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4.10 Blasius Integral Laws

In the previous section, it was argued that a lift force exists on a circular cyl-
inder that has a circulation around it. However, the magnitude of the force
can be established only by quantitative methods. The obvious way to evalu-
ate the magnitude of this force is to establish the velocity components from
the complex potential. Knowing the velocity components, the pressure dis-
tribution around the surface of the cylinder may be established by use of the
Bernoulli equation. Integration of this pressure distribution will then yield
the required force acting on the cylinder.

The difficulty with the foregoing procedure is that it would have to be
carried out for each pressure distribution and for each body under consid-
eration. The Blasius laws provide a convenient alternative. It will be shown
that if the complex potential for the flow around a body is known, then it is
possible to evaluate the forces and the turning moment acting on the body by
means of simple contour integrals. These contour integrals, in turn, may be
readily evaluated by use of the residue theorem. The Blasius laws are actu-
ally two separate laws: one for forces and one for the hydrodynamic moment
acting on the body.

In order to establish the forces acting on an arbitrarily shaped body in
a flow field, consider such a body as shown in Figure 4.10. A fixed control
contour C; of arbitrary shape is drawn around the body whose surface is
denoted by C,. The forces acting through the center of gravity, as indicated by
X and Y, are the hydrodynamic forces acting on the body in the x and y direc-
tions, respectively. Then, for the fluid contained between the surfaces C, and
C, it may be stated that the net external force acting on the positive x direc-
tion must equal the net rate of increase in the x component of the momen-
tum. In Figure 4.10, an element of positive slope of the surface C; is shown
decomposed in the x and y directions. The components of the volume flow
that pass through this element of surface are also indicated. Then, the above
statement of Newtonian mechanics for the x direction may be expressed by
the following equation:

X-( pdy-= dy - odx)u.
fcopy fcnp(uy vdx)u

In writing this equation, it has been noted that there is no transfer of
momentum across the surface C; since it is a streamline, and that the inte-
gral of the pressure around C, yields the force X, which acts in the posi-
tive direction on the body and hence in the negative direction on the fluid.
Also, the mass efflux across the element of the surface C; is p(u dy — v dx),
so that the product of this quantity and the x component of velocity, when
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udy

FIGURE 4.10

Arbitrarily shaped body enclosed by an arbitrary control surface. X, ¥, and M are the drag, lift,
and moment acting on the body, respectively.

integrated around the surfaces C,, gives the net increase in the x component
of momentum.

A similar equation may be obtained by applying the same Newtonian law
to the y direction. Thus, the statement that the net external force acting in the
positive y direction must equal the net rate of increase in the y component of
the momentum yields the equation

=Y+ pdx=f p(udy — vdx)o.
Co Co

Solving these two equations for the unknown forces X and Y yields the
following pair of integrals:

X=[ (-pdy-pu’dy +puvdx)
Co

Y =f (pdx - puvdy + po* dx).
Co

The pressure may be eliminated from these equations by use of the
Bernoulli equation, which, for the case under consideration, may be written
in the form

p+%p(u2 +7v%)=B
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where B is the Bernoulli constant. Then, by eliminating the pressure p, the
expressions for X and Y become

X=p uvdx—%(uz—vz)dy

Co

1
Y=- uody + = (u? — v*)dx
pfco b+ € )

where the fact that | Bdx= | Bdy=0 for any constant B around any
Co Co

closed contour such as C; has been used.

It will now be shown that the quantity X —iY is related to a specific com-
plex integral. Consider the following complex integral involving the complex
velocity W:

iP [ w2dz - i% (- iv)(dx +idy)

Co Co

= i% { (W =0v*)dx +2uody +i (u* - 0*)dy - 2uvdx }
Co

=p uvdx—%(tf—vz)dy +i uvdy+%(u2—v2)dx

Co

=X-iY.

The last equality follows by comparison of the expanded form of the com-
plex integral with the expressions derived above for the body forces X and
Y. That is, the complex force X — iY may be evaluated from the following
equation:

X —iY = ig W2dz (4.17a)

Co

where W(z) is the complex velocity for the flow field, and C; is any closed
contour that encloses the body under consideration. It should be noted that
X and Y were defined as the forces acting on the body through its center of
gravity.

Equation 4.17a constitutes one of the two Blasius laws. Normally, in
applying Equation 4.17a, the contour integral is evaluated with the aid of
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the residue theorem. An application of this procedure will be covered in
the next section.

In order to establish the hydrodynamic moment acting on the body, con-
sider again Figure 4.10. The quantity M is the moment acting on the body
about its center of gravity. Then, taking clockwise moments as being posi-
tive, moment equilibrium of the fluid enclosed between C, and C, requires
that

-M +f [pxdx + pydy + p(udy - vdx)uy - p(udy — vdx)ovx] = 0.
Co

The first two terms under the integral are the components of the pressure
force multiplied by their respective perpendicular distances from the center
of gravity of the body, which is at the origin of the coordinate system. The
remaining two terms under the integral represent the inertia forces, which
were evaluated in the discussion of the force equations, multiplied by their
respective perpendicular distances from the origin. These inertia forces are
equal in magnitude and opposite in direction to the rate of increase in the
horizontal and vertical momentum components.

Solving the foregoing equation for the hydrodynamic moment M gives

M =f [pxdx + pydy + p(u’y dy + v*x dx - uvy dx - uvx dy)].
G
Substituting p = B — p(u* + v?)/2 from the Bernoulli equation gives

M=p[ - %(u2 +0°)(xdx + ydy) + (u*ydy + v*xdx) - (uoydx + uoxdy)

Co

where the fact that Bxdx= | Bydy=0 for any constant B and any
Go G

closed contour C; has been used. Rearranging the preceding equation shows
that the integral for the moment M may be put in the following form:

M= _Bf [(1* - v*)(xdx - ydy) + 2uv(xdy + y dx)].
2Jc,

It will now be shown that the quantity M may be related to the real part of
a complex integral. Consider the real part, designated by Re(), of the follow-
ing complex integral:
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B 2 _ B . N2 .
Re chon dz =Re 2fco(xﬂy)(u iv)”(dx +idy)
=Re Bf [(* - v*)(xdx - ydy) + 2uv(xdy + y dx)]
2Jc,

+ iEf [(#* - v*)(xdy + ydx) - 2uo(xdx + y dy)]
2Jc,

_P

) [(* - v*)(xdx - ydy) + 2uo(xdy + y dx)]
Co

- -M.

The last equality follows from a comparison of the real part of the com-
plex integral with the expression derived for M. That is, the hydrodynamic
moment acting on a body is given by

M= —%Re f ZW?dz 4.17b)

Co

where W(z) is the complex velocity for the flow field, and C; is any closed
contour that encloses the body. It should be noted that M is defined as the
hydrodynamic moment acting on the body, and it is positive when it acts
in the clockwise direction. Equation 4.17b is the second of the Blasius laws,
and the contour integral in this equation is usually evaluated by use of the
residue theorem.

4.11 Force and Moment on Circular Cylinder

It was observed in an earlier section that a force exists on a circular cylinder
that is immersed in a uniform flow and that has a circulation around it. The
magnitude of this force may now be evaluated using the results of the previ-
ous section.

From Equation 4.14, the complex potential for a circular cylinder of radius
ain a uniform rectilinear flow of magnitude U and having a bound vortex of
magnitude I' in the negative direction is
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a> il z
Fz)=U z+— +—log—.
@) +z +2n ga

Then, the complex velocity for this flow field is

2 .
a ir
Wiz=U 1-— +—
(@) 2 2nz
WD) = L 2U%a? .\ u?a* ,iur _ iura®  1°
N z? z4 nz 2’ An

But from the Blasius integral law (Equation 4.17a),

X-iy=i? [ W?dz
2Je,

=1 g Zm'E (residues of W? inside C,)

where the last equality follows from the residue theorem. It is therefore
required to evaluate the residue of W2(z) at each of the singular points that
lie inside an arbitrary contour in the fluid enclosing the cylinder. However,
inspection of the expression derived for W?2(z) above shows that the only sin-
gularity is at z = 0, corresponding to the doublet and the vortex that are
located there. Furthermore, W2(z) is in the form of its Laurent series about
z = 0, from which it is seen that the only term of the form b,/z is the fourth
term. Hence, the residue of W?2(z) at z = 0 is iUI/x. Then, the value of the
complex force is

Xoiy=i® om UL
2 7T
=-ipUL

Equating the real and imaginary parts of this equation shows that the drag
force X is zero, as was expected, and that the value of the lift force is

Y = pUL 4.18)
Equation 4.18 is known as the Kutta—Joukowski law, and it asserts that, for

flow around a circular cylinder, there will be no lift force on the cylinder if
there is no circulation around it, and if there is a circulation, the value of the
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lift force will be given by the product of the magnitude of this circulation
with the free-stream velocity and the density of the fluid. The right-hand
side of Equation 4.18 is positive, so the negative circulation that acted on the
cylinder led to a positive, that is, upward, lift force.

In order to evaluate the hydrodynamic moment M acting on the cylinder,
the quantity zWW2 must be evaluated. From the expression for W?(z) that was
established above,

2U%a? .\ u?at .\ iur ura®> 1?

3 2

ZW?(z) =U?z - .
@) z T 7z 4n’z

But from the Blasius integral law (Equation 4.17b),
M=-PRe [ iz
2 G
= —%Re Zm'E (residues of zW? inside C,)

where again the residue theorem has been used. However, the quantity
zW?2(z), as evaluated above, is already in the form of its Laurent series about
z = 0. From this, it is evident that the only singularity is at z = 0, and the
residue there comes from the second and last terms in the expansion. Hence,

FZ

- _PRe 2 —2u2a? -
2 4

That is, as might be expected, there is no hydrodynamic moment acting on
the cylinder.

4.12 Conformal Transformations

Many complicated flow boundaries may be transformed into regular flow
boundaries, such as the ones already studied, by the technique of conformal
transformations. Before using this fact, it is necessary to study the effect of
conformal transformations on the complex potential, the complex velocity,
sources, sinks, and vortices. In carrying out this study, it will be considered
that some geometric shape in the z plane whose coordinates are x and y is
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mapped into some other shape in the { plane whose coordinates are § and n
by means of the transformation

(=1@)

where fis an analytic function. This situation is depicted in Figure 4.11.

The basis of the complex potential was that both the velocity potential
and the stream function had to satisfy Laplace’s equation. Hence, in order
to establish the effect of a conformal transformation on complex potentials,
their effect on Laplace’s equation should be studied. This will be done by
transforming the second derivatives with respect to x and y into derivatives
with respect to the new coordinates, namely, & and n. Then, considering ¢ to
be a function of £ and 7,

a9 o n o
dx dx dE  dx Jm

where 0&/0x and dn/dx will be known from the equation of the mapping,
€ =f(z). Now, in order to transform 0%¢$/dx?, each of the two terms on the right-
hand side of the expression for d¢/0x must be differentiated with respect to
x. Then, using the product rule and considering ¢ to be a function of & and n

0 GEap _9°Eog & O0E 09 n 0%
ox dx 9E  9x? 9E  dx ox 9E*  dx 9%dm

and
2 2 2
d dn dp dmadp on 9 ¢ In 9o
- = . Ty - = 4 T
2 2
dx Jx dn dx* o dx Adx 9EIM  Ax an
BRI
BRI
RRIRIIIKKS
< CIIIRLLRRS
LRSI
< R LEIIIIIRRRRLKD
[RKEEK LIRS
% s oSate s teteret kotetotatetetetedes
X R ORREIRIIIRRIKK)
RG] KIS
KR RIS [ IR
ROSXSSERE5SKE8KL K SQQEKEK X X KEKKKKERK K AL KK LKKKKEL
00002020 20 0 0 0 e 0e 0 N0 0 0 00 00 190005 %% 0000 0. 0.0, 000 0.0 008
RIS 1% %
R RRRIIKIRIRKKS % RAXRKS %
N Q.Q.Q’Q‘Q'Q‘O“ %% % o o X
0% 0.0‘ 5 ’0‘0 0.’ XX}
0.0.0 ‘Q.Q‘Q
z plane { plane

Original and mapped planes for the mapping ( = f(z) where fis an analytic function.
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Hence, the expression for 0*¢$/0dx?, in terms of derivatives with respect to &
and 1, becomes

2 2 .9 2 .9 2 2 2
Po_ 0 T, an o sk d0 0% a0 o

ox>  9x 98 ax  an®  Ox dx 9Edn  ox* 9E  ox® an’

The corresponding expression for 02¢p/dy? is

2 2 0 2 .2 2 2 2
Po_ T Fe, e i i FE a0 i

d
>  dy  9E Ay om* 9y dy 9Bm  ay® 05 ay® am

Now, since ¢ must satisfy Laplace’s equation in the original plane, that is,
the xy plane, the sum of the above two quantities must be zero. Then,

2 2 2 2 2 2 2
g e 9% am m 0% E I g am 9O
— 4+ = —+ — 4+ — —— 42 =t =
ox ay  oE? ox ay o ox dx  dy dy IEIM

2 2 2 2
OE O 9 0om om 9 _,

+
aw* gyt 9 ax® gyt am

This is the equation that has to be satisfied by ¢(E, 1) in the { plane due to
any transformation { = f(z) corresponding to 0’¢p/dx? + 0*$p/dy? = 0 in the z
plane. So far, no restrictions have been imposed on the transformation, but if
the transformation is conformal, the mapping function will be analytic and
the real and imaginary parts of the new variable { will be harmonic. That is,
0%€/0x% + 9%€/9y? = 0 and 0*1/dx> + 0*11/0y? = 0, so that the terms involving these
quantities in the equation for ¢ will be zero. Also, &(x,y) and n(x,y) must satisfy
the Cauchy—Riemann equations if the mapping function is analytic. That is,

& _m
ox oy
and
& __m
ay ax
then

dgan dgon _ asom_omag

dx dx  dy Jdy ©ox ox  ox ox
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Using this result, the equation to be satisfied by ¢ becomes

m> o P
— o, A ¢=O.

2 2 P
¥, 0 P
dx dy  9E? ax ay

Using the Cauchy-Riemann equations to eliminate first &, then n, shows
that the following pair of equations must be satisfied:

2 2

2 2
KLU K
ox ay 0E an
2 2 > >
23 ‘€ e o _
+ St =0.
ox ay 0g on

But these equations must be satisfied for all analytic mapping functions;
hence, it follows that

That is, Laplace’s equation in the z plane transforms into Laplace’s equa-
tion in the { plane, provided that these two planes are related by a conformal
transformation. Then, since both ¢ and y must satisfy Laplace’s equation, it
follows that a complex potential in the { plane is also a valid complex poten-
tial in the { plane, and vice versa. This means that if the solution for some
simple body is known in one of these planes, say the { plane, then the solu-
tion for a more complex body may be obtained by substituting { = f(z) in the
complex potential F(C).

Consider now what happens to the complex velocity under a conformal
transformation. Starting in the z plane with the definition of complex velocity,

dF(z)
dz

_dt dF(@)
dz dC

W(z) =
.19

W) = WO,
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That is, complex velocities are not, in general, mapped one to one, but they
are proportional to each other, and the proportionality factor depends on the
derivative of the mapping function.

Finally, the effect of a conformal transformation on the strength of the
basic singularities will be investigated. That is, the strength of transformed
sources, sinks, and vortices will be established. This is most readily done by
first proving the general relation that the integral of the complex velocity
around any closed contour in the flow field equals I" + im, where I" is the net
strength of any vortices inside the contour and m is the net strength of any
sources and sinks inside the contour.

To prove this relation, consider any closed contour C such as the one shown
in Figure 4.12. An element d! of this contour is shown resolved into its coor-
dinate components. Then, the net strength of all the sources inside C (sinks
being considered negative sources) and the net strength of all the vortices
inside C will be given by

m=fcu'ndl=fc(udy—vdx)

I“=fcu-dl=fc(udx+vdy).

Now, consider the integral around C of the complex velocity W(z).

&
&

dx

FIGURE 4.12
Arbitrary closed contour C with an element d/ resolved into its coordinate components.
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f W(z)dz = f (u - iv)(dx +idy)

=f(udx+vdy)+if (udy — vdx)
C C
=T'+im

where the last equality follows from a comparison with the expressions
derived for m and I'. This general result will now be applied to a single vortex
I', and a single source m, located in the z plane. Then,

L, +im, =f W(z)dz
CZ
_ dg
= fc Z WE dz
= | W@©dC
G
=T +im.

where C, is some closed contour in the z plane and C, is its counterpart in the
mapped plane. I', and m, are the corresponding vortex and source strengths
in the { plane, and the above result shows that the vortex and source strengths
are the same in the z plane as in the { plane. That is, sources, sinks, and vor-
tices map into sources, sinks, and vortices of the same strength under a con-
formal transformation.

In summary, if the complex potential for the flow around some body is
known in the { plane, then the complex potential for the body corresponding to
the conformal mapping = f(z) may be obtained by substituting this transfor-
mation into the complex potential F((). Complex velocities, on the other hand,
do not transform one to one but are related by Equation 4.19. Sources, sinks,
and vortices maintain the same strength under conformal transformations.

4.13 Joukowski Transformation

One of the most important transformations in the study of fluid mechanics is
the Joukowski transformation. By means of this transformation and the basic
flow solutions already studied, it is possible to obtain solutions for the flow
around ellipses and a family of airfoils. The Joukowski transformation is of
the following form
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CZ
z=C+ E (4.20)

where the constant ¢? is usually taken to be real. A general property of the
Joukowski transformation is that for large values of |(|, z — (. That is, far
from the origin, the transformation becomes the identity mapping, so that
the complex velocity in the two planes is the same far from the origin. This
means that if a uniform flow of a certain magnitude is approaching a body
in the z plane at some angle of attack, a uniform flow of the same magnitude
and angle of attack will approach the corresponding body in the  plane.
From Equation 4.20,

dz_, &
dt =~

so that there is a singular point in the Joukowski transformation at ¢ = 0.
Since we are normally dealing with the flow around some body, the point
¢ =0is normally not in the fluid, and so this singularity is of no consequence.
There are also two critical points of the transformation, that is, points at
which dz/d¢ vanishes, at { = +c. Since smooth curves passing through critical
points of a mapping may become corners in the transformed plane, it is of
interest to investigate the consequence of a smooth curve passing through
the critical points of the Joukowski transformation. To do this, consider an
arbitrary point z and its counterpart { as shown in Figure 4.13a. Let the point
{ be measured by the radii p, and p, and the angles v; and v, relative to the
two critical points { = ¢ and { = —c, respectively. However, according to the
Joukowski transformation, the points { = +c map into the points z = +2c.
Then, let the mapping of the point { be measured by the radii R, and R, and
the angles 6, and 6, relative to the two points z = 2c and z = -2c, respectively.
From Equation 4.20,

z+2(:=M
C
and
z—2c=7@_c)2
C
.z-2c  C-c ’

T z+2c  CTac
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FIGURE 4.13

(a) Coordinate system used to investigate the critical points of the Joukowski transformation
and (b) coordinate changes corresponding to a smooth curve passing through ¢ = c.

Thus, with reference to Figure 4.13a,

R 04 vy 2
s CRR L

Rzefez pzeivz

or

2
Ry jioron o PL i

R, P2

Equating the modulus and the argument of each side of this equation
shows that
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and
0, —0,=(; — 1.

This last result shows that if a smooth curve passes through the point{ =,
the corresponding curve in the z plane will form a knife-edge or cusp. This
may be verified by considering a smooth curve to pass through the point
¢ = c. Two points on this curve are shown in Figure 4.13b, from which it is
seen that v; changes from 3x/2 to n/2 and v, changes from 2x to 0 as the
critical point is passed. That is, the value of v, — v, changes from —n/2 to n/2,
giving a difference of n. From the result 6, — 6, = 2(v; — v,), it follows that the
corresponding difference in the value of 0, — 6, will be 2x. This yields a knife-
edge or cusp in the z plane as shown in Figure 4.13b. That is, if a smooth
curve passes through either of the critical points { = +c, the corresponding
curve in the z plane will contain a knife-edge at the corresponding critical
point z = *2c.

An example of a smooth curve that passes through both critical points is a
circle centered at the origin of the { plane and whose radius is ¢, the constant
that appears in the Joukowski transformation. Then, on this circle { = ce®,
and the value of z will be given by

z=ce™ +ce™

=2CCOSV.

That is, the circle in the { plane maps into the strip y = 0, x = 2c cos v
in the z plane. It is readily verified that all points that lie outside the circle
|C] =c cover the entire z plane. However, the points inside the circle |{| =¢
also cover the entire z plane, so that the transformation is double valued.
This is readily verified by observing that for any value of {, Equation
4.20 yields the same value of z for that value of { and also for ¢2/¢. It will
be noted that c?/{ is simply the image of the point { inside the circle of
radius c.

This double-valued property of the Joukowski transformation is treated
by connecting the two points z = +2¢ by a branch cut along the x axis and
creating two Riemann sheets. Then, the mapping is single valued if all the
points outside the circle |{| = c are taken to fall on one of these sheets and
all the points inside the circle are taken to fall on the other sheet. In fluid
mechanics, difficulties due to the double-valued behavior do not usually
arise because the points |{| < c usually lie inside some body about which
the flow is being studied, so that these points are not in the flow field in
the z plane.
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4.14 Flow around Ellipses

Applications of the Joukowski transformation will be made in an inverse
sense. That is, the simple geometry of the circle, the flow around which is
known, will be placed in the { plane, and the corresponding body that results
in the z plane will be investigated by use of Equation 4.20.

Consider, first, the constant c in Equation 4.20 to be real and positive, and
consider a circle of radius a > ¢ to be centered at the origin of the { plane.
The contour in the z plane corresponding to this circle in the { plane may be
identified by substituting ¢ = ae” into Equation 4.20:

2
. C s
z=qae" +—e"
a
2 CZ
= 44— COSV+i a-— sinwv.
a a

Equating real and imaginary parts of this equation gives

C2
X= ad+— CosVv
a

y= a ¢ sin v
; .

These are the parametric equations of the required curve in the z plane.
The equation of the curve may be obtained by eliminating v by use of the
identity cos? v + sin? v = 1. This gives

2 2
X
=1

+
a+c*/a a-c*/a

which is the equation of an ellipse whose major semiaxis is of length a +
c?/a, aligned along the x axis, and whose minor semiaxis is of length a — c?/a.
Then, in order to obtain the complex potential for a uniform flow of magni-
tude U approaching this ellipse at an angle of attack a, the same flow should
be considered to approach the circular cylinder in the { plane. However, it is
shown in the problems at the end of this chapter that the complex potential
for a uniform flow of magnitude U approaching a circular cylinder of radius
a at an angle « to the reference axis is
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2

FQ=U Ce ™+ %ei"'

Then, by solving Equation 4.20 for { in terms of z, the complex potential in
the z plane may be obtained. From Equation 4.20,

C-zZL+c*=0

2

2

z
== -c.
C 5 E

z
2

Since it is known that { — z for large values of z, the positive root must be
chosen. Then, the complex potential in the z plane becomes

a2€1cx

+
z/2 + \/(2/2)2 —-c?

2 2 2
z z e @z z ,
=U z-Z+,] 2 -c® e+ S| = ¢ e
2 2 ¢t 2 2

where the last term has been rationalized. By writing z/2 as z—z/2 in the first
term, two of the terms may be combined as follows:

Fz)=U §+ ==t e

2
2
—iat a i —ia Z
F(z)=U ze™+ —e“%-e —
c’ 2

-t 4.21a)

Equation 4.21a is the complex potential for a uniform flow of magnitude
U approaching an ellipse whose major semiaxis is a + ¢2/a and whose minor
semiaxis is a — c?/a. The flow is at an angle of attack a to the major axis. In this
form, it may be seen that the complex potential consists of that for a uniform
flow at an angle a to the reference axis plus a perturbation that is large near
the body but vanishes for large values of z. The flow field generated by the
complex potential (Equation 4.21a) is shown in Figure 4.14a together with
that for the circular cylinder in the { plane.

The stagnation points in the { plane are located at { = ae™ and { = ael*@,
that is, at { = +ae”. Then, from Equation 4.20, the corresponding points in the
z plane are
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FIGURE 4.14
(@) Uniform flow approaching a horizontal ellipse at an angle of attack and (b) uniform parallel

flow approaching a vertical ellipse.

2 2

c . C .
* 4+— COSOxl 4a—— SINOA.
a a

]
+

This gives the coordinates of the stagnation points as

X=% a+— COsa
a

c?
y=+ a-— sino.
a

Equation 4.21a includes two special cases within its range of validity. For
a = 0, it describes a uniform rectilinear flow approaching a horizontally ori-
ented ellipse, and for a = n/2, it describes a uniform vertical flow approach-
ing the same horizontally oriented ellipse. However, it is of interest to note
that the solution for a uniform rectilinear flow approaching a vertically ori-
ented ellipse may be obtained directly from the Joukowski transformation
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with a slight modification. Substitute c = ib, where b is real and positive, into
Equation 4.20

bZ
z=C- c .

Then, as with the horizontal ellipse, examining the mapping of the circle
{ = ae" gives the parametric equations of the mapped boundary:

2

X= ad-—— COSV
a

2
= a+— sinv.
Y a
Thus, the equation of the contour in the z plane is
2 2

x .\ y
a-b%/a a+b®/a

which is the equation of an ellipse whose major semiaxis is a + b*/a, which is
aligned along the y axis. Then, to obtain uniform rectilinear flow approach-
ing such an ellipse, the same flow should approach the circle in the ¢ plane.
Thus, the required complex potential, from Equation 4.13, is

a2
FQ=U T+—
© T+ C

But the inverted equation of the mapping for which { - zasz — «is

z
=+

c 2
Hence, the complex potential in the z plane is

2
F(z)y=U A a
2 |2 2/2+(z/2)? + b
. 4.21b)
2
Fz)=U z- 1+° Z_ %2 Lp
o2\ 2
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in which the same rationalization and simplification have been carried out
as before. Again, the complex potential is in the form of that for a uniform
flow plus a perturbation that is large near the body and that vanishes at large
distances from the body. Equation 4.21b describes a uniform rectilinear flow
of magnitude U approaching a vertically oriented ellipse. The flow field for
this situation is shown in Figure 4.14b.

4.15 Kutta Condition and Flat-Plate Airfoil

It was observed in Section 4.6 that the potential-flow solution for flow
around a sharp edge contained a singularity at the edge itself. This singular-
ity required an infinite velocity at the point in question, which, of course, is
physically impossible. The question arises, then, as to what the real flow situ-
ation would be in a physical experiment. Depending upon the actual physi-
cal configuration, one of two remedial situations will prevail. One possibility
is that the fluid will separate from the solid surface at the knife-edge. The
resulting free streamline configuration would be such that the radius of cur-
vature at the edge becomes finite rather than being zero. As a consequence,
the velocities there will remain finite. Examples of this type of solution will
be discussed later in this chapter.

A second possibility is that a stagnation point exists at the sharp edge.
For the flow around finite bodies, stagnation points exist, and it seems
possible that a stagnation point could be induced by the flow field to move
to the location of the sharp edge. This possibility leads to the so-called
Kutta condition, and it will be discussed below in the context of the flat-
plate airfoil—that is, a flat plate that is at some angle of attack to the free
stream.

In the previous section, the flow around an ellipse was obtained from the
Joukowski transformation (Equation 4.20) by considering the flow around
a circular cylinder of radius a > c in the { plane. Now, if the constant c is
allowed to approach the magnitude of the radius 4, the resulting ellipse in
the z plane degenerates to a flat plate defined by the strip —2a < x < 2a. The
resulting flow field, as defined by Equation 4.21a, is shown in Figure 4.15a.
Because of the angle of attack, the stagnation points do not coincide with the
leading and trailing edges of the flat plate. Rather, the upstream stagnation
point is located on the lower surface and the downstream stagnation point is
located on the upper surface at the points x = +2a cos a. Then, around both
the leading and trailing edges, the flow will be that associated with a sharp
edge, which was discussed in Section 4.6. In that section, it was observed
that infinite velocity components existed at the edge itself, a situation that is
physically impossible to realize.



Two-Dimensional Potential Flows 121

z

z plane

FIGURE 4.15
Flow around a flat plate at shallow angle of attack (a) without circulation and (b) satisfying the
Kutta condition.

The difficulty encountered above with the flat-plate airfoil does not occur
at the leading edge of real airfoils because real airfoils have a finite thick-
ness and so have a finite radius of curvature at the leading edge. However,
the trailing edge of airfoils is usually quite sharp, so that the difficulty of
infinite velocity components still exists there. However, this remaining dif-
ficulty would also be overcome if the stagnation point that is near the trailing
edge was actually at the trailing edge. This would be accomplished if a cir-
culation existed around the flat plate and the magnitude of this circulation
was exactly the amount required to rotate the rear stagnation point so that
its location coincides with the trailing edge. This condition is called the Kutta
condition, and it may be restated as follows: For bodies with sharp trailing
edges that are at small angles of attack to the free stream, the flow will adjust
itself in such a way that the rear stagnation point coincides with the trailing
edge.

The amount of circulation required to comply with the Kutta condition
may be determined as follows: In the { plane of Figure 4.15a, the rear stagna-
tion point is located at the point { = ae’®, but, according to the Kutta condi-
tion, the rear stagnation point should be located at the point z = 24, which
corresponds to the point { = 4. That is, the stagnation point on the down-
stream face of the circular cylinder in the { plane should be rotated clockwise
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through an angle a. However, from Equation 4.16, the magnitude of the cir-
culation that will do this is

I' =4xla sin o (4.22a)

in the clockwise direction (i.e., negative circulation). Then, the complex
potential for the required flow in the { plane is, from Equations 4.14 and 4.29,

2

FQ=U Te™ +%ei°‘ + ileasinoclogE .
a

But the equation of the mapping is

aZ

z=C0+—

o

and the inverse, which gives { — zas z — o, is

z z
=—+, - =-a.
¢ 2 2

Then, the complex potential in the z plane is

F@)=U 2+ )2 - e+ ae”
2 2 z/2 ++(z/2)* - a*
: 4.22b)
+i2asinolog % §+ % -a

The flow field corresponding to this complex potential is shown in Figure
4.15b. Although the flow at the trailing edge is now regular, the singularity at
the leading edge still exists. In an actual flow configuration, the fluid would
separate at the leading edge and reattach again on the top side of the airfoil.
The streamline y = 0 would then have a finite curvature, and the velocity
components would remain finite at the leading edge.

The lift force generated by the flat-plate airfoil may be calculated from the
Kutta—Joukowski law. Then, denoting the lift force by Y and using the value
of the circulation given by Equation 4.22a,
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Y = 4npU2a sin a.

It is usual to express lift forces in terms of the dimensionless [ift coefficient
C,, which is defined as follows:

where [ is the length or chord of the airfoil, which, for the flat plate under
consideration, equals 4a4. Then, the value of the lift coefficient for the flat-
plate airfoil is

C, = 2x sina. (4.22¢)

This result shows that the lift coefficient for the flat-plate airfoil increases
with angle of attack, and for small values of o, for which sin a =~ «, the lift
coefficient is proportional to the angle of attack with a constant of propor-
tionality of 2x. This result is very close to experimental observations, and
so the Kutta condition appears to be well justified. If the Kutta condition
were not valid, there would be no circulation around the flat plate and con-
sequently no lift would be generated. This would mean that kites would not
be able to fly.

4.16 Symmetrical Joukowski Airfoil

A family of airfoils may be obtained in the z plane by considering the
Joukowski transformation in conjunction with a series of circles in the ¢
plane whose centers are slightly displaced from the origin. These airfoils are
known as the Joukowski family of airfoils.

Consider, first, the case where the center of the circle in the { plane is dis-
placed from the origin along the real axis. It must then be decided in which
direction the center should be moved and what radius should be employed,
relative to the Joukowski constant c. From previous sections, it is known that
if the circumference of the circle passes through either of the two critical
points of the Joukowski transformation, { = +c, then a sharp edge or cusp is
obtained in the z plane. Then, if the leading edge of the airfoil is to have a
finite radius of curvature and if there should be no singularities in the flow
field itself, it follows that the point { = —c should be inside the circle in the {
plane. Also, since the trailing edge of the airfoil should be sharp as opposed
to being blunt, the circumference of the circle should pass through the point
{ = c. These conditions will be satisfied by taking the center of the circle to be
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at { = —m, where m is real, and by choosing the radius of the circle to be ¢ + m.
Such a configuration is shown in Figure 4.16a.
The radius a is given by

a=c+m=c(l +¢

where the parameter € = m/c will be assumed to be small compared with
unity. When e = 0, the flat-plate airfoil is recovered, so that for € <« 1, it may
be anticipated that a thin airfoil will be obtained. The significance of the
restriction € < 1 will be that all the equations may be linearized in e, which
will permit a closed-form solution to be obtained for the equation of the air-
foil surface in the z plane. Also shown in Figure 4.16a is the airfoil that is
obtained in the z plane and its principal parameters, the chord / and the
maximum thickness t. It is now required to relate these parameters to the
free parameters a and m and to establish the equation of the airfoil surface
in the z plane.

To establish the chord of the airfoil in terms of the chosen radius a and
offset m, it is only necessary to find the mapping of the points { = ¢ and
€ = —(c + 2m) since these points correspond to the trailing and leading edges,
respectively. Using the Joukowski transformation, the mapping of the point
€ =cisz =2c. Also, the mapping of the point { = —(c + 2m) = —c(1 + 2¢) is

Cc

z=—c(1+2¢)-

1+2¢

FIGURE 4.16
Symmetrical Joukowski airfoil: (a) mapping planes and (b) uniform flow past the airfoil.
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Since it was decided to linearize all quantities in ¢, the value of z will be
obtained to the first order in € only:

z = —c(1+2¢) - c[1-2¢ +O(e?)]
=-2c+O(g?).

That is, to the first order in ¢, the leading edge of the airfoil is located at
z = —2¢, so that the chord is

[ =4c.

This means that, correct to the first order in ¢, the chord of the airfoil is
unchanged by the shifting of the center of the circle in the ¢ plane.

In order to determine the maximum thickness ¢, the equation of the airfoil
surface must be obtained. This may be done by inserting the equation of the
surface in the { plane into the Joukowski transformation. However, in the {
plane, the polar radius R to the circumference of the circle is a function of the
angle v. In order to establish this dependence explicitly, the cosine rule will
be applied to the triangle defined by the radius a4, the coordinate R, and the
real { axis, as shown in Figure 4.16a. Thus,

2

a* = R* + m* = 2Rm cos(mt - v)

=R? + m? +2Rmcosv.

But a = ¢ + m, so that the equation above may be written in the form

mZ

(c+m)* = R? 1+—2+2ﬂcosv .

Now, since R > ¢, it follows that m/R < m/c so that, to first order in € = m/c,
the term m?/R? may be neglected. The equation for R then becomes

1/2
m
c+m=R 1+2icosv

=R 1+%COSV+O(82) .

Thus, to the first order in ¢, this relation becomes
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c(l1+¢)=R+mcosv
S R=c[1+¢(1l-cosv)].
This is the required equation that gives the variation of the radius R with
the angle v for points on the circumference of the circle in the { plane. Then,
in order to determine the equation of the corresponding profile in the z

plane, this result should be substituted into the Joukowski transformation
(Equation 4.20). Thus, points on the surface of the airfoil will be defined by

- ce™
z=c[l+e1-cosv)e” + —F———.
[ ( ) 1+¢(1-cosv)

This equation may also be linearized in ¢ as follows:
z=c[1+¢e(1-cosv)le™ +c[1-e(1-cosv)+O(e*)]e™
= c[2cosv +i2¢(1 - cos v)sin v + O(e?)].

Then, by equating real and imaginary parts of this equation, the paramet-
ric equations of the airfoil are, to first order in ¢,

X =2ccosv

y = 2ce(1- cosv)sinv.

Using the first of these equations to eliminate v from the second equation
gives the following equation for the airfoil profile:

2
Yy ==*2ce - h- =
2c 2c

The location of the maximum thickness may now be obtained, and this
is most readily done by using the parametric equation for the coordinate y
as derived above. Thus, setting dy/dv = 0 for a maximum value of y gives
the following equation for the value of v at the location of the maximum
thickness:

sin?v+ (1 —cosv) cos v=0.
This relation reduces to

COs 2V = cos v
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which is satisfied by v = 0, v = 21/3, and v = 4xn/3. The solution v = 0 corre-
sponds to the trailing edge and corresponds to the minimum thickness. The
solutions v = 2n/3 and v = 4xn/3 give the maximum thickness, and for these
values of v, the coordinates of the airfoil surface are

The maximum thickness ¢ will be twice the positive value of y, so that the
thickness ratio t/I of the airfoil will be

That is, the thickness-to-chord ratio of the airfoil is proportional to €, which
is the ratio of the offset of the center of the circle in the { plane to the radius
c of the critical points of the transformation. Since the thickness ratio of the
airfoil is a parameter that may be thought of as being specified, it is useful to
eliminate € in terms of this parameter. Hence,

=0.77

—
—

e
343

Then, the equation of the airfoil surface may be written in the form

2

%=:0.385 1-2% 1- 2% (4.23a)

where the maximum value of y/t will be 0.5 and the minimum value will be
—-0.5, both of which occur at x = —c.

The magnitude of the circulation required to satisfy the Kutta condition is,
from Equation 4.16, 4xla sin o, where a = ¢ + m and m = ce = 0.77 tc/l. Thus,
the required amount of circulation is

=xnlUl 1+ 0.77; sin o (4.23b)

where c has been replaced by /4. In this form, the required circulation may
be evaluated for the given free-stream velocity, angle of attack, and the chord
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and thickness of the airfoil. Using the Kutta—Joukowski law (Equation 4.18),
the lift force acting on the airfoil may be evaluated as

F =mpU’l 1+ 0.77§ sin ..
Then, the lift coefficient for the symmetrical Joukowski airfoil is
C.=2n 1+ O.77§ sin . (4.23¢)

It will be noticed that this result reduces to Equation 4.22¢ for the flat-plate
airfoil as t — 0. This indicates that the effect of thickness of an airfoil is to
increase the lift coefficient. However, this fact cannot be used to produce
high lift coefficients through thick airfoils since the flow tends to separate
from bluff bodies much more readily than it does from streamlined bodies.
This separation of the flow is a viscous effect, and it will be discussed in the
next part of the book. In the meantime, it is sufficient to say that separation
of the flow results in a low-pressure wake that destroys the lift. The same
result may occur for slender bodies, such as airfoils, that are at large angles
of attack. In this context, the separation is usually referred to as stall.

The center of the circle in the { plane is located at { = —m rather than at
€ =0. Thus, the complex potential for the flow in the { plane may be obtained
from Equation 4.29 by replacing { by { + m and adding the circulation. The
required complex potential then becomes

, a ir T+m
F =U —ioL io. iy |
© C+m)e™ + cem® T2 T, (4.23d)
where
a= L 0.77t—c
l

and

m= 0.77t7c .

The magnitude of the circulation I' is given by Equation 4.23b, and in the
Joukowski transformation, the parameter ¢ equals [/4. The flow field corre-
sponding to this complex potential is shown in Figure 4.16b.
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4.17 Circular-Arc Airfoil

It was shown in the two previous sections that, using the Joukowski trans-
formation, a circle of radius c centered at the origin in the { plane produced
a flat-plate airfoil, whereas a slightly larger circle centered a small distance
along the real axis from the origin produced a thin symmetrical airfoil. It
will now be shown that a circle whose radius is slightly larger than ¢ and
whose center is located on the imaginary axis in the { plane produces an
airfoil that has no thickness but has curvature or camber.

Referring to Figure 4.17a, consider a circle of radius a > c in the { plane such
that the center of the circle is located a distance m along the positive imagi-
nary axis. Since the trailing edge of the airfoil should be sharp, the circle
should pass through the critical point { = c as before. Then, in this case, the
circle will also pass through the other critical point, { = —c.

The equation of the airfoil in the z plane may be obtained by substituting
{ = Re" into the Joukowski transformation, where, on the circumference of
the circle in the { plane, R is a function of v. This substitution gives

2

o C
z=Re" +—e¢

R

—iv

c? c?
= R+— cosv+i R-— sinw.
R R

(b)

o
=z

/

z plane { plane

FIGURE 4.17
Circular-arc airfoil: (a) mapping planes and (b) uniform flow past the airfoil.
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Thus, the parametric equations of the airfoil profile are

C2
Xx= R+— cosv

o
Y= R_f sinv.

The variable R may be eliminated from these equations as follows:

2 2
c’ 2 2 ¢’ 2 2
xzsmzv—yzcoszv= R+— sin“vcos°v—- R--— sin“vcos” v

= 4¢? sin® vcos? v.

This is the equation of the airfoil surface in the z plane, but it still contains
the variable v. This variable may be eliminated by applying the cosine rule

to the triangle defined by the radius 4, the coordinate R, and the imaginary
¢ axis. From this, it follows that

T
a* =R*+m* -2Rmcos =-v

c® +m? =R*> +m? - 2Rmsinv
it follows that

where the fact that a> = ¢ + m? has been used. Solving this equation for sin v,

2 _ 2

2Rm

sinv =

But it was shown above that iy = [(R? - ¢?) sin 1]/R; hence, it follows that

sinv=— Y
2msin v
or
sin?v = Y

2m
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and so

y

cos’v=1--7.
2m

Using these results to eliminate v, the equation of the airfoil surface becomes

2 Y 2
J 1-
x 2m Y

Yo_ue2 ¥ oY
2m 2m 2m

Collecting like terms, this equation may be put in the form

2 2 ¢ 2
X“+y +2 —-m y=4c”.
Y Y

Completing the square in y shows that the equation of the airfoil surface is

2 2
Cc m Cc m
LR 2y 228
m Cc m C

X+ y+c

which is the equation of a circle. It should be noted that so far no approxima-
tions have been made. But to be consistent with the analysis in the previous
section and to permit superposition in the next section, the parameter € = m/c
will again be assumed to be small compared with unity. Then, linearizing in
g, the equation of the airfoil surface becomes

2 2 2

c c
Pt oy+— =c” 4+
m m

That is, correct to the first order in g, the center of the circle in the z plane is
located at iy = —c2/m and the radius of the circle is c\/4 +c¢*/m”*.

The characteristic parameters of the airfoil are the chord / and the camber,
or height, i, as shown in Figure 4.17a. Since the equation of the airfoil has
now been established, it is possible to relate these parameters to those in
the ¢ plane, namely, ¢ and m. Since the ends of the circular-arc airfoil lie on
the real axis y = 0, the foregoing equation of the airfoil shows that the cor-
responding values of x are +2c. That is, the chord of the airfoil is

[ =4c.
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This is the same chord value as for the two previous airfoils.

The simplest way of establishing the camber & of the airfoil is to use the
fact that, in view of the result that the center of the circular arc is at x = 0, the
maximum value of y will occur when x = 0. However, the parametric equa-
tion x = (R + ¢2/R) cos v shows that this corresponds to v = n/2. Then, the
other parametric equation, namely, y = 2m sin? v, shows that the maximum
value of y is 2m. That is,

h=2m.

Using the foregoing results, the { plane parameters c and m may be replaced
by the z-plane parameters //4 and /2, respectively. Then, the equation of the
airfoil surface in the z plane may be written in the form

2
piop 2
2 L
TV T4 T e

(4.24a)

In order to satisfy the Kutta condition, the rear stagnation point must rotate
through an angle greater than o, the angle of the free stream. By rotating
through the angle o, the rear stagnation point will be located on the surface
of the circle in the { plane at a point that is in the same horizontal plane as the
center of the circle, but the center of the circle is located a distance m above
the real ¢ axis. Thus, in order to be located at the point { = ¢, the rear stagna-
tion point must rotate through a further angle given by

m
tan' —=tan'¢
c
=g+ O(e?).

That is, in order to comply with the Kutta condition, the rear stagnation
point must rotate through the angle o + 1m1/c, to the first order in e. Then, from
Equation 4.16, the required circulation is

I = 4nlasin oc+E .
c

But a =+/c? + m® so that, to first order in €, a = ¢. Hence,

I = 4nlc sin 0L+E .
c
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Then, from the Kutta—Joukowski law, the lift force is
F, = 4npU2csin o+ "
c
and the corresponding lift coefficient is

c . m
CL=8n73m oa+— .
c

Using again the fact that c = [/4 and m = h/2, the lift coefficient becomes
C. =2msin o+ Zl—h . (4.24b)

Comparing this result with Equation 4.22c, the corresponding result for the
flat plate, shows that the effect of positive camber in an airfoil is to increase
its lift coefficient. As a consequence of this increased lift coefficient, a non-
zero lift exists at zero angle of attack.

Since the center of the circle in the { plane is at { = im rather than at { = 0,
the complex potential in the { plane may be obtained by replacing { by { —im
in Equation 4.29 and adding the circulation. Thus, the required complex
potential is

2

F@Q) =U C-im)e™ + a. e +i—rlog C—im

C_im Py ; (4.24¢)
where
l
a=_
4
and
h
m=—.
2

The magnitude of the circulation I is given by

I’ = nll sin oc+21—h
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and the parameter c in the Joukowski transformation is //4. The flow field
corresponding to this complex potential is shown in Figure 4.17b. As was the
case with the flat-plate airfoil, this flow field has a singularity at the leading
edge. This singularity would not exist for airfoils of finite nose radius and
would not exist even for sharp leading edges because of separation of the
flow at the nose. In spite of this local inaccuracy, the results derived above are
representative of the flow around thin cambered airfoils.

4.18 Joukowski Airfoil

The results of the two previous sections suggest that a cambered airfoil of
finite thickness may be obtained by considering the Joukowski transforma-
tion in conjunction with a circle in the { plane whose center is in the second
quadrant. Such a configuration is shown in Figure 4.18a in which the center
of the circle is displaced a distance m from the origin at an angle 6 from the
reference axis. In order that the trailing edge of the corresponding airfoil
be sharp, the circumference of the circle passes through the critical point
{ = c. The principal parameters of the airfoil in the z plane are also shown in
Figure 4.18a. These are the chord /, the maximum thickness ¢, and the maxi-
mum camber of the centerline /.

z plane

FIGURE 4.18
Joukowski airfoil: (a) mapping planes and (b) uniform flow past the airfoil.
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From the previous two sections, it follows that, to the first order in ¢, the
centerline of the airfoil will be a circular arc whose center is on the y axis and
the airfoil will be symmetrical about its centerline. Then, the equation of the
upper and lower surfaces of the airfoil may be obtained from the equation
for the circular-arc centerline plus or minus, respectively, the thickness effect.
Hence, from Equations 4.23a and 4.24a, the airfoil profile will be given by

2

I? I? , I? X x
= |— 1 -x°-——=+0385¢ 1-2- ,[1- 2~ .
y \/4 +16h2 X s ;i : (4.25a)

where the plus defines the upper surface and the minus defines the lower
surface of the so-called Joukowski airfoil.

It was observed that the effect of thickness on an airfoil was to increase
its lift by an amount 0.77t/I and that the effect of camber was to increase the
effective angle of attack to a + 2/1/1 The present airfoil has both these effects,
so that, from Equations 4.23c and 4.24b, the lift coefficient for the Joukowski
airfoil will be

CL=2n 1+ 0.77; sin o + % . (4.25b)

The complex potential for the flow around this airfoil, in the { plane, is

] ) 2 iot T c_meiﬁ
FQ=U @m0 A0 g Eome? |
© (C—me®)e +;-me’5 +, log = (4.250)

where m cos 8 =-0.77(tc/l), msin 8 = h/2, and a = [/4 + 0.77(tc/I). These results
follow from the observation that —m cos & replaces m as used in the sym-
metrical Joukowski airfoil and m sin & replaces m as used in the circular-arc
airfoil. The magnitude of the circulation I" will include both the thickness
and camber effects, and so it follows that

T =nUl 1+0.77§ sin a+¥ .

The flow field corresponding to the foregoing complex potential is shown
in Figure 4.18b. It should be remembered that there is a limit to the amount
of thickness and camber that may be introduced if the flow field is to remain
as shown. As the thickness and/or camber of the airfoil increases, the
body departs more and more from a streamlined airfoil and approaches a
bluff body. It was pointed out earlier that a consequence of this would be
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separation of the flow, which destroys the lift force and creates the so-called
stall condition.

4.19 Schwarz—Christoffel Transformation

Another conformal transformation of prime interest in the study of poten-
tial flows is the Schwarz—Christoffel transformation. This transformation is
reviewed briefly in Appendix D, from which it will be seen that the mapping
function is the solution to the following differential equation:

% _ \/mn-l-  p\B/m-1(s- _ \y/m-1
dC—K(C a)"" (-7 (G -c)

where g, b, ¢, etc., are the locations in the { plane of the vertices of a polygon
in the z plane that subtend the internal angles o, f, y, etc. The quantity Kis a
constant, and normally three of the quantities 4, b, ¢, etc.,, may be chosen arbi-
trarily. The manner in which this transformation is used will be illustrated
through its application to a simple problem whose solution may be deduced
from previously established results. This will permit a direct and indepen-
dent check on the solution obtained through use of the Schwarz—Christoffel
transformation.

The problem to be considered is that of obtaining the complex potential
for the flow around a flat plate of finite length oriented such that it is perpen-
dicular to the oncoming flow; that is, the angle of attack is 90°. The solution
to this problem may be deduced from that of a vertically oriented ellipse,
which was treated in Section 4.14, by a limiting procedure. The length of the
plate obtained in Section 4.14 was 44, and that is the length of the plate that
we will consider here. The stagnation streamline will be a line of symmetry
for this problem, so that only one-half of the plate, say the top half, need
be considered. The stagnation streamline y = 0 is shown in Figure 4.19a
for the top half of a vertical plate of length 4a. The plate itself is considered
to be made up of the line ABC, which folds back on itself. The location of
the vertices A, B, and C in the z plane are shown as the points a, b, and c,
respectively, on the { plane. The points chosen for a, b, and ¢ are -1, 0, and
1, respectively.

The equation of the Schwarz—Christoffel mapping function is the solution
to the following equation:

dz 1720 _ (- _ 1)-1/2
dc—K(CH) €-0'@E-1



Two-Dimensional Potential Flows 137

(@ p=2m

(b)

FIGURE 4.19
Flow around a vertical flat plate assuming that the flow does not separate: (a) Schwarz—
Christoffel mapping planes and (b) flow field.

where the fact that « = n/2, § = 2r, and y = n/2, as indicated in Figure 4.19a,
has been used. That is, the equation of the mapping function is

or, integrating

z=K{Z*-1+D

where D is a constant of integration that is, in general, complex. The con-
stants K and D will now be evaluated such that the points A, B, and C and a,
b, and c correspond to each other. The conditions to be satisfied are

1. When({=1,z=0.
2. When(=-1,z=0.
3. When ¢ =0, z = i2a.
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The first two conditions are satisfied by taking D = 0, while the third condi-
tion is satisfied by choosing K = 24. Then, the required mapping function is

z=2a\C%-1.

The complex potential in the { plane is that of a uniform rectilinear flow
since the streamline y = 0 has been stretched out along the real ¢ axis. To
find the magnitude of the uniform velocity, it is observed from the mapping
function that as { — o, z — 2a{. Then, from Equation 4.19, W({) — 2aW(z), so
that for a flow of magnitude U in the z plane, the magnitude in the { plane
should be 2aU. Thus, the required complex potential is

F(¢) = 2aUL.

But from the mapping function

In order that { — + as z — +9, so that the direction of the flow is correct,
the positive root must be chosen. Hence, the required inverse of the mapping
function is

= Zi\/z2 +4a”.
a

The complex potential in the z plane then becomes

F(z) =UVzZ* + 4a*.

This result may be checked by using the result for the uniform flow of
magnitude U past an ellipse of major semiaxis (a2 + b?/a) and minor semiaxis
(a — b*/a), the latter being along the x axis. The resulting complex potential is
given by Equation 4.21b, which was obtained through use of the Joukowski
transformation. As b — a in this result, the ellipse degenerates to a vertical
flat plate of length 4a. Substituting b = 2 in the complex potential confirms the
result derived above by use of the Schwarz—Christoffel transformation. The
corresponding flow field is shown in Figure 4.19b. From this figure, or from
inspection of the complex potential, it will be seen that infinite velocities, of
the type discussed in Section 4.6, exist at y = +2a. Clearly, the Kutta condi-
tion cannot be applied in such a case, and so the fluid will separate from the
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two edges of the plate. That is, the complex potential derived here does not
represent the actual flow field accurately because the fluid does not remain
in contact with the plate as was implicitly assumed here. A more representa-
tive flow configuration for this problem will be analyzed later in this chapter.

4.20 Source in Channel

The Schwarz-Christoffel transformation may be used to solve a sequence of
problems related to that of the flow generated by a line source located in a
two-dimensional channel. Then, consider a channel of width 2/ and of infi-
nite length in which a source is located midway between the channel walls.
If the origin of the coordinate system in the z plane is taken to be at the loca-
tion of the source, it is clear that the resulting flow field will be symmetrical
about both the x axis and the y axis. Then, the entire x axis and the portion
-l <y <1 of the y axis will be streamlines, so that only the first quadrant of
the flow field need be considered; the remainder will follow from symmetry.
Figure 4.20a shows the first quadrant of the flow field in the z plane in which
the source is located at z = 0.

Considering the region 0 < x, 0 <y <, to be bounded by the polygon that is
to be mapped, the vertices A and B will be chosen to correspond to the points
¢ =-1and { =1, as shown in Figure 4.20a. The interior angles corresponding

(a) (d)
R
_I
x=35 ; 21
-
\{5 T2 -1 +1 21
B a | b 2]
(b) - - 21

FIGURE 4.20
(a) Mapping planes for a source in a channel, (b) flow field for a full or semi-infinite channel,
(c) flow field for the source at the wall, and (d) infinite array of sources.
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to the vertices A and B in the z plane are n/2, so that the differential equation
of the mapping is

% _ “1/2 _ 1y-1/2
& KC+1)"""(C-1)
_ K
-1

and hence, integrating

z=Kcosh™{+D
where D is a constant of integration. The constants K and D will now be
evaluated such that the point A corresponds to the point a and the point B

corresponds to the point b. The required conditions are

1. When({=1,z=0.
2. When (¢ =-1,z =il

The first condition is satisfied by setting D = 0, while the second condition
is satisfied for K = [/x. Then, the required mapping function is

z= L cosh™' ¢
n
which has the inverse

2;=cosh%.

The flow field in the { plane now corresponds to a source located at the
point { = 1. Hence, the complex potential in the { plane is

m
F@©) = logC-1)
T
so that the corresponding complex potential in the z plane is

L |

F(z)=%log cosh ;
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This result may be simplified slightly by using the identity
cosh (X +Y) —cosh (X -Y)=2sinh Xsinh Y

for X =Y = nz/(2l). Hence,

cosh™ -1=2ginh? ™ .
I 21

Thus, the complex potential may be written in the alternative form

m nz
F(z)=—1 2sinh? —-
) 27 08 8 21

m nz om
=—log sinh— +—log2.

x 8 2l Ton 8

But the constant term may be neglected since it does not affect the velocity
components. That is, the complex potential may be taken to be

Z

m
F(z)=—1 inh
(2) . og s o

4.26)

Equation 4.26 is the complex potential for the flow configurations shown
in Figure 4.20b through d. Figure 4.20b shows the flow field due to a source
that is located on the centerline of an infinitely long channel or at the center
of the end of a semi-infinite channel. Figure 4.20c shows the flow field due
to a source that is located on one wall of an infinite channel or at a cor-
ner of a semi-infinite channel. The foregoing flow configurations are clearly
related to the largest flow field, shown in Figure 4.20b, by symmetry. The
total quantity of fluid leaving the source is 4/U, so that the source strength m
in Equation 4.26 should be 4/U in order that the channel velocity in the four
configurations shown in Figure 4.20b and Figure 4.20c will be U.

Figure 4.20d depicts an infinite array of line sources spaced a distance 2/
apart. The horizontal lines that pass midway between each pair of sources
will obviously be streamlines for such an array of sources. It follows that the
case of a source located in a horizontal channel may be thought of as only
one component of an infinite number of such channels stacked on top of each
other in the vertical direction. Mathematically, the fact that Equation 4.26
represents an infinite number of sources spread in the y direction follows
from the fact that the hyperbolic sine function repeats itself for imaginary
values of its argument.
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4.21 Flow through Aperture

One of the most impressive applications of the Schwarz—Christoffel transfor-
mation, in the field of fluid mechanics, is in the study of streaming motions
that involve free streamlines. It is not usually known where these free
streamlines lie, and this information must come out of the solution. The key
to solving such problems is the so-called hodograph plane, which uses the fact
that along such free streamlines, the pressure is constant. Two examples will
be covered in this chapter, this first example being an application to the flow
through a two-dimensional slit or aperture.

Figure 4.21a shows, in the z plane, a horizontal plate with an opening in
it. The plate contains a semi-infinite expanse of fluid above it, and this fluid
is draining through the aperture that is defined by the section BB’ of the x
axis. At the corners B and B', the flow will locally behave like that for the flow
around a sharp edge, which was discussed in Section 4.6. It was pointed out
in that section that if the fluid remained in contact with the solid boundary,
infinite velocity components would result at the edge itself. Since this cannot
be physically so, the fluid will not remain in contact with the solid boundary

(a) (b)

Cc|C
critic
I I
z plane { plane
o B A’ I
i (O
I C I
T
B A
A B CIC_B _ A
-1 I +1
{’ plane (" plane
FIGURE 4.21

(@) Mapping planes for flow through a slit and (b) geometry of one of the free bounding
streamlines.



Two-Dimensional Potential Flows 143

but will separate at the edge. In the case under discussion, the bounding
streamlines along the horizontal plate will curve toward the vertical plane
of symmetry, as shown in Figure 4.21a. The magnitude of the velocity in
the resulting jet will reach some uniform magnitude U downstream of all
edge effects. The principal streamlines in the flow field have been labeled
for identification purposes. These are the bounding streamlines on the right,
identified by the points A, B, and C, the bounding streamlines on the left,
identified by the points A, B, and C’, and the central streamline II'. The free
streamlines are BC and B'C..

The first transformation will be to the hodograph plane, which will be desig-
nated the { plane here. The transformation will be taken to be

dz _u_ U (4.27a)
dF W 2 4+ ¢

c=u
That is, the { plane is defined by the nondimensional reciprocal of
the complex velocity, and the last equality follows from the fact that

W = u-iv=+u’+v°e". The significance of this transformation is that the
free streamlines, whose positions are unknown, are mapped onto the unit
circle in the ¢ plane, as will now be shown. In so doing, it should be noted
that, by the foregoing definition, 0 is the angle subtended by the velocity vec-
tor in the z plane.

Along the free streamlines BC and B'C’, the pressure will be constant, typi-
cally atmospheric pressure, so that, from Bernoulli’s equation, the quantity
u? + v2 will be constant. The value of this constant may be determined by
noting that away from the edge effects, the velocity in the jet is U. Hence,
the value of u? + v along the free streamlines is U2 Then, along the free
streamlines, { = ¢, which is the equation of the unit circle in the { plane. To
find the portion of this unit circle that represents the free streamlines, it is
observed that along the streamline A'B’, the angle 6 of the velocity vector is
0 or 2x, while the value of 6 along AB is n. Also, along the streamline II/, the
angle 0 of the velocity vector is 3n/2. From these observations, it is evident
that the lower half of the unit circle in the { plane represents the streamlines
BC and B'C’, as shown in Figure 4.21a. The other principal streamlines may
be identified as follows: Along A'B’, the value of 0 is 0 or 2z, while u* + v2
varies from 0 at A’ to U? at B’; hence, |{| varies from infinity at A’ to unity at
B'. Likewise, along AB, the value of |{| varies from infinity at A to unity at
B, with the value of 6 being n. Finally, along the streamline II’, the value of 6
is 3n/2, while u? + v? varies from zero at I to U? at I', making || infinity at I
and unity at I' This establishes the flow configuration shown in the { plane
of Figure 4.21a. Since the flow is toward the point { = —i, which is identified
by C, C’, and I there is a fluid sink there.

Since the principal streamlines in the { plane are either radial lines or the
unit circle, the flow pattern may be mapped into a plane configuration by
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means of a logarithmic transformation. Then, a second mapping is proposed
to the ' plane, where (' is defined by

¢ =log_. 4.27b)

If a point in the { plane is represented by its polar coordinates R and 6,
where R = U/(u? +v?)"/2, then { = Re®, so that

¢ =log {=1log R +i6.

Thus, the radial lines in the { plane become the horizontal lines defined by
' =log R + i x constant in the {' plane, while the unit circle R = 1 becomes
the vertical line {' = i0. Noting that the angle 0 is the angle subtended by
the velocity vector in the z plane, it follows that the value of 6 along A'B’
is 0 or 2m. This gives the flow configuration shown in Figure 4.21a in the {'
plane, which corresponds to the flow in a semi-infinite channel due to a sink
located at the center of the end of the channel. However, it was seen in the
previous section that such a configuration could be mapped into that of a
simple source. Using the results of the previous section, a simple source flow
will result in the " plane through the mapping

" = cosh (¢’ —in)
that is,
" =—cosh (. (4.27c)

Here, the rectangle ABCC'B’A’ has been taken as the equivalent of the half
channel of width ! that was considered in the previous section. Then, the
quantity / that appeared in the transformation is = in this case, and in order
to bring the corner B to the origin in the {’ plane, the quantity {’ — in rather
than ' is the appropriate variable.

The flow field in the {" plane is shown in Figure 4.21a. The complex poten-
tial in this plane will be that for a simple sink located at {" = 0, so that

FE )= —%log?; +K 4.27d)

where the constant K has been added to permit the streamline y = 0 and the
equipotential line ¢ = 0 to correspond to a chosen streamline and equipoten-
tial line, respectively. Referring to Figure 4.21b, it will now be specified that
the streamline y = 0 be the streamline II' and that the equipotential line ¢ = 0
passes through the points B’ and B. Then, using the property of the stream
function that the difference in the values of y between two streamlines equals
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the volume of fluid flowing between these two streamlines, the value of y
along A'B'C’, which will be denoted by g, may be identified. Considering
the flow between the streamlines II' and A’'B'C’, it follows that

0 - wape =ClU

where C_ is the contraction coefficient of the jet. Similarly, if y,pc is the value
of y along the streamline ABC, it follows by considering the flow between
the streamlines ABC and II' that

Wape — 0= CIU.
That is,
Wasc = CIU
and
Wape = —CJU

Then, at point B, ¢ = 0 and y = —-CJU. Hence, the value of the complex
potential there is 0 — iCJU. Applying this result to Equation 4.27d and noting
that ¢” = -1 at the point B’ give

m
0-iCJU = - L log(-1) + K
iC, o 081+
~iC U = -i L 4 K.
2

Likewise, at point B, the complex potential is 0 + iC /U and the value of ¢’
is unity; hence,

0+iCIU =-"log1+K
2n

or
ICCZU = K.

These two equations show that K = iCIU and m = 4C_U, so that the com-
plex potential (Equation 4.27d) becomes

2CIU
T

FE)=- logt +iC.IU.
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The corresponding complex potential in the z plane may be obtained by
use of the transformations (Equations 4.27a through 4.27c). This gives

2C.IU
T

F(z)=-

log cosh log U% -in  +iCJU. (4.27e)

This result is an implicit expression for F(z) rather than an explicit expres-
sion since dF/dz appears inside the expression for F(z). However, the flow
problem has, in principle, been solved, and it is possible to obtain useful
information from the result. The quantity that is of prime interest in this
problem is the value of the contraction coefficient C,, so this value will be
determined below.

In order to evaluate the contraction coefficient C,, the equation of the free
streamline B'C’ will be established. From this result, the value of x at point
C’ should be numerically equal to the half-jet dimension C . This will enable
the quantity C. to be evaluated. The equation of the free streamline B'C’ is
most readily established in terms of a coordinate s whose value is zero at
point B" and whose magnitude increases along B'C". Then, considering a
small element of a curve, such as the streamline B'C’, whose slope is positive,
it follows that

=cos0

& &

S

SX =X +f cosBds
0

where the constant x, has been added to permit the condition x = -/ when
s = 0 to be applied. The variation of ds with 0 is now required, and owing to
the implicit nature of the mapping function (Equation 4.27a), this variation
must be obtained by indirect methods as follows: The preceding expression
for the lateral displacement x of the jet surface may be written as

0 ds d
X=X, +f2ncosﬁfd—t‘éd6

where the quantities ds/d{” and d{"/d® must be expressed in terms of 0
before the integration may be performed. The value of d{"/d6 on B'C" will be
obtained from the equations of the mappings, while ds/d¢” will be obtained
from the complex potential F({"). Considering first the value of ds/d{’, it may
be stated that, on the streamline B'C’
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u dz
1o -
w dF
|y dzde |
dt dF
But from Equation 4.27d with m = 4C U,
dr _ 2CIU 1
dg n G
dz =
S1=U—— .
dc 2C.IU <
On B'C, ¢’ <0, so that
dz) 2C11
g | = ¢’

Now, on the streamline B'C’, the value of dz may be represented by —ds ¢,
where ds is an element of the coordinate s, which was previously introduced.
Also, along B'C’, {" is increasing so that d¢” > 0. Hence,

ds 2C.1

1
e =x T’

The equations of the various mappings may now be used to evaluate {” in
terms of 6. On the streamline B'C/, the value of { is

C=ei9
~.C =logCT=1i0

and

T =-coshC =-cos6
dg

S.——=sinB
do
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and

i_ZCJ 1
dc T cosO’

Using these last two equations, the expression for the lateral displacement
x of the free streamline becomes

0

X=X, +f coseﬁsinede
2n mcos O
0

=X, + chlf sin6d6
TT 20

2C
7

=X, + g (1-cos®).

But when 0 = 2r, that is, at the point B/, x = -I. Hence, x, = -/, so that
x=-l +£‘l(1— cos 0).
7

Also, the value of x at the point C' is —C_, while the value of 0 is 3n/2. Thus,

—Ccl=—l+£°l
n

4.27f)
T

C

T2

Equation 4.27f predicts that the free jet that emerges from the aperture
will assume a width that is 0.611 of the width of the slit. This result is well
established experimentally, and the figure of 0.611 has been confirmed for
openings under deep liquids.

4.22 Flow Past Vertical Flat Plate

In Section 4.19, the complex potential for the flow around a flat plate that
is oriented perpendicular to the free stream was obtained. However, it was
pointed out that the result obtained at that time was unrealistic because it
required infinite velocity components at the two edges of the plate. It would
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therefore appear that the assumption of attached flow, which was implic-
itly made at that time, is not valid. The same problem will be treated here,
but this time, it will be assumed that the flow separates from the surface of
the plate at the two edges. The resulting free streamlines will be treated in
a manner similar to that in which the free jet was treated in the previous
section.

Figure 4.22 shows, in the z plane, the assumed flow configuration for a uni-
form rectilinear flow of magnitude U approaching a vertically oriented flat
plate of height 2I. The stagnation streamline II' splits upon reaching the plate
and forms the bounding streamlines ABC and A'B'C’, where BC and B'C’ are
the free streamlines. The region downstream of the plate between the two
free streamlines is interpreted as being a cavity that has a uniform pressure
throughout. The point I is on the surface of the plate.

As in the previous section, the free streamlines may be handled by use of
the hodograph plane. Hence, a transformation is made to the { plane, where

+1 \\
I 1 C’ I

1 -

puglsl A
2
[ _IC A B _ClIC_B C
A€ 11w
2 A
(' plane {"plane
r e, B AT rag B o
c B A A B C
+1 +1
(" plane (" plane

FIGURE 4.22
Mapping planes for flow over a flat plate that is oriented perpendicular to a uniform flow.
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coyd_u__u

dE-W i

The boundaries of the flow field in the { plane are shown in Figure 4.22.
The free streamlines BC and B'C" again become part of the unit circle in
the { plane. Since the value of 6 along ABC lies between n/2 and 0 and the
value of 6 along A'B'C’ lies between —n/2 and 0, the appropriate portion of the
unit circle is that which lies in the first and fourth quadrants. Since the flow
boundary crosses the positive portion of the real { axis, it is no longer con-
venient to consider the range of 0 to be 0 < 6 < 2x. If this range were adopted,
the multivalued functions, which require branch cuts, would divide the
flow boundary. This difficulty may be simply overcome by considering the
branch cut to lie along the negative real { axis so that the principal value of
the multivalued functions will correspond to -t <0 <.

The geometry of radial lines and circular contour is next converted to that
of a plane figure by means of a logarithmic transformation. That is, a map-
ping to the {’ plane is made where

e’ (4.28a)

¢ =log_. (4.28b)

This maps the flow boundary into that of a rectangular channel, as shown
in Figure 4.22. Since the range of 0 is now —xn < 6 < «, the lower wall of this
channel corresponds to the imaginary part of {' being — /2, and the upper
wall corresponds to +n/2. That is, the centerline of the channel corresponds
to the real {’ axis. Then, the flow field may be stretched out using the same
transformation as was used in Section 4.20 for a source in a channel. Here,
the corner B’ is located at {’ = in/2 rather than {’ = 0, and the channel half
width is m instead of . Hence, the required transformation is

€ =cosh © +ig . (4.28¢)

The principal flow lines in the {” plane may be made collinear by means
of the mapping

=@ (4.28d)

This doubles the angles subtended by the principal streamlines, so that
the flow in the {'” plane is unidirectional along the principal streamlines, as
shown in Figure 4.22. However, the flow is still not that of a uniform flow as
the principal streamlines might suggest. This may be confirmed by observ-
ing that, in the z plane, there is a source of fluid at I that flows toward a sink
at CC'. However, in the {'” plane, CC’ and I are at the same location, so that
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the flow in the {'” plane is probably that of a doublet. Rather than prove this
is so, one final transformation will be made to the {"” plane where

T o= . (4.28¢)

The effect of this transformation is to map the origin to infinity, and vice
versa, as shown in Figure 4.22. Fluid emanates from I and flows toward CC/,
as was the case in the z plane. That is, the flow in the {"” plane is that of a
uniform rectilinear flow so that the complex potential is

F(C")= K¢

The value of the constant K, which represents the magnitude of the uni-
form flow, would normally be obtained by relating complex velocities
through Equation 4.19. However, the implicit nature of the hodograph trans-
formation prohibits this being done directly, so that indirect methods must
be used, as in the previous section. The hodograph transformation involves
dF/dz, but F is known only as a function of {"”. Hence, it is proposed to start
with the hodograph transformation and express the variables in terms of {".
Also, F(¢"”) is known, so that F({") may be calculated, and so an identity will
be established in the " plane. The details now follow.

From Equation 4.28a, the following identity is established:

dz
U=t
. 7d2d£=
“UdC C.

Here, dF/d{" may be evaluated from the complex potential F({) and the
mapping functions, while { may be evaluated from the mapping functions.
Both quantities will be expressed in terms of {":

F@€ )=Kg
K
.. F =
€) c
and
K
E =
©) € )
dF 2K

& ey
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also
C=ef

-1 .
- e(cosh T —in/2)

- _iecosh‘1§ .

But cosh™ x = log(x ++/x> = 1), so that
C=-iC +@C ) -1).

Substituting the preceding expressions for dF/d{” and ¢ into the identity
established from the hodograph transformation gives

wE N ey )

In order to establish an algebraic identity that will permit the constant K to
be evaluated, the preceding expression must be integrated. It is proposed to
integrate over the region B’ to A’ so that

[ \2
Udz = iz[(%d@

)

where the upper limits of integration correspond to the point A’ and the
lower limits to the point B’ The integral on the right-hand side may be con-
veniently evaluated by use of the substitution {” = 1/sin v. Then,

0 0
Uf dz = —i2Kf (1+ cosv)cosvdv
-l /2
~illl=i2K 1+2%
4

or

2Ul
44
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2

Then, the complex potential in the {"” plane becomes

2Ul
w+4

FE )= C

The corresponding complex potential in the z plane may be obtained by
using the mapping equations (Equations 4.28a through 4.28e). Hence, using
the fact that cosh ({' + in/2) =i sinh {, the corresponding expression for F(z) is

2Ul 1
n+4 sinh?{log[U(dz/dF)]}

F(z) = - (4.28f)

This result is again an implicit expression for F(z) rather than an explicit
expression. However, since the flow problem has been solved, results may be
deduced from the solution. Here, the result of interest is the drag force acting
on the plate. Thus, if X is the drag force acting on the plate in the positive x
direction and P is the pressure in the cavity, it follows that

X=2f_l(p—P)dy

where the symmetry of the flow field about the x axis has been invoked. But
from the Bernoulli equation

E+l(u2+vz)=£+luz
P2 p 2

where the Bernoulli constant has been evaluated on the free streamlines at
a position well downstream of the plate. Thus, the expression for the drag
force X becomes

0
X = ZJ:I%p[UZ - +0%)]dy

0 0
- pl,sz_ldy—pf_l(u2 + 07 )dy.

The first integral may be evaluated explicitly, while the integrand of the
second integral may be written as v? since u = 0 on the surface of the plate.
Then W = dF/dz = —iv on the surface of the plate, so that v> = -IW? there.
That is,
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2

dy.

&[&

0
X=pu21+pf
-1

Also, x = 0 on the surface of the plate, so that z = iy there. That is, dz = idy
on the plate, so that

2

0
X=pllzl—ipf % dz.
—il

Now, since F(z) is an implicit expression, it is proposed to evaluate F({") and
to perform the indicated integration in the {” plane rather than the z plane.
This may be done as follows:

= dF dc

X=olll-ipf £ <L g
p lpf1 & @ g

where it has been noted that as z varies from —il to 0, " varies from unity to
infinity. But expressions for dF/d{" and dz/d{" were obtained earlier in terms
of " and K. Then, using these expressions and the fact that K = 2Ul/x + 4), the
expression for the drag force becomes

Lo a1 med)E )
X = plU%l - -
b lpfl T4 Q) e +yE ) -1]
4pUPl de
m+d Ji @ Pl +4E ) -1]

dc

= pU?l

The integral may now be evaluated by means of the substitution " = 1/sin v.
This gives

0
X =pU? 1+ 4 f (1-cosv)cosvdv
T+ 40
(4.28g)
X =2 ol
n+4

From the symmetry of the flow field about the x axis, it may be stated
that there will be no lift force acting on the plate. On the other hand, the
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lack of symmetry about the y axis implies the existence of a drag force, and
Equation 4.28g gives the magnitude of this drag force.

PROBLEMS

4.1 For the complex potential F(z) = ¢ + iy = cos™'z, turn this expression
around by solving it for z as a function of ¢ and y, and hence obtain
implicit expressions of the form x = x($,y) and v = y(d,y). Thus, obtain
the following results:

(@) Eliminate ¢ between the two equations obtained above to get
a single equation of the form f(x,),) = constant that defines the
points on the streamlines y(x,y) = constant in terms of the coor-
dinates x and y.

(b) Sketch three typical streamlines corresponding to y = 0, y =
vy >0, and y =y, > ;.
() Obtain an expression for the complex velocity at any location z

and hence determine the flow direction on these streamlines by
evaluating the velocity at the point x =2 and y = 0.

4.2 For the complex potential F(z) = ¢ + iy = cosh™'z, turn this expression
around by solving it for z as a function of ¢ and y, and hence obtain
implicit expressions of the form x = x($,y) and vy = y(d,y). Thus, obtain
the following results:

(@) Eliminate ¢ between the two equations obtained above to get
a single equation of the form f(x,),) = constant that defines the
points on the streamlines y(x,y) = constant in terms of the coor-
dinates x and y.

(b) Sketch three typical streamlines corresponding to y = 0, y =
vy >0, and y =y, > ;.
() Obtain an expression for the complex velocity at any location z

and hence determine the flow direction on these streamlines by
evaluating the velocity at the point x = 0.5 and y = 0.

4.3 Write down the complex potential for a source of strength m located
at z = ih and a source of strength m located at z = —ih. Show that the
real axis is a streamline in the resulting flow field, and so deduce
that the complex potential for the two sources is also the complex
potential for a flat plate located along the line y = 0 with a source of
strength m located a distance h above it.

Obtain the pressure on the upper surface of the plate mentioned
above from Bernoulli’s equation. Integrate the pressure difference
over the entire surface of the plate, and so show that the force acting
on the plate due to the presence of the source is pm?/(4nh). Take the
pressure along the lower surface of the plate to be equal to the stag-
nation pressure in the fluid.
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4.4 Consider a source of strength m located at z = —b and a sink of
strength m located at z = b. Write down the complex potential for
the resulting flow field, adding a constant term —im/2 to make the
streamline y = 0 correspond to a certain position. Expand the result
for small values of z/b and hence show that if b — « and m — « in
such a way that m/b — =nll, the resulting complex potential is that
of a uniform flow of magnitude U. That is, a uniform flow may be
thought of as consisting of a source located at —~ and a sink of equal
strength located at +co.

4.5 Write down the complex potential for the following quantities:
(@) A source of strength m located at z =-b
(b) A source of strength m located at z = —a%/b
(©) A sink of strength m located at z =0
(d) A sink of strength m located at z = a2/b
(e) A constant term of magnitude —im/2

Expand the resulting expression for small values of z/b and a%/
(bz), and hence show that if b — o and m — o in such a way that
m/b — nl, the resulting complex potential is that of a uniform flow
of magnitude U flowing past a circular cylinder of radius a.

4.6 Consider a system of singularities consisting of the following:
(@) A source of strength m located at z =-b
(b) A source of strength m located at z = —a%/b
(©) A sink of strength m located at z = a2/]
(d) A sink of strength m located at z =
(e) A constant term of magnitude —[m/(2n)]log b

Write down the complex potential for this system and let b — oo,
Show that the result represents the complex potential for flow
around a circular cylinder of radius a due to a sink of strength m
located a distance [ > a to the right of the center of the cylinder. This
may be done by showing that the circle of radius a is a streamline.

Use the Blasius integral theorem around a contour of integration
that includes the cylinder, but excludes the sink, and hence show that
the force acting on the cylinder, due to the presence of the sink, is

_ pmiat
2nl(* - a?)

4.7 A system of flow singularities consists of the following components:
(@) A source of strength m located at z = bei®+™
(b) A source of strength m located at z = (42/b)ei**?
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4.8
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410

(c) A sink of strength m located at z = (a%/b)e™
(d) A sink of strength m located at z = be'®
(e) A constant term of magnitude —im/2

Write down the complex potential for this system and expand it
for small values of z/b and a%/(bz). Hence, show that the complex
potential for a uniform flow of magnitude U approaching a circular
cylinder of radius a at an angle of attack a to the horizontal is

2

F(z)=U ze™ + & g . (4.29)
z

Determine the complex potential for a circular cylinder of radius a
in a flow field produced by a counterclockwise vortex of strength
I' located a distance ! from the center of the cylinder. This may be
done by writing the complex potential for the following system of
singularities:

(@) A clockwise vortex of strength I" located at z = b

(b) A counterclockwise vortex of strength I" located at z = —a2/b

(©) A clockwise vortex of strength I" located at z = a%/1

(d) A counterclockwise vortex of strength I" located at z = [

(e) A constant term of magnitude — [iI'/(2n)] log b

Then, let b — « and show that the circle of radius a is a streamline.
Obtain the value of the force acting on the cylinder due to the vortex
at z = | by applying the Blasius law to a contour that includes the
cylinder but excludes the vortex at z = I.

The complex potential for a uniform flow of magnitude U approach-
ing a circular cylinder of radius a that has a bound vortex of strength
I' around it is

@ T, z
Fz)=U z+— +—log—.
@ z 2m ga

Using this result, together with Bernoulli’s equation, obtain an
expression for the pressure p(a,0) on the surface of the cylinder.
Integrate the quantity —p(a,0) a sin 6 around the surface of the cyl-
inder, and hence verify the validity of the Kutta—Joukowski law for
this particular flow.

It is proposed to model the steady-state flow due to a tornado by
superimposing a sink and a counterclockwise irrotational vortex,
both of which are located at the origin of the coordinate system. If
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the strength of the sink is denoted by m and the strength of the vor-
tex by I, determine the following quantities:

(@) The velocity potential ¢ and the stream function y in terms of
cylindrical coordinates (R,0).

(b) The cylindrical velocity components u; and 1, in terms of cylin-
drical coordinates (R,0).

() The pressure p in terms of cylindrical coordinates (R,0). Take the
pressure far from the origin to be p,, and neglect gravitational
effects.

411 In order to analyze the flow around a circular cylinder that is close
to a wall, superimpose the complex potentials for the following two
configurations:

e A uniform flow of magnitude U approaching a circular cylinder
of radius a whose center is located a distance : above the x axis

* A uniform flow of magnitude U approaching a circular cylinder
of radius a whose center is located a distance / below the x axis
For the resulting superimposed flow, obtain expressions for the
following quantities:

(@ The horizontal velocity along the x axis—that is, the value of
u(x,0).

(b) The locations of any maxima and minima of the velocity
obtained in (a) above, including the values of the velocities at
these locations.

() An expression for the pressure distribution along the x axis—
that is, the value of p(x,0). Take the pressure far from the origin to
be constant and equal to p,.

412 The figure for this problem shows the (a) assumed configuration
for separated flow of magnitude U approaching a circular cylinder
of radius a. (b) Shows a system of flow elements that are proposed

()

(b)
my
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// 2\
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to model this particular flow. The model consists of the following
components:

(@) A source of strength 1, located at z = ae™
(b) A source of strength m, located at z = ae™
(© A sink of strength m, located at z = ae™

(d) A sink of strength m, located at z = ge™
(e) A sink of strength m; located at origin, z =0
For this flow model, determine the following quantities:
(i) Thestrength of the sink m; that makes the circle R = a a streamline

(ii) The magnitude of the velocity on the surface of the cylinder,
q@,)
(iii) The parameters 1, and 0, using the following two conditions:

d
d—Z(a,eS>=o

2

d
d—GZ(a,eS)=cU R}

In the above, 65 is the polar angle to the location of the separa-
tion point, ¢ is an experimentally determined constant, and R, is the
Reynolds number. Express the results in the following form:

My =1, (mlr 95/ 01, 92) f(ezz 0y, 95/ my, 1My, C, Re) =0.

That is, obtain an explicit expression for m,, but the result for 0,
may be left as an implicit expression.

4.13 A mapping function is defined by the following equation:

n

c
z=C+—
- (n-1g"
where

z=x+ iy = Re®

and

{=¢&+in=pe™.
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In the above, c is a real constant and 7 is an integer. Find the loca-
tion of the critical points of this mapping function in the { plane and
sketch their locations forn =2, 1n =3, and n = 4.

Using the Cartesian representation of z and the polar representa-
tion of ¢, find the equation of the mapping in the form:

x=x(p,v) and y=y(p, v).

From these equations, obtain expressions for the polar coordinates
R and 6 of the form:

R =R(p, v) and 0 =0(p, v).

Consider a circle in the { plane whose radius p is large so that

£=£<<1.
p

Use the results obtained above for R and 0 to find the equations for
the mapping of this circle in the z plane, working to the first order
only in the small parameter e.

Using any of the results obtained above, sketch the resulting object
shape in the z plane fore <1, n = 3.

4.14 The Joukowski transformation, in conjunction with a circle in the ¢
plane whose center lies in the second quadrant, yields an airfoil in
the z plane. It is required to prepare an accurate drawing of such an
airfoil in the z plane.

Using either (a) or (b) below, depending on the preferred system of
units, prepare a table of values of points that lie on the surface of air-
foil. Use the results quoted in Section 4.18 to calculate the points (x, )
in the z plane, ensuring that a sufficiently large number of points are
used so that a smooth curve is obtained for the airfoil surface. Use
the table of points to draw the figure that is produced in the z plane.

(@) SI units:
Joukowski parameter, ¢ = 60.0 mm
Center of circle in ¢ plane = (-5.0 mm, +7.5 mm)
Circle to pass through point = (+60.0 mm, 0.0 mm)
(b) English units:
Joukowski parameter, ¢ = 2.4 in.
Center of circle in ¢ plane = (-0.2 in., +0.3 in.)
Circle to pass through point = (+2.4 in., 0.0 in.)
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4.15 Using either (a) or (b) below, depending on the preferred system of
units, determine the lift force generated by a short span of aircraft
wing whose cross section is the same as that of Problem 4.14. Take the
air properties to be defined by the standard atmosphere at sea level.
(@) SI units:
Length of wing element = 1.0 m
Chord of wing element = 3.0 m
Flight speed =250 m/s

(b) English units:
Length of wing element = 3.0 ft
Chord of wing element = 9.0 ft
Flight speed = 750 ft/s

4.16 Find the transformation that maps the interior of the sector 0 < 0 <
n/n in the z plane onto the upper half of the { plane. Thus, by consid-
ering a uniform flow in the { plane, obtain the complex potential for
the flow in the sector 0 < 0 < n/n in the z plane.

4.17 Use the Schwarz—Christoffel transformation to find the mapping that
transforms the interior of the 90° bend shown in the z plane of the
tigure for this problem onto the upper half of the { plane as shown.

Hence, obtain the complex potential for the flow around a right-angled
bend in terms of the channel width / and the approach velocity U.

<
=

<

A x c b a 13
z plane { plane

4.18 The z plane of the figure for this problem shows one-half of a sym-
metric expansion device, or diffuser. It is assumed that the angle ¢
is not large and that the flow remains in contact with the wall. In the
¢ plane, the points specified by lowercase letters correspond to the
points indicated by the capitalized letters in the z plane. The loca-
tion of the point ¢, indicated by the value o, is undefined at this time.
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t H —U

>
=~}
<

Using the correspondence indicated, find the differential equation
of the mapping function if the angle ¢ is taken to be the ratio of two
integers times 90°; that is,

rm
b= n2
where r and # are integers. Express the result in terms of the param-

eters r, n, «, and K, where K is the scale parameter in the Schwarz-
Christoffel transformation.

Noting that the strength of the source in the { plane is m = 2UH,
obtain the complex potential for the flow field in the { plane. From
this result, obtain an expression for the complex velocity in the z
plane, expressing the result in terms of the variable { and the param-
eters U, o, 7, n, and K.

Use the result obtained above to evaluate the parameters K and o,
expressing them in terms of the remaining parameters , H, r, and n.

4.19 The figure for this problem shows a channel with a step in it in the z
plane. Show that the mapping function that maps the interior of this
channel onto the upper half of the { plane is

z—Elo 1+s —ﬁlo H/h+s
T n & 1-s H & H/h-s

where

o C-(H/MW?
-1
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Let the points A, B, and C in the z plane correspond to the points 0,
1, and o, respectively, in the { plane. The quantity a may not be speci-
tied a priori, but it should be determined from the mapping function
after the points A and B have been located as desired. Note that the
streamline ABCD may be considered to be the streamline dividing
two symmetrical regions, so that this mapping function also applies
to a channel of width 2H with an obstacle of width 2(H — h) located
along the centerline of the channel.






5

Three-Dimensional Potential Flows

Although there are no significant phenomena associated with three-dimensional
flows that do not exist in two-dimensional flows, the method of analyzing
flow problems is completely different. The method of employing analytic
functions of complex variables cannot be used here in view of the three-
dimensionality of the problems. Then, we must resort to solving the partial
differential equations that govern the variables of the flow field. These par-
tial differential equations were reviewed at the beginning of Part II, and for
irrotational motion, the equation governing the velocity potential is given by
Equation IL5. After solving the flow problem for the velocity potential ¢, the
pressure may be obtained from the Bernoulli equation, which is expressed
in Equation IL6.

The chapter begins by reviewing the equation that is to be satisfied by the
velocity potential in spherical coordinates. Then, it is shown that for axi-
symmetric flows, a stream function exists, called the Stokes” stream function.
Although the flow fields may be solved through the velocity potential, the
stream function is useful for interpreting the flow fields. Fundamental solu-
tions are then established by solving the Laplace equation for ¢ by separation
of variables. These fundamental solutions are then superimposed to establish
the flow around a few three-dimensional bodies, including the sphere and a
family of solid bodies known as Rankine solids. A study of the forces that act
on three-dimensional bodies is then made, which leads to d’Alembert’s para-
dox. This paradox contends that for a body immersed in a potential flow, no
forces exist on the body, in spite of the fact that forces are observed to exist
experimentally. The chapter ends by introducing the notion of an appar-
ent mass for a body in a potential flow. This concept allows the fluid to be
ignored if a certain additional mass is associated with the body when its
dynamics are considered.

Since bodies of interest, such as airship hulls and submarine vehicle hulls,
have an axis of symmetry, this chapter will consider only three-dimensional
bodies that are axisymmetric. In so doing, it will be found convenient to work
in spherical coordinates (r, 6, ®). These coordinates are shown in Figure 5.1.
Since the axis of symmetry of bodies is invariably in the streamwise direc-
tion, and since the approaching flow is normally taken to be horizontal, the
reference axis of the coordinate system is also taken to be horizontal. Then,
in terms of the spherical coordinates (r, 6, »), a point P may be represented by
its radius r from the origin, the angle 6 between the reference axis and the
radius vector 7, and the angle o subtended by the perpendicular line to the

165



166 Fundamental Mechanics of Fluids

Reference
axis

FIGURE 5.1
Definition sketch of spherical coordinates.

reference axis that passes through P. For axisymmetric flows, there will be
no variation in the fluid properties as o varies from 0 to 2r while ¥ and 0 are
held constant.

5.1 Velocity Potential

Although the topic of discussion is three-dimensional flows, these flow fields
are supposed to be potential. That is, the fluid motion is assumed to be irro-
tational so that a velocity potential exists, irrespective of the dimensionality
of the flow field. Then, the equation to be satisfied by the velocity potential is
Laplace’s equation, given by Equation IL.5. Hence, expanding the Laplacian
in spherical coordinates and using the fact that d/dw = 0 for axisymmetric
flows, the equation to be satisfied by ¢ is

L L N SR 5.1)
r? or ar r2sin O 90 a0

The velocity components are related to the velocity potential by Equation
I1.4, which in spherical coordinates gives

u, = 9 (5.2a)
ar
19
=——. 2b
Uy 90 (5.2b)

The third velocity component, u,, is zero for axisymmetric flows.
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5.2 Stokes’ Stream Function

In the previous chapter, a stream function was introduced that, by its defi-
nition, satisfied the two-dimensional continuity equation. In three dimen-
sions, it is not possible, in general, to satisfy the continuity equation by a
single scalar function. However, in axisymmetric flows, such a function does
exist. The continuity equation for the incompressible case under consider-
ation is, for axisymmetric flows,

——(ru)+
g )

9
9, sin @) = 0.
roing op oS

Now, consider the velocity components to be related to a function y in the
following way:

__ 1 oy
" +%sin6 90

(5.3a)

1 (5.3b)
rsin® Jr

Direct substitution shows that if the velocity components are defined in
this way, the continuity equation will be identically satisfied for all functions
y. It will now be shown that the quantity 2my corresponds to the volume of
fluid crossing the surface of revolution that is formed by rotating the posi-
tion vector OP, in Figure 5.1, around the reference axis. This statement will be
proved in the following way. First, the statement will be assumed to be true
and to constitute the definition of a quantity . Then, it will be shown that
as a result of this definition, the velocity components must be related to this
function y by Equations 5.3a and 5.3b.

Let a function y be defined such that if the position vector OP is rotated
around the reference axis, that is, if the coordinate w is varied through 2x
while r and 0 are held fixed, the quantity of fluid that crosses the surface of
revolution formed by the vector OP will be 2my. Now, apply this definition to
two points P and P’ that are close together, as shown in Figure 5.2. Then, if
the line element PP’ is rotated about the reference axis, the resulting surface
will have a quantity of fluid 2r dy crossing it per unit time. However, refer-
ence to Figure 5.2 shows that a quantity of fluid u,r d® — 1, dr crosses a unit
area of this surface so that

2ndy = 27 sin O(u,rd0 — u,dr)

oo dy = u,7* sin 0d0 — uyr sin Odr.
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Reference
axis

FIGURE 5.2
Velocity components and flow areas defined by a reference point P and neighboring point P".

But if y is a function of both r and 6, it follows from differential calculus
that

dy =% a0+ ¥ g
20 ar

Comparing these two expressions for y shows that

oy

— =u,r*sin0
99

d .

N _ —Uyr sin 0.
ar

This confirms Equations 5.3a and 5.3b, so that the definition of y agrees
with the requirements for satisfying the continuity equation. Then, it may be
concluded that for the stream function defined by Equations 5.3a and 5.3b,
the volume of fluid crossing an element of surface generated by revolving a
line element about the reference axis is 2r dyy.

The stream function defined above is known as the Stokes” stream func-
tion. It will be used here in an auxiliary way only since flow solutions will
be obtained through solutions for the velocity potential ¢. The equation
that must be satisfied by y is therefore deferred to Problem 5.1 at the end of
the chapter. It should be pointed out, however, that for rotational flows, the
velocity potential does not exist, and the stream function then offers the only
mechanism for reducing the vector equations of motion to a single scalar
equation.
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5.3 Solution of Potential Equation

The equation to be satisfied by the velocity potential ¢ has been established.
Rather than attempting to solve this equation as part of a boundary-value
problem for various physical situations that may arise, it is proposed to
obtain here a general form of solution by separation of variables. The funda-
mental solutions so obtained will subsequently be superimposed to produce
more complex solutions in a manner similar to that which was used in the
previous chapter.

The velocity potential will be a function of ¥ and 0 only for axisymmetric
tlows, and so a separable solution will be sought of the form

o(r, 0) = R(r) T(O).

Substituting this assumed form of solution into Equation 5.1 gives

Td ,dR R d . dT
— e T + —~ - sinf—— =0.
r*dr  dr  r?sin6 d6 de

This equation may now be reduced to a separated form by multiplying it
by r2/(RT).

1d ,dR 1 d . dT
—— r— =—=———— sinf—
R dr dr TsinB do de

The usual argument of separation of variables is now invoked. The left-
hand side of this equation is a function of r only, and the right-hand side is
a function of 0 only. Hence, if either r or 0 alone is changed, one side of the
equation will change while the other does not. Then, the only way the equa-
tion can remain valid is for each side to be equal to a constant, say /(/ + 1).
Then,

—— r—— =lI+1
drrr (@+1)

and

- 1 d sin@d—T =I(I+1).
Tsin® d6 de
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The significance of choosing the separation constant as [(I + 1), rather than
simply B, is that with this choice, the resulting ordinary differential equa-
tion for T(0) appears in standard form, and so a subsequent transformation
becomes unnecessary. For the time being, there is no implication that the
quantity I need be an integer. The differential equation to be satisfied by R(r) is

d ,dR
— r*— —Ill+DR=0.
dr g dr (+1)

This is an equidimensional equation, and so its solution will be of the form
R(r) = Kr~.
Substituting this form of solution into the differential equation gives
a(+ DKre=Il+ DK =0

which will be satisfied by a =l and a = —(I + 1). Then, the general solution for
R(r) will be a linear combination of these two solutions. That is,

B,

rl+l :

R/(r)=Ar' +

Since this is a valid solution for any choice of /, the arbitrary constants A,
and B, have been assigned subscripts to indicate which value of I is being
considered. Likewise, the solution R/(r) has been assigned a subscript to indi-
cate which solution is being considered.

The equation for T(6) is
1 d dT
— sin6— +[(+1)T =0.
snodo "0 HIHD

This is Legendre’s equation, and it may be reduced to its standard form by
the transformation x = cos 6, which yields

% (1—x2)% +1(1+ 1T = 0.

The solutions to this equation are Legendre’s function of the first kind,
denoted by Pj(x), and Legendre’s function of the second kind, denoted by Q; (x).
Thus, the general solution for T(0) is



Three-Dimensional Potential Flows 171

T, ©) = C,P, (cos 0) + D,Q, (cos 6).

The Qj(cos 0) functions diverge for cos 6 = 1 for all values of I. The coef-
ficient D; must then be specified as being zero since there should be no sin-
gularities in the flow field. Also, Pj(cos 0) diverges for cos 6 = +1 unless / is an
integer. Then, it must be specified that the quantity / be an integer, so that the
continuous spectrum of separation constants /(/ + 1) now becomes a discrete
spectrum.

Combining the solution for R;(r) with that for T}(6) gives the following solu-
tion for ¢, (cos 0):

1+1

o, (r,0)= A7 +ri P(cos 0)

where the arbitrary constant C, has been absorbed into the two other arbi-
trary constants A, and B,. This solution is valid for any integer [. Then, since
the partial differential equation being solved is linear, all such possible solu-
tions may be superimposed to yield a more general type of solution. That is,
¢(r, 6) may be considered to be the sum of all possible solutions ¢(r, 0). Hence,

o(r,0) = Z Ar'+ %ll P(cos0). (5.4)

The Legendre function of the first kind that appears in Equation 54 is
defined by

1 d

P _ Il
1) 2111 dy!

(x* =1).

In view of the nature of this function, it is frequently referred to as
Legendre’s polynomial of order I. The first three Legendre polynomials are writ-
ten out explicitly below for reference purposes:

B(x)=1
P(x)=x
Py(x) = l(3x2 -1).
2
Equation 54 contains certain fundamental solutions that are useful for

superimposing to establish additional solutions. These fundamental solu-
tions will now be studied.
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5.4 Uniform Flow

One of the solutions contained in Equation 5.4 corresponds to a uniform
tlow. It may be obtained by setting

B,=0 for all /

0 forl=1
u forl=1.

;=

Using the fact that P, (cos 0) = cos 6, the solution given by Equation 5.4 then
becomes

¢ (r, 0) = Ur cos 0. (5.5a)

The simplest way of confirming that the velocity potential given by
Equation 5.5a corresponds to a uniform flow is to note that the Cartesian
coordinate x is related to the spherical coordinates r and 6 by the relation x =
r cos 0. Thus, Equation 5.5a states that ¢ = Ux, which is the velocity potential
for a uniform flow of magnitude U.

The stream function for a uniform flow may be deduced from Equation 5.5a
as follows: Using the result (Equation 5.5a), it follows from Equation 5.2a that

u, = % =U cos0.
ar

Hence, from Equation 5.3a,

21. a—w—llcosﬁ
r“sin@ 90

= %l,lr2 sin® 0 + f(r)

where f(r) is any function of r. Likewise, the velocity component u, may be
evaluated from ¢ and expressed as a derivative of y, giving

1 oy
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or
1 2 .2
P =§Ur sin” 0 + g(0)

where g(0) is any function of 6. Then, comparing these two expressions for y,
it follows that f(r) = g(0) = constant at most. Taking this constant to be zero,
without loss of generality, gives

1 .
lp(r, 9) = EUT’Z sz 0. (55b)

An alternative way of evaluating y(r, 0) is simply to invoke its definition.
Then, considering an arbitrary point P in the fluid as shown in Figure 5.3,
the amount of fluid crossing the surface generated by OP due to the uniform
flow will be 2my. However, the flow area perpendicular to the velocity vector
is n(r sin 6). Hence, it follows from the definition of y that

2ny = Un (r sin 0)?

or
1., .5
w(r,6)=§llr sin” 6.

This agrees with the result obtained by the other method.

Both the methods outlined above for evaluating the stream function
are useful, and each will be used in the following sections. The particular
method employed will depend upon the complexity of the problem, and it is
evident that the second method can be conveniently employed only for very
simple flow fields.

Reference
axis

FIGURE 5.3
Geometry for evaluating the stream function for a uniform flow.
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5.5 Source and Sink

The velocity potential corresponding to a three-dimensional source or sink
is obtained from Equation 5.4 through the term whose coefficient is B,,. Then,
let

A =0 foralll

0 forl=0
F

B =
! 0 forl=0.

B,
Then, from Equation 5.4, using the fact that P, (cos 0) =1,
o(r,8) = 0.
r

The velocity components for the resulting flow field are

Uy = 0.

Hence, the velocity is purely radial, its magnitude increases as the origin is
approached, and there is a singularity at the origin. Clearly, there is a source
or sink of fluid at » = 0, and the quantity of fluid leaving or entering this sin-
gularity may be evaluated by enclosing it with a spherical control surface of
radius . Then, if Q is the volume of the fluid leaving the control surface per
unit time, it follows that

Q=fsu-nds.

However, the velocity vector is radial, so thatu-n = |u| = +By/r>and ds =
72 sin 0 dO dw. Hence,

27 T B
Q=f dwf 2% 1 sin 0d6
0 0o r
= —-4nB,.

Then, for a source of strength Q, the constant B should be set equal to—-Q/(4n).
That is, the velocity potential for a source of strength Q located at 7 = 0 is
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o(r, 0) = - L (5.6a)

4nr

It should be noted that the minus sign is associated with the source, and so
a positive sign would be associated with a sink.

In order to establish the stream function corresponding to Equation 5.6a,
the definition of y will be invoked. Referring to Figure 5.4, a source of strength
Q is shown at the origin. At any arbitrary point P, the velocity will be radial
and is indicated by u,. The quantity of fluid that crosses the surface generated
by revolving the line OP about the reference axis will depend upon whether
the source Q is considered to be slightly to the left of the origin or slightly to
the right of it. Here, the source Q will be considered to be slightly to the right
of O, so that the quantity of fluid crossing the surface generated by OP will
be 2y + Q. Then, from Figure 54, it follows that

0
2mp +Q =j; u, cos 6 2nrsin€)£i—see

where 1, cos 0 is the component of the velocity vector that is perpendicular
to O'P, and r d6/cos 6 is the element of surface area along O'P. Performing the
integration yields

Q

= (1+ cos9). (5.6b)
47

IP(V, e) ==

It now becomes evident that if the source Q had been considered to be
slightly to the left of the origin, the constant term in Equation 5.6b would
have been different. However, the velocity components would be the same.

Reference
axis

FIGURE 5.4
Geometry for evaluating the stream function for flow due to a source.
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5.6 Flow due to Doublet

As was the case in two dimensions, the flow due to a doublet may be obtained
by superimposing a source and sink of equal strength and letting the dis-
tance separating the source and the sink shrink to zero. Figure 5.5 shows a
source of strength Q located at the origin and a sink of strength Q located
a distance 8x along the positive portion of the reference axis. The distance
from the source to some point P in the fluid will be 7, and the corresponding
distance to the sink will be r - &r.

From Equation 5.6a, the velocity potential for the flow due to this source
and sink will be

..., Q
o(r,0) = 4nr+4n(r—6r)
I T

4y 1-dr/r

If the source and sink are close together, the quantity &r/r will be small,
so that the expression for the velocity potential may be expanded as follows:

2
(1)(1’,6)=—g 1- 1+g+O or
4nr r r

2
_Q o o
dnr v r

I~

Reference
axis

Origin ¢ L
+Q Sx -Q

FIGURE 5.5
Superposition of a source and a sink that become a doublet as 8x — 0.
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The quantity 6 may be eliminated in favor of dx by applying the cosine
rule to the triangle defined by the vectors r and r — &7, and the distance dx
separating the source and the sink. Thus,

(r — 87)2 = 12 + (dx)* — 2rdx cosb.

Solving this equation for cos 6 gives

r* + (dx)* - (r = or)?

cos O =
2rdx
_or_drdr bx
ox 2rdx 2r
_or 1+0 or
Oox r

SOr=0xcos® 1-0 or
r

Using this result, the expression for ¢(r, 0) becomes

& 6—xcose 1+0 6—r
4dnr r r

q)(?’, 6) =

Now, let the distance 6x — 0 and the source strength Q — o such that the
product Q &x — . Then,

(I)(rr e) =

1 cos 6. (5.7a)
nr

Equation 5.7a is the velocity potential for a positive doublet of strength 1,
that is, a doublet that expels fluid along the negative portion of the reference
axis and absorbs fluid along the positive portion.

The stream function corresponding to Equation 5.7a will be obtained

by using the equivalent expressions for the velocity components given by
Equations 5.2 and 5.3. Thus,

1
u,=@=— 5 Cos0=—— il
ar 2wy r“sin@® 90
0 .
'—w———smﬁcose

00 2mr
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and
—— a2
Y(r,0) = toor sin” 0+ f(r).

Likewise, the two expressions for u, give

106 . 1 oy
s [P &
r 90 47rr® rsin® ar
.a—w= sin® 0
or  4mr?

so that
,0)=———sin’0 0).
Y(r,0) gy S0 +8(6)
Comparing these two expressions for y(r, 6) shows that f(r) = g(6) = 0 and
P(r,0) = ———sin” 0. (5.7b)
4y

Equation 5.7b gives the stream function for a doublet that discharges fluid
along the negative portion of the reference axis and attracts fluid along the
positive part of the reference axis.

5.7 Flow near Blunt Nose

By superimposing the solutions for a uniform flow and a source, the solu-
tion corresponding to a long cylinder with a blunt nose is obtained. From
Equations 5.5b and 5.6b, the stream function for a uniform flow of magni-
tude U and a source of strength Q located at the origin is

Q

1
r,0) = —Ur*sin® 06 - =
Y(r,0) 2 -

(1+ cos0).

In order to interpret the flow field that this solution represents, consider y
to be constant and solve the preceding equation for r in terms of ©:
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\/ pAT) Q 1+cos6
r= +
Usin?6 2xU sin®0

I Q
Usin*0  4nU sin*(0/2)

where the fact that 1 + cos 6 = 2 cos? (0/2) and the fact that sin(6) = 2 sin(0/2)
cos(0/2) have been used. Then, denoting by r, the value of r for which y =0,
the radius to the surface corresponding to y = 0 is

oo 1
O\ 4xlU sin(6/2)

Thus, the radii 7, corresponding to the principal values of 6 are as follows:

when6=0, 1n=»

when6=£, r0=,/&
2 2nl

when 0 = 7, r0=1/ Q .
4nll

This defines the stream surface y = 0, as shown in Figure 5.6.
Although the polar radius 7, is infinite for 6 = 0, the cylindrical radius R, is
finite. This may be verified by noting that R = r sin 6, so that

R - - Q sinb
"\ 4nU sin(0/2)

FIGURE 5.6
Flow around an axisymmetric body created by a source in a uniform flow.
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Then,as 0 — 0

sinf  _ 0-0%/31+...
sin(0/2)  0/2-(0/2)°/3!+ ...

Hence, the cylindrical radius far from the source becomes

Ro= |2
al

The fluid emanating from the source located at the origin does not mix
with the fluid that constitutes the uniform flow. Then, a shell could be fitted
to the shape of the surface corresponding to y = 0, and the source could be
removed without disturbing the outer flow. That is, the stream function for
the semi-infinite body shown in Figure 5.6 is

Y(r,0) = %llr2 sin? 0 - %(1 + cos 0). (5.8a)

The corresponding velocity potential may be obtained from Equations 5.5a
and 5.6a, giving

¢(r,8) = Ur cos6 - % . (5.8b)

Equations 5.8a and 5.8b may be used to deduce the velocity and pressure
distribution in the vicinity of the nose of a blunt axisymmetric body such as
an aircraft fuselage or a submarine hull.

5.8 Flow around Sphere

The stream function for a uniform flow past a sphere may be obtained by
superimposing the solution for a uniform flow and that for a doublet. From
Equations 5.5b and 5.7b, the stream function for such a superposition is

Y(r,0) = 11,11'2 sin”0 - ——sin” 0.
2 4mr
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Then, the equation defining the surface that corresponds to y = 0 is

O=%Ur02 sin? @ - sin® 0

4nr,

where 7, is the value of the polar radius r that defines the surface on which
y = 0. Solving this equation for r, gives

1/3

7, =
0 2nlU

Since r,, = constant, the surface that corresponds to y = 0 is that of a sphere.
If the doublet strength is chosen to be u = 2ala?, the radius of this spherical
surface will be r, = a. Then, by choosing u = 2nla3, the stream function for a
uniform flow of magnitude U approaching a sphere of radius a is

3

1p(r,6)=%u r2-“7 sin? 0. (5.9a)

The corresponding velocity potential may be obtained from Equations 5.5a
and 5.7a, in which the doublet strength u = 2xla® is used. This gives

1a°
o(r,0)=U r+ 5 cos 6. (5.9b)

7’2

5.9 Line-Distributed Source

The stream function and the velocity potential for a source that is distrib-
uted over a finite strip will be established in this section. The result is useful
as one element in superpositions that lead to additional solutions to flow
problems.

Figure 5.7 shows a source that is uniformly distributed over the section
0 < x < L of the reference axis. The source strength, which is constant, is g
per unit length, so that gL is the total volume of fluid that emanates from the
source per unit time. An arbitrary field point P is shown whose coordinates
are 7, 6, and . One end of the line source, which is at the origin, is a distance
r from this point and subtends an angle 0 to the x axis. The other end of the
line source is a distance 1 from the point P and subtends an angle o to the
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rsin O

R=

Reference

L axis
13 dg

Source strength
L q per unit length

FIGURE 5.7
Geometry connecting a field point P to a line source of length L distributed uniformly along
the reference axis.

x axis. Also, an element of the line source of length d&, which is a distance
&, from the origin, subtends an angle v to the x axis. However, the strength
of this element of the source is g d¢, so that, from Equation 5.6b, the stream
function for the line source will be

I
1p=—f0 %(1+cos v)

where the angle v is a function of & and so will be a variable in the integra-
tion. Rather than expressing v as a function of ¢, the variable of integration
will be changed from & to v. Referring to Figure 5.7, it will be observed that
the cylindrical radius R = r sin 0 = 1) sin « remains constant throughout the
integration. Also, it may be observed that

x-&=Rcotv
. —dg = =R sc? vdo.

Hence, the expression for y may be written in the form

Y(r,0) = - %fe csc? (1 + cosv)do

gR 1 1

=—--— cotB-cota+ —--—
4r sin® sino

But from Figure 5.7, the following relations may be established:
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x=RcotB
x—-L=Rcota
R
sin 6
R
sino.’

v =

Using these relations, the expression for the stream function for a line
source of strength g per unit length and of length L is

v=-T(L+r-n) (5.10a)
47

The velocity potential corresponding to Equation 5.10a may be obtained in
an analogous way. From Equation 5.6a, it follows that

L qué

=", 4n(R/sin0)

where it has been observed that the distance from the point at £ on the line
source to the field point P is R/sin v. As before, it is observed that

x—&=Rcotv
and so
—d& = —Resc? vdo.

Then, the expression for ¢ becomes

q . 2
=—— [ sinvcsc” vdv
q) J;

_475

_ 9 ¢ do

= fe o (5.10b)
o= ql tano/2 '

-—lo
47 & tan6/2

Although the result for the stream function was more compact when
expressed in terms of lengths, the result for the velocity potential is more
compact in terms of angles, so that Equation 5.10b will be considered to be
the final result.
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5.10 Sphere in Flow Field of Source

In the Chapter 4 problems, it was established that the solution for a circular
cylinder in a sink flow could be obtained from the solutions for two sources
and two sinks, all of which have the same strength. It will be shown here that
the solution for a sphere in a source flow may be obtained in an analogous
manner, although the singularities that must be imposed are two sources, of
unequal strength, and a line sink.

Figure 5.8 shows the connection between a field point P and certain singu-
larities. The singularities are a source of strength Q, which is located at point
Q, which is a distance / along the reference axis, a source of strength Q% which
is located at point Q* which is at the image point a?/I of point Q in the sphere
of radius 4, and a uniformly distributed line sink of strength g per unit length
along section OQ* of the reference axis. It will be shown that for an appropri-
ate choice of source and sink strengths, the sphere r = a corresponds to y = 0.

If the spherical surface r = a is to be a stream surface, the total sink strength
must equal the total source strength inside the surface r = a. That is, ga2/I = Q¥
which establishes the sink strength g in terms of the source strength Q*
Then, using Equations 5.6b and 5.10a, the stream function for the singulari-
ties shown in Figure 5.8 will be

__Q _Q QLa
Y(r,0) = 4n(1+COSB) 4ﬂ:(1+coscx)+43rc " l+r ]

where o, f, and 1 are functions of r and 6. Then, for points on the surface of
the sphere r = g, the value of y will be

% 1+1 M

Q*
1+cosa)+—=
( ) 47 a g2

Y(a,0)=- Q (1+cosf)-=—
4

4

Now, if point P lies on the spherical surface r = g, a relationship will exist
among the parameters 1, 4, f, o, and 6. To establish this relationship, it will
be noted that

a*/l
a

-2
E

But the numerator and denominator of each side of this equation represent
the lengths of one of the vectors shown in Figure 5.8. Thus, it follows that

oQ* _ or
oP 0Q°
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FIGURE 5.8
Superposition of a line sink of strength g per unit length, a source of strength Q¥ and a source
of strength Q near a sphere of a radius a.

But these lengths represent corresponding sides of the triangles OPQ* and
OQP. Then, since the angle 6 is common to both these triangles, it follows
that the two triangles are similar. Then, the angle OPQ* must equal the angle
OQP, which, in turn, equals n — p. Hence, the length n may be written as

a2
N =—cos(m —a) + acos(w - B)

I

2
a
= —Tcosoc—acos[i.

Substituting this result into the expression for y(a, 6) gives

Y(a,0) = —g(l +cosfP) —g(l +Cosa.)
4 4z

* l
+=— 1+—+cosa+—cosf
4n a a

=(1+ cosp) —% Q!

dna

Thus, by choosing the source strength Q* to be equal to aQ/l, the surface
r=awill correspond to the stream surface y = 0. Then, the stream function for
a sphere of radius a whose center is at the origin and that is exposed to a point
source of strength Q located a distance / along the positive reference axis is

__Q _Qa Qa r m
Y(r,0) = 47C(1+cos[.’>) 4nl(1+cosoc)+4n l+11 . (5.11a)
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The velocity potential corresponding to Equation 5.11a may be obtained
from Equations 5.6a and 5.10b. This gives

4nC - dmn 4w tan /2

(5.11b)

Using the fact that Q* = 2Q/! and q = [Q*/a?> = Q/a, the expression for the
velocity potential for a sphere of radius a in the presence of a source of
strength Q becomes

__Q Q tana/2
6(r,0) = 47T 4m1l+4mz log tan6/2

(.11c)

The quantity { is the distance from the field point P to the source Q as
shown in Figure 5.8.

5.11 Rankine Solids

The solution for the flow around a family of bodies, which are known as
Rankine solids, is obtained by superimposing a source and a sink of equal
strength in a uniform flow field. Let the magnitude of the uniform flow be U
and the strength of the source and the sink be Q. Consider the source and the
sink to be located equal distances I from the origin as shown in Figure 5.9a.

From Equations 5.5b and 5.6b, the stream function for the configuration
shown in this figure is

Y(r,0) = 1Ur* sin” 6 - %(cos 0, — cos0,). (5.12a)
7

(b)

FIGURE 5.9
(a) Superposition of uniform flow, source and sink, and (b) uniform flow approaching a
Rankine solid.
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Then, if 7, is the radius to the surface on which y = 0, the radius r, must
satisfy the equation

1
= Urfsin>0 - g(cos 0, - cos0,).
2 4n

Working with the cylindrical radius R = r sin 0 rather than the polar radius
1, it follows that the cylindrical radius R, which corresponds to the surface
y =0, will be

R2 =2

el (cosB; — cosH,).

Then, when 0, = 0, = 0, and when 6, = 0, = &, the value of R, is zero. Also,
the maximum value of R, occurs when 6, = 0, = ©/2 or when 6, = 6, = 31/2.
Thus, the stream surface that corresponds to y = 0 defines a body as shown
in Figure 59b. The principal dimensions of this body are the half width L
and the half height /. Both these parameters depend upon the free-stream
velocity U, the source and sink strength Q, and the distance /.

The value of L may be obtained from the equation resulting from the obser-
vation that the velocity at the downstream stagnation point is zero. However,
the velocity at that point is the superposition of a uniform flow of magnitude
U, a source of strength Q a distance L + | away, and a sink of strength Q a
distance L — I away. Hence,

u+ Q - Q
4n(L+1)*  4m(L-1)?

Rearranging this equation gives the following equation to be satisfied by L
in terms of the parameters U, Q, and [

2 oy QI _
(I - Pyt = =L =0, (5.12b)

An analogous expression for the half height & may be obtained by noting
that the value of the cylindrical radius R, is # when cos 0, = —cos 6,, where
tan 0, = /i/I. Hence,

RS, : + :
2nl 242 Jp2 42
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Rearranging this expression shows that h must satisfy the following equation:
2 2 2 Ql
h \/I’l +1 —7u=0. (5].2C)
n

For various values of the parameters U, Q, and /, Equations 5.12b and 5.12c
define a family of bodies of revolution for which the stream function is given
by Equation 5.12a. The corresponding velocity potential is

¢o(r,0) = Ur cosO — Q + Q . (5.12d)
dnr, 4w,

5.12 D’Alembert’s Paradox

It will be shown in this section that if an arbitrary three-dimensional body
is immersed in a uniform flow, the equations of hydrodynamics predict that
there will be no force exerted on the body by the fluid. Experimentally, it is
known that a drag force exists on a body that is in a fluid flow, and this theo-
retical result is known as d’Alembert’s paradox.

Figure 5.10 shows a body of arbitrary shape whose center of gravity is located
at the origin of a coordinate system. The surface of the body is denoted by S,

ng

FIGURE 5.10
Spherical control surface S, enclosing an arbitrarily shaped body of surface area S. The force
acting on the body is F, and the unit normal to the body surface is n.
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and the unit outward normal to 5, locally, is denoted by n. The hydrodynamic
force, which may act on the body, is denoted by the force vector F. A spherical
control surface S, is set up around the body under consideration, and n, is the
unit outward normal to S;. That is, n, = e,, where e, is the unit radial vector.

The equation of force equilibrium will now be written for the body of fluid
contained between the surfaces S and S,. The fluid force acting on the body
through the surface S is F; hence, the force acting on the fluid through that
surface is —F. There is no transfer of momentum across the surface S since
that surface is a stream surface. Around the surface S, there will be a force
due to the pressure distribution. The magnitude of this force will be —pn, per
unit of surface area, so that the total pressure force will be the surface inte-
gral of this quantity. Across the surface S, there will be a momentum flux
corresponding to the mass flux pu - n; per unit area. Then, the momentum
flux will be pu(u - ny) per unit area, so that the inertia force per unit surface
area will be —pu(u - n,). Thus, the equation of force equilibrium for the fluid
that is bounded by the surfaces S and S is

0=-F —f pn, +pu(u-n,) ds.
So

The pressure may be eliminated from this equation through use of the
Bernoulli equation, which, for the case of steady irrotational motion under
consideration, may be written in the form

1
+—pu-u=58
p+opu-u

where B is the Bernoulli constant. Then, the force F acting on the body will
be given by the following integral:

1
F= pfso E(u-u)n0 —u(u-n,) ds.

Here, it has been observed that the surface integral of Bn, is zero for any
closed surface.

Itis now proposed to write the velocity vector u as the sum of the free-stream
velocity vector U = Ue, and a perturbation u’. The perturbation velocity u’ will
be large near the body, but it will tend to zero far from the body. Then, writing

u=U+u'

the expression for the force F becomes

F=p %U2+U'u +%u-u n,-(U+u) (U+u)'n, dS.

So
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Expand the integrand now and note that [ U’n,ds = 0 since U? is a con-
So

stant. | U-n,ds =0 since U is a constant vector and | u ‘n;,ds=0 from
So So
the continuity equation. Hence,

F=pS U-u +%u ‘u ny- u(U'ny)+u(u-ny) ds.
0

The first and third terms in the integrand may be replaced by —-U x (u x n)
in view of the vector identity

Ux (u' xng) =u(U-ny) —n, (U-u)

Hence, the expression for the hydrodynamic force F may be written in the
form

F=p . -Ux(u xng)+3(u -u)n, -u(u -ng) ds.
0

It will now be shown that each of these terms is zero.
Let ¢’ be the velocity potential corresponding to the perturbation velocity
u’. Then, from Equation 5.4, ¢’ must be of the form

O, Dlcoso
¢=2Az z(cgls)
- r

Q

cos 0 1
=-—"—4 +0 —
dnr - 4mr? r’

where the first two terms, which correspond to a source and a doublet, have
been written out explicitly, and any remaining terms must vary as 1/ or
some greater power of 1/r. Then, since u’ = V¢, it follows that

1
‘u‘=O 1’7 .

That is, the perturbation velocity varies, at most, as 1/r2. Also,

u xn;= Q. Cose+O L xe,
4 Aur? r
Q 1

e, +O — xe
4mur® 7’

re
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That is, since n, = e, and since e, x e, = 0, it follows that

1

‘u xn0‘=O - -
r

Finally, since an element of surface area dS equals #? sin 0 dO dw, it is evi-
dent that

ds = O().

Using the foregoing results, it is possible to establish the order of magni-
tude of each of the integrals that appears in the expression for F. Thus,

Ux(u xn;)dS=0 1
r

So

=

(u *u)n,dS=0

So

f u(u -n,)dS=0
So

2
1
2

r

That is, if the radius of the spherical surface S, is taken to be very large,
each of these integrals will be vanishingly small. Thus, in the limit,

F=0. (5.13)

Since it is known that any body that is immersed in a flow field experiences
a drag force, Equation 5.13 poses a paradox known as d’Alembert’s paradox.
The resolution of this paradox lies in the fact that viscous effects have been
omitted from the equations that led to Equation 5.13. It will be seen in Part
III that there is a thin fluid layer around such a body in which viscous effects
cannot be neglected. This fluid layer, or boundary layer, exerts a shear stress
on the body that gives rise to a drag force. In addition, the boundary layer
may separate from the surface of the body, creating a low-pressure wake.
This, in turn, will induce an additional drag, called the form drag, owing to
the pressure variation around the surface of the body. However, for stream-
lined bodies, Equation 5.13 is approached because of the absence of form
drag, although the viscous-shear drag will still exist.
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5.13 Forces Induced by Singularities

It was established in the previous section that, according to the equations of
hydrodynamics, no force exists on a body that is in a uniform flow field. This
is consistent with the results of the last chapter since the Kutta—Joukowski
law shows that in the absence of circulation around a body, there are no
forces acting on two-dimensional bodies. In view of the fact that V - @ = 0,
it is very difficult to establish an appreciable circulation around short bod-
ies—that is, around three-dimensional bodies. However, it was established
in the problems at the end of Chapter 4 that a force will exist on a cylinder
that is exposed to a singularity in the flow such as a source, a sink, or a vor-
tex. Likewise, it will be shown here that a force exists on a three-dimensional
body if it is exposed to a point singularity in the fluid.

Figure 5.11a shows an arbitrary body whose center of gravity coincides
with the origin of a coordinate system. The surface of the body is denoted by
S, and n is the outward unit normal to S. A singularity is assumed to exist
at the point x = x; since the polar axis may be made to pass through the sin-
gularity without loss of generality. A small spherical control surface denoted
by S; and of radius ¢ is established around the singularity. The unit outward
normal to the surface ¢ is denoted by n;. A large spherical surface, denoted
by S, is drawn around both the body and the singularity. The unit normal to
this surface is denoted by n,. The hydrodynamic force that acts on the body
and whose magnitude is sought is denoted by F.

For equilibrium of the forces that act on the body of fluid that is inside S,
but outside S and S, the sum of the forces must be zero. Hence,

Reference
axis

FIGURE 5.11
(a) Control surfaces for a body located at the origin and a point singularity at x = x; and (b) a
source and a sink close together near the body.
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0=—F=f png + pu(u-n,) d5+f pn; +pu(u-n;) dS.
So So

The first two terms on the right-hand side of this equation are identical
with those that appeared in the previous section, and the third term repre-
sents the pressure and momentum integral for the new surface S;. However,
it was shown in the previous section that the integral around S, that appears
above is zero. Then,

F= ; -n;) dS.
j;,' pnt+pu(u nz)

Since the integral that appears in this equation represents the force acting
on the surface S;, whose radius is arbitrarily small, it follows that if a force F
acts on the body S, the reaction of this force must act on the singularity. From
the Bernoulli equation, p = B — p(u - u)/2, so that

1
F= pfs,- —E(u u)n; +u(u-n;) dS. (5.14a)

In order to further reduce the integral in Equation 5.14a, it is necessary to
specify the nature of the singularity located inside the surface S,. The case of
a source, or sink, and that of a doublet will be examined.

Consider, first, the singularity at x = x; to be a source of strength Q. Then,
from Equation 5.6a, the velocity on the surface S; will be

I

= e +u;
3 i
431:82

where e, is the unit vector radial from the point x = x;, and u; is the velocity
induced by all means other than the source under consideration. Then,

Q Q

T l6n?et | 2me?

uu e u;+u;u,;

and

Hence, from Equation 5.14a,
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Q
F= + e.'u;+u;'u; e,
pf 2 16Jr2 o 2me? f

Q2e£+ui Q2+ui-eS ds
4me 4me

Q 1 Q
= e -~ u)e, +—u; +(u; - e )u,; ds.
pfsi 32m%e* 2( ) 4ge? ( )

Of these four integrals, the first is zero since it involves a constant multi-
plied by e, integrated around a closed surface. Since the radius ¢ is arbitrarily
small, the quantity u; - u, may be considered to be constant over the surface
S, and so the second integral will be zero in the limit as € — 0. The last term in
the integrand will likewise involve a quantity u;, which will be constant, and
a quantity e,, which will change direction around S,. Thus, the product u; - e,
will have equal positive and negative regions over the surface S; so that the
integral of (u; - e)Ju; over S; will be zero. Then, the expression for F becomes

u. ds

Q
F= pfﬁ‘, 47e?

where, again, u; may be considered to be constant throughout the integration
for vanishingly small value of e. Hence,

pQ 27 7T
=—uif d(uf sin6do
0 0 (5.14b)

That is, the force on the body, and on the source, is proportional to the
source strength and to the magnitude of the velocity, u,, induced at the loca-
tion of the source by all mechanisms other than the source itself. The direc-
tion of the force coincides with that of the velocity vector u;. For a sink, Q
should be replaced by —Q in Equation 5.14b.

Consider now the case when the singularity is a doublet. It was shown in
Section 5.6 that a doublet may be obtained by superimposing a source and a
sink of equal strength. Hence, consider a source of strength Q to be located
at x = x;and a sink of strength Q to be located at x = x; + §, as shown in Figure
5.11b, where 8 is a vanishingly small distance. Then, if u, is the fluid veloc-
ity at x = x; due to all components of the flow except the source and the sink
under consideration, the velocity at x = x; less that due to the source itself,
will be
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Q e +u.
43’562 x i

where e, is the unit vector in the x direction.
The velocity at x = x; + §, less that due to the sink, will be

Q

du,
Te +u, o™iy
4 ax

Then, from Equation 5.14b, the force acting on the body due to the source
will be

Q
470

pQ

S+,

and the force acting on the body due to the sink will be

ou;
-pQ '

e +u;+0

+o.
47d*

where the minus sign results from the fact that a sink is being considered.
The net force that will act on the body due to the combined source and sink
will then be

ou;

-pQd
pQ P

Now, if & is allowed to shrink to zero and Q is allowed to become infinite
such that Q8 — p, the force acting on the doublet of strength u that will result
at x = x; will be

ou;

ox

Hence, the force acting on the body due to a doublet of strength u will be

F=-p E (5.140)

As an example of an application of the foregoing results, consider a sphere
in the presence of a source, which was discussed in Section 5.10. The flow
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tield was found to consist of the source of strength Q, which was located at
x =1, an image source of strength Qua/I located at x = 4?/I, and a line sink of
strength Q/a extending over the region x = 0 to x = a?/1. Then, the velocity
u,, at the point x = [ due to all causes except the source of strength Q, will be

Qa1 fﬂ”' Q/a e,

u; = ———e, - =
4 (1- a1y 0o 4w (I-x)

=Qa/l 1 o _Q/a 1 _1 e
Sl () G O

o7

Anl(? - a?)?

X

Then, from Equation 5.14b, the force F acting on the sphere due to the
source will be

2.3
Fe 5 oo 1‘21(22 é e (5.14d)
TT —-a

That is, the sphere is attracted to the source with a force that is propor-
tional to Q2.

5.14 Kinetic Energy of Moving Fluid

It is sometimes of interest to calculate the kinetic energy associated with a
fluid disturbance. An example of the utility of this quantity in the context of
tlow around immersed bodies will be given in the next section, and an appli-
cation to free-surface flows will be made in the next chapter.

The kinetic energy associated with the fluid in the uniform flow around a
stationary body will be infinite if the flow field is infinite in extent. However,
the kinetic energy induced in a quiescent fluid by the passage of a body
through it will be finite, even if the flow field is infinite in extent. For this
reason, discussions of kinetic-energy considerations are based on a frame
of reference in which the fluid far from the body is at rest and the body is
moving.

Referring to Figure 5.12, we consider an arbitrary body of surface area S that
is moving with velocity U through a stationary fluid. An arbitrarily shaped
control surface S, is constructed around the body. The unit outward normal
to the surfaces S and S, denoted by n and n,, respectively, are indicated. If
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n,

FIGURE 5.12
Control surface for an arbitrary body moving through a quiescent fluid.

V is the volume of fluid contained between the surfaces S and S, the kinetic
energy of T of this volume of fluid will be

1
T=| ~pu-udV
[ 5 et

=§pr ¢ ¢dV

where ¢ is the velocity potential corresponding to the motion induced in
the fluid by the moving body. This volume integral may be converted to a
surface integral by use of Green’s theorem in the form given in Appendix A.
Thus, since V?¢ = 0, it follows from Green’s theorem that

1 b
== —dS
2 pf2¢ on

where X is the surface that encloses V and so consists of the surfaces S and
S, However, on the surface S, the unit normal points away from the surface
and into the volume V. Using this fact, the surface integral above may be
expanded to give

1 o 1 o
T== —+dS-= —dS.
2pf50¢6n 2pfs¢an
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The first integral that appears in this expression is zero, which will now be
shown.
From the continuity equation, if follows that

fv -udV =0.

This volume integral may be converted into two surface integrals by use of
Gauss’ theorem to yield the following expression:

u'ndS- [ u-ndS=0.
Jomas .

But u - n = ddp/dn on the surface S, and u = U, where U is a constant. Hence,

) %dS—fSU-ndS=O.

The second integral in this identity is zero since U is a constant vector over
the surface S, so that for any constant C, it follows that

c®4s_o.
s, on

Subtracting this quantity from the right-hand side of the expression for T
gives

_L e A TS S SN[
=,P) @05, 45 zpfsq’ands'

Now, since the fluid velocity far from the body is zero, the value of ¢ there
can at most be a constant. Thus, by considering the surface S, to be large and
by choosing C to be the value of ¢ far from the body;, the first integral may be
made to vanish. That is, the kinetic energy induced in the fluid by the move-
ment of the body is

__ 1 or.9
T= zpfs%nds (5.15)

where, it should be recalled, the velocity potential corresponds to the body
moving through a stationary fluid.
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5.15 Apparent Mass

When a body moves through a quiescent fluid, a certain mass of the fluid is
induced to move to some greater or lesser extent. A question that may then
be asked is, what equivalent mass of fluid, if it moved with the same velocity
as the body, would exhibit the same kinetic energy as the actual case? If the
fluid may be considered as being ideal, the mass of fluid referred to above is
found to depend upon the body shape only, and this mass of fluid is called
the apparent mass.

We define the apparent mass of a fluid M’ as that mass of fluid, which, if
it were moving with the same velocity as the body, would have the same
kinetic energy as the entire disturbed fluid. That is,

1 1~ a6
Tyvur=-tofe®4s
2 2pfs¢an

R (5.16)
M = qusq’andS'

For arbitrarily shaped bodies, the velocity potential will depend upon the
direction of the flow. That is, the apparent mass of fluid associated with a
given body will be a property of the shape of that body, and as for inertia,
there will in general be three principal axes of the apparent mass. For axi-
symmetric bodies, there will be two principal values of M’, whereas for the
sphere, there will be only one.

As an example of an application of Equation 5.16, the apparent mass for a
sphere will be worked out here. The velocity potential (Equation 5.9b) corre-
sponds to a stationary sphere of radius a with a uniform flow of magnitude U
approaching it. Then, the required velocity potential may be obtained from
Equation 5.9b by adding the velocity potential for a uniform flow of magni-
tude U in the negative x direction. This gives

1a°
o(r,0)=U r+5—2 cos 6 - Ur cos 6
7
3
=§Ua—2cos6
T
3

@ (r,0) = @(r, 0)=-U 4 cos 0.
on or

7,,3
Hence, on the surface S where r = g,

q)a—q’ = —lllzacos2 0.
on 2
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Then, from Equation 5.16, the apparent mass for the sphere is

27 T
M =—Lf doof —lllzacosze a*sin06de
UZ 0 0 2
M = %naap. (17)

That is, the apparent mass for a sphere is one-half of the mass of the same
volume of fluid. This apparent mass may be added to the actual mass of
the sphere, and the total mass may be used in the dynamic equations of the
sphere. That is, the existence of the fluid may be ignored if the apparent mass
of fluid is added to the actual mass of the body.

PROBLEMS

5.1 Use the definition of the Stokes stream function and the ® compo-
nent of the condition of irrotationality to show that the equation to
be satisfied by the stream function y(r, 6) for axisymmetric flows is
as follows:

2
L) +si116i _1 N
or? 90 sin® 90

r’ =0. (5.18)

5.2 Show by direct substitution that the stream functions obtained for
a uniform flow, a source, and a doublet, as given by Equations 5.5b,
5.6b, and 5.7b, respectively, satisfy Equation 5.18.

Y(r,0) = %uﬂ sin” 0

(5.5b)
__Q
Y(r,0) = ———(1+ cosB) (5.6b)
4
Y(r,0) = ———sin” 0. (5.7b)
4ty

5.3 Look for a separation-of-variables solution to Equation 5.18 of the
following form:

y(r, 0) = R(NTO).

Hence, show that the finite solutions for R(r) are of the following
form:

R,() = A,



Three-Dimensional Potential Flows 201

and that the equation to be satisfied by T(0) is the following:

(1-v?

where 1 = cos 0. Show that the substitution T = (1 — n?)"2t trans-
forms this equation to the following form:

dt 1

1- ) Zna+ n(n- 1)_1—1’] t=0.

This is the associated Legendre equation. Show that the nonsin-
gular solution to this equation is the following:

() = (1 )2 S
dn
In the above, P,(n) is Legendre’s polynomial of order n. Thus,
deduce that the general solution to Equation 5.18 is the following:

rn

(r,0) = 2 0L b (cost) 519

5.4 Show that setting A, = 0 for n # 1 in Equation 5.19 yields the solution
for a doublet.

5.5 The figure for this problem shows a doublet of strength u* located at
x =/ and a doublet of strength p* located at x = 42/1. Show that the sur-
face r = a corresponds to y = 0 if p*= —a® u/P. Hence, deduce that the
stream function for a doublet of strength 11 located a distance / from
the center of a sphere of radius 4 is given by the following equation:

3

Y(r,0)=-——sin’ B+ sin” a. (5.20a)

a
4nT 4nl’n

Also, deduce that the corresponding velocity potential is given
by the following equation:

a
o(r,0)=+—— e cosf - Mcosoc. (5.20b)
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5.6 Show that the force that acts on a sphere of radius a due to a doublet
of strength plocated a distance [ from the center of the sphere along
the x axis is given by the expression

. 3pall
2(l® - )

x*

5.7 A spherical gas bubble of radius R(f) exists in a liquid; that is, the
radius of the bubble is changing with time. The liquid is quies-
cent, except for any motion that is caused by the bubble itself. It
is assumed that the fluid motion does not involve any viscous or
compressible effects, and it may therefore be represented by a time-
dependent velocity potential that satisfies the following conditions:

19 L

=0
r2 or ar

2¢=
o p) =
ar(r ,H=0

i¢ _ 5
o (r=R,t)=R.

In the above, R is the derivative of R with respect to time. Obtain
an expression for the radial velocity at the surface of the bubble and
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for the velocity potential ¢(r, t), both of these expressions being in
terms of R, R, and . Also, obtain an expression for the pressure at
the surface of the bubble p (R, t), taking the pressure far from the
bubble to be p,, which is a constant.

Suppose that at time ¢ = 0 the pressure at the surface of the bubble
is p,, the radius of the bubble is R, and its initial velocity is — R; that
is, the radius of the bubble is decreasing as time increases. Obtain an
expression for the time required for the radius of the bubble to shrink
to zero.

5.8 A sphere of radius 2 moves along the x axis with velocity U(f) that
varies with time. A fixed-origin coordinate system is defined by
the location of the sphere at time ¢ = 0, so that its location at any
subsequent time will be defined by the relation

X, () =fotll(t)d‘c.

This situation is depicted in the figure for this problem. If P is
any fixed field point, its coordinates (7, f) relative to the sphere will
change with time. Obtain the velocity potential for the sphere in
a stationary fluid, first in terms of  and 6, then in terms of x, R,
and x,. If the undisturbed pressure is p., find the pressure at the
field point P in terms of r and 6. Hence, by integrating the pres-
sure around the surface of the sphere, find the force acting on it.
Compare the result so obtained with that obtained by using the
apparent mass concept in conjunction with Newton’s second law
of motion.

xo(2)
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Surface Waves

The effect of gravity on liquid surfaces is treated in this chapter. The flows
associated with surface waves will be assumed to be potential, which is a
valid approximation for many free-surface phenomena. Most of the flows
treated here will be two-dimensional. However, the treatment of surface
waves has been separated from the other two-dimensional potential flows
because of the different nature of the problems and the different approaches
to their solutions.

The formulation of surface-wave problems is discussed first. The linear-
ized version of this formulation is then presented, and this version is used
throughout most of the remainder of the chapter. The propagation speed of
small-amplitude waves is established, and the effect of surface tension on
this result is investigated. Waves on shallow liquids are discussed next, and
the manner in which waves of arbitrary form and amplitude propagate is
established.

The complex potential for traveling waves is calculated, and this result
is used to establish the pathlines for fluid particles in a body of liquid on
the surface of which waves are propagating. A superposition of traveling
waves is then used to introduce the topic of standing waves. The particle
paths for this type of wave are also established. The topic of standing waves
leads, quite naturally, to the question of what type of waves may exist on the
free surface of liquids that are contained in vessels of finite dimensions. In
particular, rectangular and cylindrical vessels are discussed. The response
of the free surface to arbitrary motions of such vessels is obtained. Finally,
the behavior of waves at the interface of the two different fluid streams is
investigated. This leads to the topics of Helmholtz or Rayleigh instability and
Taylor instability.

6.1 General Surface-Wave Problem

When a quiescent body of liquid experiences gravity waves on its free sur-
face, the motion induced by the surface waves may be considered to be irro-
tational in most instances. Then, the velocity vector may be expressed as the
gradient of a velocity potential, which, in turn, must satisfy Laplace’s equa-
tion. That is, the governing equation is the same as that for each of the two

205
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previous chapters, so that surface-wave theory introduces no new difficulties
with respect to the governing equation. The boundary conditions to be satis-
tied will now be established.

Figure 6.1 shows a body of liquid on a flat surface in which waves exist on
the free surface of the liquid. The x axis of a coordinate system is located at the
mean level of the free surface, which is defined by the equation y = n(x, z, f),
and the mean depth of the liquid is . Two boundary conditions must be
imposed on the free surface y = 1. The first of these conditions is called the
kinematic condition, and it states that a particle of fluid that is at some time on
the free surface will always remain on the free surface. Then, since the equa-
tion of the free surface is y —n = 0, it follows that

D
—(y-m)=0.
Dt (y-m)
In terms of Eulerian coordinates, this boundary condition becomes
Ly-mrue (-m=0
ot ¥y-n y-n .

However, in the Eulerian frame of reference, the coordinates x, y, z, and ¢
are independent. Also, the function n depends on x, z, and ¢t only. Hence, the
equation above may be rewritten to give

FIGURE 6.1
Coordinate system for surface-wave problems.
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where it has been noted that dx;/dx; = §;;. Finally, expressing the velocity com-
ponents in terms of the velocity potential ¢, the kinematic surface condition
becomes

m, 9 m 9o m_ 3¢
a Ixadx 9z az dy

The other boundary condition that must be imposed on the free surface is
a dynamic one. Typically, the statement amounts to specifying that the pres-
sure is constant, but in general, it may be stated that p = P(x, z, f) on y = 1. This
condition is implemented through the Bernoulli equation. The appropriate
form of the Bernoulli equation is that for unsteady, irrotational motion. Since
gravitational forces are intrinsically important in free-surface waves, grav-
ity must be included in the body-force term. Thus, from Equation II.6 with
G = —gy, the boundary condition p = P on y = 1 becomes

o P1 _
at+p+2 ¢ ¢+gn=F().

Finally, the boundary condition at the bed must be imposed. For the case
of an inviscid fluid, which is under consideration, this amounts to specifying
that the velocity component normal to the boundary be zero. For a flat bed
as shown in Figure 6.1, this simply amounts to specifying that d$/dy = 0 on
y=-h.

To summarize, in terms of the velocity potential ¢, the conditions to be
satisfied for surface-wave motions are the following:

V2h =0 (6.1a)

m, oo g o0 (6.1b)
of Odx dx 0dz 0z Jy

o P 1
[l = = Nt

o + 0 +2 o d+gn=F() ony=m 6.1¢)

gj/’ -0 ony=-h (6:1d)

The difficulty in solving surface-wave problems may be seen to be in the
boundary conditions rather than in the differential equation. Equation 6.1c
is nonlinear, and both it and Equation 6.1b are to be imposed on the surface
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y = 1. In many real situations, this surface may not be known a priori and
may be one of the quantities that evolve from the solution itself. However,
many interesting features of surface-wave flows do not depend upon these
complex features of the problem. That is, by linearizing the problem, the dif-
ficulties discussed above may be avoided while the basic features of the flow
are not destroyed. Such a linearization will be carried out in the next section.

6.2 Small-Amplitude Plane Waves

For simplicity, we consider plane waves, that is, two-dimensional flow fields
with waves on the surface. Then, without any further approximation, the
differential equation to be satisfied by the velocity potential in the xy plane is

ot =0 (6.2a)

In order to make the surface boundary conditions more tractable, small-
amplitude waves will be considered. That is, only waves for which the
amplitude is small compared with the other characteristic length scales will
be considered. The other characteristic length scales are the liquid depth &
and the wavelength of the waves. But if n is small compared with the wave-
length, the quantity dn/ox, which is the slope of the free surface, will be
small. Furthermore, the quantity d¢/dx, which is a velocity component, will
be small since surface waves do not involve high frequencies and since the
amplitude of the motion has been assumed to be small. Then, the product
d¢/0x with dn/dx, which appears in Equation 6.1b, will be quadratically small
and hence may be neglected to first order. The kinematic boundary condition
on the free surface then becomes

m 99
o, /t =1 /t'
o &b ay(xﬂ )

Although this equation is free from quadratic terms, it still contains the
difficulty that it must be imposed on y = . However, in our present approxi-
mation, 1) is small, so that a Taylor expansion may be written for the quantity
dd/9dy at y = n about the line y = 0. Thus,

d ) 92
aqy)(x,n,t)=(;;(x,O,t)may(f(x,O,tHO(nz)-
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The second term in this expansion is quadratically small and so, to the
first order, may be neglected. That is, to the first order in small quantities, the
boundary condition (Equation 6.1b) may be written in the form

99 _m
oy (x,0,t) = P (x,1). (6.2b)

The dynamic boundary condition on the free surface may be treated in
the same way. Since the fluid is essentially quiescent and any fluid motion is
induced by the waves, the nonlinear term u - u = V¢ - V¢ may be neglected as
being quadratically small. Thus, Equation 6.1c becomes

6(1) (x

1)+ +gn(x, £) = F(f).

P(x,t)
p

The quantity d$/dt may be expanded in a Taylor series about the line y = 0,
and only the first term in this expansion need be retained. This gives

a"’( 0,6+ (p PO | o, 1) = E®).

The quantity F(t) may be absorbed into the velocity potential ¢(x, y, f) by
considering ¢(x, y, f) to be replaced by

o(x, y, t) + f F(t)dt.

Thus, the linearized version of Equation 6.1c may be written in the form

P(x,t)

( 0,t)+ ——=+gn(x,t)=0.

If the time derivative of this equation is formed, the term dn/dt may be
eliminated in favor of d¢/dy from Equation 6.2b. Thus, the preferred form of
the dynamic boundary condition on the free surface is

EX) 1 aP(x 0
x,0,t)+ —(x,0,t)=0. 6.2c
oz OB g oy (x,0,1) (6:2¢)
The boundary condition on the lower boundary is unaffected by the lin-
earization and requires
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9 ) = 0. (6.2d)
ay

Equations 6.2a through 6.2d represent a much more tractable set than the
general equations presented in the previous section. However, as was men-
tioned earlier, they correctly predicted many of the features of surface waves,
and so they will form the basis of most of the remaining sections of this
chapter.

6.3 Propagation of Surface Waves

Consider a quiescent body of water or other liquid of depth /, as shown in
Figure 6.2. A small-amplitude plane wave is traveling along the surface of
this liquid with velocity c. The form of the wave is taken to be sinusoidal,
with the amplitude of the wave being € and its wavelength being A. Thus, the
equation of the free surface will be v = n(x, t), where

Nx,t)=¢ sinz%(x —ct).

This corresponds to the wave traveling in the positive x direction with
velocity c.

The question we ask is the following: Given the wave amplitude & and
wavelength A and given the depth i, what will be the propagation speed c?
The answer to this question may presumably be obtained by solving the flow
problem for the velocity potential. For the time being, surface-tension effects

FIGURE 6.2
Parameters for a pure sinusoidal wave.
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will be neglected; hence, the pressure on the surface of the liquid will be con-
stant and equal to, say, the atmospheric pressure. That is, P(x, f) = constant
in this instance. Then, from Equations 6.2a through 6.2d, the problem to be
solved for ¢(x, y, t) is

@(x, 0,f)=-¢ zicos—(x ct)
oy A A

K 3
y(x,o,th g@(x,O,t) =0

9 ) = 0.
ay

Here, the equation for n(x, t) has been used in the kinematic condition on
the free surface. The appropriate solution to the Laplace equation by separa-
tion of variables will be trigonometric in x, and hence, it will be exponential
or hyperbolic in y. This deduction follows from the nature of the value of
d¢/dy, which is prescribed on y = 0 by the kinematic boundary condition. In
fact, inspection of this boundary condition yields even stronger information.
Since d¢/dy must vary as cos 2n(x — cf)/A, then so must ¢. That is, the nature
of the time dependence is brought in through this boundary condition, as is
the nature of the x dependence. Furthermore, since the separation constant
in the x direction must be 2rn/A, the separation constant in the y direction
must also be 2n/A. Hence, the appropriate form of solution to the Laplace
equation is

o(x, y,t)—cos—(x ct) C,sinh y+C zﬂ .

Here, the hyperbolic form of solution in y has been used in preference to
the exponential form since the region in y is finite rather than infinite. This
facilitates application of the boundary conditions. Having used the form
of the first boundary condition, the third boundary condition will now be
imposed. Thus, the condition that d¢/dy must vanish on y = —h gives

cosz—n(x—ct) 2—TEC hz—nh 2—TEC mhz—nh =0.
A A A A
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Since this condition is to be satisfied for all values of x and ¢, the quantity
inside the second set of parentheses must be zero. This gives

C = gmh%

Hence, the solution for ¢(x, y, ) becomes

do(x,y,t)=C, cos (x ct) tanh@ thny osh% .

Finally, the second boundary condition, corresponding to the dynamic
condition on the free surface, will be imposed. This gives

2
C, cos2—(x ct) - 2me g—t nh@ =0.
A A A A

Again, this equation is to be satisfied for all values of x and ¢, so that the
quantity inside the square brackets must vanish. But the only unknown
quantity inside the brackets is the wave speed c. That is, imposing this final
boundary condition determines the speed ¢ with which the wave train is
traveling. In nondimensional form, the result is

A 27h
—=——tanh——. 6.3
gh 2nh A (6.32)

Equation 6.3a was obtained using a small-amplitude approximation,
which means that it is valid provided e < A and & <« h.

As a special case, we consider deep liquids, that is, liquids for which
h > ). Then, the parameter 2zh/A will be large, so that tanh (2xh/2) will be
approximately unity. Then, for deep liquids, Equation 6.3a may be approxi-
mated by

2 A
@ = ﬁ. (6-3b)

Equation 6.3b will be valid for e < A <« h.
The other obvious limit is that of shallow liquids, that is, liquids for which
h <\ In this case, the parameter 2n/i/A will be small so that
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tanh @ ~ @
A A
Then, Equation 6.3a, becomes
2
c
—=1 6.3
u (639

which will be valid for e < 1 <A

The foregoing results are presented schematically in Figure 6.3, in which
the general solution (Equation 6.3a) is shown by a solid line and the two
asymptotic limits are shown dotted.

An arbitrarily shaped wave train may be considered to be a superposi-
tion of sinusoidal waves of the type just treated. That is, waves of arbitrary
form may be Fourier-analyzed and so decomposed into a number of pure
sinusoidal waves. Thus, the foregoing results show that such waves will
not, in general, propagate in an undisturbed way. That is, the propagation
speed ¢, or celerity as it is sometimes called, depends upon the wavelength
M of its sinusoidal components. Only with shallow liquids (Equation 6.3¢) is
the propagation speed independent of the wavelength. That is, unless the
shallow-liquid conditions apply, the different Fourier components of an arbi-
trarily shaped wave will all travel at different speeds so that the waveform
will continuously change. This process is usually referred to as dispersion or,
more specifically, frequency dispersion.

“gh | Equation

/ Equation
6.3b_\y/ [ €30
10 et

/ Equation 6.3a

~=— Deep liquids Shallow liquids —=

21th

FIGURE 6.3
Propagation speed c for small-amplitude surface waves of sinusoidal form.
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6.4 Effect of Surface Tension

In the previous section, it was assumed that the pressure along the topmost
layer of the liquid was constant, corresponding to atmospheric pressure.
However, if surface-tension effects are included, the pressure along the edge
of the liquid will be different from the pressure outside the liquid, unless the
surface is flat. To establish the effect of this pressure differential, an element
of the surface is isolated in Figure 6.4 and the forces due to surface tension
are indicated.

At the reference position x, the value of the surface tension is o, and the
slope of the surface there is on/dx. Then, a short distance Ax farther from the
origin, the value of the surface tension will be 6 + (do/0dx)Ax, and the slope of
the surface will be dn/dx + (0*n/0x?)Ax. Only the first-order corrections have
been written down here since the unwritten terms in the Taylor series will
contain terms of order (Ax)?> or smaller. Then, if p, is the pressure above the
liquid and if P(x, t) is the pressure at the edge of the liquid, vertical equilib-
rium of the element of surface shown in Figure 6.4 requires that the follow-
ing equation be satisfied:

2
(P-po)Ax + 0+a—0Ax a—n+a—1;]Ax —Ga—n=0.
ax 0x  ox ox

Expanding the terms in this equation and neglecting terms that are qua-
dratic in the length Ax gives

*m 9o an
P- +o0—+——=0
(P~ po) o ox ox
o DF
o’?%*
Po /
0 —-—
P(x, t)
—X Ax ——— ==

FIGURE 6.4
Element of liquid surface showing forces due to surface tension.
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The length Ax may now be permitted to shrink to zero, so that the neglected
terms become identically zero and the preceding equation becomes exact.
The last term in this equation will be zero if ¢ is constant and will be qua-
dratically small if 6 is almost constant. Thus, to the first order in small quan-
tities, the pressure P(x, t) at the edge of the liquid becomes

2

am
P(x,t)=p, - 0?'

In the dynamic boundary condition on the free surface (Equation 6.2¢), the
pressure enters through the term oP/dt. However, if the pressure p, outside
the liquid is constant, the expression for 0P/t is

P o

ot x> ot

3
o azq) (x,0,1)
dax“dy

where the order of differentiation has been interchanged in the first equa-
tion, and on/df has been eliminated in favor of derivatives of ¢ using the
kinematic boundary conditions (Equation 6.2b). Using the preceding result,
the dynamic boundary condition on the free surface (Equation 6.2c) becomes

o 9 a9 _
ay (x,0,)+ g oy (x,0,t)=0. 6.4

K)
at(

This revised form of the dynamic boundary condition will be used to
recalculate the propagation speed of a sinusoidal wave.

The existence of surface tension does not affect the governing partial dif-
ferential equation, the kinematic surface condition, or the bed boundary con-
dition. Hence, from the previous section, the velocity potential that satisfies
these unchanged equations is

o(x,y,t)=C, cos—(x ct) tanh@ thTcy oshz% )

Application of the boundary condition (Equation 6.4) to this velocity poten-
tial results in the requirement

2 3
2% 2mc o 2n 2nth 2% 2mth

C, - ) - —— - = tnh— ““tanh—— =0.
cos 7» (x—ct) . + Y +g y a )\
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The general solution to this equation requires that the quantity inside the
square brackets vanish, which gives, in nondimensional form,

A o 2n’ 2nh
~——=—""1+— == tanh~—. (6.5a)
gh 2nh pg A A

If 6 is negligibly small, Equation 6.3a is recovered. This result shows that
the effect of surface tension is to increase the propagation speed of the wave.

For deep liquids, the parameter 2zh/A is large, so that Equation 6.5a
becomes

Y o 2’

gh  2mh +pg A

If, in addition, the parameter inside the brackets is sufficiently large that

2
o 2n
— — >1
A

P8
then the expression for the propagation speed will reduce to the following;:

2 2mo

gh pghh’

(6.5b)

Waves that satisfy the foregoing conditions and so travel at the speed
defined by Equation 6.5b are called capillary waves. It will be noted that the
propagation speed of capillary waves depends upon the wavelength A, so
that an arbitrarily shaped wave will disperse because of the different propa-
gation speeds of its Fourier components.

The propagation speed of sinusoidal waves, as predicted by Equation 6.5a,
is shown in Figure 6.5 as a function of the parameter A/(2xh). It is seen that
the effect of surface tension modifies our previous result only in the deep-
liquid end of the spectrum. This is because the condition

02752
— — >1
pg A

is realized only for small values of A, which, in turn, corresponds to deep-
liquid waves.
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FIGURE 6.5
Propagation speed for sinusoidal waves including the effects of surface tension.

6.5 Shallow-Liquid Waves of Arbitrary Form

It was deduced from the results of the previous two sections that waves of
arbitrary form will disperse unless the liquid is shallow. That is, owing to the
different propagation speeds of its Fourier components, an arbitrarily shaped
wave will decompose unless the liquid depth & is small compared with the
shortest wavelength A of the various Fourier components that constitute the
wave. The deduction that shallow-liquid waves of small amplitude will not
decompose may be verified by carrying out a detailed study of such waves.

The starting point of such a study is the equations governing the depen-
dent variables. These equations may be obtained from the continuity and
Euler equations by integrating across the fluid depth and employing a one-
dimensional approximation. However, it is no more difficult and consider-
ably more instructive to derive the equations from first principles using a
one-dimensional approach. The latter procedure will be followed here.

Figure 6.6a shows a portion of a liquid layer in which a surface wave of
arbitrary form exists. The waveform is assumed to be such that the smallest
wavelength of its various Fourier components is large compared with the
mean depth /. Then, a one-dimensional approximation may be employed.
That is, the x component of the velocity vector will be assumed to be con-
stant over the fluid depth, and the y component of the velocity vector will be
neglected as being small.

Figure 6.6b shows an element of length Ax of the fluid that extends from
the bottom to the free surface. The mass-flow rates into the element and out
of it are also indicated in Figure 6.6b. The mass-flow rate per unit depth
entering the element through the surface at x is pu(h + n). Then, the mass flow
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(@ J’T =10

pu(h +n) + a%c [pu(h +n)]

pu(n-+ ) + 2 [pu(n -+ )] Ax

1 2,071 2
SPgUr+n)” + =2 [Spelh + n)7] Ax

L]

FIGURE 6.6
(a) Arbitrary waveform on a shallow-liquid layer, (b) mass-flow-rate balance for an element,
and (c) momentum and force balance in the x direction.

leaving the element at x + Ax is indicated by the first two terms of a Taylor
series about the location x, the remaining terms, not indicated, then being of
order (Ax)? or smaller. A mass flux is shown leaving the control volume at the
top. This is due to the fact that n depends on both x and ¢, and the quantity
on/ot represents the vertical velocity of the free surface. Then, multiplying
this velocity by the density and the length Ax gives a mass efflux per unit
time per unit depth. Using the expressions for these various components of
mass-flow rate, the equation of mass conservation becomes

J 0
pulh+m)+—ou(h+ WlAx + pa—?Ax— pu(l+m) = 0.
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The first and last terms cancel each other, and the remaining terms may be
divided by pAx to give

am d
— +—[u(h =0.
ot ax[u( +m)]

The limit Ax — 0 may now be taken so that the terms that were not included
in the Taylor expansion become identically zero. In the resulting equation,
the product un will be of second order and so may be neglected. Hence, the
linearized form of the continuity equation is

m + ha—u =0.
ot ox (6.6a)

Figure 6.6c shows the same element of fluid considered above but on which
the components of the x momentum and the external forces are indicated.
The components of the x momentum are obtained by multiplying the mass-
flow components obtained above by the x component of the velocity vector,
namely, u. In so doing, it should be noted that the mass-flow component that
leaves the control volume by way of the free surface will, in general, have an
x component of velocity, so that an efflux of x momentum will be involved
although the mass flow is essentially vertical. The forces that act on the fluid
element in the x direction are due to the pressure in the fluid. This pressure,
in turn, will be hydrostatic in our linear approximation. Then, at the refer-
ence station at x, the pressure will vary from atmospheric at the free surface
to atmospheric plus pg(h + n) at the bottom. This linear variation in pressure
gives rise to a force in the positive x direction of pg(h + n)?/2, in which gauge
pressures have been used since absolute values have no consequence here.
At the location x + Ax, the first two terms in a Taylor series of this quantity
are indicated in Figure 6.6¢c. Then, from Newton’s second law, the rate of
increase in the x momentum of the fluid as it passes through the control vol-
ume is equal to the net external force acting in the x direction on the fluid.
Thus,

9 ou(h+ mdad+ - fou (4 e + pu 2 ax
ot 0x ot

a1
=— % “po(h+m)? Ax.
P ng( +)

The first term in this equation represents the time rate of increase in the
momentum of the element of fluid, whereas the second and third terms
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represent the net convective increase associated with the various mass-flow
components. The term on the right-hand side of the equation represents the
net external force that comes from the hydrostatic pressures.

Dividing this equation by pAx gives the following form of the equation of
momentum conservation:

m

L+l e+ = gl )

Here, the differentiation on the right-hand side has been carried out, and
Ax may now be permitted to tend to zero, so that the unwritten terms in the
Taylor expansion vanish. This equation will now be linearized in the small
quantities # and 1. Thus, in the first term, the product un is of second order
and hence may be neglected. The entire second term is of second order or
smaller owing to the presence of 12 Likewise, the third term is quadratically
small in # and 1. In the term on the right-hand side, the product n on/dx is
quadratically small and so may be neglected. Thus, the linearized form of
the equation of momentum conservation is

ou am
Py +g P 0. (6.6b)

Equations 6.6a and 6.6b are, respectively, the equations of mass and momen-
tum conservation. They represent two equations in the two unknowns
1 and 1. By forming the cross derivatives d%u/0xdt and 0?n/dxot, first n and
then u may be eliminated between Equations 6.6a and 6.6b. This shows that
the equations to be satisfied by u and n are

— —oh—=0
ot ox?
62n 827]
Sl_gh=1=0
ot* & ox?

That is, both # and n must satisfy the one-dimensional wave equation.
Hence, 1 and n must be of the general form

w(x,t) = fi(x=Jgh )+ g (x+gh D) (6.60)

N, 1) = fo(x=gh )+ g (x +Jgh ) (6.6d)
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where f,, g, and f,, g, are any differentiable functions. The first solution in
each of these equations represents a wave traveling in the positive x direc-

tion with velocity ./gh. The second solution in both cases represents a wave

traveling in the negative x direction with velocity \/E . Thatis, if an arbitrary
wave is traveling along the surface of a shallow-liquid layer, it will continue
to travel with velocity \/gih . This confirms the propagation speed derived
earlier for a sinusoidal wave (Equation 6.3¢c) and shows that the shape of the
wave does not change as it moves along the surface. Thus, if the shape of the
wave is known as a function of x at some time, it will be known for all values
of x and ¢.

6.6 Complex Potential for Traveling Waves

Consider, again, the case of a small-amplitude surface wave in a fluid of arbi-
trary depth. For a sinusoidal wave of the form

N(x,t) = esin 2TJI(x —ct)

it was shown in Section 6.3 that the velocity potential was

do(x,y,t)=C, cos—(x ct) tanh@ sinh —= y osh% .

The constant C, may be evaluated by completely imposing the kinematic
boundary condition on the free surface. This boundary condition was used
only to establish the functional form of the solution, but it was not completely
imposed. Then, as required by Equation 6.2b, the condition d$/dy(x, 0, t) =
on/ot(x, t) gives

G, 2:mcos—(x— t)tanhz—nh——s%c sz—n(x—ct)
A A A A

which is satisfied by setting

C, = _Cis
> tanhQ2mh/\)
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Then, the velocity potential for a traveling sinusoidal wave is

o(x, y,t)——cscos—(x ct) sinh —= ny thﬁ osh —> 2ny (6.7a)

where the propagation speed ¢ must satisfy Equation 6.3a. From Equation
6.7a, the stream function for a traveling wave may be deduced, and so the
corresponding complex potential may be obtained. This, in turn, will be
used to establish the particle paths for traveling waves.

Since u = dy/dy = 0/ dx, it follows from Equation 6.7a that

N %ssinz—n(x—ct) sinhzﬂ+coth@msh2ﬂ .
ay A A A A A

Integrating this expression shows that y(x, y, t) is of the following form:

2xh . . 2wy

Y(x,y,t)=ce sin%(x— ct) coshz%y+ cothTsth + F(x)

where F(x) is any function of x that may be added through the integration.
In principle, a function of time could also be added, but it is known that, for
a traveling wave, the time dependence above is correct. Another expression
for y(x, y, t) may be obtained from the fact that v = —dy/dx = d¢/dy. This gives

2 2
o ﬁecosz—n(x—ct) coshﬂ+cothz—nhsinhﬂ
oax A A A Iy I8

so that

w(x,y,t)=cssinz%(x—ct) coshz%Jr th@ th“y +G(y)

where G(y) is any function of y. Comparing this result with the previous
expression for y(x, y, f) shows that F(x) = G(y) = 0, so that

P(x,y,t)=ce sin%(x —ct) coshz% + coth %smhz%y . (6.7b)
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Equations 6.7a and 6.7b define, respectively, the velocity potential
d(x, y, ) and the stream function y(x, y, t) for a traveling sinusoidal wave.
Then, the corresponding complex potential F = ¢ + iy may be established
as follows:

F(z,t) = e
’ sinh(2sth/\)
X cos—(x ct) smhz—ﬂh mh2ny coshz—;fhcoshz%
—1sm—(x cf) mh@ shZ“V h@ th;‘y
-
sinh(2mh/\)

x cosh@ cosz—n(x—ct)costhh—is —(x ct) thny
A A A A

2 2
+sinh2—nh cosz—n(x—ct)sinhly—isinz—n(x—ct)coshﬂ
A A A A A

The hyperbolic functions that are inside the brackets will now be trans-
formed into trigonometric functions having imaginary arguments using the
identities sin ia = i sinh o and cos i« = cosh a. Thus, the complex potential
may be written in the form

Ce

F@D == Gnh@m/n

2my

x cosz—nh coshz—n(x—ct)cos iz—nh —sinz—n(x—ct)sin i—=
A A A A A

+sinh@ _icosz—n(x—ct)sin izﬂ —isinz—n(x—ct)cos izﬂ
A A A A A

=—.67E coshznhcos—(z ct) —is mh@s' 2—ﬂ(z ct) .
sinh(2mth/\) A
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Here, the quantities inside the brackets have been observed to be the
expansions of single trigonometric functions involving x — ct + iy = z — ct.
Again, converting the hyperbolic functions inside the brackets to trigono-
metric functions gives

Cce

FED == Gnn@ai

X COS iz—nh cosz—n(z—ct)—sin iz—nh sinz—n(z—ct)
I3 A A A

Lcosz—n(z—ctﬂh) 6.7
sinh(2mh/h) . A 6.70)

F(z,t)=-
where it has been observed that the brackets contain the expansion of a sin-
gle trigonometric function. Equation 6.7c gives the complex potential for the
traveling sinusoidal wave n(x, t) = & sin 2x(x — ct)/A.

6.7 Particle Paths for Traveling Waves

As a wave train travels across the surface of an otherwise quiescent liquid,
the individual particles of the liquid undergo small cyclical motions. The
precise trajectory followed by the fluid particles may be established with the
aid of the results of the previous section.

Consider a specific particle of fluid such as the one indicated by point P
in Figure 6.7a. The instantaneous position of this particle of fluid will be
indicated by a fixed-position vector z, and an additional vector z, that varies
with time. That is, the length and orientation of z, remain fixed, while both
the length and inclination of z; vary with time. Considering the complex
conjugate of the variable-position vector, that is, considering z, = x; — iy, it
follows that

dz, _dx _,dy;

dt  dt dt
=uU-1iv
=W
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4wy

FIGURE 6.7
(a) Coordinate system for establishing particle paths, (b) particle trajectories due to a sinusoidal
wave, and (c) trajectories in deep liquids.

Then, using Equation 6.7c,

dz, 2n/N)ece . 2n .
o REMCE g T (o et 4 i),
dt " sinh@uyn) S0y, GEH)

Integrating this equation with respect to time gives

_ € 27
=——————cos—(z-ct+ih).
Z sinh(7h/%) cos X (z—ct+ih)

Here, the constant of integration has been taken to be zero without loss of
generality. Such a constant would not affect the time dependence of z;, and
so it would not affect the trajectory of the fluid particle. Rather, it would only
change the length of the z, position vector, which is equivalent to adjusting
the choice of the constant z; position vector.
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Comparing the foregoing expression for z; with Equation 6.7c shows that
Z = - F(z,t) ‘
c

Then, it follows that x; = —¢(x, y, t)/c and y; = y(x, y, t)/c. Hence, from
Equations 6.7a and 6.7b, the coordinates x; and y, of the trajectory of our
reference fluid particle will be given by

x1=ecosz—n(x—ct) sinhzﬂ+coth@cosh2ﬂ
A A A A

y1=ssinz—n(x—ct) cosh2ﬂ+coth2—nhsinhzﬂ .
A A A A

That is, the instantaneous coordinates of the trajectory of a fluid particle
depend on both the x and y coordinates of the fluid particle and on the time.
The time t may be eliminated between these two equations to yield the tra-
jectory of the fluid particle in the following way:

sin’ 2—n(x— ct) + cos® 2—n(x—cif) =1.
A A

Substituting from the preceding equations for x; and y, into this identity gives

x;
&’ [sinh(2zy/A) + coth(2sth/\) cosh(2my/\)]*

2 69
+ Y1
g2 [cosh(2mty/\) + coth(27h/A) sinh(25y /M)

Equation 6.8 shows that the trajectory of a fluid particle depends only on
its depth of submergence, y, and that it is independent of x. Eliminating the
time also eliminated the x coordinate. This might have been expected since
each particle of fluid experiences the same waves passing above it, irrespec-
tive of its x coordinate. Thus, the motion experienced by two particles that
are separated in the x direction will be the same, but the phasing will be
different. Since Equation 6.8 is that of an ellipse, the trajectories of the fluid
particles will be ellipses whose dimensions are determined by the value of
y for the various particles. For particles that lie on the free surface, y = 0, so
that Equation 6.8 becomes
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2 2
Xy Y1

[e coth@t/ W] &2

This shows that the trajectory of particles on the free surface is that of an
ellipse whose semiaxes are ¢ in the y direction and e coth(2zh/}) in the x
direction. This result is shown in Figure 6.7b. For particles that are on the
bottom, y = —h, the semiaxis in the y direction becomes zero, and the semi-
axis in the x direction becomes &/sinh(2x/1/}). That is, the ellipse degenerates
to the line —e/sinh(2nh/A) < x, < ¢/sinh(2nh/)). For values of y that are interme-
diate to y = 0 and y = -/, the particle trajectories will be ellipses as described
by Equation 6.8 and as shown in Figure 6.7b.

For shallow liquids, the ellipses shown in Figure 6.7b merely become elon-
gated in the x direction. However, for deep liquids, the ellipses become circles.
This may be shown by observing that for deep liquids, the parameter 2zh/A
will be very large, so that coth(2nh/)) will be unity. Then, Equation 6.8 becomes

X
&*[sinh(2my /\) + cosh(2my /)]

yi
¥ ¢*[sinh(2y /A) + cosh(2my /M)

This is the equation of a circle of radius g[sinh(2ny/A) + cosh(2zy/A)]. That is,
the radius is ¢ at the free surface and decreases as y becomes more and more
negative. The particle trajectories for deep liquids are shown in Figure 6.7c.

6.8 Standing Waves

Up to this point, we have been dealing with traveling waves, that is, waves
that move along the surface of the liquid. We now consider standing waves,
which are waves that remain stationary—the surface moves vertically only.
An interesting way of obtaining the equation of a standing wave is to super-
impose two identical traveling waves that are moving in opposite directions.
Thus, consider two traveling waves n; and n, as follows:

1 21
x,t)=—¢esin—(x —ct
i h) = esin T (x-cf)

1 21
x,t)=—esin—(x + ct).
Ma(x8) = esin " (x+c)
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Let n(x, t) represent the free-surface profile that results from superimpos-
ing these two traveling waves. Then,

1 2% 2
X, t)=— in—(x-ct in— (x +ct
7( )Zasm)\( )+sm}»(+)

1 . 2mx 27ct 2nx . 2mct
=—¢ Ssin——CcoS—— — C0OS——sin ——
2 A A A A
2nx 2mct 2nx . 2mct

+sIN ———COS —— + COS —— Sin ———
A A A A

. 2nx 2ct
= gsin o cos——

That is, the superposition of two identical traveling waves results in a wave
that, at any time, is a sine function in x and that, for any value of x, oscillates
vertically in time. Such a wave, in which the entire surface oscillates in time,
is called a standing wave.

The complex potential for a sinusoidal-shaped standing wave may be
obtained by superimposing the complex potentials for two traveling waves
moving in opposition to each other. Thus, Equation 6.7c will be used to obtain
the complex potential for two waves, each of amplitude £/2 and wavelength 2,
one of which is traveling in the positive x direction with velocity c and the other
of which is traveling in the opposite direction with the same velocity. Hence,

ce/2 —COSZi(Z—Ct+ih)+C0527n(Z+Ct+ih) .

FED = Gy »

The cosine functions will now be expanded, taking z + i as one element
and ct as the other. This gives

P@ﬂ=$ﬂ$ﬁmm
X _COS%(Z+ ih)cos%—sin%(z+ih)sin?
cos 2 e+ cos - sin 2 (2 + sin 27!
P(Z/t)=—mh(;{;m/msm2;(z+ih)smzzd, ©69)
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Equation 6.9 gives the complex potential for a standing sinusoidal wave of
wavelength A that is oscillating in time with frequency 2rc/A.

6.9 Particle Paths for Standing Waves

Following the procedure employed in Section 6.7 for traveling waves, the
particle paths for standing waves may be established from the complex
potential. Using the same coordinate system as was used in Section 6.7, it
follows as before that

dz, _dF
dt  dz’

Then, using Equation 6.9,

LE - _(2/N)ece cosz—n(z + ih)sin 2met
dt sinh(2mh/\) A

Integrating with respect to time and neglecting the constant of integration
as before give the following expression for z;:

Z, = Lcosz—(zwzh)cosﬁ
' sinh(2wh/M\) A

Writing z + ih = x + i(y + h) and expanding the trigonometric function of
this argument give

_ g
7 sinh(27th/2)
cos% COSZ%xcoshf(]/+ h)—lsstmhi(y+ m)

in which the trigonometric terms having imaginary arguments have been
converted to hyperbolic terms. The quantity z, is complex and so may be
written in the polar form

Z, =ne™
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where r; and 6, are defined by

€

"= sinh(2mh/0)
(6.10a)

xcos\/co 2 Zmx Coshz—(y+h)+sm ;:xsinhzz%c(yﬂi)

0, = tan™! tanz%xtanhz%(y +h . (6.10b)

Equations 6.10a and 6.10b show that, for given values of x and y, the polar
angle 0, of the particle trajectory is constant, whereas the radius r; oscillates
in time. Thus, the particle trajectories will be straight lines whose inclination
will depend upon the location of the particle under consideration. In particu-
lar, when x = nA/2, Equations 6.10a and 6.10b reduce to

27ict cosh(2st/N)(y + h)
A sinh(27th/\)

1, = €COS
6, =0 orm.

This describes a family of horizontal lines whose length r, decreases as
the depth of submergence increases. The location x = n\/2 corresponds to
the nodes of the free surface, that is, the points of the free surface that have
no vertical motion. The horizontal motion of these points, which is shown in
Figure 6.8, is necessary to satisfy the continuity equation as the maximum
amplitude of the wave shifts from one side of the node to the other as the
surface oscillations take place.

Midway between the nodes, that is, at x = (2 + 1)A/4, Equations 6.10a and
6.10b show that

27ict sinh(2m/A)(y + h)

S T inh(2mh/ )
0, = T or 3—“
2%

This defines a family of vertical lines whose length #, decreases as the
depth of submergence increases and reaches zero on the bottom, y = —h. This
motion is also shown in Figure 6.8. As the boundary condition requires, the
vertical motion vanishes on y = —h.
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FIGURE 6.8
Particle trajectories induced by a sinusoidal standing wave of amplitude e and wavelength A.

6.10 Waves in Rectangular Vessels

The fact that standing waves may exist on the surface of an infinite expanse
of liquid raises the question of whether standing waves may exist on the
surface of a liquid that is contained in a vessel of finite extent. In this sec-
tion, rectangular vessels will be considered, and it will be shown, as might
be expected, that only standing waves whose wavelengths coincide with a
discrete spectrum of values may exist on such liquid surfaces.

Figure 6.9a shows a two-dimensional rectangular container of width 2/
that contains a liquid of average depth h. For this configuration, we ask the
following question: What type of steady-state or pseudo-steady-state waves,
if any, may exist on the surface of the liquid? Any waves that may exist will
have to satisfy the following partial differential equation and boundary
conditions:

24+5% 0 (6.11a)

0’9 )
y(x, h,t)+ g@(x, h,t)=0 (6.11b)
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FIGURE 6.9

(@) Geometry for liquid in a rectangular container and (b) first four fundamental modes of
surface oscillation.

9 20,820 6110
oy

0 p oy p) =
ax(-l,y,t) 0. (6.11d)

The first boundary condition is the pressure condition at the free surface in
which the kinematic condition has been employed, and the other boundary con-
ditions prevent normal velocity components on the bottom and side surfaces of
the container. Since the free-surface profile is not being specified a priori here,
the kinematic condition at the free surface should not be imposed separately.

Since a steady-state wave solution is being sought, the velocity poten-
tial should have trigonometric time dependence. It may be observed that
the existence of the sidewalls at x = +I eliminates the possibility of travel-
ing waves since the particle paths for traveling waves are ellipses, so that
the wall boundary conditions could not be satisfied. The time variation will
therefore be of the standing-wave type and will be chosen to be sin(2nct/A).

There is no loss in generality with this choice since any phase change merely
corresponds to a shifting of the time origin, which is of no consequence here.
Thus, the appropriate separable solution to Equation 6.11a will be of the form

o(x,y,t)= A sinz%x +A, cosz—;:x

2 2
x B, sinh il AN B, cosh ¥ sin Zct
A A A

The x dependence has been chosen to be trigonometric in view of the homo-
geneous boundary conditions at x = +[. Then, in order to satisfy Laplace’s
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equation, the y dependence must be exponential or hyperbolic. In view of
the finite extent of the domain and the homogeneous boundary condition
at y = 0, the hyperbolic form has been employed. The boundary condition
(Equation 6.11c) requires that B, = 0, so that the velocity potential becomes
of the form

o(x,y,t)= D, sm2 +D,c szix coshzny m—t.
A A A A

The pressure condition on the free surface (Equation 6.11b) then requires
that

L e .
A A I8 A
X Dlsin%+D2coszlx sinM=0.
A A A

Since this equation is to be satisfied for all values of x and all values of ¢, it
follows that the quantity inside the square brackets must be zero. This gives

> A o 2
gh 2mh A

That is, the pressure condition on the free surface establishes the frequen-
cies of the wave motion. It will be seen that this result agrees with Equation
6.3a and that each Fourier component of the waveform has a different fre-
quency of motion.

The final boundary condition to be satisfied is that of no horizontal veloc-
ity component at the vertical walls of the container. Thus, Equation 6.11d
requires that

2
2n D, cos—27d¢D2 sin—2 al cosh =¥ m—t =0
A A I I A

which will be satisfied for all values of y and ¢t if

2
D, Cosil ==D, mz—nl
A A
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This condition may be satisfied by setting D, = D, = 0, but then ¢ = 0, which
is the trivial solution. For a nontrivial solution, either D, or D, at least must
be different from zero.

Suppose, first, that D, is different from zero and D, = 0. Then,

Cosz—nz=0
A

n

!
" 2n+1

where the subscript 1 has been associated with the quantity A in anticipation
of the fact that the foregoing transcendental equation may be satisfied in an
infinite number of ways. That is, one way of satisfying the side boundary
conditions is to choose the preceding values of A, so that the corresponding
velocity potentials will be of the form

2n+1
6,(x,y,1) = D, sin (2n+ Dmx wosh (2n + Dmy sin (2n + Dmc, t
21 21
where ¢, is related to A, through the identity that resulted from imposing the
pressure condition on the free surface.
Next, suppose D, = 0 and D, is different from zero. Then,

sinz—nl=0
iy, =2,
m

Thus, another way of satisfying the side boundary conditions is to adopt
the value above for A, so that the corresponding velocity potentials will be

0 ,Y,1) = Dy cos " cosh Y sin M0t

where c,, is related to A,. The first two surface modes corresponding to ¢,
and ¢,, are shown in Figure 6.9b.

It will be seen that, out of the continuous spectrum of wavelengths that
may exist, only those waves whose particle paths are vertical at x = =/ are
permissible solutions. This gives rise to an even spectrum of modes (corre-
sponding to D, = 0) and an odd spectrum of modes (corresponding to D, = 0).
That is, there is a discrete spectrum of wavelengths whose particle paths are
vertical at x = +/ and that may therefore satisfy the boundary conditions at
the sidewalls.
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The individual solutions given by ¢, and ¢,, may be superimposed to
describe more general waveforms. Thus, a more general solution will be
obtained by superimposing all the ¢,, solutions and all the ¢,, solutions. This
gives

o(x,y,t) = 2 D, sin " ;ll)ﬂx cogh 21Dy 21+ et

2] 21
(6.12a)
+E D,,, cos M. osh ™Y sin mac,t
Ly I l l
where
2
Cu _ 21 tanh (2n + 1)mh (6.12b)
gh  (Cn+1)mh 21
and
2
U3 (6.120)
gh  mnh I

The coefficients D,, and D,,, that appear in Equation 6.12a are undeter-
mined at this point. If the initial shape and velocity of the free surface are
specified, these constants may be evaluated. An example of how this may be
utilized is to establish the response of a body of water to an earthquake. The
body of water may be an artificial reservoir or a lake whose shape may be
approximated by a rectangular container. Seismographic records for the area
would indicate the magnitude and frequency of the expected accelerations.
These data may be Fourier-analyzed and used to establish a surface wave-
form and oscillation frequency at the end of the earthquake, which would
be the beginning of the standing-wave oscillations. The constants D, and
D,,, may be used to fit these data, and then Equation 6.12a will describe the
subsequent motion.

6.11 Waves in Cylindrical Vessels

An analysis similar to that presented in the previous section may be carried
out for cylindrical containers. Figure 6.10a shows a cylindrical container of
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FIGURE 6.10

(@) Geometry for liquid in a cylindrical container and (b) Bessel functions of the first and sec-

ond kind.

radius a that contains a liquid whose average depth is i. Then, in terms of the
cylindrical coordinates R, 6, and z and the time ¢, the problem to be solved for
the velocity potential ¢(R, 6, z, ) is

1o o0 1% %

LA S A )

RoR ~oR 'R 00% @ o2 (6132)

%0 3

PR, 0,1+ g P (R,0,h,t)=0 .

P ( )+ g P ( ) (6.13b)
9 (R 6,0,t)=0 (6139
0z
%(a,e,z,t) _0. (6.13d)

The solutions to this problem will describe the possible waveforms that
may exist on the surface of the liquid in the container.
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The solution to the foregoing problem may be obtained by the method of
separation of variables. Thus, a solution is sought in the form

PR, 6, z, t) = R(R)T(0)Z(z) sin wt
Here, the time dependence has again been taken to be sinusoidal, corre-

sponding to standing waves. Substituting this expression for ¢ into Equation
6.13a and multiplying by R?/¢ give

Rd pd4 1dT R dZ
ARdR  dR T de* Z dz?

=0.

Following the usual argument of separation of variables, it is observed that
the second term in this equation contains all the 6 dependence and that itis a
function of 0 only. Then, this term must equal a constant. This constant will
be chosen to be —m?, where m is an integer. The significance of the minus sign
is that trigonometric rather than exponential 6 dependence will result, and
the significance of m being an integer is that $(0) = $p(0 + 2x) will be satisfied,
as is required. The solution for T(6) is then

T©O) = A, sin m6 + A, cos m6.

The remaining differential equation is, after dividing by R?,

1 d dz  m* 1d°Z

L )
RAZdR dR _R® ' Z 42

The separation-of-variables argument now requires that the last term be
equal to a constant. Since trigonometric z dependence does not fit the physi-
cal circumstances, this constant will be chosen as k2. Then, the solution for
Z(z) will be

Z(z) = B, sinh kz + B, cosh kz.

Here, the hyperbolic form has been used in preference to the exponential
form in view of the finite extent of the z domain.
The remaining differential equation, after multiplying by R2%, becomes

Ri Rd—%) +(K*R? -m*)A =0
dR dR



238 Fundamental Mechanics of Fluids

But this is Bessel’s equation of order m whose solution is
AR) = Dy,]u(kR) + D, Y, (kR)

where J,, is Bessel’s function of the first kind, and Y,, is Bessel’s function of
the second kind. The first two Bessel functions of each kind are shown sche-
matically in Figure 6.10b.

Since the Bessel functions of the second kind, Y,,(x), diverge for x = 0 for all
values of m, the coefficients D,,, must be zero. Thus, the radial dependence of
the velocity potential will be proportional to J,,(kR). Then, for any integer m,

the solution by separation of variables is
0,(R,8,z,t) = (A,, sinmb + A,,, cosmb)
x (B,,, sinh kz + B,,, cosh kz)],(kR)sin w t.
The boundary condition (Equation 6.13c) requires that B,,, be zero, whereas

the pressure condition at the free surface (Equation 6.13b) determines the
oscillation frequency to be

®? = gk tanh kh.
Thus, the velocity potential will be of the form
o, R, 6, z, t) = (K, sin mO + K,,, cos mb) cosh kz |, (kR) sin wt.

The remaining boundary condition (Equation 6.13d) then requires, for a
nontrivial solution,

Jon (ka) =0

where the prime denotes differentiation. This transcendental equation may
be satisfied by any of an infinite number of discrete values of k. These values
will be distinguished by employing a double subscript on k. Thus, k,,, will
denote the nth root of the J,, Bessel function in the equation

]/m (kmna) = O'

Values of k,,a that satisfy this equation may be found in tables of func-
tions. For example, the first few roots of [, (ky,a) = 0 are given below:

n 0 1 2 3
ko,a 3832 7016 10174 13.324°

From the foregoing analysis, one solution to the problem posed for the
velocity potential is

o, R, 6,z t) = (K, sin mb + K,, , cos m) coshk,,z ], (k,,R) sinw,,t.

mn
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Here, a double subscript has been associated with the oscillation frequency
o since this quantity is related to the separation constant k. The foregoing
expression for ¢,,, represents a valid solution to our problem for any inte-
ger m and any integer n. Then, a more general solution may be obtained by
superimposing all such solutions to give

© ©

o(R, 8,2, t) = E E (K., sinm + Ky, cosmb)
m=0 n=0 (614a)

xcoshk,, 2], (k,,R)sinw,,,t
where

and
o (1) = 0. (6.14¢)

As was the case in the previous section, the remaining arbitrary constants
may be defined by specifying the nature of the free surface at some value of
the time.

A simple illustration of the validity of the result above may be obtained by
use of a cup of coffee or some other liquid. If such a cup is jarred by strik-
ing the bottom squarely on a flat surface, it may be induced to vibrate in
a purely radial mode. That is, the fundamental mode in which the surface
R = a vibrates in and out may be induced. This motion causes surface waves
that will also have no 6 dependence. Then, putting m = 0 in Equation 6.14a
shows that the velocity potential will be proportional to J,(ky,R). Thus, the
surface will adopt the shape of the |, Bessel function, which is shown in
Figure 6.10b. This shape may be actually observed, and under certain condi-
tions, the peak at the axis of the vessel may become very pronounced. Of
course, the analysis is no longer valid under these conditions since large
amplitudes lead to large slopes, which violates the assumptions that were
made in the linearization.

6.12 Propagation of Waves at Interface

As a final example of surface waves, the behavior of propagating waves at
the separation of two dissimilar fluids will be investigated. Figure 6.11 shows
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FIGURE 6.11
Wave-shaped interface separating two different fluids traveling at different average speeds.

a wavy surface y = n(x, f) below which a fluid of density p, flows with mean
velocity U, in the x direction. Above the interface is a fluid of density p,
whose mean velocity is U, in the x direction.

For the foregoing configuration, we specify a sinusoidal waveform at the
interface and ask, what is the propagation speed of the wave? That is, we
specify the equation of the interface to be

n(x, t) — eei(Zn/}L)(x—Gt)'

This represents a sinusoidal wave of amplitude ¢ and wavelength A. If ¢
is real, the wave is traveling in the x direction with velocity o, whereas if ¢
is imaginary, the wave is decaying (if ¢/i is negative) or is growing (if ¢/i is
positive). The last situation represents an unstable interface.

Since the two fluids now have nonzero mean velocities, the linearization
of the boundary conditions must be reexamined here. Letting the subscript
ibe 1 or 2 for the lower and upper fluids, respectively, the velocity u; may be
written as follows:

u; = Ue, + Vo,

where ¢, is the velocity potential for the perturbation to the uniform flow
caused by the waves at the interface. Then, the material derivative becomes

D o

D_d,,.

Dt ot
RTINS

ot ax
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The second term on the right-hand side of this identity is new and is of
first order in this case. The third term on the right-hand side involves veloc-
ity components derived from the perturbation velocity potential, and hence
these velocity components will be small.

Using this result for the material derivative, the kinematic condition on the
free surface, D(y — n)/Dt = 0, becomes

S
ot ox oy

The last term on the left-hand side of this equation is quadratically small,
for small-amplitude waves, and so may be neglected. Thus, the revised kine-
matic boundary condition on the free surface is

a0; am m
L(x,0,t) = —(x, )+ U, — (x, t). 6.15
oy (x,0,t) o (x,t)+ ax(x ) (6.15a)

Comparison of this result with Equation 6.2b shows that the last term in
the preceding equation is new and that this term vanishes for U, = 0.

From Equation 6.1c, the Bernoulli equation for a constant-pressure surface
in which F(t) is absorbed into the velocity potential is, as before,

99;
ot

P; + %piui ‘u; + ;8N = constant.

Substituting our expansion for u; into this equation and neglecting qua-
dratic terms in the perturbation velocity give

9 (v 0,6)+pign(x, f) = constant.  (6.15b)

o;
! ,O,t iui
; (0Bl =

pia

Here, the term pU; /2 has been absorbed into the constant on the right-
hand side of this equation.

Using Equations 6.15a and 6.15b, we may now define the problem to be
satisfied by the velocity potentials ¢, and ¢,. In the region y < 0, the velocity
potential ¢; must satisfy the following partial differential equation:

%9, 9
b, Py (6.160)
ox ay
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where the velocities derived from ¢, should be finite. That is,
|V, | = finite. (6.16b)

Similarly, in the region y > 0, we have

3¢, 0
T T (6.16c)
X ay

|V, | = finite. (6.16d)

At the interface, which is linearized to y = 0, the kinematic condition must
be satisfied by ¢, and by ¢, separately. Thus, from Equation 6.15a,

N 0,8 = M+, M, g 6.16¢)
ay ot ox

92 (00,8 = M x5y + 1, DV (). (6.166)
oy ot ox

Finally, the pressure condition at the interface must be satisfied, which, in
the present case, amounts to equating the pressure in the two fluids at the
interface. Thus, from Equation 6.15b, since the Bernoulli constant will be the
same for i = 1 and i = 2, the pressure condition becomes

i) i)
o0 28 (x,0,6) + oL, 2L (x,0,6) + pygm(, )
ot ax
(6.16g)
_ aq)Z u aq)Z
- p2 af (X,O,f)+p2 2 3x (xrozt)*'ng”fl(x/t)-

Equations 6.16a through 6.16g represent the problem to be satisfied by any
perturbation to the uniform flows. In particular, we decided to study the
effect of a sinusoidal wave at the interface, and so the equation of the inter-
face was chosen to be

n(x, t) = eel@v/N=oh), (6.16h)

The solution to Equations 6.16a and 6.16c may be obtained by separation
of variables. In view of the shape of the interface, as defined by Equation
6.16h, the solutions should be trigonometric in x. Then, the y dependence
will be exponential. In view of the conditions 6.16b and 6.16d, the negative
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exponential should be rejected for ¢, and the positive exponential should be
rejected for ¢,. Thus, the solutions to Equations 6.16a and 6.16¢ that satisfy
Equations 6.16b and 6.16d are

¢1 (.X', Y, t) _ Ale(Zn/K)yei(Zn/)\)(x—ot)

0y (X, 1) = Aze—(Zn/k)yei(Z:r/?»)(x—ot)'

Imposing the kinematic surface conditions 6.16e and 6.16f on these solu-
tions shows that

A, =ie(-o + Uy
A, = —ig(—c + U,).
Thus, the velocity potentialsin the lower and upper regions are, respectively,
dy(x, y, £) = —ig(o — U,)eCx/Meiea/e—ob
P, (x, Y, t) = ie(o — Uz)e-(2n/x)yei(2n/x)(x—af).
These solutions satisfy all the required conditions except the pressure con-

dition at the interface. Equation 6.16g then requires

1 .27 . .21
pri(o =) 170 - puthyi(o - Uy) 17 P8
1 ;2 , 2;
= pi(0-U,) _170 +p,Uyi(0 - U,) —17 +028-

The quantity n, as defined by Equation 6.16h, has been canceled through-
out this equation as a nonzero common factor. Combining the first and sec-
ond terms on each side of this equation reduces it to the following form:

271 27
-py 7(0—111)2 +P1g =P, 7(0—112)2 +P,8-

Everything in this algebraic equation is known a priori except the quan-
tity . Then, the equation above should be looked upon as a quadratic equa-
tion for c. Solving this quadratic equation gives

2
o= PUs + iUy + Pl + iy poU +p U7 P2 @
P2 + 0y P2 + 0y P2 + Py Pr+p 27
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The first two quantities inside the square root may be combined to give the
following simplified expression for c:

o= pzuz + plul i\/ P1—P2 @ _ P1P2 - (u2 _ ul)Z . (617&)
P2+ P Pr+pP2 2 (py+py)

Equation 6.17a shows that ¢ may be real, imaginary, or complex, depend-
ing upon the nature of the free parameters. Several special cases will be
investigated.

Consider, first, the special case where U, = U, = 0 and p, = 0. This would
correspond to two stationary fluids in which the density of the upper fluid
is very small compared with that of the lower fluid. Such a condition would
closely approximate a stationary liquid over which a stationary gas exists,
for example, air over water. Then, Equation 6.17a shows that ¢ will be real,

having the values
o= :,/gl, 6.17b)
27

This agrees with Equation 6.3b, which gives the propagation speed for sur-
face waves in deep liquids. The minus sign in Equation 6.17b corresponds to
a wave traveling in the negative x direction. Since it turned out that o is real,
the waves at the interface will propagate, so that the surface of separation
will remain intact. That is, the interface is stable.

Next, consider the case where p, = 0 and the other parameters are nonzero.
Physically, this would approximate the case of a gas blowing over a liquid
surface. Under these conditions, Equation 6.17a reduces to

o=U, = % 6.17¢)

But this is just Equation 6.17b, in which a Galilean transformation of mag-
nitude U, has been applied. That is, the waves move along the surface of the
liquid at the speed of the liquid plus or minus the speed of the waves on a qui-
escent body of the liquid. Again, the interface will remain intact and so is stable.

Consider now the case in which p, = p;. Physically, the situation is a dis-
continuity in the velocity (i.e., a shear layer) in a homogeneous fluid. Then,
Equation 6.17a becomes

o= uz;ul il'uzgul. 6.17d)

This result shows that unless U, = U, (in which case there is no shear layer),
the quantity ¢ will have an imaginary part that will result in the interfacial
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wave growing exponentially with time. That is, the interface at the shear
layer is unstable. This form of instability is known as Helmholtz instability or
Rayleigh instability.

Finally, consider both fluids to be quiescent so that U, = U, = 0, but let their
densities differ. Then, Equation 6.17a reduces to

S G T VS 6.17¢)
2x Py +p,

For p, > p,, that is, for the heavier fluid on the bottom, ¢ will be real, so that
the interface will be stable. However, for p, > p;, that is, for the heavier fluid
on top, o will be imaginary, so that the interface will be unstable. This form
of instability is known as Taylor instability.

PROBLEMS

6.1 Equation 6.5a, which is reproduced below, defines the speed with
which a small-amplitude sinusoidal wave travels along the surface
of a liquid in which the effects of surface tension are included:

¢ _ g, 0 2wk o
gh 2nh pgh®> A A

Numerical evaluation of this equation, as presented in Figure 6.5,
shows that the wave speed has a minimum for nonzero values of
the surface tension and that the location of this minimum corre-
sponds to a small value of the wavelength A. Obtain an estimate
of the value of A corresponding to this minimum wave speed by
assuming that the value of tanh(2n//}) in the above equation may
be evaluated for small values of the wavelength A.

Substitute the result obtained above into the full version of
Equation 6.5a and hence obtain an analytic expression for the mini-
mum speed of the wave.

6.2 The complex potential for a traveling wave on a quiescent liquid
surface is

c

€
P& = = nn@ain)

cosz%c(z - ct + ih).

Use this result to deduce that the complex potential for a sta-
tionary wave on the surface of a moving liquid layer whose mean
velocity is ¢ in the negative x direction is
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cos 2n (z + ih).

Ce
F@) =~z = G hamiyn) < a

Show that, in very deep liquids, the result above becomes

F(Z) =—cz — Cge—fa?l/}\)z' (618&)

Use this last result to determine the stream function y(x,y) for a
stationary wave on the surface of a deep-liquid layer whose mean
velocity is c. Hence, show that the streamline y(xn) = 0 gives the
equation of the free surface as

1 = ee@/Pn smz%x. (6.18b)

6.3 The Bernoulli equation for the situation depicted in Problem 6.2 is

2

E+lu-u+gy=£+lc2 14eg2 28
p 2 p 2 A

where P is the pressure at the free surface. Use this equation,
together with the result u-u = u* + v> = WW and Equation 6.18a, to
show that setting p = P when y = n requires that the quantity c, the
velocity difference between the mean velocity and the wave-train
velocity, must satisfy the following equation:

= 2gm .
22/ 0)? + 4t/ — €2 (2 /N)2 e/

Linearize this expression for small values of &/A and hence con-
firm the following relation for deep liquids:

A

gh 2mh’

Further, show that, by neglecting only those terms that are of
fourth order or smaller, the general expression for ¢ becomes

RN (¢
1-(2n/N)?e?’
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6.4

6.5

This shows that the effect of finite wave amplitude is to increase
the wave speed on the surface of the liquid.

The complex potential for a traveling wave on an otherwise quies-
cent liquid is

F(z,t)=- cosz%c(z - ct + ih).

ce
sinh(27th/\)

Apply a Galilean transformation to the coordinate system that
was used in arriving at this expression and hence show that the
stream function for the stationary wave shown in the figure for this
problem is

e in 2™ ginh 27“ (v +H) (6.19a)

W)= =y + o S

where

2
Ul * a2 (6.19b)
gH 2nH A

Here, H is the mean depth of the liquid while € and X are, respec-
tively, the amplitude and the wavelength of the stationary wave on
the surface of the liquid.

y
€ —ssinﬂ
N
.,

Tl

H — ]
53 R EREBEEBEELEIELLES,

The result of Problem 6.4 may be used to obtain the solution to the
problem of steady flow over a wave-shaped surface. The configu-
ration for which the solution is sought is shown in the figure for
this problem. To obtain the required solution from the results of
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Problem 6.4, observe that the solution obtained in Problem 6.4 will,
at some depth /1 < H, have a wave-shaped streamline of amplitude
g, and wavelength A. This streamline may be considered to be a
surface, so that if the liquid surface is taken to be y = 0, the bound-
ary defined by y = Uh will correspond to y = — h + n, where n, =
gosin(2mx/A). In this way, show from linear theory that the ratio of

the wave amplitude to the wall amplitude is

€ 1

g cosh(2mh/M) - (gh/2nU?)sinh(2mh/2)

The various parameters in this result are defined in the figure for

this problem.

21X

J :y:—h+€osin X

B

6.6 Consider two traveling waves that are defined by the following

equations:

1 . 2=n
x,t)=— —(x—c4t
m(x,t) 25511'17\'1( 1t

1 . 2=n
x,t) = —esin — (x — C,t).
Mo (x,t) 25 7\2( ot)

Show that the equation of the free surface that results from super-
imposing these two waves will be defined by

1 1
nx,H=ecosm — - x— A_2 4
)\'1 )\’2 1 7\.2
xsinm —+— x— L% 4
1 2 Mo
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Hence, show that if A, and A, differ by only a small amount and
that if ¢; and ¢, differ by only a small amount, the resulting surface
profile may be considered to be of the following form:

1 1 c c
x,t) = A(x, t)si — b — x- L4+ 2t
n(x, 1) (x,t)sinm " + * » + r,

which represents a traveling wave whose amplitude is changing
slowly with time compared with the frequency of oscillations. This
situation is shown in the figure for this problem (y versus t for a given
x location). The phenomenon is referred to as beating, and it occurs in
situations where two similar waves or signals are superimposed.

WP\

6.7 (a) The potential energy per wavelength of a wave train is given by
the expression

A
1

Ve Loen?dr.
fongn x

Use this expression to show that the potential energy per wave-
length of the wave n = € sin2n(x — ct)/A is

1
V=" pge’h.
4 P38
(b) The kinetic energy per wavelength of a wave train is given by
A
T-1p f s
2°Jo oy

Use this expression and the velocity potential for a traveling sinu-
soidal wave

dx.

y=0

o(x,y,t) = —ce cos 2%c(x —ct)x sinh % + coth % cosh Z%y (6.7a)
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to show that the kinetic energy per wavelength of the same sinusoi-
dal wave is

1
T =—pge’A.
4988

6.8 The work done on a vertical plane due to waves on a liquid of depth
h is given by

0 (:)q)
WD = —dy.
f-hpax Y

Use the linearized form of the Bernoulli equation and Equation
6.7a to show that the work done on a vertical plane by a traveling
wave defined by n = € sin 2n(x — cf)/A is

2mh/\
sinh(2th/\) cosh(2ah/A\)

WD = %pgca2 sin? %(x —ct) 1+

Hence, show that for deep liquids, the time average of the work
done is one-half of the sum of the kinetic energy per wavelength
and the potential energy per wavelength. That is, show that

(WD)

1
= (T+V).
ave 2 ( + )
6.9 The distribution of vorticity in a lake is assumed to be represented
by the following expression:

Qx, y, t) = o(x, y, £) + Py.

In this equation, Q is the total vorticity, w is the intrinsic vorticity,
and f is a constant. x and y are coordinates that lie on the surface of
the lake, and the stream function for the liquid motion is taken to be

W(xr y, t) — Aei(kx+1y+ct).

Here, A is a constant, k and / represent the wavelengths in the x
and y directions, respectively, and ¢ represents the speed at which
the wave is traveling. Use the definition of vorticity and the defini-
tion of the stream function to show that w is proportional to y, and
find the constant of proportionality.

Using linear theory, show that the material derivative of the total
vorticity Q is proportional to the stream function y, and find the
constant of proportionality. Also, find the value of the speed ¢ that
makes the material derivative of Q zero.
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6.10 One very simple wave of representing a boundary layer is to con-
sider it to be a layer of zero fluid velocity. The figure for this problem
shows such a model of a boundary layer for uniform flow over a
flat plate. For this configuration, carry out a stability analysis of the
interface by imposing a wave on the depicted flow of the following
form at the interface:

n(x, 1) = eeiCa/m-cb),
Determine whether or not the interface is unstable to this wave,

and if it is unstable, determine the fastest growing wavelength of the
instability.

6.11 A fluid of density p, occupies the space —h <y < 0, while a different
fluid of density p, occupies the space 0 <y < k. Solid boundaries exist
at both y = —h and y = +h, and both fluids are originally at rest. A
small-amplitude traveling wave of wavelength A is introduced along
the interface separating the two fluids. Using linear theory, deter-
mine the speed at which the wave will travel along the interface and
discuss the conditions under which the amplitude of the wave will
decay with time, or grow with time.

Further Reading—Part 11

The topic of ideal-fluid flow is probably the most studied branch of fluid
mechanics, and it is well represented in the literature. Most texts on fluid
mechanics have at least one chapter on the subject, and some books are
entirely devoted to it. The following books, collectively, cover the subject in
some depth. The book by Sir Horace Lamb was first published in 1879, and
it does not utilize vector analysis or tensor analysis. However, this book has
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been a standard reference for many years, and it continues to be a valuable
source of information.

Denath, L.: Nonlinear Water Waves, Academic Press, San Diego, 1994.

Jeffrey, A.: Complex Analysis and Applications, CRC Press, Boca Raton, 1992.

Kirchhoff, R. H.: Potential Flows, Marcel Dekker, New York, 1985.

Lamb, H.: Hydrodynamics, 6th ed., Dover Publications, New York, 1932.

Lighthill, ].: Waves in Fluids, Cambridge University Press, London, 1978.

Milne-Thompson, L. M.: Theoretical Hydrodynamics, 4th ed., The Macmillan Company,
New York, 1962.

Panton, R. L.: Incompressible Flow, John Wiley & Sons, New York, 1984.

Robertson, J. M.: Hydrodynamics in Theory and Application, Prentice-Hall, Englewood
Cliffs, N.J., 1965.

Whitham, G. B.: Linear and Nonlinear Waves, John Wiley & Sons, New York, 1974.



Part 111

Viscous Flows of
Incompressible Fluids

In this section, problems will be solved and phenomena will be established
in which the viscosity of the fluid is intrinsically important. The treat-
ment is divided into four chapters. Chapter 7 covers the exact solutions to
the Navier-Stokes equations. Although these solutions are relatively few
in number, they are cherished. They are used as the basis for perturbation
schemes to solve problems that are close to the exact solution configurations,
they are used to test the accuracy of numerical techniques, and they are used
to calibrate instruments.

Chapter 8 deals with approximate solutions to the Navier-Stokes equa-
tions that are valid for small Reynolds numbers. This is achieved by reduc-
ing the Navier-Stokes equations through the so-called Stokes approximation
and by studying the solutions to the resulting equations. Such solutions are
valuable in their own right, and they have physical counterparts. In addition,
they form the basis of approximate solutions to other problems.

Chapter 9 deals with large-Reynolds-number flows. Specifically, the
Prandtl boundary-layer approximation to the Navier—Stokes equations is
examined. Some exact solutions to these equations are first obtained through
similarity methods. The Kdrméan-Pohlhausen method is then covered as an
example of an approximate solution to the boundary-layer equations. The
stability of boundary layers is also introduced.

The final chapter in this part of the book deals with buoyancy-driven flows.
The Boussinesq approximation to the Navier-Stokes and thermal energy
equations is introduced in Chapter 10. Solutions to the resulting equations
are presented for vertical isothermal surfaces, a line source of heat, and a
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point source of heat. The stability of horizontal fluid layers is also discussed
with a view to establishing the condition for the onset of thermal convection.

The governing equations for this part of the book are the continuity equa-
tion and the Navier-Stokes equations. Thus, from Equations 1.3c and 1.9b,
the vector form of the governing equations is

V-u=0 (ITL1)

1 2
— . =-= f. II1.2
o +(u- Hu 0 p+v “u+ ( )

In this last equation, we have introduced the quantity v = p/p, which is the
kinematic viscosity of the fluid. As was the case in the previous parts of the
book, these equations form a complete set for the unknown quantities p and
u. This is due to the assumption of incompressibility, which has the math-
ematical consequence of uncoupling the equations of dynamics from those
of thermodynamics.

The boundary condition that is to be imposed on the velocity vector u is
the no-slip boundary condition, which is given by Equation 1.14. This bound-
ary condition is

u = U on solid boundaries (I11.3)

where u is the fluid velocity, and U is the velocity of the solid that forms the
boundary with the fluid. This condition states that the fluid adjacent to a
solid boundary adheres to that boundary and does not slip on the boundary.
This boundary condition is much stronger than that which was used in the
study of ideal fluids. The essential difference is that the inclusion of the vis-
cous terms in Equation II1.2 has raised the order of the governing partial dif-
ferential equation by one. Thus, the true physical boundary condition may
be accommodated in this part, whereas it could not be satisfied completely in
the previous part of the book.

It was observed in Chapter 3 that the equations of momentum conser-
vation, the Navier-Stokes equations in this instance, may be alternatively
phrased in terms of the vorticity . Although we will not solve problems
here from the vorticity formulation, it is sometimes of interest to examine
solutions from the point of view of the distribution of vorticity. For such
cases, Equation 3.4a shows that the vorticity equation is

(Z—(:)+(u- Jo=(® Ju+v 2w+ xf (TI1.4)
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in which the possibility of nonconservative body forces has been included
for generality.

The solutions to the foregoing equations of viscous flow that will be estab-
lished in this part of the book will all correspond to laminar flow. Viscous
flows may be divided into two principal categories: laminar flows and tur-
bulent flows. The phenomena and treatment associated with turbulent flows
are somewhat different from the other fundamental aspects of fluid flow,
and it is usually treated separately in specialized books. This procedure will
be adopted here, so that only laminar flows will be considered.






7

Exact Solutions

In this chapter, some exact solutions to the equations governing the motion
of an incompressible, viscous fluid will be established. It is perhaps because
so few exact solutions have been found that they are so important. The basic
difficulty in obtaining exact solutions to viscous-flow problems lies in the
existence of the nonlinear convection terms in Equation III.2. Furthermore,
these nonlinear terms cannot be circumvented in this instance in the man-
ner used in the study of ideal fluids. This, in turn, is due to the inapplicability
of Kelvin’s theorem due to the existence of viscosity, and viscous flows are
not potential. In addition, the Bernoulli equations do not apply.

The exact solutions may be divided into two broad categories. In one of
these categories, the nonlinear term (u - V)u is identically zero owing to the
simple nature of the flow field. Examples of this situation that are covered
in this chapter are Couette flow, Poiseuille flow, the flow between rotating
cylinders, Stokes’ problems, and pulsating flow between parallel surfaces.

The second broad category of exact solutions is that for which the non-
linear convective terms are not identically zero. Examples presented here
include stagnation-point flow, the flow in convergent and divergent chan-
nels, and the flow over a porous wall.

7.1 Couette Flow

One of the simplest viscous-flow fields is that for flow between two parallel
surfaces. Figure 7.1a shows two parallel surfaces whose size in the z direction
is supposed to be very large compared with their separation distance /. The
flow between these plates is taken to be in the x direction, and since there is
no flow in the y direction, the pressure will be a function of x only. That is,
since there are no inertia, viscous, or external forces in the y direction, there
can be no pressure gradient in that direction. Using the fact that u = u(y) only
and v = w =0, together with the fact that p = p(x) only, Equation IIL.2 becomes,
for a force-free fluid field,

_dp, du
dx dyz'

257
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FIGURE 7.1
(a) Flow between parallel surfaces, (b) plane Couette flow, and (c) general Couette flow.

Here, the continuity equation is identically satisfied, and the nonlinear
convection terms are identically zero by virtue of the simplicity of the flow
field. The Navier—Stokes equations reduce to the preceding ordinary differ-
ential equation, which states that there is a balance between the pressure
force in the fluid and the viscous-shear force at all points in the fluid. Since
dp/dx is a function of x only, this equation may be integrated twice with
respect to y to give

2
u(y)=l% %+Ay+B

where A and B are constants of integration. The boundary condition u(0) =
0 requires that B = 0, whereas the condition u(h) = 0 requires that A = -h/2.
Thus, the velocity profile will be given by the equation

1d

u(y) = - ?apy(h -y

It is usual to introduce a dimensionless pressure parameter, which is
defined as follows:

W dp
2 Udx’




Exact Solutions 259

Here, U is any characteristic velocity such as the mean-flow velocity. In
terms of this pressure parameter, the expression for the velocity profile
between the parallel plates becomes

uy) _ _
U =P+ 1 (71a)

=<
=<

Equation 7.1a shows that the fluid flows in the direction of the negative
pressure gradient and that the velocity profile across the flow field is para-
bolic. The maximum velocity therefore occurs at the centerline between the
two plates (i.e, at y = 1/2), and the magnitude of the maximum velocity is
PU/4. For this type of flow, the pressure gradient is the driving mechanism,
so if there is no external pressure gradient, there will be no flow.

Another way of inducing a flow between two parallel surfaces, apart from
applying a pressure gradient, is to move one of the two surfaces. Figure 7.1b
depicts such a situation, which is referred to as plane Couette flow. The surface
y = 01is held fixed while the surface y = I is moving in the x direction with
constant velocity U. As before, the only nonzero velocity component will be
u, and it will be a function of y only. Also, there will be no pressure gradient
in the y direction, as before, and here, it is assumed that there is no external
pressure gradient in the x direction. Then, the governing equations reduce
to the same equation as before but without the pressure term. That is, the
velocity must satisfy the equation

0- &1,
dy?
Integrating this equation gives
u(y) = Ay +B

where A and B are constants of integration. The boundary condition #(0) =0
requires that B = 0, while the condition u(h) = U requires that A = U/h. Thus,
the velocity profile for plane Couette flow is

u(y)

u

(7.1b)

=<

This result shows that the velocity profile induced in a fluid by moving one
of the boundaries at constant velocity is linear across the gap between the
two boundaries. This is consistent with Newton’s law of viscosity in which
it was assumed that the velocity profile in one dimension was linear across
the fluid with the constant of proportionality defining the dynamic viscosity.

A more general situation is one in which either of the two surfaces is mov-
ing at constant velocity and there is also an external pressure gradient. Such
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a situation is referred to as general Couette flow. The velocity profile for gen-
eral Couette flow may be obtained by superimposing Equations 7.1a and 7.1b
since the governing equations that led to these results are linear. Thus, it
follows that

uy

u

-

=<

+

~
SRS

—_

|

=<

(71c)

The velocity profiles corresponding to this equation are shown in Figure
71c. It will be seen from Figure 7.1 that for P = 0, plane Couette flow is recov-
ered, while for P # 0, the pressure gradient will either assist or resist the
viscous-shear motion. For P > 0 (i.e, for dp/dx < 0), the pressure gradient
will assist the motion that is induced by viscous shear to overcome the shear
force at the lower surface. For P <0 (i.e,, for dp/dx > 0), the pressure gradient
will resist the motion induced by the motion of the upper surface. In this
case, a region of reverse flow may occur near the lower surface, as shown in
Figure 7.1c.

7.2 Poiseuille Flow

The steady flow of a viscous fluid in a conduit of arbitrary but constant cross
section is referred to as Poiseuille flow. Figure 7.2a shows an arbitrary cross
section in the yz plane with a steady flow in the x direction. Here again, the
transverse velocity components v and w will be zero, while u will be a func-
tion of y and z only. The pressure cannot vary in the transverse directions
since there is no motion or forces in these directions; hence, p will be a func-
tion of x only. Under these conditions, the governing equations, Equations
IIL.1 and II1.2, reduce to

2 2
0o 9, du ou
dx ay*  9z°

Again, owing to the simple geometry of the flow field, the nonlinear term
(u - V)u is identically zero and the continuity equation is identically satisfied
for any velocity distribution u(y, z). The remaining equation that has to be
satisfied is a Poisson type of equation. In standard form, this equation is

2 2
du o _1dp (7.22)
ay* oz’ dx
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= } z

FIGURE 7.2
Viscous flow along conduits of various cross sections: (a) arbitrary, (b) circular, and (c) elliptic.

where the nonhomogeneous term must be a constant at most. There is no
general solution to Equation 7.2a for arbitrary cross sections, but solutions for
a few specific cross sections do exist.

Consider, first, the special case in which the cross section in the yz plane
is circular with radius a, as shown in Figure 7.2b. With this geometry, the
preferred coordinate system is cylindrical coordinates. Then, let the cross
section of the conduit be represented by the cylindrical coordinates R and 0
rather than the Cartesian coordinates i and z, so that the independent coor-
dinates are now R, 6, and x. In this coordinate system, the axial velocity u
will be independent of 0 and x, so that Equation 7.2a will become

RdR ~dR  dx’

1d du _1dp

Since the pressure gradient is independent of R, this equation may be inte-
grated twice with respect to R to give

2
u(R) = l%RT+AlogR+B
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where A and B are constants of integration. The condition u(0) = finite
requires that A = 0, while the condition u(a) = 0 requires that B = —(dp/dx)a?/
(4p). Thus, the velocity profile in the conduit will be of the form

u(R) = - 4i % (a* - R?). (7.2b)

This result is similar to that for the flow between two parallel surfaces. The
flow depends upon the external pressure for its existence, and the resulting
velocity profile is parabolic.

For elliptic cross sections, as shown in Figure 7.2c, a proper procedure
would be to express the Laplacian that appears in Equation 7.2a in ellip-
tic coordinates and proceed as above as with the circular cross section.
However, a simpler and more direct method of solution exists and will be
followed here. The basis of this method is the observation that, for the ellipse
shown in Figure 7.2¢, the quantity (y?/a? + z2/b? - 1) is zero on the boundary.
This motivates us to look for a solution that is proportional to this quantity,
so a solution to Equation 7.2a is sought in the form

2 ZZ
wy,z)=a Z—2+b—2—1 .

Direct substitution shows that this is indeed a solution to Equation 7.2a,
provided the value of a is

Thus, the velocity profile for an elliptic conduit is

1dp a*v* y* 7°
u(y,z =275a2+b2 aT+b72_1 . (72C)

7.3 Flow between Rotating Cylinders

An exact solution to the Navier-Stokes equations exists for the case of a fluid
contained between two concentric circular cylinders either of which (if not
both) is rotating at constant speed about its axis. The cylinders are assumed
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to be long compared with their diameter, so that the flow field will be two-
dimensional. Figure 7.3 shows the geometry under consideration. The outer
cylinder has a radius R, and it is rotating in the clockwise direction with
angular velocity o,, whereas the radius of the inner cylinder is R; and its
angular velocity is ;.

Cylindrical coordinates are preferred for the geometry shown, and the
only nonzero velocity component in this coordinate system will be the tan-
gential velocity u,. Furthermore, this velocity component will depend upon
R only. For this type of velocity field, and in the absence of any external body
forces, the pressure can depend upon R only. Using these observations, the
governing equations (Equations IIL.1 and III.2) become

w, __1dp

R p dR

d’u, d oy,
= +— 9
drR?> dR R

Because of the simple geometry of the flow field, the continuity equation
is identically satisfied. The first of the equations above shows that there is a
balance between the centrifugal force that acts on an element of fluid and a
force that is produced by the induced pressure field. The second equation
above states that there is a balance between the viscous stresses in the fluid.

The foregoing equations may be readily integrated by establishing u, from
the second equation and then determining the pressure p from the first equa-
tion. Integrating the second equation twice with respect to R gives

[
NEE’

N

FIGURE 7.3
Geometry for flow between concentric rotating circular cylinders.
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ueR=A§+§

where A and B are constants of integration. The boundary conditions uy(R,) =
0,R, and uy(R;) = o;R; give

A 20,R2 - ,R?)

R? - R}
B=-R’R? ‘IZ; ';’; :

Thus, the velocity distribution in the fluid between the two cylinders will
be

2p2
L (@R -0 R)R-(0, ~w) il

uy(R) = R§ - Riz R

(7.3a)

Using Equation 7.3a and the remaining equation that is to be satisfied, it
follows that the pressure p must satisfy the equation

d_
dR ” (R: - R

2p2 44
(0.8 =0 R R=2(0, = 0,)0,RE ~0,R) T + (0, L5

Integrating this equation shows that the pressure distribution will be

PR =P (@R —oRP R 2, —w)
(Rg _ RIZ)Z o Yo it ) o i
(7.3b)
RIR?
x (0, R2 — w,R})R?R? logR - (o, — ;) 721R20 +C

where C is a constant of integration that may be evaluated in any particular
problem by specifying the value of the pressure on R = R, or on R = R;.

As special cases, Equations 7.3a and 7.3b describe the flow field due to a
single cylinder rotating in a fluid of infinite extent and a cylinder filled with
fluid that is rotating. Some aspects of these special cases will be investigated
in the problems at the end of the chapter.
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7.4 Stokes’ First Problem

The fluid-mechanics problem referred to as Stokes’ first problem has counter-
parts in many branches of engineering and physics. In the fluid-mechanics
context, the situation that is being considered is shown in Figure 74a. The
x axis coincides with an infinitely long flat plate above which a fluid exists.
Initially, both the plate and the fluid are at rest. Suddenly, the plate is jerked
into motion in its own plane with a constant velocity. Under these conditions,
what will be the response of the fluid to this motion on the boundary?

To answer this question, we appropriately reduce the equations of motion
and obtain a solution to them. Since the motion of the boundary is in the x
direction, it may be reasonably assumed that the motion of the fluid will
also be in that direction. Thus, the only nonzero velocity component will be
u, and this velocity component will be a function of y and t only. Then, the
pressure will be independent of y, and since u is independent of x, so will
p be independent of x. That is, the pressure will be constant everywhere in
the fluid. Using these properties of the flow field, the governing equations
reduce to the following linear partial differential equation:

x Ofort<0 * 0 t
u (0, t) =

Ufort>0

o

FIGURE 7.4
(a) Definition sketch for Stokes’ first problem and (b) solution curves in terms of the similarity
variable and in terms of dimensional variables.



266 Fundamental Mechanics of Fluids

The boundary conditions are

0 fort=<0
u(0,t) =

u fort>0
u(y, t) = finite.

This problem lends itself to solution by Laplace transforms and by similar-
ity methods. Since similarity solutions are the only ones that exist for some
nonlinear problems arising in boundary-layer theory and other situations,
this method of solution will be employed here to establish a base for future
considerations.

Similarity solutions are a special class of solutions that exist for problems
governed by parabolic partial differential equations where there is no geo-
metric length scale in the problem. Stokes’ first problem meets these require-
ments. It may be observed that had there been a second plate at some plane
y = h, the geometric length scale i1 would exist, and so the conditions for a
similarity solution would no longer exist. In the absence of such a length
scale, however, it may be anticipated that the fluid velocity u will reach some
specified value, say 0.4l at different values of y, which will depend upon the
value of the time f. That is, anticipating the results to be of the form indicated
in Figure 7.4b, it may be observed that at some time f,, the velocity will have
a value of 04U at some distance y, from the plate. At some later time t,, the
same velocity magnitude of 0.4U will exist at some different distance y,, and
so on. This suggests that there will be some combination of y and ¢, such as
y/t", such that when this quantity is constant, the velocity will be constant.
That is, it is expected that a solution will exist in the form

u(y,t) _
T f)
where

n= atln.

Here n(y, t) is called the similarity variable, and o is a constant of propor-
tionality that will be defined later to render n dimensionless. This assumed
form of solution has the property that when n is constant (which corresponds
to y ~ "), u = constant. If indeed a similarity solution exists to our problem,
substitution of our assumed form of solution into the governing partial dif-
ferential equation will result in an ordinary differential equation with f as
the dependent variable and y as the independent variable. That is, it will be
possible to eliminate i and ¢ in terms of 1 only.
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From the assumed form of solution, the following expressions for the
derivatives are obtained:

ou ay n
M U - un™
ot g f t f
ou o
oy U f

y t
a’u o’
—=U .
ayz t2n f

Here, the primes denote differentiation of f with respect to n. Substitution
of these expressions into the governing equation yields the identity

2
—Un?f =vll;nf.

This is not an ordinary differential equation for arbitrary values of n, but
for n = %, the explicit time dependence will be eliminated, yielding an ordi-

nary differential equation. That is, for n = %, a similarity solution is obtained.

For this value of n, the differential equation for f and the definition of the
similarity variable are as follows:

faolyf =0

2va

y
T]=(X.t17.

The quantity «a may now be determined in terms of the parameter v (and
U if necessary) to render n dimensionless. The dimensions of y/t'/2 are a
length divided by the square root of time. Since the dimensions of v are a
length squared divided by time, it is sufficient to take « equal to 1/vv. For
convenience, in the solution of the differential equation, a factor of 2 is also
included, so that the similarity variable becomes

Y
2+/ut

n=

and the differential equation to be solved becomes

f+2nf =0,
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This equation may be integrated successively by rewriting it as follows:

d
Tn(Ing )=-27

and hence
log f'=-n*+1log A
where the constant of integration has been chosen as log A. Then, combining

the two logarithmic quantities and taking exponentials of both sides of the
resulting equation give

f = Ae
A f)=A f“ e dE+ B

where B is another constant of integration, and & is a dummy variable of
integration. The boundary condition u(0, f) = U for t > 0 requires that f(0) =
1. This, in turn, requires that B = 1. The initial condition u(y, 0) = 0 for y > 0
requires that f() — 0 when n — oo. This gives

-

0=U Af e¥deE+1 =U A7”+1
0

so that the value of the constant A is ~2/+/x. Then, using the definition of the
similarity variable, the expression for the velocity becomes

But the second term on the right-hand side of this equation is the error
function whose argument is the upper limit of integration. Thus, the solution
to Stokes’ first problem may be written in the following form:

u(yl t) = 1 _ erf

y
u N

Values of the error function are presented in many tables of functions.
Figure 7.4b shows the functional form of the error function and the dimen-
sional velocity profiles generated by this single similarity curve.

(74)
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As one might expect intuitively, the disturbance caused by the impulsive
motion of the boundary diffuses into the fluid as the time from the initia-
tion of the motion progresses. An estimate of the depth of fluid affected by
the movement of the boundary may be obtained by observing from detailed
plots of the error function that u/U is reduced to about 0.04 when n = 3/2.
That is, for values of 1 greater than 3/2, the motion of the fluid is small, and
the fluid may be considered to be unaffected by the moving boundary. Then,
denoting the value of y by & at which u/U is 0.04 shows that

n=3/2=

5
2ot

That is, the thickness of the fluid layer affected by the motion of the bound-
ary is proportional to the square root of the time and to the square root of
the kinematic viscosity of the fluid. This result shows the role played by the
kinematic viscosity in the diffusion of momentum through fluids.

7.5 Stokes’ Second Problem

Another problem to which an exact solution exists is geometrically identical
to the one treated in the previous section, but the principal boundary condi-
tion is different. Stokes’ second problem differs from Stokes’ first problem
only in the condition that the boundary y = 0 is oscillating in time rather
than impulsively starting into motion. The geometry of the flow and the
nature of the boundary condition are indicated in Figure 7.5a.

Since the geometry is the same as that of the previous section and since the
motion is again in the plane of the boundary itself, the differential equation
to be satisfied by u(y, t) will be the same. That is, the problem to be solved
becomes

ou  9u

o _ Lo

at  ay?
u(0,t) = U cosnt
u(y, t) = finite.

Since the boundary y = 0 is oscillating in time, it is to be expected that the
fluid will also oscillate in the x direction in time with the same frequency.
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% X t
u(0,t)= U cos nt

AT\

0 u u(y,t)

FIGURE 7.5
(a) Definition sketch for Stokes” second problem and (b) typical velocity profiles.

However, it is to be expected that the amplitude of the motion and the phase
shift relative to the motion of the boundary will depend upon y. Thus, a
steady-state solution is sought of the form

u(y, t) = Re[w(y)e™]

where the symbol Re signifies the real part of the quantity that is inside the
brackets. Substituting this assumed form of solution into the partial differ-
ential equation for u(y, t) gives

2
Re[inw(y)e™] = v Re j—?e'm
2
" dw i"w=o0.
dy* v

Noting that Vi = +(1 + i)/y2, the solution to this differential equation is

w(y) = Aexp —(1+1) %y +Bexp (1+1) /%y .

The condition that the velocity be finite requires that the constant B should
be zero. Thus, the solution for w(y) will be of the form
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w(y) = Aexp —,/%y exp —z}/%y .

The expression for the velocity then becomes

/n . /n
=Re A - =
u(y,t)=Re Aexp oY eXpi nt oY
[ n [ n
=A - t—\| v .
exp gy 08 oY

The constant A may be evaluated by imposing the boundary condition
u(0, t) = U cos nt, which requires A = U. Thus, the velocity distribution in
Stokes’ second problem will be given by the following equation:

uy,t) _ o | |
U = exp zvy cos nt zvy . (7.5)

Equation 7.5 describes a velocity that is oscillating in time with the same
frequency as the boundary y = 0. The amplitude has its maximum value at
y =0, and it decreases exponentially as y increases. Also, Equation 7.5 shows
that there is a phase shift in the motion of the fluid and that this phase shift
is proportional to y and to the square root of n. The type of velocity profile
that Equation 7.5 represents is illustrated in Figure 7.5b.

A measure of the distance away from the moving boundary within which
the fluid is influenced by the motion of the boundary may be obtained
as follows. The amplitude of the motion at any plane y = constant may
be obtained by letting the trigonometric term in Equation 7.5 assume its
maximum value of unity. Then, if the value of y at which the amplitude of
the motion is 1/e? of its maximum value U is denoted by §, it follows from
Equation 7.5 that

Hence,
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The quantity & is a distance such that for y > §, the fluid may be considered
to be essentially unaffected by the motion of the boundary. Again, it is seen
that viscous effects extend over a distance that is proportional to yv. It is also
observed that § varies inversely as the square root of the frequency of the
motion. That is, the faster the motion becomes, the smaller the distance will
be over which the adjacent fluid will be influenced.

7.6 Pulsating Flow between Parallel Surfaces

Another type of unsteady-flow situation for which an exact solution exists
is that of an oscillating pressure in a fluid layer that is bounded by two par-
allel planes. We consider the two parallel surfaces to be located at y = +a
and consider the pressure gradient in the x direction to oscillate in time.
Then, the velocity will be in the x direction only and will also oscillate in
time. That is, the only nonzero velocity component will be u(y, t). Using these
features of the flow, the governing equations reduce to the following single
equation:

w__1p, u
2

ot p ox ay

where u(a, t) = u(-a, t) = 0. The pressure gradient is assumed to oscillate in
time so that dp/dx will be taken to be of the following form

ap

—— =P, cosnt
ax

where P, is a constant that represents the magnitude of the pressure-gradient
oscillations.

This problem may be treated in the same manner as that of the previous
section. That is, by virtue of the oscillatory nature of the pressure gradient, it
may be expected that the velocity of the fluid will also oscillate in time, and
with the same frequency, but possibly with a phase lag relative to the oscil-
lations in the pressure. Thus, the pressure gradient and the velocity may be
represented as follows:

P _ Re(P.e™)
ox

u(y, t) = Re[w(y)e™].
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Substituting these expressions into the governing equation gives
2

Re(inwe™) = - 1 Re(P.e™)+vRe j—zf em .
Y

Thus, the quantity w(y) must satisfy the following nonhomogeneous ordi-
nary differential equation:

X

d*w in P,

dy* v pv’

The general solution to this differential equation consists of a constant par-
ticular integral plus the general solution to the homogeneous equation. This
gives

w(y)=i§1’;+Acosh (1+1) %y +Bsinh (1+1) f%y

where the quantity (1 + i)/y2 has been used for Vi, and the hyperbolic form of
solution has been chosen rather than the exponential form due to the finite
extent of the flow field in the y direction. The boundary conditions u(a, t) = 0
and u(-a, t) = 0 give, respectively,

0= i£+Acosh (1+1) 4 4 Bsinh (1+9) My
pn 2v 2v
. P, L n . .| n
0=i—*+Acosh (1+i),/—a —-Bsinh (1+i),/—a .
pn 2v 2v
The solution to this pair of algebraic equations for the undetermined con-

stants A and B is

iP,

X

A= B
pricosh[(1+1)y/(n/2v)a]

B=0.

Thus, the solution for w(y) is

w(y) = ii 1- cosh[(1 + i)my]

P cosh[(1+i)/(n/2v)a]
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Then, the expression for the velocity in the fluid becomes

P, _ coshl(1+i}/(n/2v)y]

pn " cosh[(1+ inf(n/2v)a]

u(y,t)=Re i e . (7.6)

This expression may be decomposed to yield the real part explicitly.
Although the concepts are straightforward, the details are cumbersome;
hence, the compact form of Equation 7.6 will be considered to be the final
expression. It is evident from the result that the velocity oscillates with the
same frequency as the pressure gradient but that a phase lag, which depends
upon Y, exists. Thus, the motion of the fluid that is adjacent to the boundaries
will have a time-wise phase shift relative to the motion near the centerline
of the boundaries. The amplitude of the motion near the boundaries will
also be different from that near the centerline, and in order to satisfy the
boundary conditions, this amplitude will approach zero as the boundaries
are approached.

7.7 Stagnation-Point Flow

In all the foregoing flow situations, the geometry of the flow field was such
that the nonlinear inertia terms (u - V)u were identically zero. The flow in
the vicinity of a plane stagnation point is an example of a flow field in which
these inertia terms are not zero, yet one for which an exact solution exists.
Figure 7.6a shows the situation under consideration. A fluid stream whose
velocity vector coincides with the y axis impinges on a plane boundary that
coincides with the x axis. The boundary may be considered to be curved,
such as the surface of a circular cylinder, provided that the region under

(a) (b)
y P’

1.0

FIGURE 7.6
(a) Flow configuration for a plane stagnation point and (b) functional form of the solution.
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consideration is small in extent compared with the radius of curvature of the
surface. This problem was investigated by Hiemenz, and the flow field is fre-
quently referred to as Hiemenz flow. The basis of the solution is to modify the
potential-flow solution in such a way that the Navier—Stokes equations are
still satisfied and such that the no-slip boundary condition may be satisfied.

The potential-flow solution for the situation under consideration was
established in Chapter 4, and the complex potential for the flow in a sector of
angle n/n is given by Equation 4.10. Using this result and the value n = 2, the
velocity components for the potential flow are

u=2Ux
v =-2Uy.

Then, from the Bernoulli equation, the pressure distribution will be
p=po—2pU* (x> + y?)

where p, is the Bernoulli constant that corresponds to the pressure at the
stagnation point. Note that U is not an actual velocity here.

The foregoing velocity and pressure distributions satisfy the potential-flow
problem exactly, and like all potential flows, they also satisfy the equations
of motion for a viscous, incompressible fluid exactly. This may be readily
shown by observing that the difference between the potential-flow equa-
tions and the equations governing the flow of a viscous, incompressible fluid
is the presence of the viscous term vV?u in the latter. However, for potential
flows, u = V¢, so that

V2u = VA(V) = V(V2¢) = 0.

That is, the viscous-shear terms in the Navier-Stokes equations are identi-
cally zero for potential-flow fields.

Although potential-flow fields satisfy the equations of motion for a vis-
cous, incompressible fluid, they do not satisfy the no-slip boundary con-
dition. Then, for the case of stagnation-point flow, Hiemenz attempted to
modify the potential-flow field in such a way that meeting this boundary
condition would be possible. Thus, the x component of velocity is taken to be

u = 2Uxf(y)

where the prime denotes differentiation with respect to y. Then, the continu-
ity equation requires that

v u
— =—-—= =2
oy~ o If (v)
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so that the vertical component of the velocity will be of the form

v = 2Uf(y).

Defining the velocity field in this way satisfies the continuity equation for
all functions f(y), and if we stipulate that f(y) — y as y — o, the potential-flow
solution will be recovered far from the boundary.

The equations of motion have yet to be satisfied, and this will impose fur-
ther restrictions on the function f. The equations to be satisfied are

ou ou  1ap ’u  9'u
U—+0V_—=—-——+
ox  dy  pox ax* dy

v v 1ap v %o
U—+9—=-———"-+4+v 724'72
ox ay p dy ax”  dy

Substituting the expressions obtained above for u and v into these equa-
tions shows that the following pair of equations must be satisfied:

WPA(f Y -4 = -2 %P 4 otronf
p ox
1 dp
AP ff =-—-"—=2U
ff 0 3y of

The second of these equations will be used to establish the pressure distri-
bution, and this result will be used to eliminate the pressure from the first
equation. The result will be a nonlinear ordinary differential equation that
the function f(y) must satisfy.

Integrating the last equation with respect to y gives the following expres-
sion for the pressure:

p(x, y) = —2pU*(f)* — 2pUnf" + g(x)

where g(x) is some function of x that may be determined by comparison with
the potential-flow pressure distribution that should be recovered for large
values of y. Recalling that f(y) — y for large values of y shows that, for large
values of y,

p(x, v) = 2pU%y? — 2pUv + g(x)
which, by comparison with the potential-flow pressure, requires that

g(x) = po — 2pU%x* — 2pUl.
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Then, the pressure distribution in the viscous fluid will be
px, v) = po — 2pU2(f)* + 2pUn(1 - f) — 2pU>Px2,

So far, we have satisfied the continuity equation and the equation of y
momentum. From the result above, it follows that dp/dx = —4pUZ2x, so that the
equation of x momentum becomes

AUPx(f? — 4Uxff” = 4U2x + 2Uoxf".

In standard form, with the highest derivative to the left, this equation
becomes

il tF - e1=0

The boundary condition u(x, 0) = 0 requires that f(0) = 0, while the condi-
tion v(x, 0) = 0 requires that f(0) = 0. In addition, the condition that the poten-
tial-flow solution be recovered as y — « requires that f(y) — y, or that f(y) —
1, as y — . Thus, the boundary conditions that accompany the foregoing
ordinary differential equation are

f0)=f0)=0
fy) = lasy — oo
That is, the potential-flow solution may be modified to satisfy not only
the governing equations but also the viscous boundary conditions provided
that the modifying function f(y) satisfies the foregoing conditions. Clearly, it
would be preferable to solve a problem that is free of the parameter v/(2U), for
then the result will be valid for all kinematic viscosities and all flow veloci-

ties. It is possible to render the foregoing problem free from parameters by
making the following change of variables. Let

o =L £
vV

and

Then, in terms of ¢(n), the problem to be solved in order to satisfy all the
requirements is

¢+ dd - @y +1=0 (7.72)
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$(0) =$10)=0 (7.7b)
¢m) — Lasn— o (7.70)

where the primes denote differentiation with respect to n. This nonlinear
problem must be solved numerically, but this is a much easier task than solv-
ing the original system of partial differential equations numerically. For this
reason, the solution is usually considered to be exact.

To summarize, the velocity and pressure fields in stagnation-point flow
are given by

u(x, y) = 2Ux¢’ (7.8a)
u(x,y) = -v2Uv (7.8b)
p(x, y) = po — pUod? + 2pUo(1 — ¢) — 2pLiPx* (7.8¢)

where ¢(n) is the solution to Equations 7.7a through 7.7c and

2u
=y v
v

The nature of this solution is shown in Figure 7.6b in the form of a curve of
¢’ as a function of n. From quantitative plots of this type, it is found that the
value of n for which ¢"=0.99 is about 2.4.

From this qualitative figure and the supplementary quantitative data, it is
evident that ¢’ may be considered to be unity (and hence the potential-flow
solution is recovered) when n = 2.4 approximately. Then, if 6 denotes the
value of y at this edge of the viscous layer, it follows that

1224

v
§=24,]-2 .
\ou

That is, viscous effects are confined to a layer adjacent to the bound-
ary, whose thickness varies as the square root of the kinematic viscos-
ity of the fluid and inversely as the square root of the velocity-magnitude
parameter.

and hence
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7.8 Flow in Convergent and Divergent Channels

A flow field in which the continuity equation is not identically satisfied and
in which the nonlinear inertia terms are not identically zero is that of flow
in a convergent or divergent channel. For such flow fields, an exact solution
to the governing equations exists in the sense of the previous section—that
is, the system of partial differential equations may be reduced to a simple
numerical problem.

Figure 7.7a shows the flow configurations for flow in a converging chan-
nel and flow in a diverging channel. The preferred coordinate system for
such configurations is cylindrical coordinates R, 6, and z. Then, of the three
velocity components, only the radial component u; will be different from
zero, and this velocity component will depend upon R and 6 only. Thus, the
continuity equation and the Navier-Stokes equations become

1 9
——(Ruyz)=0
RGR( Ug)

ug  1ap 1o poug _ug 0%y

R9R  poaR ~ ROR R R 90>

Lap, . 2 aug
PR 90 R* 96

(b) ug Ur
uc uc
L0 1.0
Rys Ry
Ryp Ry
Ryi
Rys
- b1 m+a 0 —a 0 +a 0

FIGURE 7.7
(a) Flow configuration and (b) velocity profiles for flow in convergent and divergent channels.
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A separable form of solution will be sought to these equations. That is, a
solution for the velocity will be sought in the form

p(R, 0) = fRIF(®).

Then, the continuity equation shows that Ru; must be a constant, so that
uz must be proportional to R™. Thus, the velocity distribution will be of the
form

1 (R, 0) = %F(e) (79a)

where the kinematic viscosity has been used as a proportionality factor in
order to render the function F(0) dimensionless.

Having satisfied the continuity equation, the two components of the
Navier-Stokes equations must next be satisfied. This will impose some
restrictions on the function F(0). Substitution of Equation 79a into the
reduced form of the Navier-Stokes equations shows that the following pair
of equations must be satisfied:

v’ 10 v’
—*3(1:)2 =—*l+*3
R poR R

2

__ L v

PR3 R®

where the primes denote differentiation with respect to 6. This pair of equa-
tions may be reduced to a single equation by forming the second cross deriv-
ative of p, namely, 0’p/dR 06, and thus eliminating the pressure between the
two equations. The resulting ordinary differential equation for F(6) is

F”+4F +2FF'=0.
This equation may be immediately integrated once with respect to 6 to give
F'+4F + (F) =K
where K is a constant of integration. In order to further reduce this equation,

new dependent and independent variables are introduced. Thus, let G(F) = F’
be the new dependent variable and F be the new independent variable. Then,

(oW

dé _d
dr ~ dr
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Using this result to eliminate F’, the differential equation to be satisfied
becomes

G§+4F+(F) =
dF

That is, in terms of G(F), the differential equation is reduced to first order.
However, this equation may be integrated directly to yield G as follows.
Rewriting the equation above in the form

and integrating with respect to F give
o avkr-2r-1p
2 3

where A is a constant of integration. Solving this equation for G(F) gives

dF 1
GE) = |2 AvKF-2F2 - 1p3
") =30 \/ * 3

Although this equation cannot be solved to give an explicit expression for
F in terms of 0, the result may be put in the form of an integral expression for
0 as a function of F. The expression is the following elliptic integral:

0= f +B
Jz A+ KE-28% - 3§ (7.9Db)

where £ is a dummy variable of integration, and B is a constant of integration.
Equation 7.9b represents a fairly simple numerical problem whose solution,
when coupled with Equation 7.9a, defines the velocity distribution.

The physical boundary conditions that have to be satisfied are

ug(a) = ug(-o) = 0 (divergent)
Up(m + a) = ug(r — o) = 0 (convergent).
These boundary conditions represent the no-slip condition at the walls of

the channel. In addition, since the velocity profiles will be symmetrical about
the reference axis, it follows that
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ity (R,0)=0 (divergent)
20
aaieR(R, n)=0 (convergent).

Then, from Equation 79a, the conditions the function F() must satisfy are
F(a) = F(—a) = F(0) = 0 (divergent) (79¢)
F(@ + o) = F(nr — o) = F(m) = 0 (convergent). (79d)

These boundary conditions are sufficient to determine the constants A,
B, and K that appear in Equation 7.9b. Equations 79a through 79d describe
velocity profiles that have the form indicated in Figure 7.7b.

In Figure 7.7b, the various curves are identified by the Reynolds number
where Ry, > Ry, > Ry;. Here, the Reynolds number is defined as

where 1, is the velocity of the fluid at the centerline of the channel. It may be
seen that the nature of the velocity profile in a convergent channel may be
quite different from that in a divergent channel, particularly at low Reynolds
numbers. The adverse pressure gradient that exists in a divergent channel
may overcome the inertia of the fluid near the wall (where viscous effects
have reduced the velocity), resulting in a reversed-flow configuration. This
separation of the flow is well established experimentally, particularly at large
values of the angle .

7.9 Flow over Porous Wall

The foregoing exact solutions to the equations of viscous flow of an incom-
pressible fluid were sufficiently simple that either the nonlinear inertia terms
dropped out, or these terms were nonzero and a reduction to a nonlinear
ordinary differential equation was possible. The flow field to be studied here
is an example of a case where the nonlinear inertia terms become linearized,
but are nonzero, and a closed form of solution becomes possible.

Figure 7.8 shows a flat surface over which a steady uniform flow exists.
Rather than being impervious, the surface is porous and fluid is being
drawn off into the porous surface such that the normal component of
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R RIRR
Seteistetototeteteletetedetets

FIGURE 7.8
Uniform flow over a plane boundary with suction.

velocity at the surface is V. Porous surfaces of this type with suction
beneath them are sometimes used to prevent boundary layers from sepa-
rating (a topic that will be discussed in Chapter 9). However, it may be
stated now that boundary-layer separation on airfoil surfaces can lead to
a stalled configuration that destroys the lift generated by the airfoil. Thus,
it is natural that one of the areas of application of boundary-layer suction
has been in aeronautics.

A solution to the foregoing problem will be sought in which p is a constant
and u depends upon y only. That is, a solution to the governing equations is
being sought in which the magnitude of the suction is adjusted in such a way
that the tangential velocity component is independent of x. For this situation,
the continuity and Navier-Stokes equations become

dv _
ay
du d%u
v—=v—
dy dy
v v 0%v
U 4D — =V — 4 ——

with the boundary conditions
u@0)=0

u(x,0) = -V
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u(y) - Uasy — o

Now, the continuity equation may be integrated to show that v(x,y) is actu-
ally a constant, and the boundary condition at y = 0 shows that this constant
must be -V, That is,

v=-V.

With this information, the momentum equations reduce further to the fol-
lowing single equation:
2
_yde_ d;;'
dy dy

It may now be seen that the inertia terms are not retained in a compre-
hensive form, yet they are not zero. Rather, the intermediate case of a linear-
ized form exists in which the convection velocity is V rather than a variable,
which it would be in more general cases.

The foregoing ordinary differential equation may be integrated once with
respect to y to give

du vV
dy v v

where the constant of integration has been chosen as AV/v. The particular
solution to the remaining equation is then u = A, so that the complete solu-
tion is

u(y) = A + Be~(V/oly
where B is a constant. The boundary condition u(0) = 0 requires that B = -A,

and the condition u — U as y — « then gives A = U. Hence, the velocity dis-
tribution will be

u(y) = U(L — e=V/ow), (7.10)

Some idea of the thickness of the layer that is affected by viscosity may be
obtained by considering the value of y at which u = U(1 - 1/€5) = 0.993U to be
8. Then, from Equation 7.10, the value of & will be

§5=5".
%

That is, the distance away from the surface at which the uniform flow is
essentially recovered is proportional to the kinematic viscosity of the fluid
and inversely proportional to the suction velocity.
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It will be observed that the solution (Equation 7.10) diverges for negative
values of V (i.e., for blowing instead of suction). Some insight into the reason
for this may be obtained by studying the vorticity distribution. The equation
governing the vorticity is Equation II1.4. Here, we are dealing with steady
flow without external body forces. In addition, the flow is two-dimensional,
so that the vorticity vector will be o = (0, 0, £). Then, using the fact thatv=-V
and u = u(y), the vorticity equation becomes

2
_y9E_, dE
dy — dy’

This equation may be integrated once with respect to y to give

%
—VE—Udy.

Interpreted physically, the term on the left-hand side of this equation rep-
resents the convection of vorticity, which is toward the boundary (negative y
direction), and the convection velocity is V. The term on the right-hand side
represents the diffusion of vorticity in which the diffusion coefficient is the
kinematic viscosity of the fluid. Thus, the equation of vorticity states that
the convection rate of vorticity toward the wall due to the suction is just bal-
anced by the diffusion of vorticity away from the wall. It is this balance that
makes possible a solution of the assumed form. However, if blowing instead
of suction exists, the convection and the diffusion will both take place in the
same direction, so that a solution of the form u = u(y) only will no longer
exist.

PROBLEMS

71 To explain the manner in which Couette flow is established, find
the velocity distribution in a fluid that is bounded by two hori-
zontal parallel surfaces in which everything is quiescent for ¢t < 0
and for which the upper surface is impulsively set into horizontal
motion with constant velocity U at time ¢ = 0. (This can be done
by obtaining the solution in its asymptotic form, corresponding to
t — oo, then adding a separation of variables solution.)

7.2 To determine the manner in which plane Couette flow decays, find
the velocity distribution in a fluid that is bounded by two horizon-
tal parallel surfaces between which steady Couette flow exists for
t < 0 as described in Problem 7.1. The motion of the upper surface
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is suddenly stopped at time t = 0. Obtain an expression for the sub-
sequent velocity distribution between the two parallel surfaces.

7.3 A viscous fluid occupies the space between two stationary par-
allel horizontal surfaces defined by the lines y = -1 and y = +1.
Initially, there is no motion in the fluid, but at time ¢ = 0, a valve is
opened and the fluid begins to move in the x direction under the
pressure differential that exists between the ends of the conduit.
The problem to be solved for the subsequent fluid velocity u(y,t),
subject to the appropriate boundary conditions, is the following:

2
ou_ 2PH(t)+a—1; for-1=y =<+1.
ot ay

The velocity and the two independent variables have been made
dimensionless in this equation. The constant P is the dimension-
less value of the pressure gradient in the flow direction for ¢ > 0
(i.e., in the x direction), and H(t) is the Heaviside function (unit step
function). Obtain the solution to the foregoing problem that is valid
for all times t > 0.

74 A moving belt is inclined at an angle a to the horizontal. The lower
end of this belt is immersed in a pool of liquid, and the belt drags
some of the liquid with it as it moves upward and out of the liquid.
The liquid may be assumed to be viscous but incompressible.

(@) Using the configuration shown in the following figure, solve
the Navier—Stokes equations for the following quantities:

Gravity

(i) The velocity distribution in the liquid layer

(i) The volumetric flow rate of liquid in the x direction per
unit width

(iii) The angle a for which the volumetric flow rate is zero
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(b) Repeat part (a) of this question for the configuration shown in
the following figure.

Gravity

7.5 For Poiseuille flow through an elliptic pipe of semiaxes a and b, find
the ratio b/a that gives the maximum flow rate for a given flow area
and a given pressure gradient.

For a given pressure gradient, find the ratio of the discharge from
an elliptic pipe to that from a circular pipe that has the same flow
area. Evaluate this ratio for the specific value b/a = 4/3.

76 The figure for this problem shows a conduit whose cross section
is the shape of an equilateral triangle. For the coordinate system
shown in the figure, the equations of the three sides are

Look for a solution for the velocity distribution in this conduit of
the following form:

u(y,z)=a z+23g z+\/§y—\/7 z—\/gy—

b
3

S
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Determine the value of the constant a such that the assumed form
of solution is exact, with the value of this constant being expressed
in terms of the applied pressure gradient.

y

L.~

b Flow

PP TN

-

7.7 Itisrequired to obtain the solution for pressure-driven flow through
a duct of rectangular cross-sectional shape that is bounded by the
linesy =0,y =1,z =0, and z = a. The partial differential equation to
be solved for the velocity in the x direction, u(y,z), is

2 2
a—?+a—g—8p for 0sy<1, O<z=a.
d 0z

In this equation, the velocity u(y,z) is dimensionless, as are the
independent coordinates y and z. The parameter P is the dimen-
sionless value of the pressure gradient in the x direction, and the
parameter o is greater than unity.

(@) Show, by direct substitution, that the following expression is a
particular solution to the foregoing differential equation that
satisfies the boundary conditions on the boundaries ¥ = 0 and
y =1, and find the value of the constant A:

2

1
n@y)=-A 1-4 y--

(b) Show, by direct substitution, that the following expression is a
complementary solution to the differential equation that satisfies
the boundary conditions on the boundaries y=0,y=1,and z=0:

1 ( z)—iB Sinhﬂsinmc
2y £y " sinh nor Y

(c) Show, by direct substitution, that the following expression is a
complementary solution to the differential equation that satisfies
the boundary conditions on the boundaries y =0,y =1, and z = a:

©

sinh nw(o - z) .
u(y,z) = ZC,,S]_nh;na)smnny.
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(d) It follows from the principle of linear superposition that the fol-
lowing expression will be a solution to the differential equation:

u(y, z) = w(y, 2) + u(y, 2) + us(y, 2).

Find the values of the constants B, and C, that make this expres-
sion for u(y,z) satisfy the boundary conditions on all four boundar-
ies of the flow field.

7.8 The figure for this problem shows two parallel, vertical surfaces
and a horizontal surface. The space defined by these surfaces, 0
<y <bfor0 < z is filled with a viscous, incompressible fluid. The
horizontal surface, z = 0, is moving in the positive x direction with
constant velocity U. The other surfaces are stationary, as is the
fluid—except for the motion that is induced by the moving surface.

z

&
&

Moving
g [ surface

Derive an expression for the velocity distribution in the yz plane
if the flow is steady and if there are no body forces. Also, obtain an
expression for the volumetric flow rate of the fluid that is induced
to flow in the x direction by the moving surface.

L
y

79 The figure for this problem represents steady flow in a circular
pipe through which two Newtonian fluids of viscosities p, and p,
are simultaneously flowing in the axial direction. That is, the cross
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section of the pipe is circular in the R-0 plane, and the flow is in the
axial or z direction. The position vector is defined by x = (R,0,z), and
the velocity vector is defined by u = (uy,1,,w). The densities of the
two fluids are the same, and the applied pressure gradient, which
is a constant, is the same for both fluids. If we assume that the only
nonzero component of velocity is w and that it depends on R only,
it follows that we have the following conditions for each fluid:

w = w(R) only and up = 1,=0

d
dR

dw
where T =1z = ——.

dR

gdw 1

S

|
58

All other components of the shear stress tensor are zero.

(@) Obtain expressions for the velocity components in each of the
two fluids. That is, obtain expressions for the following velocity
components:

w,(R) valid for 0 < R < R,

w,(R) valid for R; < R < R,.

(b) From the results obtained in (a) above, evaluate the velocity
w,(0) on the axis of the pipe.

710 Two concentric circular cylinders enclose a viscous fluid. If the
inner cylinder is at rest and the outer cylinder rotates at a constant
angular velocity, calculate the torque required to rotate the outer
cylinder and that required to hold the inner cylinder at rest.

711 Using the solution for flow between concentric rotating circular
cylinders, deduce the velocity distribution created by a circular cyl-
inder that is rotating in a fluid of infinite extent that is otherwise
at rest. Compare this result with that for a line vortex of strength
I = 2nR}w; in an inviscid fluid that is at rest at infinity.

712 Obtain the velocity distribution for the modified Stokes second
problem consisting of a fluid that is contained between two infinite
parallel surfaces separated by a distance h. The upper surface is
held fixed, while the lower surface oscillates in its own plane with
velocity U cos nt.

713 The velocity profile in a fluid between two parallel surfaces created
by an oscillating pressure gradient was shown in Equation 7.6 to be
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u(y,t) = Re i P g cosh[(1+0)y(n/2vy]

pn cosh[(1 + i)4/(n/2va]

int

A Reynolds number for such a flow may be defined by the fol-
lowing quantity:

For this situation, consider the two asymptotic limits that are
defined below:

(@) For Ry < 1, it might be expected that viscous effects will domi-
nate. Expand the expression above for the velocity in this case
and obtain an explicit expression for the leading term in the
expansion. Interpret the result physically.

(b) For Ry > 1, it might be expected that viscous effects will be
small everywhere except in the vicinity of the walls. Expand
the expression for the velocity for this case, and interpret the
result that is obtained.

7.14 For potential flow due to a line vortex, the vorticity is concentrated
along the axis of the vortex. Thus, the problem to be solved for the
decay of a line vortex with time due to the viscosity of the fluid is
as follows:

®
—=v ‘o
ot

o(R,0)=0 for R>0

f o(R,t)2rRdR =T for t=0.
0

Here, o(R, t) is the vorticity, and the total circulation associated
with the vortex for any time ¢ > 0 is I. Look for a similarity solution
to this problem of the following form:

r

FR
2mut \/E ’

o(R,t) =

Thus, obtain expressions for the velocity 1,(R, t) and the pressure
p(R, t) in the fluid.
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715 The following flow field satisfies the continuity equation every-
where except at R = 0, where a singularity exists:

U = —ak
K

Uy = E

u, =2az.

Show that this flow field also satisfies the Navier-Stokes equa-
tions everywhere except at R = 0, and find the pressure distribution
in the flow field.

Modify the foregoing expressions to the following:

ug = —aR
K
=—f(R

Uy Rf()

u, = 2az.

Determine the function f (R) such that the modified expression
satisfies the governing equations for a viscous, incompressible
fluid and such that the original flow field is recovered for R — .
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Low Reynolds Number Solutions

For flow problems in which an exact solution is not known, it may be possible
to obtain an approximate solution. By an approximate solution we mean an
analytic expression that approximately satisfies the full viscous flow equa-
tions rather than a numerical approximation to these equations. In this chap-
ter, the full governing equations will be approximated for flows involving
low Reynolds numbers, and some exact solutions to the resulting equations
will be established.

The fundamental low-Reynolds-number approximation is the Stokes
approximation, and this is the first topic in the chapter. Some fundamen-
tal solutions are established and used to create more practical solutions,
similar to the procedures used in Chapter 4. In this way, the flow in the
vicinity of a rotating sphere is obtained, as is the solution for uniform
flow past a sphere. The case of uniform flow past a circular cylinder is
examined to illustrate the consequences of the Stokes approximation.
Finally, an alternative low-Reynolds-number approximation, the Oseen
approximation, is briefly discussed. A detailed study of the Oseen equa-
tions is not made, but the nature and utility of the approximation are
discussed.

8.1 Stokes Approximation

The Stokes equations are a special case of the Navier-Stokes equations cor-
responding to very slow motion of a viscous fluid. Under these conditions,
the inertia of the fluid may be neglected in comparison with the other forces
that act on it. Since the Reynolds number may be considered to be the ratio
of the inertia forces of the fluid to the viscous forces, the condition of negli-
gible inertia forces amounts to very small Reynolds numbers. The essential
feature of the Stokes approximation is that all the convective inertia compo-
nents are assumed to be small compared with the viscous forces. Then, from
Equations IIL1 and II1.2, the equations governing the Stokes approximation
for a flow field without body forces are

V-u=0 (8.1a)

293
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a—u=—l p+v ‘u (8.1b)
ot P

Equation 8.1b, which represents three scalar equations, is usually referred
to as the Stokes equation. This equation, together with the continuity equation
(Equation 8.1a), represents four scalar equations in the four unknowns u and
p. The great simplification in this approximation is that the governing equa-
tions are now linear.

The foregoing equations may be extracted from the Navier-Stokes equa-
tions in a more formal manner. Since this alternative approach must be used
when employing higher-order corrections, it will be outlined here. The first
step is to nondimensionalize the dependent and independent variables as
follows. Let

u =Uu*
pvld
p= ] P
x; =1Ix
2
L
v

Here, U is a characteristic velocity of the fluid (such as the free-stream
velocity), and [ is a characteristic length scale (such as a body dimension).
The starred quantities are the dimensionless variables where the kinematic
viscosity v has been used to nondimensionalize the pressure and the time.
Experience gained in the previous chapter indicates that the time /2/v cor-
responds to the time required for viscous diffusion to traverse the distance I.

The variables that appear above are now substituted into the Navier—-Stokes
equations to yield the equations governing the dimensionless variables. The
resulting vector equation is

vl ou* U2 .
o O W e

where the gradient and the Laplacian operators are now expressed in terms
of the dimensionless space coordinates. Multiplying this equation by /(o)
and introducing the Reynolds number Ry, = Ul/v give

a *
a‘;* + RN(u* . *)u* - _ *p* + *Zu*'
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In this form, it is evident that the Stokes equation (Equation 8.1b) may
be obtained from the Navier-Stokes equations by taking the limit Ry, — 0
while holding the coordinates fixed. This suggests that higher-order
approximations to the Stokes solution for any given problem could be
obtained by expanding the dependent variables in ascending powers of the
Reynolds number. The sequence of differential equations that would have
to be solved could then be obtained from the above form of the Navier—
Stokes equations by a limiting procedure. Thus, the Stokes equations may
be considered to be an asymptotic limit of the Navier—Stokes equations cor-
responding to zero Reynolds number, while the space coordinates remain
of order unity.

An alternative form of Equations 8.1a and 8.1b exists that is frequently use-
ful. In this alternative form, the velocity and pressure equations are sepa-
rated so that the velocity and pressure fields may be established separately.
To obtain the equation governing the velocity field, the curl of the curl of
Equation 8.1b is taken. Having done this, the identities V x (V x u) = V(V - u) -
VZu and V x Vp = 0 are employed. The resulting equation is

SLCw=- Pul=v 2w Cul

Finally, using the continuity equation, the pressure-free form of the momen-
tum equation becomes

20U _ 4y (8.2a)
ot

To obtain the equation governing the pressure, the divergence of Equation
8.1b is taken. This gives

V2p =0. (8.2b)

The advantage in the formulation above is that the pressure field has been
separated, mathematically, from the velocity field. However, the price we
pay is that the highest differentials are now fourth order instead of second
order.

Solutions to the Stokes equations may be obtained in either of two basic
ways. Using the governing equations and the appropriate boundary con-
ditions, the boundary-value problem for each geometry of interest may be
solved. Alternatively, basic solutions to the governing equations may be
established and superimposed to obtain other solutions. This is the proce-
dure that was used in Chapters 4 and 5, and it will be used again here. The
principal value of this approach is that it leads to the clear understanding
of which elements of a mathematical solution are responsible for producing
forces and torques.
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8.2 Uniform Flow

The simplest solution to the Stokes equations is that for a uniform flow. It
may be simply observed that for a constant velocity vector and a constant
pressure, Equations 8.1a and 8.2b are identically satisfied. That is, for any
constant U, the following velocity and pressure fields satisty the Stokes
approximation to the Navier—Stokes equations:

u=Ue, (8.3a)

p = constant (8.3b)

where e, is the unit vector in the x direction, which is the reference direc-
tion. Clearly, this velocity and pressure distribution does not create a force
or turning moment on the system. Some of the other fundamental solutions
that will be considered later correspond to point singularities, and some of
these singularities correspond to point forces or turning moments acting on
the fluid.

8.3 Doublet

It was pointed out in Section 7.7 that any potential flow is an exact solution
of the full Navier-Stokes equations since the viscous terms are identically
zero for potential flows. Then, for any steady potential flow, the Stokes equa-
tions will be satisfied provided that the pressure term is zero. That is, for
steady flow, all the inertia terms are zero to the Stokes approximation, and
for potential flows, the viscous term will be zero. Hence, potential flows are
also solutions to the Stokes equations provided that Vp = 0 or p = constant.
In the study of the flow of ideal fluids, it could be shown, through Kelvin's
theorem, that a flow that is initially irrotational would remain irrotational
regardless of the shape of any bodies it may flow around. Here, we consider
the nature of the velocity field assumed by an irrotational motion if it exists.
The solution to any real flow problem may contain such a component in its
solution in addition to other fundamental solutions, some of which may cor-
respond to rotational motion. It should be noted that the velocity field corre-
sponding to viscous irrotational motion is not related to the pressure field in
the manner that existed for ideal-fluid flows. The conditions that were stipu-
lated in deriving the Bernoulli equation are violated for viscous flows, so that
the pressure and the velocity are no longer connected by such a relationship.
Indeed, it was shown above that any irrotational velocity distribution had to
be accompanied by a constant pressure in order to satisfy the Stokes equations.
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If an irrotational flow field exists, the velocity will be derivable from a
velocity potential, and from the continuity equation, the velocity potential
must satisfy Laplace’s equation. Then, for three-dimensional potential flows,
the mathematical problem is the same as that of Chapter 5. That is, if we are
interested in axisymmetric flow fields, we may use the coordinate system
defined in Figure 5.1 and the solutions to Equation 5.1 that have already been
established.

But the solution that was identified as a doublet (Equation 5.7a) was of the
functional form:

cosB_A%

o(r,0) = A

2 r

where the fact that x = r cos® has been used. Then, since u = V¢, the velocity
vector will have a component along the x axis and a radial component, giving

e, 3xe,
u=A 7’73_ A (8.4a)

where e, and e, are unit vectors in the x direction and in the radial direc-
tion, respectively. This formulation of expressing variables in terms of
streamwise and radial components will be found to be useful in the present
application.

The velocity field described by Equation 8.4a cannot be proved to be valid
from upstream irrotational conditions, and it is presented here only as a pos-
sible form for a viscous fluid. Then, in order for this flow field to satisfy the
present version of the momentum equations, the pressure distribution must be

p = constant. (8.4b)

Although the solution defined by Equations 8.4a and 8.4b represents the
tlow field generated by a singularity that is located at the origin, this singu-
larity does not exert a force or a moment on the surrounding fluid. This may
be simply argued from the fact that the pressure is constant and the flow
configuration is such that there is no net momentum flux acting on the fluid.

8.4 Rotlet

In this section, a solution to the Stokes equations will be sought in which
the vorticity is different from zero and the pressure is constant. That is, a
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rotational-flow solution will be sought. The resulting solution will involve a
singularity at the origin that is known as a rotlet.
Consider steady flow fields of the form

u=rxVy
where r is the position vector. In tensor notation, this expression becomes

a
U = €3X; X
0X;
Then, the divergence of this velocity vector will be given by

o 0% oy 0
ax, ™ oy, ax, ! axox,

The first term on the right-hand side of this equation is zero since dx;/dx;
is zero unless i = j and g;; is zero when i = j. That is, the product of the pseu-
doscalar e;; and the symmetric tensor dx;/dx; is zero. Likewise, the second
term inside the parentheses in the equation above is a symmetric tensor, so
that the product of this quantity with the pseudoscalar ¢;; will be zero. That
is, the continuity equation will be satisfied identically for all forms of the
scalar y.

Since the flow is assumed to be steady and since the pressure has been
taken to be constant, the Stokes equations reduce to

VZu = 0.

But, for the form of the velocity vector introduced above, it follows that

) o, Fx; a8 oy

" ox,0x, ik ax,ox, ax, ) ox, ax,ox,

The first term on the right-hand side of this equation is zero since
0%x;/(dx; dx;) = 0 for all values of j and /. Also, the second term on the right-
hand side will be zero if V?y = 0. That is, the velocity distribution u = r x
Vy will be a valid solution to the Stokes equations for a constant pressure
distribution, provided that V?; = 0. The problem, therefore, again reduces to
that of obtaining axisymmetric solutions to the three-dimensional Laplace
equation.

From Chapter 5, the first separable solution that was obtained corre-
sponded to a source and was of the form y ~ 1/r. This gives Vy ~ e, so that
u = 0, and hence, that particular solution is of no interest in this case. The
next solution, corresponding to a doublet, was of the form
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The velocity field corresponding to this solution will be

xe,
T

X e
u=Brx - = Br x —;‘—3
r r r

Since r = re, and since e, x e, = 0, this velocity distribution may be repre-
sented by the simplified expression

e, xe,
u=B T (85&)

while the corresponding pressure distribution is
p = constant. (8.5b)

The streamlines corresponding to Equation 8.5a must be perpendicular to
both e, and e,. That is, the streamlines are circles whose centers lie on the x
axis. A typical streamline is shown in Figure 8.1a, in which the direction is

(@)

Rotlet

FIGURE 8.1
(a) Typical streamline due to a rotlet and (b) spherical control surface surrounding a rotlet.
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shown for B > 0. It is the nature of the streamlines that suggests the name
rotlet for the singularity that exists in Equation 8.5a at ¥ = 0.

The rotlet does not exert a force on the fluid, but it does exert a turning
moment on it. This may be verified by constructing a spherical control sur-
face around the rotlet, as shown in Figure 8.1b. Then, if F is the force acting
on the fluid contained within the control surface S and if n is the unit out-
ward normal to §, it follows that

L= —foi]-nde

where c; is the stress tensor. But, for an incompressible Newtonian fluid,

Equation 1.7 shows that the stress tensor may be expressed by

ou; au]'
ik
ox;  ox;

O =—pd; +

Using this result, the order of magnitude of the force F; may be evaluated
as follows. Substituting o; into the expression for F; gives

aw;  ox

ou. ou;
F = —f -pd; + i Ty ds.
Since p = constant here, the first component of the integral above will be
zero. Also, Equation 8.5a shows that u; ~ r2, and it is known that dS ~ 72 for
an element of surface of a sphere. Hence

Fi Nirz =1'
r3 7

Then, by considering the control surface S to be of a very large radius, it is
clear that F; = 0. That is, there is no force acting on the fluid due to the rotlet.

The torque, or turning moment M, exerted on the fluid by the singularity,
may be calculated as follows:

M=fsr><PdS

where P is the surface-force vector resulting from the existence of the stress
tensor ;. In tensor notation, this expression becomes

M, = f g% P, dS.
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Using the relation P, = 6,1, and the expression for o), that was used above,
the value of the turning moment becomes

ou ou
M =[e,x, -pd,+ —F+—-L ndS.
! fs i ~POu ax, ox,

Noting that, since p = constant, the first component of this integral is zero
and using the fact that 1, = x;/r for the spherical control surface of radius 7, the
expression for the turning moment of the fluid becomes

ou ou
M, = —f euXx, — -+ dS.
rJds 0x;  0xy

The preceding expression is valid for any velocity distribution whatsoever.
In this particular case, the velocity (Equation 8.5a) is a homogeneous func-
tion of degree 2. A homogeneous function of order m is one that satisfies the
condition

f

% =\"f(x,y,2)

ryz
AN

For such functions, Euler’s theorem states that

of L
x8x+yay+zaz_ mf.

Then, for the velocity distribution under consideration, which is given by
Equation 8.5a and which is homogeneous of degree 2, it follows that

— = =2u,.
ox, k

This identifies one term that appears in the integrand for the expression
for the moment M;. Another term that appears in the integrand may be eval-
uated as follows:

) ax;
I =—(qu) -y — = ~l.
0x,  oxg 0x;,

a

Here, it is noted that the first term on the right-hand side of this identity is
zero since d(xu;)/0x, = V(r - u), and the velocity vector u is perpendicular to
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the position vector r, as may be seen from Equation 8.5a. Also, dx;/0x; = , so
that u,0x,/0x, = u ;. = u,. Then, the expression for the moment exerted on the
body of fluid by the rotlet at the origin becomes

M; = 7f € X (=21 — ) dS.
Or, in vector form, this expression is
M=-3—frxuds.
rJs

But, using Equation 8.5a,

e xe
rxu=rxB—-—=*
’

_B

B X B
=—(r-e)e -—(re)e, =B—e -—e,.
r r r

<

Thus, the expression for the moment M becomes

ds

M=—BBf Te e, &
s I r

This integral may now be evaluated explicitly using the following rela-
tions, which are obtained from Appendix A:

xX=7rcos0
e, =cosbe +sinbcoswe, +sinbsinwe,

dS =72 sin 0 dO dw.

Using these results, the expression for M becomes

27 7T
M= -3B f dw [ [(cos* 6-1)e, + sin O cos O cos we, + sin 6 cos 6 sin we, | sin 6.d6.
0 0

That is,
M = 8nBp e,. (8.5¢)

Equations 8.5a and 8.5b represent a valid solution to the Stokes equa-
tions, and they correspond to a singularity at the origin called a rotlet. This
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singularity exerts no force on the surrounding fluid, but it does exert a turn-
ing moment on it. The magnitude of this turning moment is proportional to
the velocity—magnitude parameter, and it acts, according to the right-hand-
screw rule, in the x direction in the same algebraic sense as the velocity—
magnitude parameter.

8.5 Stokeslet

So far, all our fundamental solutions to the Stokes equations have corresponded
to a constant pressure. In general situations, it is to be expected that the pres-
sure distribution will not be constant, so that another fundamental solution
will be sought, and this time, the pressure will be assumed to be different from
a constant value. Then, the pressure must be a nontrivial solution to Equation
8.2b, which is the Laplace equation. Having so determined the pressure, the
corresponding velocity distribution will be obtained from Equation 8.1b.

Since the pressure p satisfies the three-dimensional Laplace equation, the
possible fundamental solutions may be written immediately from Section
5.3. The source type of solution, in which p ~ 1/7, turns out to be of no special
interest. The next highest form of separable solution, which is the doublet
type of solution, is p ~ cos 6/r% For reasons that will become apparent shortly,
the constant of proportionality will be taken as 2cy, so that the pressure field
that is being considered is

where cos 6 = x/r has been used. Then, if the flow is assumed to be steady, the
equation to be satisfied by the velocity is, from Equation 8.1b,

Of the three scalar equations represented by this vector equation, the
equation for the velocity component u is the most complicated because of
the nature of the pressure distribution; so this equation will be dealt with
last. From the expression for p, the equation to be satisfied by the velocity
component v is
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The particular integral to this nonhomogeneous partial differential equa-
tion, which is v = cxy/13, is readily obtained if one is familiar with the proper-
ties of harmonic functions of different degrees. Alternatively, this result may
be deduced from the following identities:

2 1 _ n(n-1)
rﬂ

n+2

r
@)=y o+ *Y+2 ¢ .

Here, 7> = x2 + 2 + z> and ¢ and y are any two functions. Then if ¢ = 1/7°
and y = xy, it is readily verified that

6
20—
¢="5
3
=—-— €
0 AT
2w=0

Y =ye, +xe,

o p=-6"7
;

where the first result follows from the first of the identities above. Then,
using the second of these identities,

Gu)= * Y =677

r r

That is, the particular solution to the equation for the velocity component
vis
xy
v=c—.
r3

The equation to be satisfied by the w component of the velocity vectors is

Xz
2w = -6c .
r

The particular solution to this equation is obtained in exactly the same
way as that for v and may be deduced to be
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The equation to be satisfied by the 1 component of the velocity vector is

2 X
2M=C 73—675 .
r r

In view of the solutions for v and w, it might be expected that the solution
to this equation is u = cx?/r%. This is indeed the case, as may be confirmed by

setting ¢ = 1/r* and y = x> and employing the identities mentioned above.
Thus

2 —
=—-—e,
Zw - 2
P = 2xe,
2
X
. =6
and hence
x2 2 x2
)= ? =—-6—.
r r

Solutions to the homogeneous equations VZu = 0, V2 = 0, and V*w = 0 may
be added to each of the foregoing particular integrals. Denoting these solu-
tions by u; v, and w’, respectively, the complete solution for the velocity u
corresponding to the pressure distribution p = 2cpx/r® is

2

Y Xy x
u=c e, +—e +—>e +u=ce +u
1,3 R 7’3 Y r3 2
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where u’= (1, v, w’). The quantity u, which must satisfy the equation V?u’=0,
will now be determined such that the continuity equation is satisfied. Taking
the divergence of the velocity u shows that

x
‘u=c - 731' + -u
r
but
x x x
—r =r - + = ( 1)
r r’ r’
1 3x? 3xy  3xz x
=X G- 5 Ty -z +3
oo r r r

Thus, by choosing u’ = c(e,/7), the continuity equation will be satisfied. It
will be noted that this form of u’ also satisfies the homogeneous equation
V2u’ = 0. Thus, a valid velocity distribution has been found that satisfies the
Stokes equations corresponding to a doublet type of pressure field. In sum-
mary, this solution is

p=2c 73 (8.6a)
1
u=c %e,+;ex . (8.6b)

The solution represented by the above equations has a singularity at the
origin, and this singularity is known as a stokeslet. Although the stokeslet
does not exert a torque on the surrounding fluid, it does exert a force on it.
The magnitude of this force may be established as follows: in tensor notation,
the force F; acting on the fluid will be

i=_f0ijnjds
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where, for a Newtonian fluid,

ou; 6”]'
+—1
ox;  ox;

O =-pd; +

y

and for this particular flow field, the pressure is given by Equation 8.6a.
Hence, the expression for the force F; becomes

. 0uU;
oy, n; dS.
ox; 0x

X
E=-( -2c Lo+
]
r j i

s

Now, if the surface S is considered to be a sphere of radius 7, then n; = x]-/ T,
so that the expression for F; becomes

But in this particular case, the velocity distribution, which is given by
Equation 8.6b, is homogeneous of degree 1. Hence, from Euler’s theorem, it
follows that

A second quantity that appears in the integrand of the foregoing integral
may be evaluated as follows:

X =) U=
a i axi 0x 8xi

i

(r-u)-u

where the fact that u,dx;/dx; = u,d; = u; has been used. Hence, using Equation
8.6b,

u, 9 X xx; 9,
_l_ Lo o Tl
X; = 2c c —+
Jx;  ox; r r r
2 xx; xx; 94
=c —0,;-2 - i
r r r r
_ 6711_3xxl-
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Using these results, the expression for the force acting on the fluid due to
the stokeslet becomes

! 4 3

XX; xx; 0, 0, XX;
F=—f 2¢ Tr-c— TSh+ o4 370 dS
s r ror r ror r

- 6c fx—’j‘ds.

r

Alternatively, in vector notation, this expression becomes
X
F = 6c f e, dS.
st

This integral may be evaluated explicitly using the following relations:
X =cos0
e, =cos 0 e, +sin 0 cos we, + sin 0 sin we,

dS =72sin 0 dO dw.

Thus, the force acting on the fluid due to a stokeslet is

2

F=6c f du)f cos b(cosBe, +sin b coswe, +sinOsin we,)sin 6 d6.
0 0

That is,
F = 8ncpe,. (8.60)

Thus, a stokeslet exerts a force on the surrounding fluid with a strength
that is proportional to the pressure parameter c and whose direction is in the
positive x direction for ¢ > 0.

8.6 Rotating Sphere in Fluid

The foregoing fundamental solutions are sufficient to establish more practical
solutions for low-Reynolds-number flows. One of these solutions corresponds
to a solid sphere that is rotating in an otherwise quiescent fluid. Consider
such a sphere to be rotating about the x axis with constant angular velocity Q.
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The nature of the resulting flow field may be expected to be similar to that of
a rotlet. Hence, let the velocity distribution correspond to Equation 8.5a and
see if the boundary conditions are, or may be, satisfied. Then

e, xe,
r2

u=_8

This velocity distribution gives a finite velocity as r — « as required. The
other boundary condition is, on r = 4, u = Qae, x e,. This condition is satisfied
for B = Qa?3, so that the required velocity distribution is

aS

u= —e€

xXe
r
rZ

- (8.7a)

Since the rotlet is located at = 0 and since the surface of the sphereis r =g,
there is no singularity in the fluid.

Since the rotlet was found to exert a moment on the surrounding fluid, the
surrounding fluid will exert a moment on the surface r = a4 in this case. The
magnitude of this moment is given by Equation 8.5c. Hence, using B = Qa?,
the moment acting on the sphere will be

M = -8npuQade,. (8.7b)

This moment acts in a direction that opposes the motion of the sphere, as
might be expected.

8.7 Uniform Flow Past Sphere

The solution corresponding to uniform flow past a sphere may be obtained
by superimposing the solutions for a uniform flow, a doublet, and a stokeslet.
Hence, from Equations 8.3a, 8.3b, 8.4a, 8.4b, 8.6a, and 8.6b, the assumed forms
of the velocity and pressure fields are as follows:

e, ., xe, X e,
u=Ue, +A —-3—" +c e +—

r r r r

Far from the origin, this velocity field reduces to that of a uniform flow as
required. The simplest way of imposing the near boundary condition is to
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observe that at the rear stagnation point, u = 0. Hence, substituting x =r =a
and setting u = 0 in the foregoing expression for the velocity give

e e e, e
O=Ue, +A —-3—F +c —+—
a a a a

Setting the coefficients of e, and e, equal to zero separately yields the fol-
lowing pair of equations:

0=LI+£+£
a’ a
0-34,¢
a® a

The solution to these algebraic equations is

A=-u®

4

c=—§Ua.
4

la a* 3ax a*
u=U ex—Z? rT+3 ex+2r72 ?—1 e, (88a)
3 ax
- ™ _
p=-yull’ (:5)

In this form, it is readily confirmed that Equation 8.8a satisfies the bound-
ary condition u = 0 over the entire surface r = a.

Neither the uniform flow nor the doublet exerts a force on the fluid.
However, the stokeslet exerts a force on the surrounding fluid, and the mag-
nitude of this force is given by Equation 8.6c. Then, since the stokeslet is
inside the spherical surface r = a, the surrounding fluid will exert an equal
but opposite force on the sphere. Thus, from Equation 8.6c and using the fact
that ¢ = — 3Ua/4, the magnitude of the force acting on the sphere will be

F = 6rulae,. (8.8¢0)
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This is the famous Stokes” drag law for a sphere in a uniform flow, and
it is valid for low Reynolds numbers. Since the direction of this force
is clearly in the direction of the uniform flow, this result is frequently
quoted in terms of the dimensionless drag coefficients, which involves
the scalar magnitude of the force only. This drag coefficient is defined as
follows:

where A = na? is the frontal area of the sphere. Thus, using Equation 8.8c

24

Cp=—-.
D Ry,

(8.8d)

Here, Ry = pU2a/p is the Reynolds number of the flow. This result is
shown in Figure 8.2, which shows the form of the drag coefficient as a func-
tion of the Reynolds number for a sphere over a large range of Reynolds
numbers. Of the entire range of Reynolds numbers, Equation 8.8d is the
only closed-form analytic solution that exists. Strictly speaking, Equation
8.8d is valid only for Reynolds numbers that are small compared to unity,
but it is found experimentally that the result is valid for Reynolds numbers
that are less than unity, as well as for Reynolds numbers that are small
compared to unity.

\

\\ c _%
\/_ b= Ry

Ry

FIGURE 8.2
Drag coefficient as a function of Reynolds number for flow around a sphere.
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8.8 Uniform Flow Past Circular Cylinder

It will be shown in this section that the solution obtained above for a sphere
in three dimensions has no counterpart in two dimensions. This will be
shown by attempting to solve the Stokes equations for uniform flow past a
circular cylinder. Since the fundamental solutions have not been established
for two dimensions, the alternative approach of solving the boundary-value
problem for the stream function will be adopted.

For steady flow, the Stokes equation is given by

0=- 1 p+v ‘u
p
Taking the curl of this equation gives the following equation for the vortic-
ity to the Stokes approximation:
Vie = 0.

But in two dimensions, the only nonzero component of the vorticity vector
is ¢, which is the vorticity in the z direction. Hence

V2 =0
but

2 2
cw o Fy Py
ax ay at oyt

where the stream function that satisfies the continuity equation has been
introduced. Thus, the equation to be satisfied by the vorticity is

P
g-}-ayiz ’ll)=0

This is the biharmonic equation, and in cylindrical coordinates (R, 0), it
becomes

2
? 19 1 o
dR?> R AR R?* 90?
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Noting that the stream function for a uniform flow is y = Uy = UR sin 6, we
look for a solution to the preceding equation of the form

Y(R, 6) = f(R) sin 6

where f(R) — UR as R — . Substituting this form of solution into the bihar-
monic equation gives

d2 1d 1
— S _ = f-o
a®RTRaR RS

This is an equidimensional equation whose solution is of the form
3 D
f(R)= AR’ + BRlogR + CR + R

Thus, the stream function is of the form
3 D .
PY(R,0)= AR’ +BRlogR +CR + ® sin 6.

In order to recover a uniform flow far from the cylinder, y(R, 6) must tend
to UR sinB as R — «. Hence

A=B=0

and

C=U

Hence, the expression for the stream function reduces to
D .
P(R,0)= UR+ R sin .

The near boundary condition requires that on the surface of the cylinder,
both the tangential and radial velocity components should vanish. That is,
on R = g, both dy/d0 and dy/dR should vanish. Since dy/d0 should be zero
for all values of 0, the tangential velocity component being zero is equivalent
to requiring that (s, 0) = constant, where the constant may be taken to be

zero. That is, the no-slip boundary condition on the surface of the cylinder
requires that
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Y(a,0)=0

P
—(a,0)=0.
R @9

It is now evident that there is no choice of the constant D in our solution
that satisfies these two boundary conditions. If we had satisfied the near
boundary conditions first with the solution, it would have been found that it
was impossible to satisfy the far boundary condition. Thus, we conclude that
there is no solution to the two-dimensional Stokes equations that can satisfy
both the near and far boundary conditions. The lack of such a solution is
known as Stokes” paradox.

The difference between the two-dimensional Stokes equations and the
three-dimensional Stokes equations is best explained by reexamining the
Stokes approximation. In terms of dimensionless variables, the Navier—
Stokes equations were shown to be

ou*

ot*

+ RN(u* . *)u* - _ *p* + *Zu*

so that the Stokes equations correspond to the limit Ry — 0. Thus, a more
accurate solution for low Reynolds numbers could be sought in the form

Y =Y, + Ryp, +O(R12\1)

which represents an asymptotic expansion for the stream function, which is
valid for low Reynolds numbers. Then, by employing a limiting procedure,
the problem for y, may be solved, then the problem for y;, and so on. It has
been shown here and in Section 8.7 that a solution corresponding to y, exists
for a sphere but not for a cylinder. However, it is found that the problem for
y; in the case of the sphere has no solution, which is known as Whitehead’s
paradox. Thus, a basic difficulty has been encountered, and this difficulty
appears in the first-order problem in two dimensions and in the second-
order problem in three dimensions.

In mathematical terminology, the difficulty encountered above is referred
to as a singular perturbation. That is, the Stokes approximation is really the
first-order problem arising out of a perturbation type of solution to the
Navier—Stokes equations, and the inability of this type of solution to match
the required boundary conditions renders the perturbation singular. In two
dimensions, the difficulty associated with this singular perturbation appears
immediately, whereas in three dimensions, the difficulty is postponed to the
second-order term in the expansion.

In more physical terms, the difficulty encountered is associated with the
neglect of the convection of momentum of the fluid, an assumption that
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is invalid far from the body. The limit Ry — 0 is equivalent to completely
neglecting the convection in the fluid in comparison with the viscous dif-
fusion in the fluid. Because of the nature of the viscous boundary condi-
tion near the body, viscous diffusion will be large near the body, whereas
convection will be small because of the retardation of the velocity by the
body. However, far from the body, the velocity gradients will die down, so
that viscous diffusion will be reduced. At the same time, the fluid velocity
will be close to that of the free-stream velocity. That is, the convection in the
fluid will become more important, while the viscous diffusion will become
less important. This means that the conditions required to satisfy the Stokes
approximation will be violated.

The nature of the failure of the approximation introduced by Stokes is that
of a nonuniform representation in space. The approximation is valid close to
the body but is invalid far from the body. Thus, singular perturbations are
sometimes referred to as nonuniform expansions. The mathematical diffi-
culties encountered in singular perturbations may be overcome by matched
expansions, which will be described in the next section.

8.9 Oseen Approximation

An alternative low-Reynolds-number approximation is the Oseen approx-
imation. Oseen recognized the discrepancy that was inherent in the
Stokes approximation far from the body under consideration. He noted
that the Stokes approximation corresponds to convection at zero veloc-
ity and recognized that far from the body momentum will be convected
with a velocity that will be close to the free-stream velocity. Thus, Oseen
proposed linearizing the Navier-Stokes equations such that momentum
is transported not with the local velocity (as in the exact case) or with zero
velocity (as in the Stokes approximation), but with the free-stream veloc-
ity. Thus, if the free stream flows in the x direction with velocity U, the
equations that represent the Oseen approximation to the Navier-Stokes
equations are

V-u=0 (8.9a)
a—u+ a—u=—l p+v ‘u (8.9b)
ot ox p

Solutions to the equations above may be established in a manner similar to
that used to obtain solutions to the Stokes equations. The results so obtained
will be valid far from the body but will fail close to the body. This is exactly
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the opposite of the solutions to the Stokes equations. Thus, two independent
solutions are obtained, one being valid near the body and the other being
valid far from the body. By matching these two solutions, a uniformly valid
expression will result that will be valid for small Reynolds numbers. The
details are considered to be beyond the scope of the fundamentals that are
being treated in this book, but they may be found in the book by Van Dyke
referenced at the end of Part IIl. This method of overcoming the difficul-
ties encountered owing to the singular perturbation is called the method of
matched asymptotic expansions.

PROBLEMS

8.1 Using the Stokes solution for uniform flow over a sphere, inte-
grate the pressure around the surface of the sphere to establish
the pressure drag that acts on the sphere. Hence, deduce what
portion of the total Stokes drag is due to the pressure distribution
and what portion is due to the viscous shear on the surface of the
sphere.

8.2 A liquid drop whose viscosity is p’ moves slowly through another
liquid of viscosity p with velocity U. The shape of the drop may be
taken to be spherical and the motion may be taken to be sufficiently
slow that inertia of the fluid may be neglected. The boundary condi-
tions at the surface of the drop are that the velocity and tangential
stresses in the two fluids are the same.

Show that the solution to the problem posed above exists in which
the pressure inside the drop is proportional to x and that outside the
drop is proportional to x/r%. From the solution, calculate the drag of
the drop and show that it is smaller than that for a rigid sphere of the
same size, the drag ratio being

1+2/3( / )
1+ / '

8.3 A flow field is represented by the following equations:

Show that this representation satisfies the Stokes equations. Using
this solution, find the velocity and pressure fields for a fluid that is
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contained between two concentric spheres of radii r; and 7, > r; in
which the outer sphere is rotating with angular velocity €, about
the x axis and the inner sphere is rotating with angular velocity Q,
in the same direction. Calculate the torque that acts on each of the
spheres.

From the results obtained above, deduce the velocity and pressure
fields in a fluid that is contained inside a rotating sphere and find the
torque that acts on the sphere.

84 A flow field is represented by the following equations:

u=VxxQ
p=0.

Here, Qis a vector of constant magnitude. Show that this represen-
tation is a solution of the Stokes equations provided that y satisfies
the following equation:

1 ay

2

-—==0
% v ot

Solve this equation for x, and hence, find the velocity field gener-
ated by a sphere of radius a that is rotating with a periodic angular
velocity |Q| e,

8.5 Show that in spherical coordinates the Stokes equations for axisym-
metric flows may be written in the following form:

L(L2y) = 0
where

@ sin0 0 1 9

2
[ = Tt — - .
ar r> 00 sinB 40

8.6 Obtain a solution to the Stokes equations in spherical coordinates for
the stream function y(r, 6) in the following form:

W, 0) = £O)
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Show that this solution can be used to represent Stokes flow in a
right-angled corner in which the vertical surface x = 0 is stationary
and the horizontal surface y = 0 is moving in the negative x direction
with constant velocity U. Estimate the range of validity of this solu-
tion by evaluating from it the order of magnitude of the inertia terms
and the viscous terms as indicated below and determining the value
of the radius r that ensures a small value of the Reynolds number:

N U
inertia: pu, p L

2

. °u
viscous: ——-.
r° 00

r

Figure 8.3 shows how modern experimental instrumentation and
techniques permit the visualization of complex fluid flow situation.
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0.4 ms 1.4 ms

0.5 ms 1.8 ms

FIGURE 8.3
Heptane droplet striking a stainless steel surface at a temperature of 160°C. (Courtesy of
Professor Sanjeev Chandra, University of Toronto, Toronto, Canada.)






9

Boundary Layers

Boundary layers are the thin fluid layers adjacent to the surface of a body in
which strong viscous effects exist. Figure 9.1 shows the nature of the flow
field that would exist around an arbitrary body at a Reynolds number that
is not small or of order unity. The nature of such a flow field is known from
information gathered from a large number of experiments.

A dotted line is shown in Figure 9.1 that originates at the front stagnation
point and moves downstream near the top and bottom surfaces of the body.
Outside of this dotted line, relative to the body, the velocity gradients are not
large, and so viscous effects are negligible. Then, if compressible effects may
be ignored, the fluid may be considered to be ideal, and the results of Part II
of this book may be employed. Thus, if the flow field far upstream is uni-
form, it is also irrotational there, so that Kelvin's theorem guarantees us that
the flow outside the dotted line is everywhere irrotational. This potential-
flow field is frequently referred to as the “outer flow.”

Between the dotted line and the body, there are strong viscous effects due
to the large velocity gradient that exists. These large velocity gradients are
necessitated by the no-slip boundary condition on the solid boundary, which
reduces the large velocities that exist in the outer flow to zero on the surface.
This is the so-called boundary layer or “inner flow.” Here, the vorticity is not
zero. Vorticity is generated along the surface of the body, and it is diffused
across the boundary layer and convected along the boundary layer by the
mean flow.

Toward the rear of the body, the boundary layer will encounter an adverse
pressure gradient, that is, an increasing pressure. This usually causes the
boundary layer to separate from the body, forming a so-called wake region
behind the body. The velocity gradients are not large in the wake, so viscous
effects are not too important. However, all the vorticity that exists in the
boundary layers is convected into the wake, so that the flow in the wake is
not irrotational. If the boundary layer is still laminar at separation, a shear
layer will exist of the type discussed in Section 6.12. Such shear layers were
found to be unstable, and over a wide range of Reynolds numbers, this insta-
bility manifests itself in the form of a periodic wake that is the well-known
Kérman vortex street.

The coverage of boundary layers begins here with the derivation of the
boundary-layer approximation to the Navier-Stokes equations. Some exact
solutions to these equations are then discussed, including the Blasius solu-
tion for the boundary layer on a flat surface and the Falkner—Skan similarity

321
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Viscous effects negligible

Outer flow Boundary layer separates
Vorticity zero /

Viscous effects not important
7 Vorticity nonzero

Inner flow

Strong viscous effects
Vorticity generated

FIGURE 9.1
Nature of flow around an arbitrarily shaped bluff body.

solutions. The Karman—-Pohlhausen approximate method is then introduced
and applied to a general boundary-layer problem. Finally, the separation of
boundary layers and their stability are discussed.

9.1 Boundary-Layer Thicknesses

Prior to establishing the boundary-layer equations, it is useful to establish
the three types of boundary-layer thicknesses that are in common use. The
most widely used of the boundary-layer thicknesses is simply referred to as
the boundary-layer thickness, and it is denoted by &. Its usual definition is that
distance y = § from the solid boundary where the local value of the velocity
reaches 0.99 of the free-stream or outer-flow value. That is,

y =8 when u = 099U (0.1a)

Figure 9.2a shows the boundary-layer thickness & for flow over a flat
surface.

Another type of boundary-layer thickness that is useful under certain
circumstances is the displacement thickness, which is denoted by &' This
thickness is defined as the distance by which the undisturbed outer flow
is displaced from the boundary by a stagnant layer that removes the same
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099U

FIGURE 9.2
Definition sketch for (a) boundary-layer thickness and (b) displacement thickness.

mass flow from the flow field as the actual boundary layer. That is, §" is the
thickness of a zero-velocity layer that has the same mass-flow defect as the
actual boundary layer. This thickness is illustrated in Figure 9.2b. In math-
ematical terms, the volume of fluid that is absent owing to the presence of
the boundary-layer model is U&". Equating this to the volume of fluid that is
absent owing to the actual boundary layer gives the equation that defines the
displacement thickness. Thus

U6*=fw(U—u)dy
0

andhence
8 = 3 1_l d 9.1b
j; u Y. (©.1b)

A third type of boundary-layer thickness that is frequently used is the
momentum thickness, denoted by 6. The momentum thickness is defined as
that thickness of layer that, at zero velocity, has the same momentum defect,
relative to the outer flow velocity, as the actual boundary layer. Thus, the
momentum thickness is a layer similar to that illustrated in Figure 9.2b,
except that momentum fluxes rather than mass flows are compared with the
actual boundary layer. The mass flow that would exist through the momen-
tum thickness at the outer velocity would be pU6. Hence, the momentum
defect due to this layer will be pU?20. Equating this to the momentum defect
in the actual boundary layer gives

u?e = i u-u)d
p pﬁ u(U - u)dy
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and hence

U u
0= — 1-— dy. .
o U u Y (©1q)

The various thicknesses defined above are, to some extent, an indication
of the distance over which viscous effects extend. Each of these thicknesses
will be used in later sections of this chapter, but in the meantime, it may
be stated that the boundary-layer thickness & is usually larger than the dis-
placement thickness 6°, which, in turn, is usually larger than the momentum
thickness 6.

9.2 Boundary-Layer Equations

The boundary-layer equations may be derived from the Navier-Stokes
equations by either a physically based argument or a limiting procedure as
Ry — o. The original derivation used by Prandtl was physical in nature and
will be followed here.

Figure 9.3 shows a typical boundary-layer configuration on a plane sur-
face or on a curved surface for which 6 is small compared with the radius
of curvature of the surface. The only geometric length scale in such prob-
lems is the distance x from the leading edge of the surface. For all points in
the boundary layer except those near the leading edge, the boundary-layer
thickness will be small compared with the distance x. That is, except near the
leading edge, 6/x < 1. The x component of velocity u is of order U, the outer
flow velocity, and d/dx is of order 1/x in the boundary layer. Thus, du/0x is of
order U/x, and hence, in order to maintain the importance of both terms in

FIGURE 9.3
Development of a boundary layer on a plane surface.
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the continuity equation, dv/dy must also be of order U/x. Since v will be much
smaller than u in the boundary layer and since d/dy will be much larger than
d/0x, the required order of dv/dy may be met by considering v to be of order
Ud/x and d/dy as being of order 1/8. Thus, within the boundary layer, the
following order of magnitudes will exist:

u~U
v~U§

X
o1
x x
91
ay d

The Navier-Stokes equations for the steady two-dimensional flow under
consideration are

ou ou  1ap  u  du
P m— =y

u

ox dy  pax  axt oy’

o v 1 v 90
U—— 40— === 4V +V .
ox ay p Iy ox ay

Using the order of magnitudes established above, the various terms in
these two equations will be of the following order of magnitude:

ur ur 1ep U U
=T hv— v
X X pax x* §

du*  du? 1adp ou u
Tt =-—+U—+U__.
X X p dy X xd

No attempt has yet been made to estimate the order of magnitude of the
pressure terms, so they are carried along as they are.

In the first of these two equations, the two inertia terms are of the same
order, but the second viscous term (vd*1/dy?) is seen to be much larger
than the first viscous term (v0*u/0x?). Hence, the latter viscous term may
be neglected in boundary layers. Since fluid particles may be accelerated in
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boundary layers, and since strong viscous effects exist, the dominant viscous
term is assumed to be of the same order of magnitude as the inertia terms.
This gives

u? u
S0
X 8?
or
&~ |2
0

That is, from purely order-of-magnitude considerations, it may be deduced
that the boundary-layer thickness will increase as x. Furthermore, the con-
dition §/x <« 1 becomes

x? Ux
M e |
8 v
or
RN=%>>1
v

that is, the boundary-layer assumption that §/x < 1 is equivalent to the con-
dition Ry > 1, where the length scale used in the Reynolds number is x.

From the foregoing discussion, it is obvious that, provided the Reynolds
number based on x is large, the x component of the Navier—Strokes equations
may be approximated by the following equation:

ou ou 1  du
U—+V —=-— 4V —.

0x ay p dx Yy

Furthermore, from the order-of-magnitude balance for the y component of
the Navier-Stokes equations, it is evident that the inertia in the y direction
is of order §/x smaller than that in the x direction and so may be neglected
by comparison. Also, the viscous terms in the y direction are of order &/x
smaller than those that act in the x direction, and so the former may be
neglected. Thus, the y component of the Navier-Stokes equations becomes

1o
pay
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That is, p is independent of the transverse coordinate i in boundary layers,
so that p will be a function of x only. Thus, in boundary layers, the continuity
equation and the Navier-Stokes equations become

o _

0 9.2
ox ¥ ay -22)

2
gou, ou_ ldp 9w ©.2b)
ax dy  pdx ay?

It will be noticed that the loss of the highest derivative in x now makes
the governing equations parabolic, whereas the Navier-Stokes equations
are elliptic. This mathematical change has physical consequences, as well as
mathematical implications, which will be exposed in later sections.

Since the pressure p is independent of the transverse coordinate y in
boundary layers, the pressure distribution along the boundary layer will be
the same as that of the outer flow. But the outer flow is a potential flow, and
so the Bernoulli equation is valid there. Hence

P

+ lLIZ = constant
p 2

where the outer velocity U will be a constant for flow over a plane surface,
but in general, it will be a function of x. Thus, from the Bernoulli equation,
considering U to be a function of x,

1dp du
-——=U—-.
p dx dx

Substituting this result into Equation 9.2b gives the following alternative
form of the Prandtl boundary-layer equation:

ou ou  dU  o'u
U—+o—=U—+v—. (9.2¢)
ox oy dx ay’

The boundary conditions that accompany the boundary-layer equations
are the no-slip conditions on the surface and the condition that the outer-
flow velocity is recovered far from the bounding surface. That is, the follow-
ing boundary conditions must be satisfied:

u(x, 0)=0 (9.3a)
v(x,0)=0 (9.3b)
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u(x, y) = U(x) as y — oo. (9.3¢0)

The last condition in effect matches the inner flow to the outer flow, so
that the potential-flow solution must be known before the boundary-layer
problem can be solved.

The alternative way of deriving the boundary-layer equations from the
Navier—Stokes equations involves a limiting procedure similar to that which
was used to extract the Stokes equations from the full Navier-Stokes equa-
tions. The Navier-Stokes equations are first written in terms of dimension-
less variables, which results in a coefficient 1/Ry appearing in front of the

viscous terms. Stretched coordinates X = x and Y = /Ry are then intro-

duced, which removes the coefficient 1/R, from one of the viscous terms. If
the limit Ry, — o is taken, while holding X and Y fixed, the boundary-layer
equations will be obtained. This approach is useful if higher approximations
to boundary-layer theory are required, that is, if an expansion type of solu-
tion is sought. However, the nature of the coordinate stretching is not obvi-
ous without appealing to the results obtained using the physical approach.

9.3 Blasius Solution

An exact solution to the boundary-layer equations corresponding to a uni-
form flow over a flat surface was obtained by Blasius. The flow configuration
for a flat boundary is shown in Figure 9.3, where it is understood that U is a
constant and & is a function of x. Since U is constant, the pressure term in the
boundary-layer equation is identically zero. Thus, the continuity equation
and the boundary-layer approximation to the Navier—Stokes equations are

ou dv
o L 2r
dax dy

ou  gu_du

In order to reduce this pair of equations to a single equation, a stream func-
tion defined by u = dy/dy, v = —dy/dx is introduced. This satisfies the conti-
nuity equation identically for all stream functions y and yields the following
form of the boundary-layer equation:

LA A I
dy oxay ox ay> oy’
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Since this is a parabolic partial differential equation and since there
is no geometric length scale in the problem, a similarity type of solu-
tion will be sought. Similarity solutions were discussed in Section 7.4 and,
in the context of the problem in hand, were shown to be of the following
form:

w(x, y) ~ fn)

where

n~-=-.
xn

The value of n for the case of a flat surface is 1, so that the similarity vari-
able n is chosen to be 2

Y

n= Jux/U .

Here, the parameters v and U have been used to render the similarity vari-
able dimensionless. For this choice of 1, the x component of velocity will have
the following functional form:

_w U
u= oy \/;f(n)

where the prime denotes differentiation with respect to n. But when n = con-
stant, u should be constant, so that the proportionality factor in the equa-
tion y(x, y) ~ f(n) should include a Vx. Then, since the units of y are a length
squared divided by time, the dimensions will be correct if the proportional-

ity factor also includes vvU . That is, a similarity solution of the following
form is sought:

U)(x,]/) = Wuxf(ﬂ)
where

.Y

Jux/Uu '

From these expressions, the various derivatives that appear in the bound-
ary-layer equation may be evaluated as follows:
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w_ Uy, 1,71
= 2xf+2 vl f

ox x1/2

1 |jolU 1 vl
B AT

o
= U]

ay f

Py U uy u
AL x3/2f __anf
92 u

Cueuy S

ay vXx

Py _ur

ay’  vx

Substituting these results into the equation for the stream function gives
u u’ u’
o W -f)f =
or
u’ u’
e

Since x may be canceled from this equation, the existence of a similarity
type of solution is confirmed. That is, a solution of the assumed form exists
provided that the function satisfies the following conditions:

f o+ % ff =0 (94a)
fl0)=£(0)=0 (94b)
f) = Tasn— e (94¢)

The boundary conditions 9.4b and 9.4c follow from the no-slip boundary
conditions and the matching condition with the outer flow as described by
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Equations 9.3a through 9.3c. From the solution to this problem, the stream
function may be obtained using the relationship

w(xly)= V"UUXf \/U.ZW . (94d)

The problem represented by Equations 9.4a through 9.4c is a well-posed
problem. It is shown in the problems at the end of this chapter that the dif-
ferential equation may be reduced in order. However, numerical integration
is eventually required. In spite of this, the Blasius solution to the boundary-
layer equations is considered to be exact since the partial differential equa-
tion has been reduced to an ordinary differential equation that, together
with the appropriate boundary conditions, may be solved numerically to a
high degree of accuracy.

The results of interest that should be extracted from the solution are the
shear-stress distribution along the surface, the drag acting on the surface,
and the boundary-layer thickness. The shear stress on the surface is given by

Ty(x) = Z‘(x, 0)

BRI
= x,0
a]/2( )

3
- 1o

vXx

Nondimensionalizing this surface shear stress by means of the kinetic
energy in the free stream gives

() _2f (0)

%pu2 \/a

where the Reynolds number is based on the distance x from the leading edge
of the surface to the location under consideration. But the value of f/(0) is
found numerically to be 0.332, so that the shear-stress distribution along the
surface will be given by the expression

To(x)  0.664
%puz VRy (9.5a)

This result shows that the shear stress falls off as \x along the surface.
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The drag force acting on the surface may be evaluated by integrating the
shear stress. That is, the drag force acting on the surface up to the location x
will be given by

Fo = [ o(®)de.
Thus, the drag coefficient of the surface will be

Cp = fDi _1 f Fﬂdg.
- UZ XJo - UZ

2P 2P

Using the result obtained above for the surface shear-stress distribution

gives

0.664 0 dE
C. = 95
P« j; JRy
9.5b)
CD=1.328

e

Strictly speaking, the shear-stress distribution given by Equation 9.5a
should not be used near the leading edge of the surface since the boundary-
layer assumptions are no longer valid there. However, any difference
between the actual shear stress and that given by Equation 9.5a is not likely
to create any significant discrepancy because of the relatively short distance
involved. The shear stress actually has a singularity at x = 0, but this singu-
larity is integrable, so that the drag force is not singular. Indeed, Equation
9.5b shows that the drag force varies as \Vx, where x is the point up to which
the accumulated drag is being considered.

To obtain the boundary-layer thickness, it is observed from the numerical
solution that u = 0.99U when 1 = 5.0. Then, using the definition of n and the
fact that y = 6 when u = 0.99U gives

)

Jux/U )

5.0

and hence

\/ﬁ (9.5C)
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where, again, the length used in the Reynolds number is the distance x to
which the boundary-layer thickness applies. In the same way, the following
expressions are obtained from the numerical results for the displacement
thickness and the momentum thickness:

6: 172 95d
X /RN 5d)
6 0.664

; = \/E . (958)

These results show that the various boundary-layer thicknesses grow as
Jx and that 0 < § < 8.

9.4 Falkner-Skan Solutions

A whole family of similarity solutions to the boundary-layer equations was
found by Falkner and Skan. These solutions are obtained by seeking general
similarity-type solutions and interpreting the flow field for each solution so
obtained.

Look for general similarity solutions of the form

u(x, y) = U)f ()

where

e

Here, U(x) is the outer flow and &(x) is an unspecified function of x that will
be determined later. For the form of velocity given above, the stream func-
tion, which may be obtained by integration, must be of the following form:

w(x, y) = UXER)fN)-

But, from Equation 9.2¢, the equation to be satisfied by y is

2 2 3,
oy Py oy Py _ AU o'y

dy oxdy  ox ay’ dx ay’
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The various terms that appear in this equation may be evaluated as follows:

T rus

dx
Ef u® f u s nf

W _

ay_f

oy _du .,y dg

oxdy  dx g% dx
_du ud
dx E dx

oy U

ay’ %f

Py U

y3 EZf

Substituting these results into the equation to be satisfied by y gives

du u dg dU u
uf af —Eaﬂf Ef s f u® ﬂf f EH)?
Hence
d d
U P -U - R U e f

Here, the second term in the first bracket has been canceled with the third
term in the second bracket. Combining the second and third terms of this

equation gives

U (- S - ey
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This equation may be put in standard form by multiplying by &2/(oll),
giving

=g g =Y

fo+ o dr  dx 1-(f)}=0.

If a similarity solution exists, this should now be an ordinary differential
equation for the function fin terms of n. Thus, the two coefficients inside the
square brackets should equal a constant at most, say o and f, respectively.
That is, for a similarity solution, we must have

Ed .
de(ug)_a
g2du
v P

where aand p are constants. A convenient alternative to one of the preceding
equations may be obtained as follows:

d dg du
—(UE*) =2UE—>+E>—
dx( 5) depé dx
=2E ug_,_gdiu - Zdil’[
dx dx dx

d du
=28—(UE)-&>—
El ug)-g 1
that is,

gga@%=vaa—m.

This equation, together with either of the foregoing two equations, is suf-
ficient to relate U and & to the constants o and p. In terms of a and p, the dif-
ferential equation to be solved for the function f becomes

£+ aff’ + Bl1 - (f)2] =0.

The boundary conditions that accompany this differential equation are the
same as for the flat surface. If the problem so obtained is a solvable one, then
we have found an exact solution to the boundary-layer equations.
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From the foregoing analysis and discussion, it is evident that exact solu-
tions to the boundary-layer equations may be obtained by pursuing the fol-
lowing procedure:

1. Select values of the constants o and f.
2. Find the corresponding values of U(x) and &(x) from the relations

&g - va-p) 960
X
g? %l = pu. (9.6b)

3. Determine the function f(n) that is the solution to the following

problem:
frroff +pl- (=0 (9.60)
A0)=f(0)=0 9.6d)
f) = lasn— . (9.6€)

4. The stream function for the flow field in the boundary layer is then
given by

3 v
V() =UWEWS o 9.66)

Having chosen the constants a and f, a particular flow configuration
is being considered. This flow configuration will not be known a priori,
but it will become evident when step 2 is completed. The function U(x) is
the outer-flow velocity, which is the potential-flow velocity for the geom-
etry under consideration. Then, when U(x) is established through step 2,
comparison with the results of Chapter 4 will reveal the geometry of the
problem. Since a and p have been chosen, the problem to be satisfied by
the function f(n) is now explicit, so a solution may be sought. This solu-
tion, together with the quantities U(x) and &(x), completely determines the
stream function for the problem from which all properties of the flow field
may be derived.

Several exact solutions to the boundary-layer equations may be obtained
by the foregoing method. The solution corresponding to a flat surface, which
has already been established, will be obtained from the Falkner—Skan
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solutions to illustrate the procedure and to verify the result. Some new solu-
tions to the boundary-layer equations will then be established in the follow-
ing sections.

It should be noted that for a = 1, numerical solutions to Equations 9.6a
through 9.6f show that f(0) — 0 as p is decreased. The value for which
f(0) = 0is p =-0.1988, and for values of f that are smaller than this value,
fn) > 1 at some location. This corresponds to u > U, which is physically
realized only in certain types of wall jets. Therefore, for a = 1, we consider
p >-0.1988. 1

The solution corresponding to a flat surface is obtained by choosing o = >

and p = 0 in the Falkner—Skan solutions. Then, from Equations 9.6a and 9.6b

d

— (UE*)0 =

M(@ v
dUu

2
—=0.
E d

Since &(x) cannot be zero, the second of these equations shows that U(x) = c,
where c is a constant. Then, the other equation shows that

() = [ .
C

The fact that U(x) is a constant in this case identifies the geometry as a
flat surface rather than a curvilinear one that may be thought of as being
stretched out into a plane. Using the values of a and § chosen above, the
problem to be solved by the function f(n) is

oy f =0

fO=f©=0
fm—=1 asn— o

The stream function is then given by

1P(xr]/)=\/af \/ny—/c .

These results are seen to agree identically with the form of the Blasius solu-
tion, which is given by Equations 9.4a through 9.4d.
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9.5 Flow over a Wedge

The solution to the boundary-layer equations corresponding to flow over a
wedge may be obtained from the Falkner—Skan equations by setting a = 1
and keeping f arbitrary. Then, from Equations 9.6a and 9.6b with o = 1, U(x)
and &(x) will be defined by the following equations:

£ @ =v2-p

au _

€ -V

Integrating the first of these equations gives
€2U = (2 - P)x.
Dividing the second of the foregoing equations by this last result gives

ldu_ g 1
Ude 2-fx

This equation may be integrated directly to give

p
2-f

where ¢ is an arbitrary constant. Hence, the outer-flow velocity correspond-
ing to our choice of a is

logU = log x + log ¢

Ulx) = cxb/e-h (9.7a)
but
2 dU _
g o uvp
) Ezci X 20B/CP) _ g
o 2 p
and hence

E(x) = ch_ B) a-ren, ©.7b)
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Equation 9.7a shows that the outer flow is that over a wedge of angle nf. This
may be shown by using the potential flow solution for flow in a sector whose
angle, measured in the fluid, is n/n. The result, as given by Equation 4.10, is

F(z) = Uz"
- Wi(z) = nUz™"
that is,
u — v =nl(x + iy)" L

Hence, on the surface y = 0, the velocity components are

u=nlx""
v=0.

That is, the velocity given by Equation 9.7a has the same form as that

near the boundary of the flow in a sector of angle n/n. To find the angle of

the wedge corresponding to Equation 9.7a, the exponents of x in these two
expressions are equated. Hence

This gives the angle of the half wedge measured in the fluid. Then, from
the symmetry of the flow field, the angle of the wedge will be 2(r — n/n). From
the equation above, this angle is nf,, which is shown in Figure 9.4.

FIGURE 9.4
Boundary-layer flow over a wedge.
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From Equations 9.6¢ through 9.6e, the problem to be solved for the function

fis

fr+ff+BL - (f)1=0 ©.79)
f0)=7(0)=0 0.7d)
f) - Tasn— co 9.7¢)

This problem may be solved numerically. Having obtained the solution for
f(n), the stream function will be given by Equation 9.6f, where U(x) and &(x)
are given by Equations 9.7a and 9.1b. This gives

W(x,y) = e~ xEP f T —yB e arRen 9.7f)

9.6 Stagnation-Point Flow

Another exact solution to the boundary-layer equations that may be obtained
from the Falkner-Skan similarity solution is that corresponding to a
stagnation-point flow. The values of the constants a and f that yield this solu-
tion are a = = 1. But this is equivalent to letting f be unity in the solution for
the flow over a wedge. Then, the angle of the wedge becomes n, which means
that the flow impinges on a flat surface yielding a plane stagnation point.

The solution may be obtained by setting § = 1 in Equations 9.7a through
9.7f. This gives

U(x) = cx (9.8a)

E(x) = % (9.8b)

e ffr e 1-(FR=0 9.80)
f0)=f(0)=0 9.8d)

f) —Tasn— e (9.8¢)
W(x,y)=Jev xf —L— 9.8f)

Ju/e
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It will be noticed that this is precisely the exact solution to the full Navier—
Stokes equations that was obtained by Hiemenz for a stagnation point. This
solution is given by Equations 7.7a through 7.7c. Thus, the exact solution to
the boundary-layer equations is also an exact solution to the full Navier—
Stokes equations in this instance.

9.7 Flow in Convergent Channel

The boundary-layer solution for flow in a convergent channel may be
obtained from the Falkner—Skan solution by choosing o = 0 and p = 1. For
these particular values of the constants, Equations 9.6a and 9.6b become

4 ety
s =-v

Integrating the first of these equations gives
UE? = —ox

and dividing the second equation by this last result gives

Integrating this equation shows that the outer-flow velocity is of the form
c

U(x)=-— (99a)
x

where c is a constant. Then, from one of the results above, it follows that

E(x) = \/Ex. (99b)

Equation 9.9a is the potential-flow velocity for flow in a convergent chan-
nel. That is, the solution obtained here corresponds to a boundary layer on
the wall of a convergent channel in which the flow is directed inward to the
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FIGURE 9.5
Boundary-layer flow on the wall of a convergent channel.

apex of the channel walls. This flow configuration is shown in Figure 9.5. It
will be noted that for c < 0, that is, for outward flow from the apex, Equation
99b shows that no solution exists. This may be interpreted as follows: for
flow in a divergent channel, the adverse pressure gradient will cause the
boundary layers to separate, and hence a reverse flow will result. This situa-
tion was encountered in Section 7.8 when the exact solution for viscous flow
in convergent and divergent channels was studied.
For « = 0 and p = 1, the problem to be satisfied by the function f(n) is

7 +1-(fR=0 (99¢)
f0)=£(0)=0 (99d)
f) — lasn— . (9.9¢)

Then, using Equation 9.6f and the results 9.9a and 9.9b, the value of the
stream function will be

W(x,y) = —eof JI% . 09f)

9.8 Approximate Solution for Flat Surface

The foregoing solutions have all been exact in the sense that a similarity
form of solution reduced the governing partial differential equations to a
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nonlinear ordinary differential equation that could be solved numerically
to a high degree of accuracy. For situations where an exact solution to the
boundary layer equations does not exist, an approximate solution must be
sought. One of the classical approximate methods that are used widely was
introduced by von Karmén and refined by Pohlhausen. The basic procedure
will be presented in this section in the context of boundary-layer flow on a
flat surface, and the procedure will be generalized in the next section.

For flow over a flat surface, the outer velocity U is constant, so that the
boundary-layer equations are

o
ox dy
ou  ou  du
UtV =V
x oy ay?

Normally, a functional form of solution to these equations is sought that
satisfies the equations identically at each point in space, that is, at each value
of x and y, and which tends to the appropriate values on the boundaries. If
such a solution cannot be found, it may be possible to satisfy the basic equa-
tions on the average rather than at each and every point in the fluid. That
is, if the boundary-layer version of the momentum equation is integrated
with respect to y across the boundary layer, the resulting equation will rep-
resent a balance between the average inertia forces and viscous forces for
each x location. Then, a velocity distribution may be obtained that satisfies
this averaged balance of forces but which does not provide a balance at each
point across the boundary layer. The results that are extracted from such
approximate solutions are found to be reasonably accurate in most instances.

Prior to integrating the boundary-layer equations, it is useful to recast them
in a slightly different form. The term u du/dx may be rewritten as follows:

L i(uz)— w
dx  dx ox
=)+ il

ay

in which du/dx has been replaced by —dv/dy from the continuity equation.
Thus, the boundary-layer form of the equation for the x momentum may be
written in the following form:

., o ou  9'u
— (W )+Uu—+v_—="v
ox ay ay
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or

This equation is still exact within the boundary-layer approximation. This
local balance of forces will now be reduced to an average balance across the
boundary layer by integrating with respect to y fromy =0toy =&:

f ) dy + ol = Sy

0

But u(x, 0) = v(x, 0) = 0 from the no-slip boundary condition and u(x, 8) = U,
which is the outer-flow velocity. Also, p du/dy = 1, the surface shear stress,
when y = 0, and since the boundary-layer velocity profile should blend
smoothly into the outer-flow velocity at y = &, du/dy = 0 there. Hence, the
integrated boundary-layer equation becomes

f 9 ?)dy + Un(x, 8) = _F

The quantity v(x, §) may be evaluated by integrating the continuity equa-
tion between the limits y = 0 and y = 8. This gives

2 du
—d 5=0
foax y+[ol
2 du
S Uo(x,d)=-Uf| —d
v(x,0) foax

Then, the integrated boundary-layer equation may be written in the fol-
lowing form:

Finally, these two integrals involving derivatives may be expressed as
derivatives of integrals through the rule of Leibnitz. For any function f(x, y),
this rule states

() 9
& 2,y y)dy = f(x y)dy - f(x, B)fﬁ+f(x,oc)dfa-
d dx dx

a(x) ax
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Using this rule, the integrals that appear above may be rewritten as follows:

f—(u) y—dfu dy - Uzji
ﬁ:’aaxd —%f udy - Uji

Thus, the integrated boundary layer equation becomes

Cicfudy U—fudy FO

Since U is a constant, it may be taken inside the derivative and the integral,
and the two integrals may be combined. This gives

d A T
el U-udy=-2
dxﬁu( uydy =

This equation is known as the momentum integral, and it is valid for bound-
ary layer flow over a flat surface. Physically, this equation states that the rate
of change of the momentum in the entire boundary layer at any value of
x is equal to the force produced by the shear stress at the surface at that
location.

The manner in which the momentum integral is used is as follows: A form
of velocity profile is first assumed, typically a polynomial in y. The arbitrary
constants in this expression are used to match the required boundary condi-
tions. These boundary conditions are

u(x,0)=0
u(x,8)=U

ou
—(x,0)=0.
ay(x )

The first of these conditions is the no-slip boundary condition at the sur-
face, the second condition matches the boundary-layer velocity to the outer-
flow velocity, and the third condition ensures that the matching is smooth at
y = 0. It should be noted that all the higher derivatives should also be zero at
y = & for a smooth transition from the boundary layer to the outer flow. The
number of conditions that can be satisfied, of course, depends upon the num-
ber of free parameters in the assumed velocity profile. It should further be
noted that a series of boundary conditions should also be imposed at y = 0.
The boundary-layer equation for this case is
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ou _du  du
R

Hence, the no-slip boundary condition at y = 0 would automatically result
in ?u/dy? = 0 at y = 0 if our velocity profile was the correct one. However,
since we know that our assumed velocity profile is not the correct one, this
boundary condition must be imposed separately. Likewise, by differentiating
the boundary-layer equation, conditions for the third and higher derivatives
will be obtained, and these should be imposed separately in our approxi-
mate solution. The number of boundary conditions out of this infinite array
at y = 0 and y = & that can be accommodated depends upon the number of
free parameters that are available in the assumed expression for the velocity.
Normally, the three conditions mentioned above are included in the order
of priority in which they are written down, then the condition d%1/dy? = 0 at
y = 01is imposed, then 0%u/dy? = 0 at y = §, and so on.

Typically, the velocity profile is taken to be a polynomial in y, and the
degree of this polynomial determines the number of boundary conditions
that may be satisfied. For the case under consideration, we propose the fol-
lowing form:

2

u Yy
9

E=a0+a1%+a2

Then, three boundary conditions may be satisfied, and since these bound-
ary conditions all involve constants in the case of a flat surface, the quantities
a,, a;, and a, will be constants. Thus, the velocity profile represented above
will be similar at various values of x and so represents a similarity type of
profile. The boundary conditions u(x, 0) = 0, u(x, 8) = U, and du/dy(x, 8) = 0
give, respectively,

0=a,
l=ay+a,+a,

0=a, + 2a,.

The solution to these equations is 2, =0, 4, = 2, and a, = 1, so that the veloc-
ity profile becomes

2

(9.10a)

o<
o=

u_
u

Using the assumed velocity profile across the boundary layer will
reduce the momentum integral to an ordinary differential equation for the
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boundary-layer thickness 5(x). The terms that appear in the momentum inte-
gral may be evaluated as follows:

S

fu(U—u)dy=U2f6u 1-2 dy
0 u u

0

2 2

)
=1J? 21_ ¥ 1_21 ¥ d
uj; 5 0 R

1

=au2L @n-n*)(1-2n+n*)dn

_2

== 8U?
15

Substituting these results into the momentum integral gives

d 36112 =2UU
dx 15 )

8dd =15 dx.
u
Integrating this equation and setting 8 = 0 when x = 0 give
vx
O =+30,.
u

In nondimensional form, this result becomes

5.48

R

5
- (9.10b)
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where the length scale that has been used in the Reynolds number is the dis-
tance x. Equation 9.10b compares favorably with Equation 9.5¢, which is the
exact solution for a flat surface. The relation t, = 2upl/8 shows that the shear
stress on the surface is given by

T 073
%puz JRx (9.10c)

This result also compares favorably with the exact solution, which is given
by Equation 9.5a.

It is evident that the momentum-integral approach is capable of producing
meaningful results, even when it is used in conjunction with a rather crude
approximation to the form of the velocity profile. In the case under consider-
ation here, a second-degree polynomial was used. An even more crude rep-
resentation of the velocity profile would be a straight line that matches only
the boundary conditions u(x, 0) = 0 and u(x, ) = U. On the other hand, third-
or higher-degree polynomials could also be employed that would yield more
accurate results. The second-order profile used here, expressed by Equation
9.10a, gives 0%1/dy?(x, 0) = -2U/8? instead of zero as required. By employing a
third-degree polynomial, the correct velocity curvature at the wall could be
imposed, which would yield more accurate results. This will be confirmed in
the problems at the end of the chapter.

9.9 General Momentum Integral

The momentum integral that was developed in the previous section for flat
surfaces will be generalized here to include outer flows whose velocities are,
in general, functions of x. The boundary layer may still be considered to be
stretched out in a plane, provided that the radius of curvature of the body is
large compared with the boundary-layer thickness, and centrifugal effects are
negligible. In such cases, the outer-flow velocity U(x) will not be constant but
will be defined by the potential-flow solution for the body under consideration.

Performing the same manipulation on the term u du/dx as was carried out
in the previous section, the boundary-layer equations may be written in the
following form:

ou dv
—+—=0
ax dy

2
T )
ox ay dx pay
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Integrating the second equation across the boundary layer and utiliz-
ing the boundary conditions u(x, 0) = 0, u(x, §) = U, p du/dy(x, 0) = 7, and
ou/dy(x, 8) = 0 give

O 0
a , , du T
— @) dy +Uo(x,8)=—— [ Udy -2
[ 3 @)y +Uotx,0) =" - [ U dy -~
The outer-flow velocity U(x) at the edge of the boundary layer depends
upon x only, and dU/dx has been taken outside the integral while U has
been kept inside the integral in the pressure term. This is purely a matter of

convenience that will permit two integrals to be combined in the subsequent
analysis. Integrating the continuity equation shows that

® du
u(x,d) = - . ady

so that the momentum-integral equation becomes

f—(u )ydy - uf “dy ——fUd

Using Leibnitz’s rule permits the order of the integration and the differen-
tiation to be interchanged, yielding the following result:

d > dd ,dd  dU 1,
wdy-u-—-Uu udy +U Udy-—.
dxf Y- dx dxf dr ~ dx f Y P
The second integral may be rewritten as follows:

U;xﬁéudy=§xﬁalludy—iljﬁaudy.

Thus, the momentum integral becomes

;xfu dy - flludy+ ljjjudy=§jjfUdy—E.

The first and second integrals may now be combined, and the third and
fourth integrals may be combined to give

d du o T
Lt -wdy+ - wydy ="
dxj;u( u)y+dxfo( u)dy 0
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But the integrands of these two integrals are essentially zero for y > §, so
that the upper limits of integration may be taken to be infinity. This gives

%uzfoal-—d uf 1-— dy_—

Now, the first integral is the momentum thickness 6, and the second inte-
gral is the displacement thickness &°, as may be seen from comparison with
Equations 9.1c and 9.1b, respectively. Then, the momentum integral may be
rewritten in the form

d du T
—UPe)+— - Ud =
dx o)+ dx P
Expanding the first derivative and dividing the entire equation by U? yield
the following alternative form of the generalized momentum integral:

de 1dU =
+(20+8%)—— 0,
dx + )U dx pll2

(9.11)

For any assumed form of velocity profile across the boundary layer, 6,
8", and 71, may be evaluated from their definitions. Then, Equation 9.11
will provide an ordinary differential equation that may be solved for the
boundary-layer thickness. The manner in which the solution is carried out,
for the case of a fourth-order polynomial for u, will be covered in the next
section.

9.10 Karman-Pohlhausen Approximation

The general momentum integral, when used in conjunction with a fourth-
order polynomial to represent the velocity profile, is known as the Karman-—
Pohlhausen method. The velocity profile is taken to be of the form

%=a+bn+cn2 +dn’ +en’
where

Y

n(x, y) = o)’
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The coefficients a, b, ¢, d, and e will, in general, be functions of x, so that
solutions that are not similar may be obtained. The foregoing velocity profile
can satisfy five boundary conditions, and these are taken to be

u(x,0)=0
u(x,d) = U(x)

ou
—(x,0)=0
ay(x )

Pu g UG AU
ay? v dx
2

IU 8y =0

The fourth boundary condition comes from the boundary-layer form of
the momentum equation and the no-slip boundary condition. In terms of the
dimensionless velocity #/U and the dimensionless coordinate n = y/5, these
boundary conditions become

2 2
% _g ang TW/E) (u/zll):_(ldiu:_/\(x) onn=0
u om v dx
and
2
Y1 and 8(u/U)=6(u/2U)=0 onn=1.
u an o

The quantity A(x) that has been introduced here is a dimensionless vari-
able that is a measure of the pressure gradient in the outer flow.

Imposing the foregoing boundary conditions on the assumed form of veloc-
ity profile gives the following set of algebraic equations for the unknown
coefficients:

O=a

-A=2c
l=a+b+c+d+e
0=b+2c+3d+4e
0=2c+6d+12e.
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The solution to this set of equations is

a=0
b=2+A
6
A
C=——
2
d=—2+A
2
e=1—é.
6

Thus, the assumed form of velocity profile that satisfies the principal
boundary conditions is

It is advantageous to separate the right-hand side of this expression into
terms that are independent of A(x) and terms that depend upon A(x). This
gives

A
=2n-2n*+n')+ g(n—3n2 +3n° -n*)

Si=

(912a)

S

A
=1-(+m(d-n’+ md-m)"
Equation 9.12a is now in the form
2~ F)+ AG()
U n n

where the functions F() and G(n) are shown schematically in Figure 9.6a.
The function F(n) is seen to be a monotonically increasing function of n that
ranges from zero at n = 0 to unity at n = 1. The function G() increases from
zero at 1 = 0 to a maximum of 0.0166 at n = 0.25, after which it drops off to
zeroatn =1

Figure 9.6b shows the velocity profiles corresponding to various values
of the pressure parameter A. For A = 0, the velocity profile corresponds
to a flat surface in which the representation is a fourth-order polynomial.
For values of A greater than 12, the resulting velocity profiles show that
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FIGURE 9.6

(a) Form of functions F(n) and G(n), and (b) velocity profiles for various values of the parameter
A(x).

u/U > 1. Since the boundary-layer velocity is not expected to exceed that
of the outer flow locally, it is concluded that A < 12. Also, for values of A
less than —12, the velocity profiles show negative regions that correspond
to reverse flow. Although reverse flows do occur physically, under these
conditions, the basic assumptions upon which the theory is based cannot
be justified. Thus, the parameter A should be greater than —12. Combining
these two results, it is concluded that the parameter A should lie in the
range

~12 < A < 12. (9.12b)

Having established a suitable approximation to the velocity profile, the
boundary-layer thicknesses and the surface shear stress may now be evalu-
ated. Substituting the velocity profile 9.12a into Equation 9.1b gives the follow-
ing expression for the displacement thickness in terms of the boundary-layer
thickness:

1
u
8* =9 1-—d
ﬂ u

1
A
=df (Weml-m’ = =)’ dn 0.120)
R AL
10 120

Similarly, Equation 9.1c yields the following expression for the momentum
thickness:
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Dy u
o=o = 1-+- 4d
j;ll u M
=% 11-(1+ x1—)3+§—a- )
J; (- + o n(l-n
3 A 3
x a+mu—n>—gna—m dn

37 A A?
315 945 9072 °

9.12d)

The shear stress on the surface for this velocity distribution will be

u o(u/U)
Ty= —
& an =0
uo 3
- S om 1-1+m)(1-n)? +*7](1 m) o (9.12¢)
‘EO = E 2 + A .
) 6

The foregoing expressions relate the displacement and momentum thick-
nesses and the surface shear stress to the boundary-layer thickness, which
as yet, is unknown. These relations follow purely from the velocity profile
under consideration. It is now required to determine the values of these vari-
ous quantities as constrained by the momentum integral.

Multiplying the momentum integral (Equation 9.11) by 6/v gives the fol-
lowing additional relation connecting 6, &, and t,:

UGdB oedu =,0
+(20 4+ 8% =0
+ o v dr u
or
2 * 2
lyd 00, ,,0 00dU_ b
2 dx v 6 v dx u

Expressions for the various quantities that appear in this equation
will now be established as functions of the pressure parameter A(x). Recall
that
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and hence

But 6/8 may be evaluated in terms of A through Equation 9.12d. This gives

2

0° du 37 A A?

v dx 315 945 9072

or
2
6” du = K(x)
v dx
where
2
37 A A?

Kx)= -2 _
315 945 9072

The term §°/6 that appears in the momentum integral may be similarly
evaluated using Equations 9.12¢c and 9.12d. Thus

3 A

5 10 120

6 37 A A
315 945 9072

or
6*
— = f(K
)= f)
where
3_A
10 120
K) =
fK) 37 A A

315 945 9072
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The function f depends on A and hence upon x. However, K is also a
function of x, so that f may be considered to be a function of K. The other
parameter that appears in the momentum integral is t,0/(uU). Multiplying
Equations 9.12d and 9.12e gives the following expression for this parameter:

T,0
- (K
U 8(K)

where

2
W 22N T AN
6 315 945 9072

These results will now be substituted into the momentum integral. For the
time being, the leading term in the momentum integral will be retained in its
existing form. Substituting from the above results for §°/6, 8*(dU/dx)/v, and
7,0(uU) then gives

1,.d o
—u— — 2 K)K = ¢(K
JUS 2+ K = §(K)
where
2
g odu
v dx

It is now proposed to take Z = 0?/v as a new dependent variable so that

K=Zd—u.
dx

Then, the momentum integral becomes

dz

dr 2{g(K)-[2 + f(K)IK}
or
% = H(K)
where

H(K) = 2{g(K) — [2 + AK)IK]}.
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Then, substituting for g(K) and K from their definitions shows that H(K) is
related to A(x) through the following identity:

A 37 A A®

H(K)=2 2+=> =2L_ 2 _
6 315 945 9072
3_ A ,
”. 10 120 37 A A?
37 A A 315 945 9072
315 945 9072
2
_, 37 A A Je, 2 1 . 2 s
315 945 9072 315 945 120 9072

where the quantity K is related to A by the expression

2

7 _ AN
315 945 9072

From these two expressions, both K and H(K) may be evaluated for any
value of the pressure parameter A(x). Thus, a curve of H(K) as a function of
K may be constructed. The form of this curve is shown in Figure 9.7. The
momentum integral has been reduced to the ordinary differential equation
U dZ/dx = H(K) where the functional form of the quantity H is sufficiently
complex that this integral cannot be evaluated explicitly. However, it may be

H(K) \

0 0.0783 K

FIGURE 9.7
Exact form of the function H(K) (solid line) and straight-line approximation (dashed line).
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seen from Figure 9.7 that the function H is approximately linear in K over the
range of interest.
Thus, the function H may be approximated by the equation

H(K) = 047 — 6K.

Substituting this result, the momentum integral becomes

Ud—z=047 6K
dx
=047 - 6Zd—u
dx
or
1 d
T (ZU6) 0.47.

In this form, the momentum integral may be expressed in terms of the fol-
lowing quadrature:

0.47
U

f U(g) de.

Then, since Z = 02/v, the value of 0 will be

0.47v

0°(x) = UG

L xu(E)S dE. 9.12f)

For any given boundary shape, the approximate solution to the bound-
ary layer equations may be obtained as follows: For the specified boundary
shape, the potential-flow problem should be solved to yield the outer velocity
U(x). Then, Equation 9.12f may be used to evaluate the momentum thickness
0(x). The pressure parameter A(x) may then be evaluated from the relation

2 2 2
7 AT, P o128
315 945 9072 v dx

Having established A, the boundary-layer thickness  may be evaluated
from Equation 9.12d, and the displacement thickness 6" may then be obtained
from Equation 9.12c. The velocity distribution across the boundary layer will
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be given by Equation 9.12a, and the shear stress at the surface will be given
by Equation 9.12e. In practice, it is difficult to evaluate the quantity A(x) from
Equation 9.12g unless A = constant. It is therefore much simpler to choose
specific functions A(x) and use the foregoing equations to determine the
outer-flow velocity and hence the nature of the boundary shape.

As an example, the Karman-Pohlhausen approximation will be applied to
the case of flow over a flat surface. For a flat surface, the outer-flow velocity
U will be constant, so that Equation 9.12f gives

0% = 047 %
u

or

0 0.686

x \/E (9.13&)

Since U = constant, dU/dx = 0, so that Equation 9.12g will have the solution
A =0. Then, from Equation 9.12d,

N (9.13b)

Equations 9.12c and 9.12e may now be employed to evaluate the displace-
ment thickness and the surface shear stress. This gives

5 175
N
T, 0686
Lowr Ry (9.13d)
2

These results compare favorably with the results obtained from the Blasius
solution that are given by Equations 9.5a through 9.5e. The principal result
of physical interest is the surface shear stress. The exact solution has a coef-
ficient of 0.664, whereas the Karman—Pohlhausen approximate solution has
a coefficient of 0.686. Thus, the shear-stress distribution obtained here is
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within 3.5% of the exact solution. It should also be noted that there is con-
siderable improvement in the fourth-order polynomial velocity distribution
used here over the second-order polynomial analysis, which yielded a coef-
ficient of 0.73 as was established in Equation 9.10c.

9.11 Boundary-Layer Separation

It is known from experimental observations that boundary layers have a
tendency to separate from the surface over which they flow to form a wake
behind the body, as shown in Figure 9.1. The existence of such wakes leads
to large streamwise pressure differentials across the body, which results in a
substantial pressure drag or form drag. Indeed, for bluff bodies such as circu-
lar cylinders, the form drag constitutes almost all the total drag at Reynolds
numbers of 10* or higher. That is, the shear stress along the surface of a cylin-
der produces a drag force that is negligible compared with the form drag for
large Reynolds numbers. For lifting bodies such as airfoils, separation of the
boundary layer can destroy the bound vortex on the body, thus destroying
the lift that the airfoil generates. This is the so-called stall condition.

A simple qualitative explanation for the existence of boundary-layer sepa-
ration on a bluff body may be given as follows: The pressure gradient along
a boundary layer is determined by that of the outer flow, as was established
earlier. Then, if a region of adverse pressure gradient exists in the outer flow,
this pressure gradient will exert itself along the surface of the body near
which the fluid velocity is small. The momentum contained in the fluid lay-
ers that are adjacent to the surface will be insufficient to overcome the force
exerted by the pressure gradient, so a region of reverse flow will exist. That
is, at some point, the adverse pressure gradient will cause the fluid layers
adjacent to the surface to flow in a direction opposite to that of the outer flow.
Such a flow configuration means that the boundary layer has separated from
the surface and is deflected over the reverse-flow region.

Figure 9.8 shows the qualitative form of the velocity profile in a boundary
layer in the vicinity of the separation point. Prior to separation, the velocity
gradient at the surface is positive, so the shear stress there opposes the outer
flow. After separation, the velocity gradient at the surface is negative, so the
shear stress has changed its sign and direction. This observation leads to the
classical definition of a separation point as a point at which the shear stress
vanishes. That is, separation is said to occur at the point where the velocity
gradient vanishes.

%(x, 0)=0 for separation. (914)
y
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Separation point

FIGURE 9.8
Velocity profiles in a boundary layer in the vicinity of separation.

Using this definition of separation, it may be shown that separation can
occur only in a region of adverse pressure gradient. Along the surface y =0,
the boundary-layer equations reduce to

due to the no-slip boundary condition. Thus, the curvature of the velocity
profile is proportional to the pressure gradient along the surface. Then, if
dp/dx is negative, the curvature of the velocity profile is negative and will
remain negative at the surface just as it is at the edge of the boundary layer.
That is, separation will not occur in a region of favorable pressure gradient.
On the other hand, if dp/dx is positive, the curvature of the velocity pro-
file will be positive at the surface. Since 0%11/9y? must still be negative at the
edge of the boundary layer, the velocity profile must go through an inflection
point somewhere between y = 0 and y = . Such a velocity profile may lead to
separation if the curvature at y = 0 is sufficiently positive to yield a reverse-
flow configuration, as shown in Figure 9.8. Thus, it may be concluded that
separation can occur in a region of positive pressure gradient.

Calculating the location of the separation point is not an easy matter. The
obvious way of proceeding is first to solve the potential-flow problem for the
body in question. The pressure so obtained could then be substituted into
the boundary-layer equations, which could then be solved by either an exact
solution or an approximate solution. From the solution to the boundary-layer
equations, the location of the point of zero shear stress could then be located.
The obvious difficulty with such a procedure is that as soon as the boundary
layer separates, the pressure distribution will differ from that predicted by
the potential-flow solution since the latter applies to a different streamline
configuration from that which exists physically.

There are two principal approaches that are used to overcome the diffi-
culty outlined above. The approach used by Hiemenz involved determining
the pressure distribution around the body in question experimentally. The
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resulting pressure curve may then be represented analytically by a polyno-
mial that permits it to be used in the boundary-layer equations. The results
obtained by this method show good agreement with experimental observa-
tions. However, the disadvantage of this approach lies in the fact that the
pressure distribution must be established experimentally for each body
shape and for each Reynolds number of interest. Measuring the pressure
distribution around circular cylinders is not difficult since a single pressure
tap may be rotated to sense the pressure at different angles from the front
stagnation point. On the other hand, measuring the pressure distribution
around noncircular cylinders is not such a simple matter.

The second approach that is used to determine analytically the location of
the separation point is to modify the potential-flow model from which the pres-
sure distribution is obtained. Several flow models exist, each of which takes
into account the separated configuration of the outer flow. The difficulty with
this approach is that empirical constants exist in the potential-flow model, and
experimental results must be consulted to establish these constants.

From the foregoing discussion, it is evident that the subject of boundary-
layer separation is one that is not well understood analytically. Indeed, it is
still not clear whether the boundary-layer equations are regular at separa-
tion. One school of thought claims that the boundary-layer equations are
regular at separation by virtue of the appropriate pressure distribution.
Some recent results even question the validity of condition 9.14 at separa-
tion. Evidence suggests that the location of the point where dp/dx vanishes,
that where du/dy vanishes, and that where separation occurs are all distinct.
However, no length scales could be established; thus, all these points could
possibly concur within the appropriate macroscopic length scales. What is
known is that boundary layers will separate in adverse pressure gradients, so
the magnitude and extent of such pressure gradients should be minimized.
This means that bodies should be streamlined in shape rather than bluff
and should be oriented at small angles of attack. Also, it is known that sharp
corners that bend away from the fluid become separation points; thus, such
corners should be avoided if separation is to be delayed as far as possible.

9.12 Stability of Boundary Layers

Like any fluid-flow situation, boundary layers may become unstable. Usually,
instabilities in boundary layers manifest themselves in turbulence. That is, a
laminar boundary layer that becomes unstable usually becomes a turbulent
boundary layer. The properties of laminar and turbulent boundary layers are
quite different. For example, the angle to the location of the separation point
on a circular cylinder, measured from the front stagnation point, is about
82° for a laminar boundary layer and about 108° for a turbulent boundary
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layer. This significant change in location of the separation points results in
an appreciable drop in the drag coefficient, as shown in Figure 8.2. It is there-
fore of some interest to investigate the stability of boundary layers.

The basis of our stability calculation will be to introduce a small distur-
bance into the boundary-layer variables and determine whether this distur-
bance grows or decays with time. If the disturbance grows with time, the
boundary layer will be classified as unstable, and if the disturbance decays
with time, the boundary layer will be classified as stable. Intermediate to
these two situations is the case of marginal stability, in which the distur-
bance neither grows nor decays.

Figure 9.9a shows the velocity profile in a narrow strip of a boundary layer.
For such a narrow strip, the velocity in the horizontal direction may be con-
sidered to be a function of y only, for example, V(1), and the vertical velocity
may be considered to be zero. The undisturbed boundary-layer velocity V(y)
acts in the horizontal direction, although the symbol V has been used. This
symbol has been employed to avoid confusion with the outer-flow velocity
U(x) at the edge of the boundary layer.

A small but arbitrary disturbance is introduced to this boundary-layer
velocity profile so that the velocity components and the pressure become

ulx, y, ) =V(y) + u'x, y,
ox, y, ) =0+0xy 1)

px, y, ) = po(x) + p'x, y, 1)

()

a a8’ Stable
Im(c) <0
} Im(c) > 0 0.34 Unstable
Im(c) <0 Stable
|4 520 uos*

FIGURE 9.9
(a) Undisturbed boundary-layer velocity profile, (b) stability calculation results for fixed V, and
(c) stability diagram.
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where |u/V|, |v/V|, and |p/p,| are all small compared with unity.
Substituting these instantaneous local values of the velocity components and
the pressure into the continuity and the Navier-Stokes equations gives

v

ox  dy
MWy, AV
ot dx dy ay

1 dp, dp Pu  dV u
=—— 24" 4y + +
p dx ax axr  dy* oy
9o 1ap v v
- =T 4u +

al+(V+u)ai+v—= > 5
ot ox ay p dy ox ay

As a special case, when the perturbation is zero, the foregoing equations
reduce to

_ldpy &V

0= .
p dx dy?

Hence, these terms may be removed from the equation of x momentum.
Furthermore, since the perturbation is assumed to be small, products of all
primed quantities may be neglected as being small. Thus, the linearized
equations governing the motion of the disturbance are

v _
ox 9y
ou ou v 1dp Pu  0u
—+V—40 —=—-—"—+v
ot ax dy  pox w* oy’

2 2
ai+vav 1ap+vav+6v
ot x  pay axt dy

These three equations may be reduced to two by introducing a perturba-
tion stream function defined by

N N

U =-

u .
ay x
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In terms of this stream function, the governing equations become

2 2 3 3
Py, e _wdv_ dp oy 'y

dyot  xdy ox dy  p ox oy ay’

2 2 3 3
0w o 1o 0w 9y
dxot ox? p dy ox®

axdy®

Finally, by forming the mixed derivative d’p’/dx dy from each of the fore-
going equations and equating the two expressions so obtained, the two equa-
tions above may be reduced to one, which may be written in the following form:

2 2 4 4 4
i+Va 61p+871p ——dva—w=vaw+2 I +81p
ot ax gyt ax®  dy? ox ayt  axPay®  ax?

The stream function for the disturbance must satisfy this linear, fourth-
order, partial differential equation.

Since the disturbance under consideration is arbitrary in form, it may be
Fourier-analyzed in the x direction. That is, the perturbation stream function
may be represented by the following Fourier integral:

Y(x,y,t) =j: lp(y)eiﬂ(x—ct) do

where « is real and positive. The time variation has been taken to be e, so
that if the imaginary part of c is positive, the disturbance will grow, and if it is
negative, the disturbance will decay with time. For ¢ =0, the disturbance will be
neutrally stable. Substituting the preceding expression for the stream function
into the governing equation yields the following integrodifferential equation:

f [(—iOLC + lOLV)(Ip - 0,21[!) — oWV ]eia(x—ct) da
0

0

= f [P -20*F +o'W)e“tNda
0

where the primes denote derivatives with respect to y. Since this equation
should be valid for any arbitrary disturbance whatsoever, it should be valid
for each individual value of the inverse wavelength . Thus, the integrand in
the preceding equation should vanish. This gives

(V-o)W —c®W) -V W= "W —202W +0'W), (9.15a)
{0
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Equation 9.15a is known as the Orr—Sommerfeld equation. The boundary
conditions that accompany this equation may be derived from the condition
that the disturbance should vanish at the surface y = 0 and at the edge of the
boundary layer. Thus

u'(x,0,t)=01(x,0,t)=0
uxy =0y t)>0asy — o
In terms of the stream function ¥(y), these boundary conditions become
Y(0)=710)=0 (9.15b)
Y(y) =P(y) = 0asy — . (9.15¢)

Solutions to the Orr-Sommerfeld equation are obtained as follows. For a
given undisturbed velocity profile and disturbance wavelength, both V(y)
and o will be known. Then, Equations 9.15a through 9.15¢ represent an
eigenvalue problem for the time coefficient c. Then, if each possible wave-
length in turn is treated, results of the form indicated in Figure 9.9b may
be established. That is, regions that are stable (corresponding to the imagi-
nary part of ¢ being negative) and regions that are unstable (correspond-
ing to the imaginary part of c being positive) may be identified. Then, by
considering all possible values of the undisturbed boundary-layer velocity
that are less than the outer-flow velocity, a stability diagram may be con-
structed. That is, by considering all possible values of V(y) in the range
0 < V(y) < U(x), the stability boundaries for that particular x location may be
established. Figure 9.9c shows the results of carrying out such a procedure
for flow over a flat surface. Here, the Reynolds number has been based on
the displacement thickness of the boundary layer, and the inverse wave-
length a has been nondimensionalized by the same quantity. It may be
seen that the lowest Reynolds number for which an instability may occur
is 520. Thus

%

Wait _ 52, (9.16)
(Y]

Hence, an arbitrary disturbance having a Fourier component whose wave-
length is such that ad” = 0.34 will lie on the stability boundary. Thus, it may be
expected that for Reynolds numbers in excess of 520, arbitrary disturbances
will be unstable. Such instabilities will manifest themselves in the form of
turbulence at Reynolds numbers slightly larger than this critical value.
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PROBLEMS

9.1 It was stated in the text that the boundary-layer equations are para-
bolic. Show that they may be put in the form of the one-dimensional
diffusion equation, or heat conduction equation, by taking i = p +
pu - u/2 as anew dependent variable with £ = x and n = y as the new
independent variables.

9.2 The Blasius solution for flow over a flat surface involves solving a
third-order, nonlinear, ordinary differential equation. It will be
noticed that this differential equation is invariant to the following
transformations:

(@ f—f n— n+constant

(b) f— f/a,n — an

Show that transformation (a) enables the Blasius equation to be
reduced to a second-order equation by taking F = df/dn as a new
dependent variable and f as a new independent variable. Then, show
that transformation (b) enables the resulting second-order equation to
be reduced to a first-order equation by taking G = (dF/df)/f as a new
dependent variable and & = F/f? as a new independent variable. Find
the resulting ordinary differential equation for G(§).

9.3 The solution to the boundary layer equations corresponding to flow
in a convergent channel resulted in the following ordinary differen-
tial equation:

Fr+1=(fR=0.

Show that this third-order, nonlinear, ordinary differential equa-
tion may be integrated once by multiplying by the integrating factor
f, then integrated again to give

f(n)=3tanh®> L +1.146 -2

N

where the primes denote differentiation with respect to 1.

94 The figure for this problem illustrates a two-dimensional jet enter-
ing a reservoir that contains a stationary fluid. A solution is sought to
the laminar boundary layer equations for this situation. Assuming
that there is no pressure gradient along the jet, look for a similarity
solution for the stream function of the following form:

w(x, y) = 6owx'/*f(n)
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where

In the expressions above, a is a dimensional constant and v is the
kinematic viscosity of the fluid. Obtain an expression for the func-
tion f(n) in this solution and the boundary conditions that it has to
satisfy. From the solution for f(), obtain the solution for the stream
function y(x,y).

9.5 A boundary layer develops on a surface over which the outer (invis-
cid) flow velocity is represented by the following expression, in
which k is a constant:

U(x) = kx/5.

Look for a similarity solution to the boundary-layer equations for
this flow of the following form:

3 - /Zk
Y(x,y)= Ekvx f(m) where n= 30 ]1//3.

X

Verify that such a similarity solution exists by finding the value of
the exponent m and establishing the ordinary differential equation
to be satisfied by the function f(n).
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9.6 The Blasius solution to the laminar boundary layer equation for uni-
form flow over a flat surface was shown to be as follows:

Y(x,y)=vvolx f(n)  wheren= Y

Jux /U
and f +%ff =0
with f(0)=f(0)=0 and f (n)—1lasn— .

The solution to the ordinary differential equation above has the
following asymptotic forms for small and large values of the simi-
larity variable n:

f)=n*/6+- form <1

f(n)zn—\/§+~- forn > 1.

(@) Using the above results, obtain an expression for the nondi-
mensional surface shear stress in terms of the Reynolds number
based on the distance x.

(b) Obtain an expression for the nondimensional vertical compo-
nent of velocity v(x, y)/U in the boundary layer in terms of the
Reynolds number, based on the distance x, and which is valid for
Y = oo,

(c) Obtain an expression for the value of the vorticity at any loca-
tion in the boundary layer in terms of the similarity variable 1.
Integrate the result so obtained over all values of y and hence
obtain an expression for the total amount of vorticity in the fluid
at any location x, per unit width in the z-direction.

9.7 Use the momentum integral and the velocity profile

—=a+b

u Yy
u d

to evaluate the boundary-layer thicknesses 9, §°, and 6 and the sur-
face shear stress t, for flow over a flat surface.

9.8 Repeat Problem 9.7 using the following velocity distribution:
2 3
Y

Y _gipyec Y 4 Y
u d d d
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9.9 Repeat Problem 9.7 using the following velocity distribution:

910 Repeat Problem 9.7 using the following velocity distribution, in
which a is a constant:

L

9.11 The velocity distribution and the shear stress associated with the
turbulent flow in the boundary layer on a flat surface are found
from experiments to be represented by the following expressions:

wx,y) _y
u

u? v b
tO(X)=p44 us

Use the first result to express 0 in terms of § and the second result,
together with the momentum integral, to express 6 in terms of x.
Hence, evaluate §, &', 6 and 7, in terms of the Reynolds number for
the turbulent boundary layer.

9.12 Use the Karman-Pohlhausen approximation to obtain the solution
for the boundary layer that develops on a surface for which the
outer flow velocity is defined by the following expression:

U(x) = Ax1/°

where A is a constant. From the solution, evaluate the boundary-
layer thicknesses §, &, and 0 and the surface shear stress ,.

9.13 Figure 9.10 shows a viscous, incompressible liquid flowing down
a vertical surface. A boundary layer develops on the vertical sur-
face and grows to approach the free surface. Taking into account
the force due to gravity, write down the boundary-layer equations
for this flow configuration. From these equations, obtain the cor-
responding momentum integral. Hence, by employing a second-
order polynomial for the velocity distribution, obtain an expression
for the boundary-layer thickness 6(x).
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Free surface

FIGURE 9.10
Liquid flowing down a vertical surface.

Figure 9.11 shows the wake behind an airfoil at a large Reynolds number. It
illustrates the difference in streamline pattern between laminar flows, which
are treated herein, and turbulent flows.

FIGURE 9.11
Wake behind a NACA 0025 airfoil at 5° angle of attack and Reynolds number of 10°. (Courtesy
of Sebastian Goodfellow and Professor Pierre Sullivan, University of Toronto, Toronto, Canada.)
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Buoyancy-Driven Flows

There exists a large class of fluid flows in which the motion is caused by
buoyancy in the fluid. Buoyancy is the force experienced in a fluid due to
a variation of density in the presence of a gravitational field. According
to the definition of an incompressible fluid, as was presented in Section
1.6, variations in the density normally mean that the fluid is compressible,
rather than incompressible. That being the case, one might expect that the
material content of this chapter would be presented in Part IV of the text,
rather than Part III. The rationale for this apparent contradiction is dis-
cussed below.

For many of the fluid flows of the type mentioned above, the density
variation is important only in the body-force term of the Navier-Stokes
equations. In all other places in which the density appears in the govern-
ing equations, the variation of density leads to an insignificant effect. That
is, compressibility of the fluid is not a prime consideration. However, vis-
cous effects are of first-order importance. Buoyancy results in a force acting
on the fluid, and the fluid would accelerate continuously if it were not for
the existence of the viscous forces. The viscous forces oppose the buoyancy
forces and cause the fluid to move with a velocity distribution that creates
a balance between the opposing buoyancy and viscous forces. Therefore, if
buoyancy-driven flows are to be classified as being viscous flows of incom-
pressible fluids or compressible flows of inviscid fluids, the former is the more
appropriate classification.

The situation depicted above occurs in natural convection. The other type
of convection is forced convection, in which the fluid moves under the influ-
ence of forces other than the buoyancy force. Since density variations exist
in buoyancy-driven flows, the density is no longer a known quantity. This
means that the continuity and Navier—Stokes equations no longer constitute
a complete set of equations from which the solution to a flow problem may
be obtained. The energy equation is required in order to yield a complete set
of equations, and this adds to the complication of solutions to this class of
problems.

The equations that are most commonly used to solve buoyancy-driven
flow problems employ the Boussinesq approximation. This is the first topic that
is addressed in this chapter. The balance of the chapter is devoted to a pre-
sentation of some of the solutions of the governing equations, as defined by
the Boussinesq approximation.

373
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10.1 Boussinesq Approximation
The equations governing the flow of an incompressible fluid in which grav-
ity provides the only significant body force are written below:

V-u=0

Ju 5
pgw(u- Ju=- p+ “u-pge,.

Here, e, is the unit vector acting in the positive z direction, and it is
assumed that gravity acts in the negative z direction. In the absence of any
motion, these equations reduce to the following form:

0= -Vp, - poge. (10.1)

where p; and p, are, respectively, the pressure and density distributions that
exist under static equilibrium. Then, we may adopt the following notation
for the pressure, density, and velocity distributions in the fluid during con-
vective motion:

p=po+p*
p=py+p*
u=0+u*

where p* is the pressure in the fluid relative to the static value, p* is the den-
sity measured relative to the static value, and u* is the velocity of the fluid
during the convective motion. Substituting these values into the equations
quoted above yields the following result:

‘ut =0

*
*

Ju
(o + p*)? +(p +p)* Jut=- p*+  Pu*-p'ge,.

In the above, the equation of static equilibrium, Equation 10.1, has been
subtracted.

The equations presented above are exact for an incompressible fluid that
has a density variation, or stratification, throughout it. The Boussinesq
approximation consists of neglecting any variation of density except in the
gravitational term. The latter term is of prime importance since it represents
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the force that causes the motion that is being represented. However, the vari-
ation of density is assumed to have only a minor effect on the inertia forces.
This may be considered to be reasonable where relatively small density dif-
ferences exist over moderate distances. Then, considering p to be constant,
the Boussinesq approximation to the governing equations is

V-u=0 (10.2a)
Ju )
Pt plu- Ju=- p+ “u-Apge,. (10.2b)

In Equations 10.2a and 10.2b, it is understood that the pressure p is mea-
sured relative to the static pressure distribution. The quantity Ap is the den-
sity difference relative to the static distribution, and it is positive when the
density is greater than the static value.

Strictly speaking, the equations presented above are valid only for a fluid
in which the density varies, but which is incompressible. However, the idea
behind the Boussinesq approximation may be extended to also include com-
pressible fluids. Provided that the variation in density is small, it may be
assumed that in buoyancy-driven flows, the variation in density is negligible
in all of the terms in the governing equations except the gravitational term.
This means that the variation in density may be neglected in the continuity
equation as well as in the equations of dynamics.

10.2 Thermal Convection

In thermal convection, the density variation is caused by temperature varia-
tions in the fluid. This is to be contrasted with the case of density variations
caused by such effects as salinity variations in water. In thermal convection,
the density is usually expressed in terms of the temperature by the following
relationship:

p=po[l = (T - Ty)l (10.3)

where f is the coefficient of thermal expansion of the fluid, and T is the tem-
perature of the fluid that exists at static equilibrium.

The representation of the density given by Equation 10.3 is valid for mod-
erate departures of the temperature T from the static value T, for an incom-
pressible fluid. In general, the thermal equation of state may be written in the
form p = p(p, T). Hence, it follows, without invoking the condition of incom-
pressibility, that
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J J
p=py+(p- Po)af;)(POr To)"'(T—To)%(pOI Ty)

In the above, only the linear terms in the pressure difference and the tem-
perature difference have been retained in a Taylor series expansion. Now, if
it is assumed that compressible effects are negligible, the second term on the
right-hand side of this equation will be negligible. This is equivalent to say-
ing that the density is a function of the temperature only, rather than being
a function of both the pressure and the density. The third term on the right-
hand side of the equation above may be evaluated for the case of an ideal gas,
for which p = p/RT, giving the result

—P
=p,+(T-T
p=po+( 0) RT(Z)

_ (T-Ty)
Po = Po T, .

This is the same form as Equation 10.3, and it shows that for an ideal gas,
the thermal expansion coefficient assumes the value p = 1/T. In general, the
value of p is determined experimentally, and it is a property of the fluid in
the same sense as the viscosity is a property.

From Equation 10.3, it follows that Ap = —p (T — T,) = —pp(T — T;), where the
density p is assumed to be constant and equal to the value that exists when
there is no motion. Then, substituting this value into Equation 10.2b yields
the following form of the equations governing the motion that results when
thermal convection occurs:

V-u=0 (10.4)

p(z—l; +pu- Ju=- p+ u+pgB(T-Tpe,. (10.5)

Equations 10.4 and 10.5 constitute four scalar equations for five unknown
quantities. The unknown quantities are the velocity vector u, the pressure
p, and the temperature T. Then, in order to achieve a closed mathematical
system, the thermal energy equation must be employed. This means that the
dynamics of the system, and its thermodynamics, are no longer independent
of each other. That is, permitting the density of the fluid to vary with its
temperature in the buoyancy term has coupled the system’s dynamics and
thermodynamics.

The appropriate form of the equation of conservation of energy was
derived in Problem 3.1, and it is given by Equation 3.6, which is rewritten
below:
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p(;—};+p(u' )h=%+(u- p+ (k T)+® (10.6)

where
h=h(p, T). (10.7)

In the foregoing equations, /1 is the enthalpy of the fluid, and ® is the dissi-
pation function. In accordance with the Boussinesq approximation, the den-
sity is assumed to be constant in the energy equation.

In general, the enthalpy & is a function of the pressure p and the tempera-
ture T. However, if we restrict our discussion to ideal gases, it follows that &
may be considered to be a function of T only. Then, for cases where  may be
considered to be a function of T only, including all fluids that are ideal gases,
Equations 10.4 through 10.7 may be rewritten in the following form:

Vou=0 (10.8)
Ju 2

po, +p Ju== p+ u+pgh(T-Tye. (10.9)

o0, TapCyw =P e Do 0m)

In the foregoing equations, the density p is assumed to be constant, and
the pressure p is measured relative to the static value. The quantity C, is the
specific heat at constant pressure, and @ is the dissipation function.

10.3 Boundary-Layer Approximations

Buoyancy-driven flows that comply with the general Boussinesq approxima-
tion are governed by Equations 10.4 through 10.7. For the case of thermal
convection in which the density may be considered to be a function of the
temperature only, the simplified form of the governing equations is given by
Equations 10.8 through 10.10. In this section, we further simplify the govern-
ing equations by applying the boundary-layer approximation to the latter
set of equations and by assuming that the fluid properties remain constant.
In the interests of consistency with Chapter 9, we consider two-dimensional,
steady flow in the x—y plane, in which the main flow is in the x direction.
Since the flow is buoyancy driven, this requires that we adopt the configura-
tion illustrated in Figure 10.1. This is the same situation as depicted in Figure
9.3 except that there is no externally driven flow, and the coordinate system
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Heated
surface

FIGURE 10.1
Development of thermal and momentum boundary layers on a vertical heated surface.

has been rotated through an angle of 90°. In Figure 10.1, the quantity 5; is the
thermal boundary-layer thickness, which is assumed to be of the same order of
magnitude as the boundary-layer thickness 6.

For boundary layer-like flows, the dynamic equations are approximated
in the same way that they were in the previous chapter. That is, the equa-
tions of the dynamics are the same as those derived in Chapter 9 and given
by Equations 9.2a and 9.2b, except that the buoyancy term that appears in
Equation 109 acts in the x direction. Then, it remains to arrive at a consistent
version of the energy equation (Equation 10.10). Writing this equation explic-
itly for steady, two-dimensional flows for which the viscosity coefficient and
the thermal conductivity are both constant yields the following result:

2 2
pC u£+v£ = ua—p+va—p +k ¥+£ +®
ax ay ox ay ox®  dy

where

2 2
au v ou Jv
— o+ — o+ —+—
ox ay ay ox

The following observations may be made regarding the various terms that
appear in this equation.
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1. The convective terms on the left-hand side of this equation are both
of the same order of magnitude—as was the case for the convection
of momentum in the boundary-layer equations.

2. In the first bracketed term on the right-hand side, the pressure gradi-
ent across the boundary layer is negligibly small. This fact is verified
by the y component of the momentum equation in the boundary-
layer equations.

3. In the heat conduction term, the component involving the second
derivative with respect to y is considerably larger than that with
respect to x. This is the same approximation as was made with the
viscous terms in the boundary-layer equations.

4. For moderate velocities induced by thermal convection, the dissipa-
tion of energy by the action of viscosity is negligibly small. That is, ®
may be neglected.

Applying these observations and assumptions to the energy equation, as
written above, results in the following reduced form:

pC, u£+v£ = ua—p+k—2.
ox ay dx ay

Combining this result with the continuity and momentum equations
results in the following set of equations for buoyancy-driven thermal con-
vection according to the boundary-layer approximation:

w9y (10.11a)
ax ay
u u_ 1dp du
R TH BT -T,
" ax T ay p dx U ay’ i gh( ) (10:116)

ox ay T pC, dx oy (10.11)

i3

In Equation 10.11c, the quantity k = k/pC, is the thermal diffusivity of the
fluid.

Equations 10.11a and 10.11b are the same as Equations 9.2a and 9.2b, except
that the buoyancy term exists in Equation 10.11b. This additional term
involves the local value of the temperature of the fluid, which, in general, is
unknown. This requires the inclusion of the energy equation in order to yield
a closed set of equations. Equation 10.11c is the form of the energy equation
that is consistent with the boundary-layer approximation and that is valid



380 Fundamental Mechanics of Fluids

for moderate temperature differentials from the ambient value. Equations
10.11a through 10.11c are to be solved subject to the no-slip boundary condi-
tion on the surface y = 0 and subject to the condition that the velocity should
be zero far from the heated surface. In addition, either the temperature of the
heated surface or the heat flux on its surface must be specified.

10.4 Vertical Isothermal Surface

In this section, we apply the equations derived above to the flow induced by
a vertical surface that is isothermal at a temperature that is elevated relative
to the ambient. The situation is as depicted in Figure 10.1 in which the tem-
perature of the vertical surface is everywhere T, while the temperature far
from the surface is T, both of which are constants. For such a configuration,
there is negligible pressure gradient in the x direction. From Equations 10.11a
through 10.11c, the mathematical problem to be solved becomes

KL (10.12a)
ox dy
ou  ou  du
Mo T T -T, .
u P +0 o v oy + gPB( o) (10.12b)

T oT  9°T
U——+0 =K. (10.12¢)
x o ay  ay

The boundary conditions that accompany these differential equations are

u(x, 0)=0 (10.13a)
v(x,0) =0 (10.13b)
u(x,y) >0 asy— o (10.13¢)
T(x,0) =T, (10.13d)
Tex,y) = Ty, asy — oo, (10.13¢)

The first two of these conditions are the usual no-slip boundary condi-
tion, whereas the third condition ensures that the effect of the heated surface
does not extend far from the surface. The last two conditions specify that
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the temperature in the fluid is T, at the vertical surface and T, far from the
surface.

In order to facilitate the solution to Equations 10.12a through 10.12¢c, two
changes will be made. First, the stream function y(x, ) will be introduced
as was done in Chapter 9. This will permit Equations 10.12a and 10.12b to
be replaced by a single equation involving the stream function. Second, the
temperature T(x, y) will be replaced by a dimensionless temperature differ-
ence 6(x, y), defined as follows:

0(x, y) - T (;,y)T— To
s~ 40
This dimensionless temperature varies in value from zero, far from the
surface, to unity at the surface. This makes it preferable to the alternate
dimensionless temperature defined by the quantity p(T — T;). In terms of
these new variables, Equations 10.12a through 10.12c become

o 9° np 9* 9°
WA TS gB(T, - 7,0

dy oxdy ox ay®  dy (10.142)

ap a0 dp a0 9%0
ii_i’fﬂj, (10.14b)
dy dx dx dy ay

Following the methods employed in Chapter 9, we now look for a similar-
ity solution to this problem of the following form:

w(x, y) = Cx"fn)

and

6(x, y) = F(n)

where

ey =G
x

where m and n are undetermined exponents, although not necessarily inte-
gers, and C; and C, are constants whose values will be chosen to render the
functions f, F, and n dimensionless. From the definitions of these quantities,
the various derivatives that appear in the differential equations are evalu-
ated as follows:
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0 -
S - cxnf - mmf )

a m-n
% = C,Cx" M f

0%y
axoy

=CCx" " (m-n)f —nnf }

R
ay*

- Clchm—an

3
‘Zy‘f -G f

3—2 = —nx"'mF
z—e =Cx™"F

Y

2

2 ? =Cx™"F .

The primes in the expressions above represent derivatives with respect to
the similarity variable n. Substituting these expressions into Equations 10.14a
and 10.14b produces the following two equations:

GG Hm=n)(f ) -mff }=vCCGx""f +gB(T, - T))F

-mC,x""' f F =xCyx™"F .

If these equations are to be reduced to a pair of ordinary differential equa-
tions, the powers of x on each side of the first equation must be zero, and the
powers of x on each side of the second equation must be equal. That is, the
following relations must be satisfied:

2m-2n-1=0
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Although it is generally not possible to satisfy three conditions with only
two quantities, the three equations above are satisfied by the following val-
ues because two of them are linearly dependent:

Using these values, the ordinary differential equations derived above sim-
plify to the following form:

Cl gB(Ts - TO)

3FF —2(f Y]+ F=0

f e B AT
F+2S g o

4 xC,

Having selected the values for the exponents m and 7 in order to produce
a similarity solution, it is now possible to define explicitly the constants C;
and C, in such a way that the functions f, F, and n are dimensionless. The
quantities available for this purpose are v, g, and K. Then, it is sufficient, from
dimensionality considerations, to define C; and C, as follows:

1/4

8
C, ~ =
2 Uz

However, including dimensionless constants of proportionality in the def-
initions of these two quantities permits two of the coefficients that appear
in the differential equations to be normalized to unity. Noting that the
quantity p(T, — T;) is dimensionless, we define the constants C, and C, as
follows:

1/4
c _v 4gB(T-T)
1~ 4 2
v

1/4
C2 - 4g|3(Ts _TO)

UZ

With this choice of values for the constants C, and C,, the differential equa-
tions for the functions f and F become
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Fr+3ff —2fR+F=0 (10.15a)
F// + 3PrfF/ — 0 (1015b)

In the above, the parameter P, = v/x is the Prandtl number. Numerically,
the Prandtl number is about 0.7 for air and about 7.0 for water. In terms of
the functions f and F, the boundary conditions defined by Equations 10.13a
through 10.13e become

f0)=£0)=0 (10.15¢)
f) —>0asn— e (10.15d)
FO)=1 (10.15¢)

F) — 0asn— e (10.15f)

Once the solution to the ordinary differential system defined by Equations
10.15a through 10.15f has been obtained, the corresponding solutions for the
stream function and the dimensionless temperature are given by the follow-
ing relations:

1/4
Y(x,y) =% w ¥ f(n) (10.16a)
0(x, y) = F(n) (10.16b)
where
1/4
ney)= 28P0=To) Y (10.160)
LY X

The problem defined by Equations 10.15a through 10.15f was solved by
Pohlhausen for P, = 0.733. The physical result of greatest interest is the rate at
which convective heat transfer takes place between the vertical surface and
the ambient fluid. The result so obtained is usually quoted in the following
nondimensional form:

N, = 0.359(G )4 (10.17)

where



Buoyancy-Driven Flows 385

and

_ 813(Ts -Tp)

2
vT,

G,

The parameters in Equation 10.17 are the Nusselt number N,, which is the
nondimensional heat transfer rate, and the Grashof number G,, which is the
nondimensional temperature differential that drives the convection. In these
quantities, 1 is the convective rate of heat transfer per unit area per unit time,
k is the thermal conductivity of the ambient fluid, and / is the length of sur-
face over which the heat transfer takes place.

10.5 Line Source of Heat

Figure 10.2 shows the physical situation that exists when a line source of heat
is immersed in an otherwise stationary fluid. The situation is similar to that
of the vertical surface, except that there is no physical surface involved and
no characteristic temperature differential.

FIGURE 10.2
Thermal convection from a line or point source of heat.



386 Fundamental Mechanics of Fluids

The equations governing the motion that is induced in this situation will
be the same as those of Section 10.4, and they are given by Equations 10.12a
through 10.12c. However, the boundary conditions are different in this case.
Since there is no physical surface in the present case, conditions 10.13a and
10.13d are no longer relevant. The first of these conditions must be replaced
by a statement that the x axis is a line of symmetry and the second condition
by a statement ensuring that the total heat rising from the source is the same
at all streamwise locations. These new conditions are expressed by the fol-
lowing equations:

ou

—(x,0)=0

ay 0
f puC, (T -Ty)dy = Q

where Q is the value of the total amount of heat that leaves the source per
unit time per unit length of source.

With the changes noted above, the problem to be solved consists of the
differential system defined by Equations 10.12a through 10.12c, subject to the
following boundary conditions:

I 0)=0 (10.18a)

ay
o(x, 0) = 0 (10.18b)
f g puC,(T - Ty)dy =Q (10.18¢)
ﬂ(x, 0)=0 (10.18d)

ay
T(x,y) >0 asy— £ oo (10.18¢)

As in the case of the isothermal surface, we recast the differential equa-
tions in terms of the stream function and a dimensionless temperature. The
former is defined to satisfy the continuity equation as before, but the dimen-
sionless temperature has to be redefined. The surface temperature no longer
needs normalizing to unity, so the appropriate definition of the dimension-
less temperature in this case is

0(x, v) = P{T(x, y) — Tp}.
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In terms of the stream function and the new dimensionless temperature,
Equations 10.12a through 10.12¢ reduce to the following form:

oty oty 3y
ki k. YA LA 4N 0
oy axay  ox ay? v oy +g (10.19a)

ap a0 dp 00 9%0
i’i_i’fﬂj_ (10.19b)
ay dx dx dy ay

In order to obtain solutions to these equations, we seek a similarity solu-
tion that is suggested by that found in the previous section, but in which the
dimensionless temperature also has a coefficient that is a function of x. The
form of the solution that is sought is

w(x, y) = Cx"f(n)

where

y

nx,y)=C,—-
X

and

6(x, y) = Cx'F(n).

In these expressions, m, 1, and r are undetermined exponents, and the quan-
tities C,, C,, and C; are constants that render the functions f, F, and n dimension-
less. It is not to be assumed that any of these quantities have the same values as
they did in Section 10.4. The derivatives of the stream function are the same as
those in Section 10.4, and the derivatives of the dimensionless temperature are

®_ Cyx" M (rF —=nnF)
ax

9 _ C,C,x"™"F
9y

2
T8 _cexE
ay
Using these results, Equations 10.19a and 10.19b reduce to the following form:

CiCx™" " H(m-n)(f ) -mff }=vC,Cx""f +gCyx'F

C,CCox™ ™ Nrf F=mfF ) =xC;C;x>"F .
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For a similarity solution to exist, the x dependence in these equations
must cancel. This leads to the following equations relating the exponents
m, n, and r:

2m—-2n—-1=m-3n
2m-2n—-1=r
m-n+r—-1=r-2n.

The first two of these relations come from the momentum equation, while
the last relation comes from the energy equation. It will be seen that the
first and last equations are the same, so that the requirement of reducing
the partial differential equations to ordinary differential equations is met
by satisfying the first two of the equations presented above. Rewriting these
equations shows that the similarity condition is met provided that

1 1
m=z(3+r) n=Z(1—r).

It will be noted that for the special case r = 0, the solution obtained in
Section 10.4 is recovered. In order to determine the value of r for the case
under consideration, the condition given by Equation 10.18c must be invoked.
This condition specifies the following:

Q- [ _puC,(r=Ty)ey

=

np . 6 x"
= —C, ——dn.
pray "BGC, b

In the above, it has been noted that

x" 1
dy="—d "1 _— qdx.
y G, M+ nx C ndx

2
However, the integration indicated above is carried out in a plane x = con-

stant, so that dy will be proportional to x"dn. Substituting the values estab-
lished for the quantities in the integrand produces the result

C o
Q-CC, pﬁ—”xw f f Fdn, (10.20)
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Since the quantity Q should be independent of x, it follows that (m + r)
should be zero. This additional requirement, coupled with the results
obtained above, leads to the following values for the exponents m, 1, and r:

For these values of the exponents, the differential equations for the func-
tions f and F become

f

C
2y ugcié?’F -0 (10.21a)

12

3C

d
* e d (f) 0. (10.21b)

In order to render the functions f, F, and n dimensionless, we choose

8
C ~v =
1 Uz
g 1/5
C ~ 9
2 ’[)2
g -1/5
C ~ 9
3 Uz

As was the case in the previous section, we may include dimensionless
constants of proportionality in the definitions of the preceding quantities.
The purpose of doing this is to simplify the parameters that appear in the
resulting differential equations and boundary conditions. The differential
equations are given above, and the condition relating to the total amount
of heat being Q is given by Equation 10.20. The boundary conditions may
be considerably simplified through normalization, and in this context, it is
noted that the following quantity is dimensionless:

puC,
BQ

With this observation, the following definitions of the constants C,, C,,
and C; are adopted in order to simplify the coefficients in the resulting
problem:
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1/5

-1/5

Using these definitions, the differential equations, as given by Equations
10.21a and 10.21b, reduce to the following form:

(f +3ff -(f)*+F=0 (10.22a)
F +30 3 (fp) 0. (10.22b)
dn

The boundary conditions that accompany this differential system are
given by Equations 10.18a through 10.18e. In terms of the new variables and
parameters, these equations become

f0)=f10)=0 (10.23a)

f " fFRdn=1 (10.23b)
F(0)=0 (10.23¢)

Fm) =0 as 1 — oo (10.23d)

The solutions to Equations 10.22a and 10.22b are of the following form:
f(n) = A tanhan
F(n) = B sech?an.

This form of solution satisfies Equations 10.23a, 10.23c, and 10.23d for all
finite values of the constants A, B, and a. Then, substitution of the assumed
form of solutions into Equations 10.22a, 10.22b, and 10.23b produces restric-
tions on the values of A, B, and a. These restrictions are, respectively,

5 _50

a=—A and B

= A
6 27
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a=23PA
B=3A.
4

These four conditions cannot be satisfied by the constants A, B, and « alone,
and the solution only exists for a particular value of the Prandtl number P,.
The solution to the preceding equations is

1/5

A 8
200
1/5
53 200
4 81
5 81 7
)
6 200
N
18

In summary, a similarity solution has been found for a particular value of
the Prandtl number only, and the solution is as follows:

1/5

BQ g

w(x,y)=6?av oG, o x*% tanh am (10.24a)
5 p'Qt o - /5 atn?

G(x,y)=87x p'Cl g x~"sech®am (10.24b)

where

1/5
n(x,y)—é pi%é xzyﬁ (10.240)
and
1/5

o =§ 2%10 (10.24d)

The above solution is valid for a Prandtl number of P, = 5/18. It shows that
the centerline temperature (T(x, 0) — T) varies as x%/°.
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10.6 Point Source of Heat

A solution analogous to that obtained in Section 10.5 may be obtained for the
case of a point source of heat. The physical situation that will exist is illus-
trated in Figure 10.2, it being understood that in the present case, there will
be angular symmetry about the x axis. In recognition of this fact, the pre-
ferred coordinate system involves circular cylindrical coordinates in which
the coordinates y and z are replaced by R and 6. Under these circumstances,
the coordinate system will be (R, 6, x), which is different from the usual situ-
ation in which the axis of symmetry is the z axis.

Noting that there will be no 6 dependence due to the symmetry already
noted, the governing equations, which are described by Equations 10.12a
through 10.12¢, may be rewritten in terms of the preferred coordinate system
as follows:

9 9
—(Ru)+——(Rug) =0
o R+ o (Rug)

ou ou v d ou
U—+uUg —=—— R— +gB(I -T
ox PR TR o N TSPT-To)

+ Uy =K
ax R R IR IR

oT , T _ 19 T

where u is the velocity in the x direction, and uy; is the velocity in the radial
direction, perpendicular to the x axis. In order to facilitate obtaining a solu-
tion to this set of differential equations, a Stokes stream function and a
dimensionless temperature are introduced as follows:

Ru = a and Ruy = - il
R ax

0=p(T - T,).

In terms of these new dependent variables, the differential equations
quoted above assume the following form:

lop o9 19 lop 0 194 v J o 19
Ao d Loy Loy d 1y _vi 09 1dp +gB(T - Ty)
R 0R 9x R dR R dx o0R R R RAOR 4R R dR

o 9T oy oT 9 poT
OR 9x d9x OR  OR ~ 9R
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The boundary conditions that accompany these differential equations are
the following:

ou o 1oy
S0 = =0
R (x,0) R RaR (10.25a)
1 oy
L0)=-— "5 =0
ug(x,0) R or (10.25b)
f puC,(T - Ty)2nR dR = f p—Cp—2ndR =Q (10.25¢)
0 o OR " P
T 1 08
Z(x,0)=— — =0
3R (x,0) B R (10.25d)
Tx,00 >0 and 6(x,0) >0asR — * o, (10.25¢)

Solutions to the differential system are sought of the following form:

w(x, R) = Cx"f(n)

where
R
nx,R)=C, —
X

and
0(x, R) = C,x"F(n).

Substitution of these assumed forms of solution into the differential sys-
tem shows that a similarity solution exists for the following values of the
exponents m, 1, and r:

m=1 and 4n+r=1
The third equation that is required to define the solution is obtained from

the condition defined by Equation 10.25c. In terms of the new variables, this
condition becomes

C o0
21pC,C, pr"’” fo fFdn=0Q. (10.26)
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Since the quantity Q, the heat leaving the source per unit time, must
be independent of x, the additional requirement is that (i1 + r) = 0. This results
in the following values of the exponents for a similarity solution to exist:

m=1 n=l r=-1.
2

For these values of the exponents m, 11, and r, the differential equations for
the stream function and the dimensionless temperature become

G, d f, 3G
v’ dn mn  WCG

fo-1- nF=0

F+Sfroo.
KN

Also, the values of the constants C,, C,, and C; that preserve the cor-
rect dimensions of the stream function and the dimensionless temperature are

C, ~v

1/6

c,~ &

-1/3

C,~ S

For a point source of heat, as we have here, the quantity Q has the dimen-
sions of quantity of heat per unit time. Then, a dimensionless parameter for
this case is

1/3
pUCP 1)2

BQ g

Then, in order to simplify the coefficients in the differential equations and
in Equation 10.26, we choose the following values for the constants:

C =v
1/4 1/4
c. . PQ 8
2= 2
vap v
c, - BQ
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Using these values for the constants of proportionality in the expressions
for the stream function and the dimensionless temperature, the system
reduces to the following ordinary differential system:

f —(1—f)i lf +nF=0 (10.27a)
dn n

F+pl fF =0. (10.27b)
M

In arriving at Equation 10.27b, the energy equation has been integrated
once, and the boundary condition 10.25d has been employed. The boundary
conditions that accompany these two differential equations are

f0)=f(0)=F(©0)=0 (10.28a)

f fF dn= 1 (10.28b)
0 2

Closed-form solutions to the problem posed above exist, and they will be
explored in the problems at the end of the chapter.

10.7 Stability of Horizontal Layers

When a horizontal layer of fluid is heated from below, or cooled from above,
a buoyancy force exists that can result in convective motion. However, if the
buoyancy force is not sufficiently large, no motion occurs. This situation may
be qualitatively explained as follows.

Consider a horizontal layer of fluid as shown in Figure 10.3. The fluid is at
rest, and heat is passing through the fluid by conduction from the lower sur-
face to the upper surface. For simplicity, the two horizontal surfaces are con-
sidered to be isothermal, although they have different temperatures. Under
these circumstances, the buoyancy force will tend to cause the fluid to rise
from the lower surface, resulting in natural convection.

Suppose that while the fluid is still at rest, a small-amplitude disturbance
is introduced. It may be that the viscous forces that act on the disturbing
motion exceed the buoyancy force, which would cause any convection to
take place. Under these circumstances, the disturbance will decay and the
motion will cease. On the other hand, if the buoyancy force exceeds the vis-
cous forces, the disturbance will grow and convective motion will result.
These observations suggest that a stability analysis of the situation depicted
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FIGURE 10.3
Horizontal layer of fluid heated from below.

could identify the existence of a minimum value of the buoyancy force below
which no motion will exist. The situation described above may be analyzed
quantitatively in a manner similar to that used in Section 9.12, which dealt
with the stability of boundary layers. However, the governing equations will
be different in the current case because of the existence of heat addition. The
relevant equations are also different from those used in the previous few
sections since the disturbance mentioned above will not, in general, satisfy
the assumptions of the boundary layer approximation.

The situation depicted in Figure 10.3 involves heat conduction in a station-
ary fluid. A small-amplitude disturbance is assumed to be introduced into
this situation. The equations that govern the motion involved in this distur-
bance will be unsteady and three-dimensional. Following the Boussinesq
approximation, we consider variations in density to be important only in the
gravitational term. We further consider the fluid properties to be constant,
and in the energy equation, we neglect the viscous dissipation of energy
and the effects of pressure variations in the transfer of energy. Then, using
the density variation defined by Equation 10.3, the equations governing the
motion associated with the disturbance will be

‘u=0

M )u=—i prv u-g[1-BT -T))le,
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In the preceding equations, the pressure is measured relative to its abso-
lute value; that is, it is no longer measured relative to the static value. But,
from Figure 10.3, the static temperature distribution, T,(x), is represented by

L)=T,-(-T,)) .

Before the disturbance is introduced, the velocity vector u in the preceding
equations will be zero. Then, using the temperature distribution specified
above, the equations reduce to the following form:

1 dp, X
0=~ 4o 8 1B (G-T)-(G, T

The pressure distribution that exists in the stationary state has been labeled
p,, and as before, the density is understood to be evaluated at the reference
temperature T,

When the disturbance is introduced, the field variables are assumed to be
perturbed in the following manner:

ux, v,z H)=0+ulx,yzt)
PO Y, 2, 1) = pol) +px, Y, 2, 1)
T, y,z ) =T,(x) + T(x, y, z t).
Here, the primed quantities are, by assumption, small perturbations caused

by the disturbance. Then, products of primed quantities may be neglected.
Thus, the linearized form of these equations is

V-u=0
du 1 5
— = +v ‘u +gpTe
ot "ot ShT e,
o (=T ap
ot h

The pressure may be eliminated from this system of equations by taking
the curl of the momentum equation. Then, it is proposed to take the curl of
the resulting equation and to use the identity

Vx(Vxu)=V(V-u)-Vu'=-Vu’
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in which the continuity equation has been utilized. In this way, the momen-
tum equation becomes

SRR
v ot v ax

The y and z components of velocity may now be eliminated by taking
the dot product of this equation with the unit vector e,. Thus, the problem
reduces to the following two equations:

, 19 2, B . d

The disturbance that is represented by the perturbation quantities #’and T”
is arbitrary in its form. Therefore, it may be represented by Fourier integrals
in the y and z directions. Thus, we represent the velocity and temperature
perturbations by the following expressions:

u (x, y, z, t) =f f u (x’ tki(kyy+k22) dky dkz

T (x/y,Z, i’) =f f 0 (x/ t)ei(kyy+k22) dky dkz

Substituting these expressions into the equations that govern the distur-
bance and using the fact that the result must be valid for all wavelengths of
disturbance result in the following two differential equations:

2 2
O it T ey 8Py
v ot 2 v

Ao NP (I Y I
ax* K ot Kh

where
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We next use the fact that the coefficients in the equations above are con-
stants. Then, we can seek solutions to the differential equations of the fol-
lowing form:

U (x,t) = U(x)e®®/")

0 (x, t) = B(x)e”/",

In the above, the time f has been made dimensionless by dividing it by
the quantity h?/, which is the time required for heat to diffuse across the
fluid layer. Substituting this representation of the disturbance into the two
governing equations gives

D —o?- % (D —a?yu =3P o2 (10.29)
P, vh?
(D*~o?~c)o = _G=T)y (10.29b)
xh
where
o = hk
and
p-nd,
dx

In these equations, a is a dimensionless wave number, and D is the dimen-
sionless derivative with respect to x. Eliminating the temperature 6 between
these two equations yields the following stability equation:

(D* -a*)D*-0a*-0) D*-a’- Pio +a’R, U=0 (10.30)

r

where
R — gh3B(T1 - TZ) .
! KU

The parameter R, = P,G, is the Rayleigh number, where P, is the Prandtl
number, and G, is the Grashof number. G, is a measure of the strength of the
buoyancy force that tries to initiate convective motion.
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For the configuration depicted in Figure 10.3, the boundary conditions
require that the velocity and the temperature perturbations vanish at the
boundaries, x = 0 and x = h. The first of these conditions requires that U = 0,
while the second condition requires that

D? D2-2a2-§ u=0.

7

The latter result follows from Equation 10.29a and the fact that U itself
vanishes at the boundaries. In addition, the no-slip condition at the bound-
aries requires that not only u"but also v”and w’ must vanish on the bound-
aries. With reference to the continuity equation, this condition will be
satisfied if DU vanishes on the boundaries. Putting these boundary con-
ditions together produces the following set of conditions that are to be
satisfied:

U=DU=D? D2—2a2—Pg U=0 on %=0,1. (10.31)

r

The problem posed by Equations 10.30 and 10.31 represents an eigenvalue
problem. For given values of the Rayleigh number R, the wave number of
the disturbance o, and the Prandtl number P,, the eigenvalue will be the time
coefficient 6. That is, for given values of R,, o, and P,, there will be a value of
the quantity o that satisfies the conditions specified above. As the value of
the wave number is varied, different values of o will be obtained. The largest
real value of ¢ will define the Fourier component of the disturbance that is
the fastest growing.

It was stated earlier that there was a qualitative reason to expect that there
was a minimum value of the buoyancy force for convection to start. If this
is so, there will be a minimum value of the Rayleigh number below which
no convection will take place. In order to identify this minimum value, we
note that the situation that will exist in such a case will correspond to the
wavelength of the fastest-growing component for 6 = 0. All other compo-
nents will be decaying. Then, at the onset of instability, the time coefficient
in the preceding equations will be zero. For this situation, Equations 10.30
and 10.31 become

[(D? - 02 + 2R JU = 0 (10.32a)

U=DU=(D*-a’)’U=0 on ~=0,1. (10.32b)

=R

The eigenvalue is now the Rayleigh number R, which can be determined
from the equations above for any given value of the wave number a. Then,
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the minimum value of R, with respect to «, will be the critical Rayleigh num-
ber. This corresponds to the magnitude of the smallest temperature gradient
for which all disturbances, that is, all possible wave numbers, will decay in
time rather than grow in time and produce convective motion.

The problem posed by Equations 10.32a and 10.32b has a solution that
yields a value of 1707.8 for the critical Rayleigh number. When one of the
boundaries is free, the appropriate boundary condition is that the surface
be free of stress. In this case, the value of 1100.7 is obtained for the critical
Rayleigh number. For two free boundaries, the value of the critical Rayleigh
number is 657.5.

PROBLEMS

10.1 A similarity solution exists to the problem posed by a point source
of heat. The solution is of the following form:

(6, R) = Cx £ ()
N, R)=C, X
X

0(x, R) = Csx"F(n).

Carry out the analysis of this solution by substituting the
assumed form of solution into the differential equations for the
stream function and the temperature to verify that a similarity
solution to the equations exists provided that

m=1and 4n +r=1.

The total amount of heat leaving the point source per unit time at
any x location is given by the following relation:

C 0
2mpC,C, pr’"*’fo f Fdn.

Use this result and the fact that the total amount of heat leaving
the point source is a constant to determine the values of the param-
eters m, n, and r.

10.2 The problem posed by convection from a point source of heat was
shown, after using similarity methods, to reduce to the following
problem:
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f —(1—f)i lf +nF=0 (10.27a)
dn m
1
F+P=fF=0 (10.27b)
n
where
f0)=f(0)=F(©0)=0 (10.28a)
and
wf Fdn= - (10.28b)
[ fEdn= . .
Look for a solution to this problem of the following form for the
case P, =1:
2
fop=A-"
a+m
1
Fm)=B———
™ (@+n’)

where the quantities A, B, and 4, are unspecified constants.

(@) Show that the assumed form of solution satisfies Equation
10.27b provided that A = 0.

(b) Show that it also satisfies Equations 10.27a and 10.28b provided
that

N

i=6y2r  and B= V2T
3n

10.3 Show that for a point source of heat in a fluid for which P, =2, a
solution exists in the following form:

nz
a+n

fm)=A

1

=B



Buoyancy-Driven Flows 403

Find the values of the constants A, B, and a that satisfy Equations
10.27a, 10.27b, and 10.28b.

104 The problem of marginal stability of a layer of fluid that is heated
from below is represented by the following equations:

[(D? - 02 + 2R JU = 0 (10.32a)

U=DU=(D*-a??U=0 on % -0,1. (10.32b)

The general solution to these equations is of the following form:
U(x)=Cie" + Coe " + Coe " + Cpe™" + Coe™"" + Cee ",

(@) Find the values of the constants y; that satisfy the differential
equation (Equation 10.32a).

(b) The existence of a nontrivial solution that satisfies the bound-
ary conditions 10.32b leads to a certain determinant being zero.
Find this determinant. It is not required to solve the problem of
setting this determinant equal to zero.

10.5 Replace the boundary conditions defined by Equation 10.32b for
the case of two free surfaces at x = 0 and x = h. That is, find the
equivalent of Equation 10.32b for the case of two free surfaces.

Further Reading—Part III

The topic of laminar viscous flows is fairly well covered in books, particu-
larly the boundary layer section of the material. The following books cover
and extend the material treated in Part III of this book.

Batchelor, G. K.: An Introduction to Fluid Dynamics, Cambridge University Press,
London, 1967.

Gebhart, B., Jaluria, Y., Mahajan, R. L., and Sammakia, B.: Buoyancy-Induced Flows and
Transport, Hemisphere Publishing Corporation, New York, 1988.

Rosenhead, L. (ed.): Laminar Boundary Layers, Oxford University Press, London, 1963.

Schlichting, H.: Boundary-Layer Theory, 6th ed., McGraw-Hill Book Company, New
York, 1968.

Van Dyke, M.: Perturbation Methods in Fluid Dynamics, Academic Press, New York,
1964.

Yih, C.-S.: Fluid Mechanics, McGraw-Hill Book Company, New York, 1969.






Part 1V

Compressible Flow
of Inviscid Fluids

In this part of the book, some phenomena associated with the compress-
ibility of fluids will be discussed, and some methods of solving the govern-
ing equations will be established. In order to do this, the viscosity of the
fluid will again be neglected, but owing to the high speeds associated with
most compressible effects, the inertia of the fluid will be retained. That is, the
fluids under consideration and the flow fields associated with them will be
considered to be such that viscous effects are negligible but such that com-
pressible effects are important.

Part IV of the book encompasses Chapters 11, 12, and 13. Chapter 11 deals
with the propagation of disturbances in compressible fluids and shows how
shock waves are formed. This is followed by a treatment of both normal and
oblique shock waves. Chapter 12 deals with one-dimensional flow situations
and shows how pressure signals react upon reaching interfaces between dif-
ferent fluids and also solid boundaries. Nonadiabatic flows, including heat
addition and friction, are also included. The final chapter, Chapter 13, deals
with multidimensional flow fields, both subsonic and supersonic. These
include the Prandtl-Glauert rule for subsonic flow and Ackeret’s theory for
supersonic flow.
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IV.1 Governing Equations and Boundary Conditions

When the density of the fluid is not constant, the equations of continuity
and momentum conservation are no longer sufficient to permit a solution for
the velocity and pressure fields to be obtained. This is because the density,
which is now a dependent variable, appears in these equations. To close the
system of equations, the conservation of thermal energy must be utilized.
Thus, from Equations 1.3a, 1.9a, and 1.11, the equations governing the motion
of an inviscid fluid in which there are no body forces are

%?+ “(pu)=0 IV.1)
Ju
pg;+mu')u=— 4 (IV.2)
de
0§+Mu'k=—p'u+-w 7). (IV.3a)

In addition, equations of state must be included. These equations will be
of the general form

p=pp 1)
e=e(p, T).

The foregoing set of equations represents seven scalar equations for the
seven unknowns u, p, p, ¢, and T.

Two useful alternative forms of the thermal-energy equation exist. One of
these was derived in the problems at the end of Chapter 3 and is given by
Equation 3.6. This equation, which introduces the enthalpy & of the fluid in
preference to the internal energy e, is

M a e .
pat+mu )h—at+m. y+ -k T). (IV.3b)

It should be noted that Equation IV.3b follows directly from Equation IV.3a
without further approximation. If the form Equation IV.3b is employed, the
caloric equation of state for e should be replaced by the following caloric
equation of state for /:

h = h(p, T).
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The second alternative form of the thermal-energy equation is obtained
from Equation IV.3a as a special case. For situations in which heat conduc-
tion is negligible, Equation IV.3a may be written in the form

De __
Py =P

If, in addition, the fluid is a perfect gas, it follows from the results of ther-
modynamics that

de
=e(T d —=C,.
e=e(T) an T ;

Also
p = pRT.
Thus, the thermal-energy equation may be written in the following form:

DT

C == - _
vat p

It should be noted that De/Dt = (de/dT) DT/Dt = C, DT/Dt, so that the result
above is valid even if C, is not constant.

Using the continuity equation, V - u may be replaced by —(Dp/Dt)/p in the
equation above. Also, T may be replaced by p/(pR) from the thermal equation
of state. Thus, the energy equation may be rewritten as follows:

pC, 1Dp _p Dp _pDp

R pDt p*Dt pDt

1Dp_ R+C, 1Dp
p Dt C p Dt

v

o =
Sis

The last result follows from the thermodynamic relations

C,-C,=R and y=C,/C,
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The thermal-energy equation is now in the form of logarithmic derivatives
that may be combined as follows:

%(logp) = %(IOg pY)

D P
— log— =0
Dt 8 p’
% = constant along each streamline. (IV.3¢0)
P

The foregoing result will be recognized as the isentropic law for thermo-
dynamic processes. This is compatible with the assumptions of an inviscid
fluid in which heat conduction is negligible. The latter assumption means
that the flow is adiabatic, and the absence of viscosity eliminates any irrevers-
ible losses. Equation IV.3c states that the quantity p/p* is constant along each
streamline, which means that the entropy is constant along each streamline.
But if the flow originates in a region where the entropy is constant every-
where, then the constant in Equation IV.3c will be the same from streamline
to streamline. That is, p/p? will be constant everywhere for adiabatic flow of
a perfect gas that originates in an isentropic-flow field or reservoir.

The boundary conditions that accompany the foregoing equations may
specify the velocity and the temperature or the heat flux. Since inviscid
fluids are again being considered, the no-slip boundary condition cannot
be imposed at rigid boundaries as it was in Part III. Rather, the condition
u - n = U, which was used in Part I, must again be employed, for the same
reason as before.
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Shock Waves

This chapter establishes the relationships for shock waves that occur in
supersonic flow. First, the propagation of infinitesimal internal waves is
studied, which establishes the speed of sound. It is then shown how this
acoustical result is modified in the case of finite-amplitude disturbances.
That is, it is shown how nonlinear effects grow to cause a shock wave to
form. The remainder of the chapter is devoted to the study of steady flows
involving standing shock waves.

The famous Rankine-Hugoniot relations for a normal shock wave are first
derived. These relations show, among other things, that the flow through a
shock wave is nonisentropic. From the second law of thermodynamics, it is
then shown that shock waves can occur only in supersonic flow and that,
in the case of a normal shock wave, the downstream Mach number will be
less than unity. This is followed by derivation of the working equations for
both normal shock waves and oblique shock waves. That is, relationships are
established that permit the conditions downstream of a shock wave to be
calculated if the upstream conditions are known and, in the case of oblique
shock waves, the angle of the boundary that is inducing the shock wave, rela-
tive to the flow direction.

11.1 Propagation of Infinitesimal Disturbances

By studying the equations of motion for a small-amplitude internal dis-
turbance in a gas, the speed at which such disturbances propagate may be
established. This speed is, of course, the speed of sound since sound is a
small-amplitude disturbance. Thus, we consider a perfect gas that is origi-
nally at rest and through which a one-dimensional or plane disturbance is
traveling. It will be assumed that this disturbance travels at a sufficiently fast
speed that heat conduction may be neglected. That is, it is assumed that the
flow is adiabatic. Then, from Equations IV, IV.2, and IV.3¢, the fluid vari-
ables must satisfy the following conditions:

p 0
—+—(pu)=0
at+ax(p)

409
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ou ou 1 dp
Y
ot ox p ox

r
p’

= constant.

The flow field under consideration is isentropic, so that the pressure may
be considered to be a function of one thermodynamic variable only, say the
density p. That is, p may be considered to be p(p) only where the particular
function that applies is defined by the energy equation written above. Then,
the pressure term in the Euler equation may be rewritten as follows:

dp _dpop
ox dp ox’

Using this relation, the continuity and momentum equations may be
rewritten as follows:

ad ad ad
—p+u—p+p—u=0
ot ox ox

a—u+ %+1dp6p=0

u
ot dx pdp ox

So far, the preceding equations are exact within the assumptions of one-
dimensional motion of an inviscid fluid in which the flow is adiabatic. In
order to utilize the assumption of a small-amplitude disturbance, the field
variables will now be written in terms of their undisturbed values plus a
perturbation that is caused by the passage of the disturbance. The undis-
turbed velocity is zero, and the undisturbed pressure and density will be
denoted by the constants p, and p, respectively. Then, the instantaneous
field variables may be written as follows:

p=po+p
p=po+p
u=0+u.

Substituting these expressions into the two equations derived above gives

P

ap ou
+u —+(Ee+p)—=0
ot gx TPOFP)
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w w1 dpop
ot ax (pp+p)dp ox

The quantities p’/p,, p’/po, and u’ will be small for a small-amplitude dis-
turbance, and so products of all primed quantities may be neglected as being
quadratically small. The meaning of the statement “u’ is small” will be clari-
fied later. Thus, the linearized form of the foregoing equations is

ap ou
=0
at P ax
w 1 dpodp
ot py dp , ox

It has been considered that the quantity dp/dp has been expanded in a
Taylor series, and the quantity (dp/dp), is the leading term in such an expan-
sion. The meaning of the subscript zero is that the quantity dp/dp should be
evaluated using the properties of the undisturbed gas.

From these equations, it follows that

*u #¥p  dp dp

Poaxat ™" o T dp , ax?

so that the equation to be satisfied by the density perturbation is

ot? dp  ax?

Likewise, by eliminating p’, the equation governing the velocity perturba-
tion is found to be

Pu dp  u

o’ dp

0

Thus, both the density perturbation and the velocity perturbation will have
the same functional form, so that " may be considered to be a function of p’
only. That is, whatever the dependence of p’ is on x and ¢, #’ will have the same
form of dependence, so that a simple relationship must exist between u" and p'.
The foregoing partial differential equations will be recognized as being one-
dimensional wave equations. Thus, the solution for p’ will be of the form

/dp dp
px,t)=f x- | == t +g x+ t
dp 0 dp 0
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where f and g are any differentiable functions of their arguments. The first
term in this expression represents a wave traveling in the positive x direc-
tion with velocity +/(dp / dp), , and the second term represents a wave travel-

ing in the negative x direction with the same velocity. Thus, the speed with
which the density perturbation travels, and also that with which the velocity

perturbation travels, is 4/(dp / dp),. Since the disturbance was assumed to
be small and since sound is a small disturbance, this will be the speed with
which sound travels. That is, if a, denotes the speed of sound in a quiescent

gas, it follows that
e[
dp o

The foregoing result may be put in a different form by evaluating the indi-
cated derivative through use of the isentropic relationship and the ideal-gas
law. From Equation IV.3c

P _Po

[V
dp -1 Po
7=va ro
dp [

.

P

Hence, employing the gas law p = pRT gives

dp = yRT.
dp

Thus, the speed of sound in a quiescent gas may be written as

a0 = \YRT, = /Y% (11.1a)
0

where T is the temperature of the undisturbed gas. This familiar result
shows that the speed of sound in a gas may be considered to be a function
of the temperature of the gas only and that the speed increases as the square
root of the temperature of the gas.

It is now possible to be more precise concerning the assumption made ear-
lier that the perturbation velocity u’ is small. A quantitative interpretation of
this assumption may be obtained from our original linearized form of the
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momentum equation together with the solution just obtained. The linearized
form of the momentum equation that was used above is

2
G aydp _
of  py ox

But it was shown that for a wave traveling in the positive x direction, the
solution for u’ was f(x — a,t), so that

u ou
—=—q X —ayt) =—-a, —
ot of (= aof) * ox

where f’ is the derivative of f with respect to its argument. Thus, the linear-
ized form of the momentum equation may be written in the following form:

2
ou  aydp _,

0 ox Py 0x

or
. _ 4y op.
ox P, 9x

Integrating this equation with respect to x and noting that #’ = 0 when
p’ =0 gives the following algebraic relation between the velocity and density
perturbations:

- (11.1b)

Equation 11.1b shows that the meaning of the assumption “u’ is small” is
that u'/ay <1 since it was already assumed that p’/p, < 1. Equation 11.1b also
exposes the simple relationship between ' and p’ that was deduced to exist.

Another result that may be deduced from Equation 11.1b concerns a funda-
mental difference between compression and expansion waves. For compres-
sion waves, the density perturbation p’ will be positive. Then, Equation 11.1b
shows that the velocity perturbation #’ will also be positive. That is, the fluid
velocity behind a compression wave will be such that the fluid particles tend
to follow the wave, as shown in Figure 11.1a. On the other hand, p’ will be
negative for an expansion wave, so Equation 11.1b shows that u” will also be
negative. That is, the fluid behind an expansion wave will tend to move away
from the wave front, as shown in Figure 11.1b. This fundamental difference
between compression and expansion waves will be discussed further in later
sections.
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(a) (b)

Compression Expansion
wave wave
front front
u' u’
—— > ﬂo — E— ﬂo
p'>0 p':u’:O p'<0 p':u':O

FIGURE 11.1
Fluid velocity induced by (a) compression wave front and (b) expansion wave front.

11.2 Propagation of Finite Disturbances

Consider, again, the passage of a plane wave through an otherwise quies-
cent fluid, but this time, no assumption will be made about the infinitesimal
nature of the wave amplitude. By retaining the effects of finite amplitude, the
phenomena associated with finite-amplitude disturbances may be exposed.
It will be shown that finite-amplitude waves do not propagate undisturbed
and that they form shock waves.

The continuity and momentum equations are the same as those that were
the starting point of the previous section:

9 a d
—p+u—p+p—u=0
ot ox ox

J
o ou__1dp

ot 9x  pox

In the previous section, it was found that, for infinitesimal waves, u was a
function of p only and p was a function of p only. Although the flow will con-
tain finite-amplitude effects here, it will be assumed that u and p will again
be functions of p only. Then, from u = u(p), it follows that

9p _dpou
ot du ot
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dp _dp ou
ax  dudx’

Also, from p = p(p) only,

dp _dp dp ou
ox dpdu ax’

Thus, the continuity and momentum equations may be rewritten in the
following form:

dp ou ou ou
— —+u_— +p_.—=0
du ot ox ax

%wiu_ 1dp dp du
ot dx  pdpduax’
The bracketed term in the first equation also appears in the second equa-
tion and so may be readily eliminated between these two equations. The
resulting relation is

duau_1dpdpau
pdp ox pdpduox’

Canceling ou/dx from this equation and solving for du gives

du== d—p@
\dp p

For convenience, the quantity dp/dp will be denoted by a?, although no
physical significance will be attached to the quantity a at this time. However,
it is known that a—a, as the amplitude becomes infinitesimal. In terms of this
quantity a, the preceding equation becomes
du _ d

a p’

The analogous equation that was obtained in the previous section was
du/a, = dp/p, for a forward-running wave. Thus, in order that the result
obtained here may reduce to the linear result for weak waves, the plus sign
must be associated with a forward-running wave, and the minus sign should
be associated with a backward-running wave. This gives a fluid-particle veloc-
ity that follows a compression wave and moves away from an expansion wave
as before. The foregoing relation shows that for a forward-running wave
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—=— (11.2a)

This result will be used in the momentum equation as follows:
ou ou 1adp
U=
ot x p ox

__1ldpdpiu
p dp du ax

__lapau
P aodx

That is, the momentum equation for a forward-running wave may be writ-
ten in the following form:

%+(u+a)a—u=0.
ot ox

Solutions to this equation are of the form
u(x, ) = flx — ( + a)f] (11.2b)

where fis any differentiable function. It should be noted that in this instance,
both u and a are functions of the two independent variables x and t.

The foregoing solution represents a wave traveling in the positive x direc-
tion with velocity

U=u+a.

The wave speed U may be related to the speed of an infinitesimal wave,
that is, to the speed of sound a,, by use of the isentropic law p/p' = p,/p}.
From the definition of the quantity g, it follows that

a= %
dp
= it B
Po
(y-1)/2
_ e
Po Po
(y-1)/2
o P
Po
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where the definition of the speed of sound has been employed from Equa-
tion 11.1a. Using this result and Equation 11.2a, the local value of the fluid
velocity may be related to the local speed of sound. From Equation 11.2a

du=a%

p

__ % g3
=P dp
Po

where the relation between a and g, established above has been employed.
This equation may be integrated to give
(v-1)/2
U= %pj + constant.
Po (Y - 1) / 2
Using the fact that when u = 0, p = p, shows that the value of the constant
of integration is 2a,/(y — 1), so that the expression for u becomes

(v-1)/2

2
= a—a,).
=)

Here, the relation that was established between a and 4, has again been
used. That is, the quantity a is related to the local fluid velocity u in the fol-
lowing way:

y-1

a=a,+ u.

This result shows that a > a, for u > 0 and that the difference between a and
a, is proportional to the local fluid velocity u. Using this result, the speed
of propagation of a finite-amplitude disturbance may be evaluated. It was
shown from Equation 10.2b that such a disturbance travels with velocity

U(x,t)y=a+u

S U, ) =ay + ;“ L (11.20)

where the relation between a and u established above has been used. Equation
11.2¢ shows that the speed of propagation of a finite-amplitude disturbance
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is greater than the speed of sound for u > 0. That is, the speed of propagation
is no longer constant, and its value depends upon the value of the local fluid
velocity.

Since the propagation speed defined by Equation 11.2c¢ depends upon
both x and ¢, it is not an equilibrium speed. That is, the speed at which a
finite-amplitude signal travels will change continuously according to Equa-
tion 11.2c. It is instructive to deduce the manner in which a given wave
front will change its characteristics as a result of this fact. In time 7, Equa-
tion 11.2¢ shows that a disturbance will travel a distance L that is given by
the expression

u T

Then, relative to an observer who is moving at the speed of sound g4, the
distance traveled by the wave will be

S=y+1

urt.

That is, relative to the observer, the wave will travel a distance that is
dependent upon the magnitude and the sign of the local fluid velocity in
the disturbance. Thus, regions of high local velocity will travel faster than
regions of low local velocity. Then, a smooth disturbance of arbitrary form
will develop as shown in Figure 11.2.

At time 71, a smooth velocity profile is considered to be traveling in the
positive x direction. Then, at some later time t, > 7,, the regions of higher
velocity will have advanced further, relative to an observer moving at con-
stant velocity a,, than the regions of lower velocity. At time t; > 7,, the wave
front is shown to be vertical as higher-velocity regions continue to advance
faster than the slower regions. Finally, at time 7, > 75, the higher-velocity

T Ty T3 Ty

FIGURE 11.2
Progression of finite-amplitude disturbance.
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regions are shown as having overtaken the portion of the signal that is mov-
ing at the sonic velocity a,. It is seen that this is an impossible configuration
since three values of u exist at a given location. Thus, it is concluded that the
wave front will steepen as indicated until the situation depicted at time T
is reached. At this stage, a sharp discontinuity in the field variables exists,
which is called a shock wave. For times greater than t,, this sharp wave front
or shock wave will propagate in an equilibrium configuration.

To summarize, if a smooth, finite-amplitude compression wave is gener-
ated, it will travel in a nonequilibrium configuration. Different parts of the
wave will travel at different speeds in such a way that the wave front will
steepen as it progresses. Eventually, the steepening of the wave front will
reach the point where the changes in velocity, pressure, etc., take place across
a very narrow region. That is, a shock wave has been formed, and this shock
wave will continue to travel at an equilibrium speed.

It should be noted that, in the foregoing argument, the fluid velocity u was
taken to be positive, which corresponds to a compression wave. For an expan-
sion wave, 1 will be negative for a forward-running wave so that, according
to Equation 11.2c, the wave front will move more slowly than the speed of
sound. Also, the more intense parts of the wave move the most slowly, so
that the wave front will spread out rather than steepen. That is, compression
waves steepen as they propagate, but expansion waves spread out.

11.3 Rankine-Hugoniot Equations

In the previous section, it was shown how shock waves develop from finite-
amplitude compression waves. In this section, the variation of some of the
fluid properties across a shock wave will be established. In particular, the
Rankine-Hugoniot equations relate the density ratio across a shock wave to
the pressure ratio and the fluid-velocity ratio.

Shock waves are very thin compared with most macroscopic length scales,
so that they are conveniently approximated as line discontinuities in the fluid
properties. For purposes of analysis, it is convenient to adopt a frame of ref-
erence in which the shock wave is stationary and in which fluid approaches
the shock wave in one state and leaves in another state. Figure 11.3 shows
such a situation in which the incoming velocity, pressure, and density of the
fluid are, respectively, u;, p;, and p,. The corresponding outgoing values are
Uy, p,, and p,. Since the shock wave is oriented normal to the velocity vector,
it is called a normal shock wave.

The quantities u,, p;, py, U, p,, and p, will be related to each other through
the equations of mass, momentum, and energy conservation. Since the shock
wave represents a discontinuity in the fluid properties, differential equa-
tions cannot be used across it. Thus, either the differential equations must be
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FIGURE 11.3
(@) Shock-wave configuration and (b) results from the Rankine-Hugoniot and isentropic
relations.

integrated to yield algebraic equations or the governing equations must be
rederived in algebraic form.

Adopting the latter procedure, the equation of mass conservation may be
readily written down by inspection from Figure 11.3a:

pitty = Paolds. (11.3a)

Multiplying these mass flow rates by the corresponding velocity magni-

tudes gives the change in momentum across the shock wave as p,u; — p;u;.

This change in momentum must be caused by the pressure force p, — p, per
unit area, so that the equation of momentum conservation becomes

plu% +p= pz”% + P2 (11.3b)

Finally, the energy balance may be established as follows: The enthalpy per
unit mass will be

Here, the ideal-gas law has been used together with the identity C, - C, = R.
The total energy per unit mass will be the sum of the kinetic and internal
components, so that the equation of energy conservation is

1
—up o+

) Y P 1u2+ Y P2 (11.30)

v-1p, 27 y-1p,
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In deriving Equation 11.3¢, it has been implicitly assumed that the flow is
adiabatic, although it has not been assumed that it is isentropic. Since shock
waves involve high speeds and since heat conduction is a slow process, the
adiabatic condition is well justified.

Equations 11.3a through 11.3c represent three equations in the six quanti-
ties uy, py, py, Uy, Py, and p,. Hence, two of these quantities may be eliminated,
leaving an equation connecting the remaining four quantities. The quan-
tities u; and u, will be eliminated as follows: Dividing Equation 11.3b by
Equation 11.3a gives

u, + PL_ u, P2
P1th P2l
. P1— P2
Sl — Uy =
P1th

where p,/p,u, has been rewritten as p,/p,u;, which follows from the continu-
ity equation. Multiplying the preceding equation by u, + u; gives

P1—P>

-y =2 14
P1 U

Uy

But, from the continuity equation, u,/u, = p;/p,, so that

1 1
”%‘”f =(p1-p2) —+— .
1 P2

The left-hand side of this equation may be replaced by a function of the
pressures and densities only through use of the energy Equation 11.3c. The
resulting equation is

2 PP =(p, - p») 1.1
y-1 p P P P2

This is the required equation that relates the pressures and densities across
the shock wave. Solving this equation for the density ratio results in the fol-
lowing alternative form of the equation above:

P, pit(r+1)/(y-Dp,

pp (+D)/(y=-Dp+p,

From the continuity equation p,/p, = u,/u,, so that, combining this result
with the equation above, the following conditions will apply across a normal
shock wave:
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P _1+0+ D/ =Dp/py) _ (11.4)

o WD/ =Dp/p

Equations 11.4 are called the Rankine—Hugoniot equations, and they relate
the density ratio across a shock wave to the pressure ratio and the velocity
ratio.

In the derivation of the Rankine-Hugoniot equations, it was not assumed
that the flow was isentropic, and indeed, it will now be shown that it is not
isentropic. If the flow had been isentropic, the density ratio across the shock
wave would have been

1/y

Pr_ P2

P1 P1

Thus, in a plot of log(p,/p;) versus log(p,/p,), the isentropic law will be
a straight line of slope 1/y. The corresponding curve obtained from Equa-
tions 11.4 is a curved line, as shown in Figure 11.3b.

From the foregoing results, it may be concluded that shock waves depart
from the isentropic law unless p,/p, and p,/p, are close to unity. That is,
unless the shock wave is very weak, it will not be isentropic.

11.4 Conditions for Normal Shock Waves

It will be shown in this section that, as a consequence of the second law of
thermodynamics, only that portion of Figure 11.3b that lies in the first quad-
rant has physical significance. This restriction will be shown to result in the
requirement that the upstream Mach number M, must be greater than unity
for shock waves to occur, and the resulting downstream Mach number M,
will be less than unity.

Using the results from thermodynamics that are quoted in Appendix E,
the entropy difference across a shock wave s, — s, will be given by

s, -5, =C,log % -Rlog Z—z .
1 1

Using the ideal-gas law, the temperature ratio in the equation above may
be eliminated in favor of the pressure and density ratios. Thus, the entropy
change may be rewritten as follows:
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$, =5, =C,log %% -Rlog %

P2 P2
=(C,-R)log == -C, log ~= .
g 8 P ro8 P1

But C, - R = C,, so that the entropy change across the shock wave, which
will be denoted as As, may be evaluated from the following equation:

é£=log P2 -ylog P .
v 1 1

Using the preceding result, the entropy change and the density ratio will
be compared, for a given pressure ratio, for two processes. The first process
will be a shock wave that must obey the Rankine-Hugoniot equations, and
the second process will be a hypothetical isentropic one for the same pres-
sure ratio as the shock wave. Then, from the equation above, the entropy
changes in each of these processes will be

el =log P2 -y log P2
Co rn P1 P1 rey
P2 P2
O=log = -ylog =
P P1

where the subscript R-H indicates the entropy change and the density ratio for
a shock wave in which the pressure ratio is p,/p;, and the subscript I denotes
the density ratio for an isentropic process that spans the same pressure ratio.
Subtracting these two equations to eliminate the common pressure ratio gives

A =y log L -log P2

v R-H 1 P ren

But the second law of thermodynamics requires that As > 0, so that

log Pz = log Ll

pl I pl R*H‘

Figure 11.3b shows that this inequality can be satisfied only in the first
quadrant of the diagram, which corresponds to log(p,/p;) > 0 and log(p,/p:)
> 0. That is, in order to satisfy the second law of thermodynamics
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P2 o9, (11.5a)
P1

This means that the gas must be compressed as it goes through a shock
wave. The continuity equation then shows that

Mg (11.5b)
U,

so that the fluid is slowed down as it passes through a shock wave.

Conditions 11.5a and 11.5b may be put into the more meaningful condition
M, = 1. In order to achieve this alternative formulation, it is first necessary to
derive a relationship that is known as the Prandtl or Meyer relation. In deriv-
ing this region, the subscript " will be used to denote the value of a variable
when M = u/a = 1, where u is the fluid velocity, and a is the local value of the
speed of sound. Then, it follows that u. = a..

The starting point in the derivation is the equation obtained by dividing
the momentum equation 11.3b by the continuity equation 11.3a:

P1 P2

=u, +
P1ty P2l

U, +

Using the definition of the speed of sound to introduce a; = yp;/p, and
a; = 1p,/p, gives

ay a;
U+ =1 +—=
1 YU,
2 2
_n M
Uy =ty = —> -
Y, YUy

The right-hand side of this equation may be replaced by an equivalent
expression that is obtained from the energy equation in the following
form:

a,  y+1 2
y-1 2 y-1 2(y-1) "

where the fact that u;. = 1, = a;. = a,. = a. has been used. Thus, the velocity
difference 1, — u, may be rewritten in the following form:

" 1 y+1 y-1
P, 2 2 yu, 2 -2
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This equation simplifies considerably to the form
u, = a? (11.6)

which is the Prandtl or Meyer relation.

The preceding result will be used in the conditions that were established
for a normal shock wave to obtain an alternative form of these conditions.
Multiplying both the numerator and the denominator of the inequality 11.5b
by u, gives

=1
Uty
Then, using Equation 11.6
u—lz =1
a;

The left-hand side of this inequality may be evaluated from the energy
equation as follows:

u  al v+1

+ = a:.
2 y-1 2(y-1

Dividing this equation by 1] gives

1 1 y+1 a
y=1M7 2(v-1Duf

1
—+
2

cu_ (y+DM;
B

a2 2+(y-1DM?
Substituting this expression into the condition for a shock wave gives

(v + DM
2+(y -1M?

which reduces to

M, >1. (11.7a)
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That is, a shock wave can occur only if the incoming flow is supersonic.
Furthermore, in view of the Prandtl or Meyer relation (Equation 11.6), the
inequality 11.7a implies that

M,<1. (11.7b)

To summarize, in order that the second law of thermodynamics is not
violated, a normal shock wave can occur only in supersonic flow, and the
resulting downstream flow field will be subsonic. That is, the fluid will be
compressed as it passes through the shock wave.

11.5 Normal-Shock-Wave Equations

The results of the last two sections were intended to establish the funda-
mental phenomena of shock waves and the principal consequences of the
existence of shock waves. However, the relationships established in these
sections are not suitable for evaluating the conditions downstream of a
shock wave in terms of the upstream conditions. It will be recalled that
the three conservation equations connect six quantities, three upstream
values and three downstream values. Then, it should be possible to elimi-
nate any two of the downstream conditions and so obtain an equation
that relates the remaining downstream condition to the three upstream
conditions. In this way, equations may be established for each of the
downstream quantities in terms of the upstream conditions, which are
presumably known. Rather than considering the velocity to be one of the
quantities, the Mach number M will be considered. Thus, for supersonic
flow in which a shock wave exists, the known quantities may be consid-
ered to be p;, p;, and M;, while the unknown downstream quantities will
be p,, p,, and M,.

To evaluate the downstream Mach number M,, the energy equation
involving the upstream conditions and the sonic conditions is employed
as follows:

aj y+1

= a*
2 y=-1 2(v-1

and hence

@ _,v-11 1
@+l 27 (g-pME
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Similarly, from the energy equation involving the downstream conditions
and the sonic conditions, it follows that

2
a—*=2y_1 l+ ! .
w; v+l 2 (y-1)M;

These expressions will be used in the Prandtl or Meyer equation (Equa-
tion 11.6) as follows:

v-1 1+ 2 > 1+ 2 5
y+1 (v -DM; (v =DM,

Solving this equation for M, gives

2
M% _ 1+[(2Y -1)/2]M; _ (11.8)
yMi-(y-1)/2
That is, the downstream Mach number is a function only of the upstream
Mach number and the specific-heat ratio of the gas. The variation of M, with
M,, as defined by Equation 11.8a, is shown schematically in Figure 11.4a. It will
be seen that as the upstream Mach number increases, the downstream Mach

number decreases. As M, — o, Equation 11.8a shows that M; — (y - 1)/2y,
which defines the asymptotic limit.

() (b) (©)

M P2
2 P1
y+1 %
1.0 y-1[ T _— 1
Y_l\
o [T T T T T T
0 1.0 1.0
1.0 M 1.0 M, 1.0 M,
FIGURE 11.4

Conditions downstream of a normal shock wave: (a) Mach number, (b) density, and (c) pressure.
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The density ratio across the shock wave will be obtained by first evaluating
the velocity ratio and then using the continuity equation. From Equation 11.6

where the last equality was established earlier in this section. Simplifying
the right-hand side of this equation gives

u, (y-1HM;+2
Uy (y + DM}

But, from the continuity equation, u,/u; = p,/p,, so that the expression for
the density ratio across the shock wave is

2

P (e DM (11.8b)
P (y-DMj +2

The form of the density ratio as a function of the upstream Mach num-
ber is shown in Figure 11.4b. The density ratio is a monotonically increasing
function of M, and reaches an asymptote that, as shown by Equation 11.8b,
is (y + )/(y - 1).

The pressure ratio across a normal shock wave may be readily evaluated
from the Rankine-Hugoniot equations (Equations 11.4) and the density ratio
as given by Equation 11.8b. This gives

L+[(y+ /(= Dlpa/py) _ (v + DM}

Y+ -D+p/pn (-DMF+2

Solving this equation for the pressure ratio gives

Prq, 2T 2oy, (11.80)
P y+1

The form of this result is shown in Figure 11.4c. It will be seen from this
curve, and it may be verified from Equation 11.8c, that the pressure ratio
increases without limit as the upstream Mach number increases.

The foregoing relations (Equations 11.8a through 11.8¢) give each of the
principal downstream quantities in terms of the upstream Mach number
and the specific-heat ratio of the gas. The functional form of these results is
shown qualitatively in Figure 11.4, and quantitative data may be obtained
from tables and figures that appear in the references at the end of this part
of the book.
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11.6 Oblique Shock Waves

Oblique shock waves are shock waves that are inclined to the free stream
at an angle different from n/2. Such a shock wave is shown in Figure 11.5,
in which both the incoming and outgoing velocities have been decomposed
into components that are perpendicular to the shock wave and those that
are parallel to the shock wave. The shock wave is inclined at an angle p to
the incoming flow direction, and the velocity vector is deflected through an
angle 8 by the shock wave.

The components of the velocity vectors that are normal to the shock wave
are u, sin p and u,sin (p — d) for the incoming and outgoing flow, respectively.
These velocity components must obey the normal-shock-wave equations, so
that

U, sin (B —8) <u, sin .

On the other hand, the tangential-velocity components must be equal since
there is no pressure differential or other force acting in the tangential direc-
tion. This shrinking of the normal-velocity component and preservation of
the tangential-velocity component result in the downstream velocity vector
i, being bent toward the shock wave, as shown in Figure 11.5.

The equations that determine the downstream values of the pressure, den-
sity, and Mach number need not be established from the governing equa-
tions in the same manner as was done in the previous section. Rather, the
observations that have already been made regarding normal- and tangential-
velocity components may be utilized in conjunction with the normal-shock-
wave equations. Since the upstream normal-velocity component is now
u, sin p rather than u;, the upstream Mach number M, should be replaced by

My, p1 P1

!
AN
AN
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j&’/}\\ )
N S &E
N 7N

FIGURE 11.5
Configuration of an oblique shock wave.
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M, sin B in Equations 11.8a through 11.8c. Likewise, the downstream Mach
number M, should be replaced by M, sin(p — &). Thus, the equations for the
downstream Mach number, density, and pressure become

1+[(y - 1)/2]M? sin* B

20200 8y _
M s (B =0) = e sinp - (y - 1)/2

P, (y+1)Misin’p
pr (Y =DM sin®p+2

P21, 2 (MPsin?B-1).
P y+1

The foregoing equations express M,, p,, and p, in terms of M,, , and &.
Although M, is usually known, only one of the angles p and § is typically
known. If the shock wave is generated by the leading edge of a body, the
angle 6 will be known since the downstream velocity vector must be tangent
to the surface of the body. Then, the angle f is typically the unknown quan-
tity. However, one more equation exists to close the system of equations. The
conservation equations have been applied only to the normal components of
the upstream and downstream velocity vectors. Since there are no forces act-
ing along the shock wave, the conservation of mass and momentum in that
direction are satisfied by equating the components of the velocity vectors in
the tangential direction. This gives

U, cos = u, cos (f — d)
and hence

U cos(B-9)
U, - cosp

Since this equation is supposed to determine the shock-wave angle f, the
velocity ratio should be eliminated in favor of known quantities. The conti-
nuity equation, which involves the normal-velocity components, gives

ﬂ=p72sin([3—6).
u p; sinf

Equating these two expressions for the velocity ratio results in the identity

P _ tanp
p; tan(B-9) '
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Now, the density ratio may be eliminated from the results that were deduced
above from the normal-shock-wave equations. The result is the relation

(y + )M? sin’ B _ tanfp
(y-DM:sin*f+2 tan(B-9) "

This equation is sufficient to determine the angle p since both M, and &
are known. However, the result is an implicit expression for f§ rather than
an explicit expression. Although the equation is not readily rearranged to
express f in terms of M, and §, it is possible to solve for M, in terms of  and
8. Solving directly for M, gives

2 _ 2tanf
" sin?B [(y + ) tan(B-8)— (y - I)tanp]

This result may be simplified by first rearranging the numerator and
denominator to give

M2 = 2 cos(p - 0)
' sinB [(y + 1)sin(B - 8) cosp - (y - 1)sin P cos(B - 8)] °

Next, the trigonometric identities for multiple-angled functions may be
employed to reduce the expression to the following form:

2 2 cos(p - 9)
' sinB[sin(2B - 8) - y sind]

(11.9a)

Equation 11.9a connects three quantities, two of which will be known in any
flow configuration. The form of the solution represented by Equation 11.9a
is shown in Figure 11.6a. These results show that, for given values of M, and
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FIGURE 11.6
Oblique-shock-wave relations: (a) shock-wave inclination p, (b) downstream Mach number M,,
and (c) pressure ratio across the shock wave.
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the deflection angle 8, two shock-wave angles f are possible. The limiting
values of p may be established by recalling the condition for a normal shock
wave, M, > 1, which here becomes M, sin § > 1. Then, p must lie in the follow-
ing range:

. !
sin M, =f= 5 (11.9b)
where the upper limit corresponds to a normal shock wave. The lower limit
will be recognized as the angle of a Mach wave, that is, the angle to the lead-
ing edge of a sound wave that is being continuously emitted by a source of
sound in which the source is moving with Mach number M,. Mach waves, of
course, represent the sonic end of the shock-wave spectrum, so that the pres-
sure ratio and the density ratio across Mach waves are unity. On the other
hand, normal shock waves exhibit the maximum pressure and density ratio
for a given approach Mach number. These observations lead to the classifica-
tion of oblique shock waves as being either strong (if the value of p is close to
n/2) or weak [if the value of f is close to sin™ (1/M,)]. It will be shown shortly
that the downstream flow is subsonic for a strong shock wave and super-
sonic for weak shock waves. The dotted line in Figure 11.6a corresponds to
M, =1, which does not coincide with the minimum value of M, for fixed J,
although these two values do not differ substantially.

The value of the downstream Mach number may be obtained from the
equations that were already deduced from the normal-shock-wave equa-
tions. The expression is

. 1+[(y-1)/2]M}sin’p
* sin?(B-8) [yM2sin®p-(y-1)/2]

(11.9¢)

Since M, and & will be known from the problem definition and since f
will be known from Equation 11.9a, the value of M, may be established from
Equation 11.9c. The results of such a solution are shown schematically in
Figure 11.6b. The figure clearly illustrates the possibilities of having either
subsonic or supersonic flow downstream of the shock wave. In the case
of normal shock waves, it was found that the downstream flow had to be
subsonic, but for oblique shock waves, the unaffected tangential-velocity
component, when added to the subsonic normal component, may result in
supersonic flow, particularly for shallow angles .

The expression for the pressure ratio across an oblique shock wave was
also deduced from the normal-shock-wave equations and was shown to be

P14 2T (MPsin?B-1). (11.9d)
P y+1
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The form of the curves that are generated from this equation is shown in
Figure 11.6¢c. This diagram brings out the significance of the terminology
“strong” and “weak” as applied to shock waves. The strength of a shock wave
is defined by the nondimensional pressure difference (p, — p;)/p;, which is
seen to be larger for the strong shock waves than for the weak shock waves.

The downstream Mach number and pressure ratio are two quantities of
principal interest in shock-wave flows. The equation for the density ratio was
also deduced from the normal-shock equations and was shown to be

[ (y + )M? sin’ B
P (Y -DMisin®f+2

(11.9¢)

The foregoing equations are sufficient to completely determine the condi-
tions downstream of an oblique shock wave, provided that the type of shock
wave is known (i.e., strong or weak). There is no mathematical criterion for
determining whether the shock wave will belong to the strong family or the
weak family. The configuration that will be adopted by nature depends on
the geometry of the projectile or boundary inducing the shock wave.

Figure 11.7 shows two different shapes of leading edge that are considered
to be immersed in the same supersonic flow field. For the blunt-nosed body;,
the boundary condition on the solid surface requires that the velocity vec-
tor be close to the vertical in the vicinity of the front stagnation point. This
boundary condition may be realized only if a detached shock wave exists in
front of the body as shown. Since the angle f is close to n/2 for this shock
wave, it will be of the strong variety, so that the downstream Mach number
will be less than unity. The corresponding subsonic flow may then satisfy

M;>1

FIGURE 11.7
Supersonic flow approaching a blunt-nosed body and a sharp-nosed body.
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the required boundary condition in the usual way. Moving away from
the front stagnation point along the surface of the body, the angle 5 of the
downstream velocity vector is continuously changing. Thus, some point is
eventually reached where the value of § is such that matching the boundary
condition by deflecting the flow through a weak shock wave is possible. The
shock wave will therefore bend back with the flow far from the body so that
the downstream flow becomes supersonic. Thus, a region of subsonic flow
will exist in the vicinity of the nose of the body, and the rest of the flow field
will be supersonic.

In the case of a sharp-nosed slender body, an attached shock wave will
exist, as shown in Figure 11.7. With this configuration, the velocity vector
will be deflected by the shock wave through just the correct angle to sat-
isfy the boundary condition that the surface be a streamline. Since the shock
wave will belong to the weak family in this case, the flow downstream of the
shock wave will remain supersonic.

PROBLEMS

111 In general, the enthalpy / depends on both the pressure and the
temperature, that is, h = h(p, T). However, if p = pRT, it follows that
h = h(T) only. To show this, obtain the first law of thermodynamics
in the form

Tds=dh-vdp

where v is the specific volume of the gas. Then, by considering s =
s(p,T) and h = h(p,T), show that

as 1o
oT T dT

o5 _ 1 oh

ap T ap

These are the reciprocity relations that are quoted in Appen-
dix E. By eliminating s from these equations and utilizing the gas
law p = pRT, show that oh/dp = 0 so that h = h(T) only.

11.2 In general, the internal energy e depends on both the specific vol-
ume and the temperature, so that e = e(v, T). Show that if p = pRT, it
follows that e = e(T) only.

11.3 Show that, for a calorically perfect gas, the entropy change involved
in some event may be related to the temperature ratio and the
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pressure ratio, or to the temperature ratio and the density ratio, by
the following expressions:

§-5,=C, logTZ—Rlogﬁ
0

Po
T P
=C,log— +Rlog>.
T, p

114 The equation governing the fluid velocity induced by a finite-
amplitude forward-running disturbance was shown to be

ad d
—u+(u+a)—u=0
ot x

in which both # and 2 depend on both x and . Show, by direct sub-
stitution, that

u=flx—@u+a)t]

is the general solution to this equation, where fis any differentiable
function.

11.5 The equation to be solved for u in Problem 11.4 is

Du_,
Dt
where
B=—+(u+ ) —
Dt ot 0.
and
29
dp

Show that the steepness of the wave du/dx satisfies a relationship of
the form:
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and find the constant of proportionality. If the steepness of the
wave front at time ¢ = 0 is denoted by

u

axo

=S

find the time required for du/dx to become infinite, and thus show
that S must be negative for a shock wave to form.

11.6 The entropy increase across a shock wave may be calculated from
the following expression:

Y
P2 P1

b5 _ log
G, P1 P2

Using the normal-shock-wave equations, express As/C, as a function
of M, and y only. Denoting (M7 - 1) by ¢, express As/C, as the sum of
three terms, each of which has the form (1 + ae), where « is a function
of y only. Expand the result for small values of &, and hence show that
As/C, ~ €3, which shows that weak shock waves are almost isentropic.

11.7 A normal shock wave occurs in a fluid that is not a perfect gas and
for which the pressure and the density are related by the following
expression:

where c is a constant.

(@) Using the continuity and momentum equations, together with
the foregoing relation and the general expression for the speed
of sound, show that the upstream and downstream Mach num-
bers are related as follows:

M2 2 2
logﬁl2 =M -M;.

2

(b) The pressure across the shock wave (p, — p;) can, in principle, be
expressed in terms of M; and c. The relation is implicit rather
than explicit, but it can be solved for M as a function of (p, - p,)
and c. Find this expression.
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11.8 The equation to be solved in the propagation of sound waves is the
same as that to be solved for shallow-liquid waves. This leads to an
analogy between sound waves in a gaseous medium and waves
on the surface of a liquid. Find the corresponding physical quanti-
ties in this analogy, and find the value of y that makes the analogy
complete.

119 The equations governing a wave that is approximately one-dimen-
sional are as follows:

o
ax dy

u ou ou 1 ap

+U
ot ax ay p dx

Lop
p dy
Look for a progressive-wave solution to these equations in which the

pressure p(x, t) is dependent on x and ¢ only, and the velocity compo-
nents follow the same x and f dependence as indicated below:

g
ot 0x

u(x,y,t)=U(p,y)

J
o(x,y,t) = V(P,y)ﬁ.

In the above, the wave speed c(p) is considered to be a function of
the pressure p. Without linearizing, determine the equations to be
satisfied by the functions U(p,y) and V(p,y).

Look for a similarity solution to the equations obtained above in
the following form:

Ulp, y) = p*U*(y)
Vip, y) = p™/2VX(y)

C(P) — pl/zc*

where C* is a constant. Find the equations to be satisfied by U*(p,y)
and V*(py).
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One-Dimensional Flows

This chapter deals with flow fields that are essentially one-dimensional
and that are compressible, either subsonic or supersonic. Most of the topics
considered involve sonic flow and so constitute a continuation of the topics
treated in the previous chapter.

The topic of weak shock waves or sonic waves is treated from a general
viewpoint by means of Riemann invariants. In this way, the manner in
which acoustic waves react in various situations is established. Particular
situations that are treated include the release of waves in a shock tube, the
reflection of waves at a solid boundary, reflection and refraction of waves at
the interface of two gases, and waves generated by a moving piston. In order
to show the quantitative differences due to finite-strength waves, the non-
linearized shock-tube problem is also treated. Nonadiabatic flows are also
treated through the technique of influence coefficients. This allows not only
heat addition but also friction and area changes to be handled. Finally, the
flow through convergent-divergent nozzles is treated.

12.1 Weak Waves

The topic of weak shock waves or acoustic waves will be further investigated in
this section. The Riemann invariants for the governing equations will be estab-
lished, which permits the treatment of general problems involving weak waves.

It was shown in Chapter 11 that weak waves are isentropic, so that p may
be considered to be a function of p only. Thus

The continuity and momentum equations for a plane wave may therefore
be written in the following form:

d J d
P, 0
ot ax ox

=0

ou ou 2 0P
pP—+uU—=-a"—.
ot dax 0x

439
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For a fluid that is originally at rest before the wave passes through it, the
density, pressure, and velocity may be written as their quiescent values plus
a perturbation. That is,

p=po+p
p=po+p
u=0+u

where, for a weak wave, p'/p, < 1, p'/p, < 1, and u'/a, <« 1, where 4, is the
speed of sound in the undisturbed gas. Thus, the linearized form of these
equations, which will describe weak waves, is

o

ot P o T

dx
ou , 0P
—+a; ——=0.
Po"ar T o
Since p, is a constant, it may be added to p’ when it appears inside a deriva-
tive. Thus, the equations above may be rewritten in the following form:

9 9
E(po+p)+poa(0+u)=0

) 9
905(04'”)"'”35(90“'9):0-

Using, again, the expansions for p and u shows that

ap ou
—+p—=0
at Py

ou 5, 0p
—+a;——=0.
Poar T ax

The quantities inside the differential operators will now be nondimension-
alized by dividing the first equation by p, and the second equation by pa,.
This gives

0P +aoi oo
ot Py 0x a,

u ad
— — +ay— P
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Finally, the desired form of the governing equations for weak waves is
obtained by first adding, then subtracting, these two equations:

K L A
ot a, o ox a, Po

u p a u p
— == —gy— ——— =0.
o a, Py X a, Po

Both these equations are of the form of a material derivative of some quan-
tity being zero. The material derivative is one in which the convection veloc-
ity is the speed of sound, and in the first equation, the convection is in the
positive x direction. Then, integrating these two equations gives

u_p

— + — = constant along x - a,t = constant (12.1a)
a4y Po
L _ P _ constant along x + a,t = constant. (12.1b)
a9 Po

The lines x — a,t and x + a,t are called the characteristics, and the quantities
u/ay + p/py and u/a, — p/p,, which are constant along the characteristic lines,
are called Riemann invariants. Figure 12.1a shows the characteristics that pass
through a typical x location and the Riemann invariants for these character-
istics. It will be noted that one of the characteristics is forward-running and
the other is backward-running.

The Riemann invariants may be expressed in terms of the pressure and the
velocity rather than the density and the velocity. Depending on the problem
being considered, this alternative formulation may be desirable. To obtain
the alternative formulation, the density ratio is replaced by the pressure ratio
through use of the isentropic gas law as follows:

P _Po
P P
Y
r_ P
Po Po
Y
= 1+p
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FIGURE 12.1

(@) Characteristics and Riemann invariants in the xt plane and (b) basis of evaluating the field
variables at an arbitrary point P.

Since p’/p, < 1, this expression may be linearized to give

£=1+yp—.

Po Po

In order to eliminate the density ratio from the Riemann invariants, this

expression must be rearranged to yield the density ratio in the form p/p,.
This may be done as follows:

p_1 P 4
Po Y Po
but
£=1+p
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and hence

PPy

Po Y Po
_1p  v-1
Y Po Y

That is, the density ratio p/p, may be replaced by the pressure ratio as indi-
cated so that, from Equations 12.1a and 12.1b, the Riemann invariants may
be written in the form

Ly 1pr_ constant along x - a,t = constant (12.10)
4 Y Po
w_1lpr = constant along x + a,t = constant. (12.1d)
4 Y Po

Figure 12.1a shows the two characteristics that pass through a typical point
(x, 0) in the xt plane and the alternative Riemann invariants along these
characteristics.

Equations 12.1a through 12.1d may be used to evaluate the velocity, the
density, and the pressure at any value of x and any value of t if the values of u,
p, and p are known as functions of x at some time such as t = 0. The manner in
which this is done may be explained with reference to Figure 12.1b. A typical
point P(x, t) is shown in the xt plane together with the two characteristics that
originate along the t = 0 axis and that pass through the point P. Associated
with these two characteristics are Riemann invariants whose constants may
be evaluated from the known conditions at t = 0. Then, at the point P, the
Riemann invariants for 1 and p provide two algebraic equations for the two
unknowns. Alternatively, the Riemann invariants for u and p provide two
algebraic equations for these two unknowns. The following sections will uti-
lize this approach to obtain solutions for particular flow situations.

12.2 Weak Shock Tubes

The first application of the foregoing theory will be made to a shock tube
in which a weak wave is released. A shock tube is a relatively long tube
fitted with a diaphragm, as shown in Figure 12.2a. The gas on one side of
the diaphragm is maintained at a pressure different from that on the other
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FIGURE 12.2

(@) Shock tube, (b) initial pressure distribution, (c) xt diagram, and (d) typical pressure distri-
bution for t > 0.

side. In general, the gases on either side of the diaphragm may be different
and so may have different properties and states. In this instance, the gases
will be considered to be the same, and only the states are assumed to differ.
The initial pressure distribution, which is taken to be an equilibrium state,
is shown in Figure 12.2b. The diaphragm may be designed to burst at some
predetermined value of the pressure p;,. A pressure wave is thus released
from the vicinity of the diaphragm as the two regions tend to equalize their
pressures. The problem is to determine the pressure and the velocity in the
gas at any location and at any time.

The xt diagram for the shock tube is shown in Figure 12.2c. The time at
which the diaphragm bursts is taken to be ¢ = 0 for convenience, and the loca-
tion of the diaphragm is chosen to be x = 0. Then, a compression wave will
emanate from the origin and will travel into the lower-pressure region, while
an expansion wave will emanate from the origin and travel into the region
of higher pressure as indicated. Since weak waves are being considered, the
two waves mentioned will travel at the speed of sound a,. Then, the slopes
of the waves in the xt plane are a, and — g, for the compression wave and the
expansion wave, respectively. The xt diagram in Figure 12.2c is divided into
three regions that are defined by the waves emanating from the origin of the
diagram. Region 1 is that portion of the positive x axis that has not yet been
affected by the oncoming compression wave. In this region, the velocity is
zero and the pressure is p,. Region 2 is defined as that portion of the negative
x axis that has not been influenced by the expansion wave. Here, u = 0 and
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p = p;- The third region, denoted region 3, is that part of the x axis that has
been influenced by the compression wave and the expansion wave. Since the
pressure and the velocity must be continuous across x = 0, both the positive
and negative portions of the x axis in region 3 will experience the same pres-
sure and velocity.

In order to determine the pressure and the velocity in region 3, an arbi-
trary point P is considered, as shown in Figure 12.2c. The two characteristics
that originate on the x axis at t = 0 and pass through the point P are indi-
cated and are denoted by & = constant and n = constant. The values of the
Riemann invariants along these characteristics may be determined from the
known distributions along the x axis at ¢ = 0. Thus, along the characteristic & =
constant, Equation 12.1c shows that

w, 1lp _1p
a4y Y Po Y Po .

Here, the fact that u = 0 and p = p, at t = 0 for x < 0 has been used. From
1 = constant:

u

) YPO_

1
Y

The fact that u = 0 and p = p, at t = 0 along the positive x axis has been used
here. The solution to these two algebraic equations shows that the velocity
and the pressure behind the compression and expansion waves are

L (12.2a)
ay 2y Po
p_lpg (12.2b)
Po 2 Po

Equation 12.2a shows that for p,/p, > 1, u/a, > 0, so that the fluid moves along
the positive x axis. This agrees with our previous finding that the fluid parti-
cles tend to follow compression waves and move away from expansion waves.
Equation 12.2b shows that the pressure in region 3 is the arithmetic mean of
the pressures in regions 1 and 3. The pressure distribution along the tube is
shown for some time ¢ > 0 in Figure 12.2d. This figure illustrates that a com-
pression wave of amplitude (p; — p,)/2 moves along the positive x axis with
speed a,, whereas an expansion wave of the same amplitude moves along the
negative x axis at the same speed. The expansion waves may be considered
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to be a discontinuity since only weak waves are being considered. The analo-
gous problem for finite-strength waves will be considered in a later section.
The analysis that is presented above may be used to solve a variety of prob-
lems involving weak waves interacting with walls, interfaces with other flu-
ids, etc. The section title “Weak Shock Tubes” appear to be self-contradictory
in the sense that the words “weak” and “shock” are used together. However,
the physical system described above is an excellent model for determin-
ing the physics of what happens when weak waves, such as sound waves,
encounter the situations that will be discussed in the following sections.

12.3 Wall Reflection of Waves

The behavior of a weak pressure wave when it strikes a solid boundary will
be established in this section. This will be done by considering a shock tube
that has a closed end so that the wave that travels along the tube will impinge
upon it. In this way;, it will be shown that a compression wave is reflected by
a wall as a compression wave of the same strength, and an expansion wave
is likewise reflected as an identical expansion wave.

Figure 12.3a shows a shock tube similar to that which was considered in
the previous section, except that the tube is closed at one end. The xt diagram

(a) (c)

P

Diaphragm %(pl + pg)
N B =3
end Po

N N

P

1 Motion
21+ Po)

FIGURE 12.3
(a) Shock tube, (b) xt diagram; (c) pressure distribution at some time and (d) at a later time.
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for the four conditions that result from bursting the diaphragm at time t = 0
is shown in Figure 12.3b. As in the previous section, the outgoing waves
divide this diagram into distinct regions numbered 1, 2, and 3. Upon striking
the closed end of the shock tube, the wave that was traveling in the positive
x direction will be reflected as a wave of some form. It is known that this
reflected wave will travel at the speed of sound 4, but it is not known a priori
whether it will be an expansion wave or a compression wave and what the
strength of this wave will be in relation to that of the incident wave. Thus, the
properties of the gas in region 4, which is the region that has been influenced
by both the incident and reflected waves, are not known.

Region 1 of Figure 12.3b has not yet been influenced by the outgoing wave
from the origin and so maintains its initial conditions of u = 0, p = p,. Likewise,
region 2 maintains its undisturbed condition of u = 0, p = p;. Region 3 has been
influenced by the outgoing waves, and so the velocity and the pressure there
will be given by Equations 12.2a and 12.2b. In order to determine the state of
the gas in region 4, an arbitrary point P(x, t) and its two characteristics are
indicated in Figure 12.3b. The £ = constant characteristic comes from region 3,
where the velocity and the pressure are known. Hence, this characteristic
may be terminated at any point in region 3 where the value of the Riemann
invariant may be established. The n = constant characteristic runs parallel
to the line of the reflected wave and eventually reaches the x location of the
closed tube end. Since the pressure here is unknown, the &, = constant char-
acteristic from this point is drawn into region 3, where the velocity and the
pressure are known. This permits the Riemann invariant to be established for
the n = constant characteristics.

Denoting the pressure at the wall or closed end in region 4 by p,, the
Riemann invariant along the n = constant characteristic gives

Here, the fact that u = 0 and p = p,, on the wall has been used. To evaluate
P the Riemann invariant for the n = constant characteristic will be used.
This gives

W lp_ dpe_1op_ 1 m,

a4 Y Po Y Po 2Y Po 2y po

where the values of the velocity and the pressure in region 3, as given by
Equations 12.2a and 12.2b, and at the wall in region 4 have been used. The
last equality is satisfied by

pw = pl'
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Then, the equation along the 1 = constant characteristic becomes

uw 1p__1p
4 Y Po Y Po

A second equation is obtained from the & = constant characteristic:

w, lp _1p
ay Y Po Y Po.

Here, the constant for the Riemann invariant has been evaluated in region
3 by again using Equations 12.2a and 12.2b. The solution to the last two alge-
braic equations is

u=0 (12.3a)

p=p: (12.3b)

Equation 12.3a shows that the velocity of the gas in region 4 is zero. This
result is due to the fact that region 4 includes the closed end of the tube, and
the boundary condition there requires zero velocity. Equation 12.3b shows
that the pressure in region 4 equals the pressure in region 2. This means that
the pressure at any value of x > 0 varies as follows: initially, the pressure is
po, and as the first wave passes toward the closed end, the pressure jumps
to (p; + py)/2. Finally, as the reflected wave passes, the pressure jumps to p;,
as shown in Figure 12.3a and b. That is, the pressure differential across the
incident wave is (p; — py)/2, which is also the pressure differential across
the reflected wave. Since this result is valid for either p, > p, or p, < p,, it fol-
lows that compression waves are reflected as compression waves of the same
strength by solid boundaries, and expansion waves are reflected as expan-
sion waves of the same strength.

12.4 Reflection and Refraction at Interface

When a wave encounters an interface between two different gases, part of
the wave is transmitted through the interface and part of it is reflected by
the interface. This conclusion may be verified by modeling the situation by
a shock tube in which an interface between two different gases exists part
way down the tube. Such a configuration is shown in Figure 12 4a. Initially,
the velocity is zero everywhere while the pressure is p, for x < 0 and p, for
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FIGURE 12.4
Shock tube with gas interface and xt diagram subsequent to bursting the diaphragm.

x > 0. Partway down the positive x axis, the physical properties of the gas
are assumed to change abruptly because there are two different gases in the
tube, or the same gas may have two regions that are at different tempera-
tures. In either case, the speed of sound is taken to be ay, to the left of the
interface and a, to the right of the interface. Likewise, the specific-heat ratio
is denoted by v, to the left of the interface and v, to the right.

The xt diagram describing the sequence of events that results from burst-
ing the diaphragm at time ¢ = 0 is shown in Figure 12.4b. It is assumed that
the wave that emerges from the burst diaphragm and that travels in the posi-
tive x direction toward the gaseous interface is partially transmitted and
partially reflected at the interface. That is, in general, it is assumed that part
of the incident wave passes through the gaseous interface and is refracted,
while the other part of the wave is reflected by the interface. Thus, Figure
12.4b is divided into four distinct regions as indicated. Region 1 represents
the initial state of the gas located to the right of the diaphragm, and although
the physical properties of the gas will be discontinuous at the gaseous inter-
face, the velocity will be everywhere zero, and the pressure will be every-
where p, in this region. In region 2, the velocity will be zero and the pressure
will be p;. In the regions marked 3, the gas will be influenced by passage
of the waves that result from bursting the diaphragm, and the velocity and
pressure there will be given by Equations 12.2a and 12.2b.



450 Fundamental Mechanics of Fluids

In order to determine the velocity and the pressure in the regions marked 4,
an arbitrary point P(x, t) that lies on the interface between the two gases is
considered. From this point, the & = constant and n = constant characteristics
are drawn, and by virtue of the fact that the point P lies on the interface,
each of these characteristics lies entirely in the domain of one gas only. The
& = constant characteristic may be terminated anywhere in region 3 where
the velocity and pressure are known, whereas the n = constant characteristic
may be terminated anywhere in region 1. Since the velocity and the pressure
must be continuous across the interface at all times, the two regions labeled 4
must have the same velocity and pressure. Since it is realized that, in general,
the interface may move after being struck by the incident wave, the line that
represents the interface in the regions 4 does not necessarily correspond to
X = constant.

Using the Riemann invariant along the & = constant characteristic shows
that the velocity and the pressure in region 4 must satisfy the condition

u+1p_ 1 ﬂ—1+iﬂ+1
Ay Y1 Po 2Y1 Po AT
_1lp
Y1 Po

Here, the known conditions for region 3 have been employed from
Equations 12.2a and 12.2b. Along the n = constant characteristic, we get

w_1p__1

Ay Y2 Po Yo

where the undisturbed conditions for region 1 have been used. The solution
to these two equations is

o P/po=1 (12.4a)

Ay Y1+ Y201/ A

P _ p1/Po + (V1/Y2 Mg /ag,) . (12.4b)
Po 1+ (v, /v, )0 /agy)

Equation 12.4a shows that for p,/p, > 1, the velocity u in region 4 will be
positive, so that the interface will move to the right in the positive x direc-
tion. As in the previous section, this confirms the result that the flow tends
to follow compression waves since for p,/p, > 1, not only will the incident
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wave be a compression wave but so will the reflected wave. It was shown in
the previous section that, for a solid interface, the reflected wave was of the
same strength as the incident wave. In the present case, it may be anticipated
that the gaseous interface is not as efficient in reflecting waves as the solid
interface. The actual strength of the reflected wave may be calculated from
the solution given by Equation 12.4b. If the pressure differential across the
reflected wave is denoted by Ap,, it follows that

R

Po Po 2 Po

where the solution for the pressure in region 3 has been used. Then, using
the pressure given by Equation 12.4b for the value in region 4, the pressure
differential across the reflected wave becomes

Ap, _ (1= (v1/v2)(@0p /30))(p1 /o = 1)_
Po 20T+ (y1/Y2 @02/ ag1)]

(12.5a)

If the speed of sound a,, becomes very small compared with ay, which
corresponds to a high-density gas beyond the interface, this result reduces to
Equation 12.3b for a solid boundary. That is, as the density difference at the
interface increases, the foregoing result reduces to that for an impermeable
boundary corresponding to perfect reflection.

If the pressure differential across the transmitted or refracted wave is
denoted by Ap,, it follows that

o _p g

Po  Po

where the fact that p = p, in region 1 has been used. Then, from Equation
12.4b,

%= (p/po = 1) 12.5b
Po 1+(Y1/Y2)(ﬂ02/’101). (12.5b)

Equations 12.5a and 12.5b show that the strength of the reflected wave
and that of the transmitted or refracted wave depend on the nature of the
interface between the two gases. For v, = v, = y and ay, = a4, = a,, Equations
12.5a and 12.5b show that there is no reflected wave and that the transmitted
wave is identical to the incident wave. For a,,/a, — 0, the reflection becomes
total, as was discussed earlier. Intermediate to these two limiting cases, both
a reflected wave and a transmitted wave will exist.
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12.5 Piston Problem

The so-called piston problem is a classical one and may be stated as fol-
lows: Figure 12.5a shows a cylinder or a tube inside which a piston slides.
Initially, the piston and the gas ahead of it are stationary, when the piston
is suddenly jerked into motion at some constant velocity. The problem is to
find the velocity and the pressure ahead of the piston after the motion has
started.

The xt diagram for such a situation is shown in Figure 12.5b. One of the
two lines on this diagram corresponds to a wave front that is generated by
the impulsive acceleration of the piston and that travels down the cylinder
ahead of the piston with velocity a,. The second line represents the instan-
taneous location of the piston, which is moving with constant velocity U
for t > 0. Since, according to our linear theory, U/a, < 1 so that the piston
will always be close to x = 0, compared with the location of the wave front,
the boundary condition that u = U on the piston face may be imposed on
x = 0 rather than x = Ut. This yields the modified xt diagram shown in
Figure 12.5¢c.

The xt diagram shown in Figure 12.5c is divided into two distinct regions
by the wave that leaves the piston face. Region 1 contains the undisturbed
gas, which is stationary and whose pressure is p;. In order to determine the
velocity and pressure in region 2, a typical point P(x, t) is considered. The
n = constant characteristic from this point enters region 1, where the val-
ues of u# and p are known. The & = constant characteristic runs parallel to
the wave front and eventually encounters the position of the piston face at
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FIGURE 12.5

(a) Piston and cylinder, (b) actual xt diagram, and (c) linearized xt diagram.
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x = 0. Although the velocity there is known, the pressure is not; hence, the
1, = constant characteristic is drawn from the point where the & = constant
characteristic terminates. These three characteristics permit the values of the
velocity and the pressure at the point P to be evaluated.

Denoting the value of the pressure at the piston face by p,, the Riemann
invariant along & = constant gives

W 1p_U 1P,
QG Y Po 4 Y Po

The pressure at the piston may be evaluated from the Riemann invariant
for the n, = constant characteristic, which is evaluated first on the piston face
and then in region 1. The resulting equation is

w_ lp U 15 _1

QG Y Po 4 Y Po Y.

Using the last equality to evaluate p,, the equation for the & = constant
characteristic becomes

W lp U1
4 Y Po a Y

Another equation connecting u and p may be obtained from the Riemann
invariant for the n = constant characteristic, which yields

w_ lp __1

4 Y Po Y

The solution to the last two equations is

u=U (12.6a)
pﬂ —y ag +1 (12.6b)
0 0

Equation 12.6a shows that the gas velocity in region 2 is everywhere the
same as that of the piston. Equation 12.6b shows that the pressure ahead of
the piston but behind the outgoing wave is greater than the initial value by
an amount that is proportional to the piston speed U.
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12.6 Finite-Strength Shock Tubes

In the previous four sections, some properties of internal waves have been
exposed through reference to weak shock tubes. In reality, finite-strength
waves exist, and their properties may be established through reference to
finite-strength shock tubes. Although the qualitative behavior of strong
waves is the same as that for weak waves, the quantitative results are dif-
ferent. The nature of these differences will be established in this section by
carrying out an analysis for the problem that is analogous to that treated in
Section 12.2 for a weak shock tube.

Figure 12.6a shows a shock tube in which the initial velocity is everywhere
zero and in which the initial pressure distribution is p, to the right of the
diaphragm and p, to the left of the diaphragm. The initial pressure distribu-
tion is shown in Figure 12.6b, and it is assumed that the pressure differential
p, — p; is substantial so that a linear theory is no longer valid. Because of the
substantial pressure differential and/or the fact that the gases may differ,
the specific-heat ratio and the speed of sound will be different on either side
of the diaphragm and are denoted by v,, 4, and vy,, a,, as indicated in Figure
12.6a.

The form of the pressure distribution for ¢ > 0, corresponding to the dia-
phragm bursting at t = 0, is shown in Figure 12.6c. A compression wave
of finite strength will travel down the tube in the positive x direction. It is

(a) () ‘
oS Interf:
Diaphragm §%\je / nterrace
QS% i
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u=0,p=py \ u=0,p=p | C,o“‘Q{eqe
Y =Yoo a=ay g=gpa=a @ w2 @
x
(b) (d)
p r
Py Py

t<0 l t>0
P P

FIGURE 12.6
(a) Shock tube configuration, (b) initial pressure distribution, (c) xt diagram, and (d) typical
pressure distribution for t > 0.
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known from Section 11.2 that this wave will steepen as it travels and will
develop into a shock wave, as shown in Figure 12.6c. An expansion wave
will travel in the negative x direction, and it is known that such waves tend
to smooth out as they propagate. The px diagram for this situation is shown
in Figure 12.6d. The shock wave may be represented by a single-line discon-
tinuity, but the expansion wave will extend over a substantial portion of the
x axis and is thus represented by an expansion fan. This consists of a series
of lines that emanate from the origin of the xt diagram and may be thought
of as a very large number of weak waves.

The location of the interface between the two bodies of gas is also shown
in Figure 12.6¢ so that the possibility of two different gases may be covered.
Thus, the xt diagram is seen to be divided into four distinct regions. Region
2 consists of gas 2 and represents those locations that have been affected by
the passage of the compression wave. Region 3 consists of gas 4 in those loca-
tions that have been affected by the expansion wave. The principal quantities
of interest are the strength of the shock wave that results from bursting the
diaphragm, for given values of p, and p,, and the speed with which the shock
wave moves along the tube.

The boundary conditions at the interface between regions 3 and 2 are
us = u, and p; = p,. These conditions guarantee continuity of the velocity and
the pressure, and they may be used to determine the strength of the shock
wave as follows. By employing a Galilean transformation to a stationary nor-
mal shock wave, the results obtained in Chapter 11 may be used to obtain an
expression for u, in terms of p,/p;. By an analogous procedure, the velocity
u; may be expressed in terms of the pressure ratio across the expansion wave
ps/ps- The matching conditions at the interface will then give an equation
that relates the pressure ratio p,/p; across the shock wave to the initial pres-
sure ratio p,/p, across the diaphragm.

Considering first the compression wave, let u,, and u,, be the gas velocities
u, and u,, respectively, expressed in a frame of reference in which the shock
wave is stationary. Then, in order that #; may be zero here, a Galilean trans-
formation of magnitude u,, must be made on the velocities. The relationships
between the normal shock velocities u,, and 1,,, which refer to Figure 11.3a,
and the present velocities 1, and u,, which refer to Figure 12.6d, are

Uy =y, — Uy, =0

Uy = Uy, = Uy, = Uy, 1=

This now represents a shock wave that is moving with velocity u,, through
a stationary gas in which the velocity of the gas behind the shock wave is u,.
But u,,/u,, may be evaluated from the Rankine-Hugoniot relation (Equation
11.4), which gives
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_ i+ )/ (v, =D +p/pa
T+ (v +D/(vi = D(py - p2)

Uy = Uy,

where p; and p, now refer to Figure 12.6d. But u,, = a,M,,, where M,, is
the Mach number of the flow approaching a stationary shock wave. From
Equation 11.8¢c

M=t Py
21 P2

where, again, the pressures now refer to Figure 12.6c. The expression for the
velocity in region 2 then becomes

1/2
nlop o . D/ -Dep/p
2v1 P 1+, + )/ (v = 1) (pi/ps)

This result may be simplified to give

2(p,/p, _1)2
Doyl =D+ (g + D(p /p)]

U, =a

Consider next the expansion wave. It was established in Chapter 11 that
expansion waves, contrary to compression waves, tend to smooth out and
spread themselves over substantial distances. Thus, the expansion from p, to
ps takes place in a continuous manner that may be thought of as consisting
of a very large number of weak expansion waves, each of which is isentropic.
Thus, from Equation 11.2a, it follows that at any point in the expansion wave

where the minus sign denotes that the wave is traveling in the negative x
direction. But

‘12=\{4£=Y74p4pY4
PP opy
va-1
-2 P
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where the isentropic law has been used to relate p to p. Thus, the expression
for du becomes

- ai‘i (v4-3)/2
du = pﬁl“'l)/zp dp.

Integrating this expression and noting that # = 0 when p = p, yield the
following expression for the local value of the velocity u in the expansion
wave:

(v4-1)/2
U= _ZA P -1 .
Ya-1 py

The local density p may be replaced by the local pressure p through use of
the isentropic law. This gives

(v4=1)/2v4)
2a
U= =74 1_ L

Ya-1 Pa

In particular, at the trailing edge of the expansion wave, p = p; and u = 1,
so that

2 (v4=1)/274
Uy =4 1_ &

Ya-1 P4

The expressions obtained above for u,, from the compression wave, and u;,
from the expansion wave, will now be used in conjunction with the interface
matching conditions. Thus, setting u; = 1, and at the same time replacing p;
by p, yields the following identity:

(v4=1)/214 1/2

2a, 1_ P2 a 2(p,/py - 1)

va-1 Pa Oy =D+ (g + D(py /)]

Although this equation cannot be solved to yield an explicit expression for
the shock-wave pressure ratio p,/p; in terms of the initial diaphragm pres-
sure ratio p,/p;, the converse is not true. Thus, solving this equation for p,
gives
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-2y4/(v4-1)

Py _P2 1- (Y4 =Wa,/a,)(pi/pa = 1)
PP 20l - )+ + D /)]

(12.72)

If p,/p, is replaced by 1 — ¢, Equation 12.7a shows that p,/p; = 1 + 2¢ to the
first order in €. That is, for weak waves, the result obtained in Section 12.2 is
recovered.

It was shown earlier in this section that the results for normal shock waves
could be related to those of a shock wave moving with velocity u,, into a
stationary gas in which the gas velocity behind the shock wave is u;, — u,,.
Thus, if M, denotes the Mach number with which the shock wave propagates
through fluid 1, it follows that

M, =M,,.
Then, from Equation 11.8a
1/2

1+[(y, - 1)/2]M3,
Y1M12n -(y,-1)/2

But, from Equation 11.8¢

M2, o1t Py
2y, b

where p, and p, refer to the problem at hand. Thus, the expression for the
Mach number of the compression wave becomes

v = 1O = D/2+ (00 =D+ /A /e =D

| Vi + [0+ D/2)pi /P2 -1 - (v -1)/2

This result may be simplified to yield the following equation:

1/2

M. = (v, -D+ (;1 +D(p/p) (12.7b)
Y1

As p,/p, approaches unity, M, also approaches unity. That is, for weak
shock waves, the front travels at the speed of sound, which confirms the
results of Chapter 11. Equation 12.7 also shows that the Mach number of the
shock wave can be considerably greater than unity for strong shock waves.
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12.7 Nonadiabatic Flows

The physical situations to be treated here differ from those of the previous
sections in several ways. As the heading suggests, the most significant dif-
ference is that flows in which heat is being added to the gas, or removed
from it, will be covered. In addition, external body forces such as frictional
forces may be included. Another difference from previous treatments is that
variations in the flow area may be included, provided that the flow may be
considered to be essentially one-dimensional. In the previous sections of this
chapter, the flow configurations have been exactly one-dimensional. Here,
converging and diverging boundaries will be permitted, and the velocity
that exists at any streamwise location will be considered to be the average
value at that location. Finally, the flow situations considered here differ from
those of the previous sections in the sense that the flow varies continuously
rather than abruptly, as with Mach waves or shock waves. That is, the heat
addition, external forces, and area changes will be assumed to be such that
they vary the flow properties continuously rather than abruptly, so that the
use of derivatives will be valid.

Figure 12.7a shows a typical flow configuration of the type to be consid-
ered here. At the location defined by x, the flow area is A, and at the location
x + dx, the flow area is A + dA. The element of length dx is subjected to an
external force 8f(x), and an amount of heat 84(x) is added to it.

The equations of motion for the gas may be readily derived in differential
form as follows: The continuity equation requires that puA = constant where
p and u are, respectively, the average density and velocity of the gas at the
location x. Then

d(pud) =0

FIGURE 12.7
(a) Element of one-dimensional flow field and (b) flow through a typical nozzle.
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so that performing the indicated differentiation and dividing by puA give
e e el (12.8a)
Equation 12.8a supplies one equation connecting three of the variables.
The momentum balance for the element shown in Figure 12.7a is
pudu=dp+5f
Dividing this equation by p and using the fact that a? = yp/p reduce it to the

following form:

yM2%+%=g (12.8b)
u.op oy

where the local Mach number M = u/a has been introduced. The thermal-
energy balance for the case of a perfect gas may be written in the form

CpdT + u du = 8q

where T is the local temperature of the gas. Dividing this equation by C,T
gives

dT udu_

. o
T CT CT°

But T may be replaced by p/pR for a perfect gas and RCp = (y — 1)/y, so the
energy equation may be written in the form

dT ,du g
=DM — =
7t -DM = (12.8q)

C,T

Here, again, the expression for the speed of sound and the definition of the

Mach number have been employed. Finally, the equation of state for an ideal
gas gives

P _R (12.8d)



One-Dimensional Flows 461

and hence

Equations 12.8a through 12.8d represent four algebraic equations for the
differentials du, dp, dp, and dT in terms of the local values of the variables
u, p,p, T, M, f, q, and A. Then, these equations may be solved to yield expres-
sions for each of the differentials separately. To obtain the expression for du,
for example, use Equation 12.8d and eliminate dp/p by using Equation 12.8b
and dT/T by using Equation 12.8c. This gives

—YM2%+g _@4_ (Y—l)Mzﬁ ﬂ

- =0.
u p p u CT

Adding Equation 12.8a to this result to eliminate dp/p results in the follow-
ing expression:

du 1 dA ¥ &

u M*-1 A p CT°

(12.9a)

Equation 12.9a gives the change in speed du that is associated with a
change in area dA, application of an external force &f, and the addition of
an amount of heat 8. The coefficients of the quantities dA/A, &f/p, and dq/
CpT are called influence coefficients since they represent the influence of some
external process, such as heat addition, on some flow variable, such as the
gas velocity. Thus, with respect to the normalized velocity du/u, the influ-
ence coefficient for dA/A and that for &f/p is 1/(M 2 - 1), whereas the influence
coefficient for 6q/C,T is —-1/(M? - 1). As a special case, consider adiabatic flow
without external forces. Then, Equation 12.9a becomes

du 1 dA

U ME_1A°

This equation expresses the familiar result that in order to accelerate
(du > 0) subsonic (M < 1) flow, the flow area should be decreased (dA < 0). On
the other hand, for supersonic flow, the area should be increased to acceler-
ate the flow. This leads to nozzle shapes of the form shown in Figure 12.7b
where the throat, which corresponds to dA = 0, has sonic conditions.
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An expression similar to Equation 12.9a may be obtained for dp/p by using
Equation 12.8a and by eliminating du/u from it through use of Equation
12.9a. This gives

dp YM? %_1+(y—1)M2ﬁ+ YM?* g
p M*-1 A M*-1 p M,-1CT’

(12.9b)

For adiabatic flow without external forces, Equation 12.9b becomes

dp __ wM* dA
p M*-1 A

This result shows that in order to expand a gas in a nozzle continuously,
the flow area should decrease when the flow is subsonic and increase when
the flow is supersonic. This agrees with our conclusion regarding accelerat-
ing gases in nozzles. For flow in a constant-area channel in which there are
no external forces acting on the gas, Equation 12.9b becomes

dp __ M ¥
p M -1CT’

This equation shows that in order to expand a gas in a pipe by thermal
means, heat should be added if the flow is subsonic, whereas heat should be
removed if the flow is supersonic.

The variation of the temperature may be obtained by using Equation 12.8a
to eliminate du/u from Equation 12.8c. This gives

2 2 2
dl=_(Y _1)M %_ (Y_l)M g_'_ YM” -1 6‘7 . (12.9C)
T M*-1 A M-1 p M-1CT

For adiabatic flow without external forces, Equation 12.9¢c reduces to

dr __(v-1)M* dA
T M>-1 A’

Hence, for y > 1, the temperature will drop in a converging-flow area if the
tlow is subsonic or in a diverging-flow area if the flow is supersonic. For flow
in a constant-area channel in which there are no external forces, Equation
12.9¢ becomes

dT  yM?-1 &
T M*-1CT
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The influence coefficient here changes sign at M =1/ \/? and at M = 1.
Then, for y > 1, the effect of heat addition will be to increase the tempera-

ture for M < 1/\/; and for M > 1, but the temperature will decrease in the

range 1/ \/; <M <1. For adiabatic flow in a constant-area duct, Equation
12.9¢ becomes

dr _ (v -1)M* of
T M*-1 p

Hence, for y > 1, the effect of an external force such as wall friction is to
increase the temperature of subsonic flow and to decrease the temperature
of supersonic flow.

Two well-known results that may be established in this way are the
Fanno line and the Rayleigh line. In each case, the variation of temperature
or enthalpy is considered as a function of the entropy. The resulting curve
for the case of adiabatic flow in a constant-area duct is called a Fanno line,
whereas the curve for the case of a constant-area duct without external forces
is called a Rayleigh line. That is, the Fanno line shows the effect of friction in
a constant-area duct, whereas the Rayleigh line shows the effect of heat addi-
tion. One of the most practical and significant results that may be deduced
from these diagrams, or from the equations established above, is that chok-
ing takes place at M = 1. Thus, adding heat to a constant-area flow will accel-
erate it until M = 1, and no more heat can be added beyond this point. For
long pipelines, the effect of friction is similar, so that a point may be reached
beyond which no more gas can be pumped through the pipe without remov-
ing some heat.

12.8 Isentropic-Flow Relations

For isentropic flows, simple and useful relations exist between the local
value of some variable, such as the temperature or pressure, and the local
Mach number. These relations may be obtained from the thermal-energy
equation as follows.

For steady, isentropic flow, the thermal-energy Equation IV.3b becomes

pu-Vyh=(u-Vp.

But, forming the scalar product of u and the Euler equation (Equation IV.2)
shows that
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(u- )p=-ulpu- )u]

1
=—p(u- ) ~uu.
plu- ) Su
Hence, the energy equation may be written in the form
(u- ) h+ lu ‘u
p 5 -

This means that the quantity / + u-u/2 is constant along each streamline,
so that

h+1u-u=hO
2

along each streamline. The quantity /, is called the stagnation enthalpy, and
it corresponds to the enthalpy that the fluid would have at zero velocity. Of
course, it may be known that in some part of the flow field, the stagnation
enthalpy is constant, which is usually the case, so that &, will have the same
constant for every streamline.

The foregoing result for the enthalpy may be recast in terms of the tem-
perature, the pressure, or the density. For a perfect gas whose physical prop-
erties are constant, i = C,T, so that

1
C,T+ Euz =CTy.

Here, the fact that we are dealing with one-dimensional flows only has
been used, so that u - u = #2. The quantity T, that has been introduced here is
the stagnation temperature, which corresponds to the temperature the fluid
would have if it were brought to rest. Solving this equation for the tempera-
ture ratio shows that

The quantity 42/T may be rewritten as yRu?/(yRT) = yRM 2 since yRT, which
appears in the denominator, is the square of the local value of the speed of
sound. Then, since R/C, = (y — 1)/y, the expression for the temperature may
be written in the following form:

E=1+Y_1

M>. (12.10a)
T 2
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The following relations are known from thermodynamics to be valid for
isentropic flows:

Using these identities and Equation 12.10a shows that the following rela-
tions hold:

Po_ 1,0 1hp (12.10b)
p 2
-1 Ny-1)
Po_ 47002 . (12.100)

Here, p, and p, are, respectively, the stagnation pressure and the stagnation
density, and M is the local Mach number.

12.9 Flow through Nozzles

It was shown in Section 12.7 that if a nozzle is required to expand a subsonic
flow to supersonic speeds, its shape should be of the form shown in Figure
12.7b. Such a flow configuration will be considered here in which the flow
reaches sonic conditions at the throat and is supersonic downstream of the
throat. Since the flow is adiabatic and if frictional losses may be considered
to be negligible, the flow will be isentropic. This means that the results of the
previous section may be employed.

The notation that was introduced in Section 11.4 to indicate sonic condi-
tions will again be used here. Thus, the temperature, pressure, and density
corresponding to M = 1 will be denoted by T, p., and p., respectively.

Then, from Equations 12.10a through 12.10c

(12.11a)

Po_ ¥+l (12.11b)
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+
—_

N v 1/(v-1)
Lt U
o 5 ) (12.110)

That is, the temperature, pressure, and density at the throat of the nozzle
may be determined if the stagnation values are known. The stagnation condi-
tions will be known directly if the flow originates in a large reservoir where
the fluid speed is zero and its other properties are known. If the fluid proper-
ties and nonzero speed are given at the inlet to the nozzle, then the stagna-
tion properties may be calculated from Equations 12.10a through 12.10c.

The variation of the Mach number of the flow with the flow area of the
nozzle may be established as follows. The continuity equation, written for an
arbitrary section and for the throat of the nozzle, gives puA = p.u.A.; hence

A p.M.a.
A, pMa

But M. =1 by definition and a./a =/T./T , so that

A ppp 1 |TT,
A pp M\ T,T

Now p./p, is given by Equation 12.11c, while py/p is given in terms of the
local Mach number by Equation 12.10c. Likewise, T./Tj is given by Equation
12.11a and T,/T is given by Equation 12.10a. Thus, the expression for the area
ratio may be written in the form

1/(v-1) 1/(v-1) 1/2
A_ 2 14 1- 1 M2 1
A. y+1 2 M yv+1
1/2
X 1+Y7_11\/I2
2

This result may be simplified to yield the following expression for the ratio
of the local flow area for which the Mach number is M to the area of the
nozzle throat:

(v+1)/2(y-1)

1 y-1.,
= 1+7 " M
ATMy+1 T2 ' (12.122)
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Equation 12.12a relates the local flow area to that of the throat, so it is of
interest to obtain an expression that relates the throat area A. to the mass
flow rate through the nozzle. Denoting this quantity by #z, it follows that

1 = p.itA.

= PPo (MaA..
Po

Using Equation 12.11c again together with the facts that M. = 1 and
a= \/yR"E = \/yRTOR /T, this equation becomes

1/(y-1) 1/2

2
PoYRT, Vel A

where Equation 12.11a has been used for T./T. Using the ideal-gas law to
write p, = po/(RTy), the final form of the expression for the mass flow through
the nozzle becomes

7=
y+1

T v+1

(y+1)/(y-1)
_ pA 2
-b \/ v . (12.12b)

m=
VI

As might be expected, the mass flow rate through the nozzle is propor-
tional to the throat area A.. Equation 12.12b further shows that the mass flow
rate is proportional to the stagnation pressure of the gas and inversely pro-
portional to the square root of the stagnation temperature of the gas.

The foregoing expressions are sufficient to design convergent—divergent
nozzles. Typically, the conditions in the gas at the entrance to the nozzle are
given together with the mass flow rate (or the inlet area) and the exit pressure
to which the gas must be expanded. From the inlet conditions, the stagna-
tion properties may be evaluated from Equations 12.10a through 12.10c. The
required throat area for the nozzle may then be calculated from Equation
12.12b. Since the stagnation properties of the gas are constant through the
nozzle, Equation 12.10b permits the exit Mach number to be determined.
Equation 12.12a then determines the exit flow area of the nozzle.

PROBLEMS

12.1 Consider a weak shock tube that has a finite length in the positive x
direction downstream of the diaphragm. The end of the shock tube
that is downstream of the diaphragm is open to the atmosphere.
Draw the xt diagram for the wave that will result from bursting
the diaphragm. If the pressure at the open end of the shock tube is
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maintained at the value p,, obtain expressions for the velocity and
the pressure behind the wave that is reflected from the open end of
the tube, back toward the burst diaphragm.

12.2 The figure in this problem shows a piston located at x = 0 in a cyl-
inder of length L. Two different gases occupy the space between
the piston and the cylinder head, and these two gases meet at a
free interface that is located at x = aL. At the time ¢ = 0, the piston
is impulsively set into motion with constant velocity U, where the
velocity U may be assumed to be small compared with the acoustic
velocities ay, and ay,. At the interface between the two gases, part
of the resulting wave is reflected and part of it is refracted. If the
reflected wave reaches the piston at time ¢ = T and the transmitted
wave reaches the end of the cylinder at the same instant, draw the
xt and use linear theory to find the following:

(@) The temperature ratio Tj,/Tj,; in terms of «

(b) The velocity and the pressure for 0 < x < L and for 0 <t < 7 in
terms of U, py, v, 4y, and «

(©) The ratio of the strength of the reflected wave to that of the
transmitted wave

Piston Interface
—1V u=0,p=pg / u=0,p=py
Y> To1s do1 Y> To doa
x=0 x=al x=1L
Closed end

12.3 Show that, for a finite-strength shock tube, the Mach number of the
flow behind the shock wave is given by the following expression:

-1/2
M2=i &_1 1_,_717_1 &_1 P2
Y1 P 2yy py P

where the subscript 1 corresponds to the region ahead of the shock
wave, and the subscript 2 corresponds to the region immediately
behind the shock wave.

12.4 Heat is being added to a perfect gas that is flowing in a constant-
area channel. Neglecting all external forces, determine the influ-
ence coefficient f in the equation
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Use the resulting equation, together with the following equation:

dp _ aM* ¥
p M -1CT

to establish a differential relation between p and M. Integrate this
relation to obtain an expression for the pressure ratio p,/p; in terms
of the Mach numbers M, and M, at any two locations.

12.5 Use the results of Problem 12.4 and the isentropic flow relations
to obtain the temperature ratio T,/T; and the density ratio p,/p; in
terms of the Mach numbers M, and M,.

12.6 The entropy change between two flow conditions of a perfect gas
may be determined from the following relation:

Y
s-s, =C,log PE %1
1

where the subscript 1 denotes inlet conditions for the flow. Apply
this to the case of steady, adiabatic flow in a channel of constant
cross-sectional area to establish the equation of the Fanno line. To
do this, use the following forms of the continuity and energy equa-
tions, and the equations of state:

PUA = m
uZ
h+7=h0
p =pRT
h=C,T.

Here, m is the mass flow rate of the gas and h, is the stagnation
enthalpy. Hence, show that the equation of the Fanno line is

s—s pl R 2A b
2T _oe h(h, — )YV2 41 1
C, og h(hy - h) +1og p C, i
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From this last result, show that the entropy reaches a maximum
when M = 1.

12.7 To obtain the equation of the Rayleigh line, use the same equations
that were used in the previous problem except, since heat addition
is involved, replace the energy equation with the following form of
the momentum equation:

pu+p=p;
Hence, obtain the equation of the Rayleigh line. From the result,

show that the entropy reaches a maximum where M = 1 and that
the enthalpy reaches a maximum when M =1/ \/; .



13

Multidimensional Flows

This chapter deals with some steady two-dimensional and three-dimensional
flow problems, both supersonic and subsonic. The governing equations are
first established for irrotational motion, and then solutions to these equa-
tions are sought. The Janzen-Rayleigh expansion is discussed first. This
expansion is a parameter expansion in powers of the Mach number squared,
and so it is valid only for Mach numbers that are somewhat less than unity.
Small-perturbation theory is next discussed. This approximation assumes
that the body about which the flow is sought disturbs the free stream in a
minor way only. Since many real flow situations satisfy this condition, small-
perturbation theory is widely used.

Small-perturbation theory is then used to study some specific subsonic
and supersonic flows. The Prandtl-Glauert rule for subsonic flows is cov-
ered. This rule relates subsonic compressible flows to the corresponding
incompressible flows. Ackeret’s theory for supersonic flows, which is also
based on small-perturbation theory, is discussed next.

Leaving the topic of small-perturbation theory, the chapter ends with a
discussion of an exact solution. The flow treated is that of supersonic flow
turning around a corner that bends away from the free stream. This flow is
known as the Prandtl-Meyer flow.

13.1 Irrotational Motion

As was the case for incompressible flows, many of the flow fields of inter-
est are irrotational because they originate in a uniform flow. According to
Crocco’s equation, the flow will then also be isentropic. Then, the pressure
term in the momentum equation may be rewritten as follows:

dp )

p=d7p p=a p

where the fact that p is a function of p only, owing to the isentropic nature of
the flow, has been used. Then, the momentum Equation IV.2 becomes

471
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ou a?
—+(u Ju=-— p.
Py (u- ) P

Forming the scalar product of the velocity vector u with this vector equa-
tion gives

%%(u-u)+u-[(u- Jul = —iu- p.

The term on the right-hand side of this equation may be recast by use of the
continuity equation in the form

u p=-—_ =P u

Thus, the foregoing form of the momentum equation becomes

2
li(u-u)+u-[(u- )u]=a—@+a2 .
29t 0 of

The density may be completely eliminated from this equation by taking
the time derivative of the Bernoulli equation. Thus

2
Po 1o o0 (b
ot 2 ot ot J p
_9 dpde
oot Jdp p
o pa*
-—— [*d
atfp b

d pd? ap
- (Tap P,
dp fp P ot

Now, the two inverse operations, differentiation with respect to p and inte-
gration with respect to p, cancel one another, so that the Bernoulli equation
becomes

2 2
PO 10 gy TP
or* 2 at p ot
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Using this equation to rewrite the nonsteady term on the right-hand side
of the momentum gives
19 g 19

Eg(u‘uﬂu-[(u‘ yu] = P —Eg(u‘u)ﬂz ‘.

Solving this equation for V - u yields the following equation governing the
irrotational motion of a compressible fluid:

1 a 90
‘u=— u-f(u J)ul]+— —+u-u
a* C ot ot

In terms of the velocity potential ¢, this equation becomes

29 L 6 o 900, .
¢—a2 o-IC ¢ )¢]+at YR (13.1)

Equation 13.1 is the differential equation to be satisfied by the velocity
potential ¢ for irrotational motion of a compressible fluid. The equation differs
drastically from the Laplace equation, which was shown to be the governing
equation for incompressible flow. In fact, Equation 13.1 becomes V2 = 0 as
a? — oo, which corresponds to p = constant. This may be verified by noting that

:_dp
a _dp

where the derivative is evaluated at constant entropy. But for p = constant,
dp =0, so that a2 — . Thus, it may be concluded that for constant density, the
governing equation for irrotational motion is linear, but for variable density,
the governing equation becomes nonlinear. It should also be noted that the
nonlinearity must be dealt with directly here since the equations of kinemat-
ics and dynamics are no longer separable.

Cleary, Equation 13.1 represents a formidable analytic problem for any spe-
cific flow problem that is to be solved. The difficulty of obtaining exact solu-
tions has led to the development of approximate methods, and two of these
will be discussed in the following sections.

13.2 Janzen-Rayleigh Expansion

The Janzez-Rayleigh expansion is an expansion of the steady-state form of
Equation 13.1, which is valid for any shape of body but only for Mach num-
bers less than about 0.5. For steady flow, Equation 13.1 becomes
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In tensor notation, this equation is

2
o _ 1905 A (13.2a)

As the speed of sound becomes infinite, that is, as the Mach number tends
to zero, this equation reduces to the Laplace equation. That is, the right-hand
side of the foregoing equation represents compressible effects, so that these
effects vary as a. It seems reasonable, then, that an approximate solution for
slightly compressible flows could be sought in which the first correction due
to compressibility varies as the Mach number squared.

The foregoing remarks form the basis of an expansion for ¢ of the follow-
ing form:

9(x,y,2) = UZ M2, (x,y,2) (13.2b)

It is assumed here that a uniform flow of magnitude U approaches the
body under consideration, and M., = U/a., is the Mach number far from the
body, where a., is the speed of sound there. Substituting Equation 13.2b into
Equation 13.2a gives

UE 2n a ¢n - i;z oZon aq)n EMZ" aq)n E 0Zon az¢n .
6x,8x, a c')xiaxj

The significance of including the coefficient U in Equation 13.2b is now
apparent; the coefficients in the series may be made dimensionless, yielding
the following equation:

2
o

EMZn azq)n =LM22M2n %EMM %EMZH a2¢n .
LT ex; b L T L T x4 T axox;

]

Although the coefficients are all dimensionless, the quantity a? /a” is not
constant and should also be expressed as a series in M? . This may be done
by using the energy equation in the form
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where the constant has been evaluated far from the body. Solving this equa-
tion for the ratio of the local value of the speed of sound to that far from the
origin gives

N

=1+

y-1 Mi_u-u
2 a?

©

m‘&
8 N

Substituting u = V¢ and inserting the expansion for ¢ yield the following
expansion for a*

2

2 =
%=1+L1Mi 1= E:Min%
ax 2 - ox;

2
S a2 s M ooy,
2 0x

i

Inverting this expression defines the quantity that appears in the govern-
ing equation:

2 2
T Y= yp il % omn),
a? 2 a

i

Substituting this result into the expanded form of the governing equation
for ¢ yields

2 2
e e VR B VR B R e
X;0X; x

i

x EMZn %E MZn %E M2n azq)n .
*ox; ©oox; © ox;0x;
n n n

The expansion (Equation 13.2b) is assumed to be uniformly valid in M.
This means that the coefficients of like powers of this quantity must balance

n
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on each side of the foregoing equation. This gives the following sequence of
equations that represents the coefficients of the quantities MY, M*, M*, etc.:

99,

=0 13.2¢c
0x;0X; ( )

01 _ 99 90y 99y

(13.2d)
ox;0x;  0x; 0x; 0x,0x;
2 2 2 2
s __y=1 by oo by 3, by by 3,
dx;0x; 2 ax; dx; dx; dx;0x;  9x; Ox; 9x;0;
(13.2¢)

L %0 001 00, 39y 300 0%,
ox; dx; ox;0x;  9x; 0x; 9x;0X;

and so on.

The equation to be solved for ¢, (Equation 13.2c) represents the incompressible-
flow problem corresponding to M., — 0. The problem for ¢,, represented by
Equation 13.2d, is a linear one, although the differential equation is non-
homogeneous. Having solved the problem for ¢, the right-hand side of
Equation 13.2d will become an explicit function of the spatial coordinates.
Likewise, having obtained expressions for ¢, and ¢,, Equation 13.2e repre-
sents a linear, nonhomogeneous equation for ¢,. In this way, solutions for
o, d1, d,, etc, may be obtained sequentially, and each of these solutions rep-
resents a term in the perturbation solution (Equation 13.2b). It may be seen
from the equations for ¢, and ¢, that the differential equation to be solved
becomes complicated rapidly, and it is not practical to carry out the solution
beyond the first two or three terms. This means that the solution so obtained
will be valid only for Mach numbers that are of the order of 0.5 or smaller.
The advantage of the Janzen—Rayleigh expansion, on the other hand, is that
it is valid for any shape of body, not just slender bodies.

13.3 Small-Perturbation Theory

An alternative approximate method of solution to the equation for compress-
ible potential flows is small-perturbation theory. This approximation is valid
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for supersonic flows as well as subsonic flows, but it is restricted to relatively
slender bodies.

Suppose that a uniform flow approaches a body that is sufficiently slender
that it causes only a small perturbation to the free stream. Then, the velocity
potential may be written in the form

b=Ux+® (13.3a)
where

l @ < 1.

U |dx;

Then, for steady flows, Equation 13.1 becomes
P = iz(llex + D) [(Ue,+ D) JUe,+ D).
a

This expression is still exact within the inviscid theory, but now, the fact
that the perturbation velocity potential ® leads to small-velocity components
will be used to eliminate quadratic and higher terms. Thus, the linearized
form of the governing equation is

2 42
2¢=U78(I)

a* ox?

This simplified equation retains only one term out of the compressible cor-
rection terms, and the retained term corresponds to the direction of the free
stream. In its present form, the compressible correction term contains a vari-
able coefficient 2, which should also be linearized for consistency. This may
be done by appealing to the energy equation in the form

1 a2 1,, a
STuu+ =
2 y-1 2 y-1

but
u-u=U+u)P+ @)?+ W)

where 1, v}, and w’ are the velocity perturbations to the free-stream velocity
U. Thus, the linearized form of the kinetic-energy term is
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u-u=U*+2Uu'
Using this form, the energy equation becomes

2 2

Uu +

y-1 vy-1

or

a? =a? 1—(y—1)u—1; .
a

©

Substituting this expression into the simplified equation for the perturba-
tion velocity potential gives

-1
u? Uobd 0’0
2p="" 1-(y-1)—— )

%

This equation, in turn, may now be linearized to yield the following linear
equation with constant coefficients:

2 a2
b
Zq)_uia

az ox?

o

In Cartesian coordinates, this equation becomes

PP PP PP
(1-M32) st ot =
0x ay 9z

0. (13.3b)

Equation 13.3b shows that for subsonic flow, the governing equation is
elliptic and so will have no real characteristics. On the other hand, for
supersonic flow, the governing equation is hyperbolic and so will have
real characteristics. This observation is compatible with our previous
result that shock waves can occur only in supersonic flow. Equation 13.3b
is valid for supersonic flows as well as subsonic flows, but as will be dem-
onstrated later, it is invalid near M., = 1. Also, by virtue of the lineariza-

tion, the equation is valid only for flows that involve relatively slender
bodies.
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13.4 Pressure Coefficient

The principal quantity that will be of interest in the solution to specific problems
in this chapter is the pressure distribution in the fluid since the integral of the
pressure around the surface of a body defines the lift and drag forces acting on
the body. The usual way of expressing the pressure is by means of the dimen-
sionless pressure coefficient. It is therefore of interest to obtain a linear expression
for the pressure coefficient that will be compatible with the linearized equation
derived in the previous section. Such an expression will be derived now.
The pressure coefficient C, is defined in the following way:

PP
P %pwuz

Here, p.., p., and U are, respectively, the pressure, density, and fluid veloc-
ity far from the body around which the flow is being studied. The pressure
coefficient may be readily expressed in terms of the pressure ratio as follows:

C =2 P~ P _4

’ pwuz pOC
_ 2 p_
M P

In order to relate the pressure ratio to the velocity, the energy equation will
be employed:

v
y-1

luZ _,_Lpl
2

1
—u-u+ .
2 Y -1p,

P_
p

Since the flow is irrotational, it is also isentropic; hence, the quantity p/p
may be expressed in terms of p only by use of the isentropic law. Thus

1/y

p=p. -
poc
-1/
p_r r
P Pe Po
lli 1-1/y
Y P
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Substituting this expression into the energy equation gives

(v=1)/y 5

v =—U*+ a
2

0

v-1

From this equation, the following expression is obtained for the pressure
ratio:

1 v/(y-1)
£= 1+Y72(U2 —u‘u)

0 00
Using this result, our expression for the pressure coefficient becomes

v/(y-1)

ZENS O Bl ) C R -1 (134a)

C =
Y 242

Equation 13.4a, which expresses the local value of the pressure coeffi-
cient in terms of the local velocity;, is still exact within the inviscid, adiabatic
assumptions. In order to obtain an expression for the pressure coefficient
that is compatible with small-perturbation theory, Equation 13.4a will now
be linearized in the perturbation velocity. The velocity term in the foregoing
equation may thus be rewritten as follows:

W2—u-u=U>-[(U+u)+ @)+ W)
==2Uu".

Substituting this linearized expression into Equation 13.4a yields the fol-
lowing simplified expression for the pressure coefficient:

v/(v-1)
2 Uu
CP=YM2 1—('Y—1)1172 —1 .

0 ©

But, to the first order in the perturbation velocity,

v/(y-1)
Uu Uu
1-(-D= =l-y—.
a a

L3 o
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Thus, the linearized form of Equation 13.4a is

C,=-21-.
u (13.4b)

This simple result will be used in conjunction with approximate solu-
tions to the compressible-flow equations that are established through use of
Equation 13.3b.

13.5 Flow over Wave-Shaped Wall

The first application of small-perturbation theory will be made to flow over
a sinuous wall. This flow is relatively simple, and it has the property of illus-
trating clearly the distinctions between subsonic and supersonic flows.

Figure 13.1a shows a sinusoidal surface over which a uniform flow of mag-
nitude U is assumed to flow such that compressible effects are not negligible.
The equation of the wavy surface is taken to be as follows:

. 2mx
y=m(x)= ssmT

where g/) is assumed to be small compared with unity, so that the linear the-
ory will be valid. The differential equation to be solved is the two-dimensional
form of Equation 13.3b. The boundary condition to be satisfied on y = n(x) is

v _dl_&“ 2mx

= =—"—gcos .
U+u dx M A

In view of our linear theory, the quantity v7/(U + u’) may be reduced to
v’/U, so that this boundary condition becomes

@(x,n) = Uz—nscoszﬂ.
dy A A

The left-hand side of this equation may be expanded in a Taylor series and
linearized so that the linear form of this boundary condition is

@(x,O) = Uz—nscoszix.
ay A A
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FIGURE 13.1
(a) Wave-shaped wall, (b) flow for M., <1, and (c) flow for M, > 1.

From the foregoing discussion, it is evident that, within the confines of

small-perturbation theory, the problem to be solved for the perturbation
velocity potential ®(x, y) is the following;:

o 9D

1-M2 + =0 13.5a
( ) oy (13.5a)
@(x, 0) = Uz—ns coszﬂ (13.5b)
ay A
gg(x, y) = finite  asy — . (13.5¢)
x

Since Equation 13.5a may be either elliptic or hyperbolic, depending on
whether M., is less than unity or greater than unity, it is convenient to dis-
cuss the solution to Equations 13.5a through 13.5¢ for subsonic flow and
supersonic flow separately.

Consider, first, the case of subsonic flow. It will be found convenient to
replace x by a new coordinate ¢ that is defined by
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X
1- M2

Then, in terms of the coordinates £ and y, Equations 13.5a through 13.5¢
become

w7
9> ay’

—(E 0) = U— £ COS —\/1 MZ2E

L)
— = finif — 00,
P &) te asy

The governing equation is now seen to be Laplace’s equation in this new
coordinate system. Solving this equation by separation of variables, the solu-
tion should be trigonometric in & in view of the first boundary condition.
Then, the y dependence will be either exponential or hyperbolic, and in view
of the semi-infinite domain in the y direction, the exponential form will be
used. However, the second boundary condition rules out the possibility of a
positive exponential, so the required solution will be of the form

D, y) = (A cos a& + B sin ak)e.

But the first boundary condition has been used only to obtain the func-
tional form of ®. Thus, imposing this boundary condition completely gives

—(E,O)——oc(A cosaf + B sinag) = U—ecos —\/1 M2E .

Equating coefficients and arguments of the two trigonometric terms
involved in the last equality gives

2
—0A=U="
a )\s
B=0
2
== J1-M>.
4=
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Thus, the complete solution for @ is

(D(E,y)=—$cos z—ﬂ\ll—MiE exp —z—ﬂwll—Miy .
J1-M? A A

Returning now to the original coordinate system gives the following solu-
tion for the perturbation velocity potential:

2 oMy . (13.6a)

D(x,y) = —Lcoszixexp -
Ji-m2  h A

Equation 13.6a shows that the perturbation to the free stream is in phase
with the wall, and it leads to a flow pattern, as shown in Figure 13.1b. It is
also evident that the perturbation dies exponentially with distance from the
surface. Since it was assumed that the perturbation velocity should be small
compared with the free-stream velocity, Equation 13.6a shows that

Max 1 _@n/Me o

u Ji-m?

Using the expression 13.4b and the solution 13.6a yields the following
expression for the pressure coefficient in the fluid:

(O G2 ey IRy . (13.6b)
M2 )\, )\.

©

Cp(x’ ]/) =

This result shows that the maximum pressure on the wall corresponds to
the bottom of the troughs, and the minimum pressure corresponds to the top
of the humps. That is, the pressure is symmetrically distributed about the
humps on the wall, so that there will be no induced drag on the wall. This
result will be further discussed later in this section.

Considering now the case of supersonic flow, the governing partial dif-
ferential equation is

Fo 1 PO
ot (M2 -1) ay*

This is the so-called one-dimensional wave equation, whose general solu-
tions will be of the form
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d)(x,y)=f(x— Mi—ly)+g(x+ Mi—ly)

where fand g are any differentiable functions. The first solution represents a
wave that slopes downstream and away from the wall, so that perturbations
generated by the wall will travel downstream only according to this solu-
tion. On the other hand, the second function in the solution above represents
signals that travel upstream as they move away from the wall. Since such a
solution has no physical meaning in supersonic flow, it must be rejected here,
so that the general solution becomes

(I>(x,y)=f(x— M —1y).

The function f may be evaluated by imposing the surface boundary condi-
tion 13.5b. This gives

gg(x,O) = —JM2 -1f (x) = Uz%scosz%
Y

and hence

Ue . 2mx

f(x)= —7(@51117

Thus, the perturbation velocity becomes

Ue 21
D(x,y) = ————sin —(x- M§-1y) ) (13.7a)
JM? -1 A

This solution satisfies the remaining boundary condition (Equation 13.5c).
From this solution and Equation 13.4b, the value of the pressure coefficient
is found to be

(4 / Ne 21
= ———"C08

x4 M2 -1 ) 13.7b
14 'Mi—l }\(x © y ( )

The solution represented by Equations 13.7a and 13.7b shows that the
velocity components and the pressure are constant along the lines:

C
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x — M2 -1y = constant.

The slope of these lines is given by

=tan©

dy _
2-

.
|

where 0 is the inclination of the lines with respect to the x axis. Hence

This result shows that the lines along which the flow parameters are con-
stant are actually the Mach lines. That is, signals are propagated along the
Mach lines undisturbed. The resulting flow field is illustrated in Figure 13.1c.

Equation 13.7b shows that the pressure on the wall is proportional to cos
(2nx/2), which means that the pressure peaks are 90° out of phase with the
geometric peaks of the wall. It follows, then, that a drag force will exist on
the wall for the case of supersonic flow. This is quite different from the result
that was obtained for subsonic flow. Figure 13.2a shows a section of the
wall, whereas Figures 13.2b and 13.2c shows the pressure distribution on the
wall for subsonic flow and for supersonic flow, respectively. In this figure,
the value of the pressure coefficient C, evaluated on the wall is denoted by

C,. Because of the symmetrical pressure distribution about each geometric
peak, there is no drag force in subsonic flow. However, the lack of symme-
try in supersonic flow leads to a drag that is called the wave drag. Thus, the
linearized theoretical drag on bodies in compressible flow is as shown in
Figure 13.2d. The theoretical drag becomes infinite for Mach numbers close
to but greater than unity because the linearized theory breaks down in sonic
flow. A transonic theory exists and shows, as it should, that retaining some
important terms that the linear theory neglected results in a finite drag coef-
ficient. The actual drag indicated in Figure 13.2d illustrates this result and
also shows that, owing to viscous effects, a drag force exists even for sub-
sonic flows. However, this viscous drag is relatively small for slender bodies
when it is compared with the wave drag.

The foregoing solution for a wave-shaped wall is significant in its own
right, and it illustrates some important features of subsonic and super-
sonic flows. Also, since the theory being used is linear, superposition is
valid. Thus, by use of Fourier integrals, the solution obtained here may be
extended to obtain solutions for compressible flow over arbitrarily shaped
surfaces.
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y
(b) x (d)
Linearized

x

(C)i Linearized

(N theory
x 0 Mo,
FIGURE 13.2

(a) Wave-shaped wall, (b) surface pressure coefficient for subsonic flow and (c) supersonic flow,
and (d) drag coefficient versus Mach number.

13.6 Prandtl-Glauert Rule for Subsonic Flow

Using small-perturbation theory, it is possible, by means of a simple transfor-
mation, to reduce all subsonic-flow problems to equivalent incompressible-
flow problems. The rule that results from such a transformation is called the
Prandtl-Glauert rule.

For subsonic flow over a body whose surface is defined by y = f (x), the per-
turbation velocity potential must satisfy the following problem:

;e 1 PP
ax*  1-M2 oy’

9 o-udY
oy (x,0)=U (%)

%(x,y)=ﬁm’te asy — oo.
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Introduce a new velocity potential ®”and a new vertical coordinate n that
are defined as follows:

oo 1

J1- M2
n=y1-Mzy.

Then, the problem to be solved for ®/x, n) is the following:

PO 9D

o om? =0

od df
(x,00=U2L

P, (x,0) dx(x)
&(x,n)=ﬁnite as 1 — oo,
ax

That is, in the xn plane, the problem to be solved is that of irrotational
motion of an incompressible fluid about a body whose surface is defined by
n = f{x). Assuming that the ideal-fluid flow problem can be solved, the cor-
responding pressure coefficient may be evaluated from Equation 13.4b. Thus,
denoting the incompressible pressure coefficient by C,, it follows that

D
c,--292
U ox
But the compressible pressure coefficient is given by

c - 20®
U ox

1 2040

N /1_]\/102o E?
that is,

_ Gy

2

0

C,(x,y) (13.8)
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That is, the pressure distribution around a body in subsonic compressible
flow may be obtained from the corresponding incompressible pressure dis-
tribution. The rule that connects these two pressure distributions (Equation
13.8) is known as the Prandtl-Glauert rule. It establishes the effects of com-
pressibility for subsonic flows and illustrates that, within the linear theory,
any subsonic compressible-flow problem may be solved provided that the
corresponding incompressible-flow problem may be solved.

13.7 Ackeret’s Theory for Supersonic Flows

Small-perturbation theory may also be used to establish a theory for super-
sonic flows. The resulting theory is known as Ackeret’s theory. The situation to
which this theory applies is shown in Figure 13.3a. Supersonic flow approaches
a thin, cambered airfoil that is at an angle of attack « to the free stream whose
Mach number is M... The chord of the airfoil is denoted by ¢, t is the maximum
thickness, and # is the maximum camber of the airfoil. The equation of the
upper surface of the airfoil is i = 1,(x), whereas that of the lower surface is 1,(x).

According to the linearized theory, the problem to be solved for the pertur-
bation velocity potential is the following;:

’D 1 o
2 T a2 2 =0
ox® M, -1y

0P dn
—(x,0=U—
oy 0 =U G

%(x,y)=ﬁnite as y— .

(@) (b)

y= nu(x)

FIGURE 13.3
(a) Parameters for supersonic airfoil and (b) definitions of the half-thickness function t(x) and
the camber function y(x).
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Since, in general, the surfaces y = n,(x) and y = n,(x) will be different, the
boundary condition on y = 0 will be different for the upper and lower sur-
faces, so that the corresponding values of ® will be different. Denoting these
solutions by ®, and @, it therefore follows that the two solutions will be

CI>M(x,y)=f(x—,leo —1y)
<I>,(x,y)=g(x+./M§, —1y).

Here, the left-running solution has been omitted from ®,, and the right-
running solution has been omitted from ®,. This satisfies the condition that
signals can travel downstream only in supersonic flow, so that the lines
along which signals travel must slope downstream as they move away from
the airfoil.

The functions f and ¢ may be evaluated by imposing the boundary condi-
tions at the surface of the airfoil. Thus, the boundary conditions at y = 0+ and
y = 0—, together with the corresponding solutions, give

___ U dn,
u dn,

g (x)= Wa(@-

It is not necessary to integrate these expressions in order to evaluate the
pressure coefficient. From Equation 13.4b

2 9P

Pouoax

Thus, denoting the pressure coefficient on the upper surface by CTM and
that on the lower surface by C,, it follows that

2 2 dn
C,=-2 =< Tl
ET R ME 1 dx
= 2 2 dn
C,=-=% - I
pl ug(x) g i
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These results show that the local value of the pressure coefficient is pro-
portional to the local slope of the airfoil surface.

Using these values of the surface pressure coefficient, the lift coefficient of
the airfoil may be evaluated as follows:

C 1 fo(rfz-m)dx
L

1 c
= muz
29

Here, p., is the density of the fluid far from the airfoil, and p, and p, are,
respectively, the pressure on the lower and the upper surface of the airfoil.
Then, from the definition of the pressure coefficient, it follows that

CL-%LC(C?—CTM)dx

f dn, dm dx

J 2 _q
#[m +m,]o-
M2 -1 '

But n,(c) = n,(c) = 0 and 1n,(0) = n,(0) = ac. Hence, the value of the lift coef-
ficient is

c - 2 (13.9a)

JM2 -1

Equation 13.9a shows that the lift force that acts on a supersonic airfoil
depends only on the Mach number of the flow and on the angle of attack of
the airfoil. That is, the lift force is independent of the camber and thickness
of the airfoil. This result is quite different from the corresponding result for
subsonic flow. Indeed, Equations 4.25b and 13.8 show that the lift coefficient
for a Joukowski airfoil at subsonic speeds is

C=—2"_ 1+077% sin as2”
1-M2 ¢ ¢

Thus, the lift of subsonic airfoils is greatly affected by airfoil thickness and
camber, but the lift of supersonic airfoils is not affected by these parameters.
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The drag coefficient for the airfoil may be evaluated in a similar manner
as follows:

o

[, (-p)ay

1
= wUZ
2p

Cp =

This integral may be converted to one in x by noting that dy = (dy/dx)dx,
where dy/dx = dn,/dx on the upper surface and dy/dx = dn,/dx on the lower
surface of the airfoil. Hence

1 4% =dn dn
== C,=—L_C, =" dx.
o Cﬁ i’ dx ™ dx dx

Using the expressions that were derived for the pressure coefficient on the
upper and lower surfaces shows that

2 2

dn, dn, dx.

.
Com i) @t w

Since the integrand of this integral is positive definite, it is evident that a
nonzero drag will exist for nontrivial airfoil shapes, a result that was deduced
for the wave-shaped wall in Section 13.5.

The manner in which airfoil thickness and camber affect the wave drag
may be established by writing the equations of the upper and lower surfaces
of the airfoil in terms of the corresponding parameters. Thus, let the thick-
ness parameter and the camber parameter be defined, respectively, by

Qs o=

Here, t and & are the maximum thickness and maximum camber, respec-
tively, as shown in Figure 13.3a. Then, a half-thickness function t(x) is defined
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such that the local value of the airfoil half thickness is dct(x), as shown in
Figure 13.3b. Likewise, a camber function y(x) is defined such that the local
value of the airfoil camber is ecy(x), which is also shown in Figure 13.3b.
Thus, the upper and lower surfaces of the airfoil may be defined in terms
of the angle of attack, the half-thickness function, and the camber function.
From the definitions of these two functions, they must lie in the following
range:

Osr(x)s%

0<y®) <l

In terms of the functions defined above, the equations of the upper and
lower surfaces of the airfoil are

n,(x) = alc — x) + ecy(x) + dct(x)

,(x) = alc — x) + ecy(x) — det(x).
That is, the upper and lower surfaces are defined by the line through the
mean thickness of the airfoil, plus or minus the half thickness, respectively.

Thus, the integrand of the integral that defines the drag coefficient may be
evaluated as follows:

%=—a+scy +0ct
dx
m=—oc+scy -dct
dx
2 2
dn,  dn 500, 2e%c*(y ) +28%c* (v )* - doecy .
dx dx

In the foregoing equations, the primes denote differentiation with respect
to x. Substituting the last result into the expression for the drag coefficient
yields the following result:

C
Cp = #I 202 + 232c2(y )2 + 262C2(‘c )2 —4oecy dx.
Mz -1Jo

C\/i
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The first term in the integrand is a constant and may be integrated directly.
The last term in the integrand integrates to zero since y(0) = y(c) = 0. Thus, the
drag coefficient becomes

40* 4¢%c

¢ 49°c
D=\/Mi_1+\/Mi_1L(Y )2dx+ML(T)2dx' (13.9b)

C

Equation 13.9a showed that the lift coefficient of supersonic wings is inde-
pendent of the camber and the thickness of the airfoil. On the other hand,
Equation 13.9b shows that both camber and thickness increase the drag coef-
ficient of such wings. Hence, it may be concluded that supersonic wings
should be as straight as possible and as thin as possible. However, it is evi-
dent that for structural reasons, there is a limit to the minimum thickness
to which such wings may be made. Apart from the general guidelines of
minimizing the camber and thickness of wing sections, application of the
foregoing theory to specific airfoils shows that sharp corners are prefera-
ble to rounded corners in supersonic flight. The investigation of the perfor-
mance of specific airfoil sections is deferred to the problems at the end of the
chapter.

13.8 Prandtl-Meyer Flow

In this section, an exact solution to the equations of two-dimensional flow of
a compressible fluid will be derived. The flow situation to which this solution
refers consists of supersonic flow approaching a sharp bend in a boundary in
which the boundary bends in such a direction that an expansion, rather than
a compression, is required to turn the fluid. The resulting flow field is called
the Prandtl-Meyer flow.

Figure 13.4a shows the flow configuration that is under consideration.
Supersonic flow whose Mach number is M, flows parallel to a boundary
that suddenly changes direction, as shown in the figure. In order to satisfy
the boundary condition at the surface, the velocity vector must be deflected
in the direction indicated. Since this deflection is opposite in sense to that
which was shown to be necessary for shock waves, it may be concluded
that an expansion, rather than a compression, is required. Although expan-
sions are continuous processes as opposed to the discrete processes of shock
waves, the expansion is illustrated as consisting of a large number of very
weak expansion waves. This is known as a Prandti-Meyer fan.

An arbitrary point in the expansion fan is indicated in Figure 13.4a. The
Mach number at this point is M, and the deflection of the velocity vector at
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Agsin (f + )

(B +6)
Ag cos (p +0)

FIGURE 13.4
(a) Prandtl-Meyer fan and (b) velocity change through a typical Mach wave.

this point, relative to its original direction, is denoted by 0. The inclination of
the Mach wave that passes through the point under consideration is denoted
by B. Then, it is known that the leading Mach wave will subtend an angle
defined by

g1
=sin~ — .
B1 M,

Since the pressure gradient will be normal to each of the Mach lines, the
changes in the velocity must also be normal to the Mach lines. Thus, if g
denotes the magnitude of the velocity vector as it approaches our reference
Mach wave and if Aq denotes the change in the value of g that is caused by
the Mach wave, the velocity diagram will be as shown in Figure 13.4b.

The velocity vector that emerges from the Mach wave will have a magni-
tude g + dg, and it will have been deflected through an angle d6. Since Ag
will be infinitesimally small as the limit of an infinite number of Mach waves
is approached, the deflection of the velocity vector may be approximated by

_ Aq cos (B +0)

de
q+dg

In the limit, all second-order terms will vanish identically, so that this
expression may be further reduced to

de = Aqqcos B +0).
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But, from Figure 13.4b,
q+dg=q+ Agsin (p +6).

Using this result to eliminate Ag yields the following equation for dé:
de = ﬂcot(|3 +0).
q

But it is known that the inclination of the Mach wave under consideration
is given by

sin(p +0) = %

and hence
cot(p+0)=vM?> - 1.

Thus, the turning angle of the velocity vector becomes
do=vM*-1 dq
q

In order to complete the solution, this equation must be integrated to yield
an expression for 0 in terms of either g or M. Choosing the latter as being
more relevant, § must be expressed in terms of the local Mach number M.
This may be done by use of the definition of the Mach number in the form

g =aM
and hence

dg _da dM
g a M’

To eliminate a from this equation, the energy equation will be employed
in the form




Multidimensional Flows 497

Or, multiplying this equation by (y — 1)/a* gives

2
\{_1M2+1=g—O
2 a?
2 ”3

a- =
1+ (y-1)/2 M?

and

—ay(y -1)M dM '
{1+ (v =1)/2 1\/12}2

2ada =

Thus, the expression for dg becomes

dg__y-1_ MdM_ dM
q 2 1+ ¢-1)/2 M* M
1 dM

T14 (y-1/2 M* M

Using this result to eliminate q from our expression for the element of
turning angle do gives

0= M’ -1 ﬂ
1+ (y-1)/2 M* M~

This equation may now be integrated to give

0 = o(M) — v(M,) (13.10a)

o(M) = /Htan‘l /E(Mz—l) —tan‘l(\/Mz—l). (13.10D)

Equation 13.10b defines the so-called Prandtl-Meyer function, which is
denoted by v(M). The solution 13.10a gives 6 = 0 for M = M, and represents

where
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a monotonically increasing function of M for values of M > M. Then, the
minimum value of 0 occurs at M, = 1, with the value of 6 being zero, and the
maximum value of 6 occurs when M tends to infinity. From Equations 13.10a
and 13.10b, this gives the following maximum:

0 y+1

7T
z 1.
max 2 (1310C)

y-1
For y = 14, this gives a maximum deflection of approximately 130°.

PROBLEMS

13.1 Show that the equation to be satisfied by the velocity potential for
steady, two-dimensional, irrotational motion of an inviscid fluid is

2 2 2 2 2
12 00 Lm0 T g w0,
a- ox a” dxdy a- oy

13.2 Introduce a stream function that is defined as follows:

d d
p”=p()7w pv=p()7w
ay 9x

where p, is a constant reference value of the density. Show that this
stream function identically satisfies the continuity equation for
steady two-dimensional motion of a compressible fluid, and that
for irrotational motion, the equation to be satisfied by y/(xy) is

2 2 2 2
(P L
a- ox

612 axay (12 ayz

13.3 The stream function that is defined in Problem 13.2 may be consid-
ered to be a function of the magnitude of the velocity vector g and
its angle 0. That is, we can consider the stream function to be y (g, 6)
where u = g cosb and v = g sin6. To obtain the differential equation
to be satisfied by y/(g, 6), proceed as follows:

(@ Obtain expressions for d¢ and dy in terms of dx and dy in
which the coefficients are functions of g, 6, and the density ratio
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only. Invert these equations to express dx and dy in terms of
d¢ and dy.

(b) Use the fact that the velocity potential is ¢(g, 6) and that the
stream function is (g, 0) to eliminate d¢ and dy in the expres-
sion obtained above in terms of their derivatives with respect to
g and 6.

(o) Considering both x and y to be functions of g and 6, obtain
expressions from differential calculus for dx and dy. By equat-
ing these expressions with those obtained in (b) above, obtain
expressions for the partial derivatives of x and y with respect to
g and 6.

(d) Eliminate both x and y from the results obtained in (c) by form-
ing the second mixed derivatives of x and y with respect to g and
0. In this way, obtain expressions for the following derivatives:

% %
90 aq

in terms of derivatives of the stream function (g, 0). In deriv-
ing this result, it should be noted that the density p is a function
of the magnitude of the velocity g, but not of its direction 0.

(e) Use the Bernoulli equation for steady flow and the definition of
the speed of sound to obtain the result

d py pq
dg p po

From this result, and one of the expression obtained in (d),
show that

ai):_& 1 MZ)aw
aq p g 20

Finally, eliminate ¢ from this expression, using a result
obtained in (d), to show that the following equation is to be sat-
isfied by the stream function:

, Y ER)
1+ M? +(1- M? =0.
7 o +t1( + ) P +( e
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13.4 In the Janzen—Rayleigh expansion, find the differential equation to
be satisfied by the function ¢; in the series.

13.5 The linearized form of the pressure coefficient is defined by the
equation

u
Cp = —ZU

Find the next correction term in the expression for C,. That is,
find the approximate expression for the pressure coefficient that is
valid to the second order in small quantities.

13.6 Using the results of the linearized theory for compressible flow
over a wave-shaped wall, integrate the pressure over one wave-
length of the wall and so verify that the drag is zero for subsonic
tlow. Also, calculate the drag per wavelength per unit width of the
wall for supersonic flow over the wall.

13.7 An infinitely long cylinder of radius a + & sin(2xx/}) is exposed to a
uniform axial flow of compressible fluid, as shown in the figure for
this problem. If the flow is subsonic and if the following conditions

apply

E<<1 and E<<1
A a

then the conditions to be satisfied by the perturbation velocity
potential are as follows:

2 2
(1= M2 z—‘f+a—¢+1 % _o
X

ar* v oor

ad dr
—(x,a)=U—.
ar (x,a) dx
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The flow velocity is also subject to the condition that it should be
everywhere finite. In the foregoing expressions, r is the value of r
that corresponds to the surface of the cylinder.

Using a linearized theory, find an expression for the pressure
coefficient on the surface of the cylinder. Form the ratio of this pres-
sure coefficient to that for a wave-shaped wall, and by expanding
this ratio in powers of A/[a(1- M2)], establish the effect of wall
curvature.

13.8 Use Ackeret’s theory to find the drag coefficient of the double-
wedge airfoil shown in the figure for this problem for zero angle of
attack in supersonic flow.

1
2%

‘ c ‘

139 The half-thickness function t(x) for the biconvex circular-arc airfoil
shown in the figure for this problem is given by the expression

o) = ).

N =

\4—
=
&

The equation of the upper surface is defined by the equation

2 2
1
ral+ x-S = as—dc
(+a) 7 7
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Using Ackeret’s theory, evaluate the drag coefficient of a bicon-
vex circular-arc airfoil at zero angle of attack in supersonic flow.
Compare this result with that for the double-wedge airfoil dis-
cussed in the previous problem.

Further Reading—Part IV

The compressible-flow area of fluid mechanics is not as well endowed with
books as the other areas. However, the following materials adequately cover
and extend the material treated in Part IV of this book.

Ames Research Staff: Equations, Tables and Charts for Compressible Flow, National
Advisory Committee for Aeronautics, Report 1135, 1935.

Anderson, J. D.: Modern Compressible Flow: With Historical Perspective, McGraw-Hill,
New York, 2003.

Liepmann, J. W., and A. Roshko: Elements of Gas Dynamics, John Wiley & Sons, New
York, 1957.

Lighthill, J.: Waves in Fluids, Cambridge University Press, London, 1978.

Shapiro, A. H.: The Dynamics and Thermodynamics of Compressible Fluid Flow, Vols. 1
and 2, The Ronald Press Company, New York, 1953.

Thompson, P. A.: Compressible-Fluid Dynamics, Rensselaer Polytechnic Institute Press,
1988.



Part V

Methods of Mathematical
Analysis

In Part V, we summarize the methods of mathematical analysis that are
used extensively in this book. The coverage is by no means complete or even
extensive. Only a few of the commonly used methods of analysis are typi-
cally used in fluid flow problems, and we restrict ourselves to the methods
that are employed in this book. Within the topics that are discussed, the cov-
erage is not intended to be complete. For the fundamental topics, the pre-
sentation is essentially a summary. For the more advanced topics, a slightly
more detailed coverage is presented.

Part V consists of a single chapter, Chapter 14. This chapter consists of the
selected topics as outlined above and as described in more detail in the intro-
ductory comments relating to the chapter.






14

Some Useful Methods of Analysis

The topics that have been selected for coverage herein are as follows:

¢ Fourier series

¢ Complex variable theory

e Separation of variable solutions
¢ Similarity solutions

¢ Group invariance methods

When solving differential equations in general, and partial differential
equations in particular, it is necessary to express both the solution to the dif-
ferential equation and the applied boundary conditions in the same form or
“language” as each other. For example, in Chapter 6, we solved surface wave
problems in which both the solution to the Laplace equation and the bound-
ary condition at the free surface were expressed in terms of trigonometric
functions. Sometimes we can obtain the solution to the differential equation
in the form of an applied boundary condition, and on other occasions, we
write the boundary condition in the same form as the solution to the dif-
ferential equation. In both cases, we end up with a solution to the differen-
tial equation and to the applied boundary condition, both being in the same
form or “language.”

Fourier series has been selected as a relevant topic because it is widely
used to express boundary conditions in the same language as the solution to
the differential equation. We can express a given function that may appear in
a boundary condition in several ways. For example, the function sin x may be
expressed as a series, as a derivative, or as an integral as follows:

2n+1

sinx = ;(—1)"(23:“_1)! = —%(cosx) =fcosxdx.

Frequently, the solution to the differential equation is in the form of trigono-
metric functions, and Fourier series permits us to also express boundary condi-
tions in the form of trigonometric functions. One of the most common examples
involves graphical representations of a variable on a boundary. Fourier series
permits us to convert the graphical representation to a trigonometric series in
order to match the form of the solution to the differential equation.

505
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Complex variable theory is used extensively in Chapter 4, where it leads
to a variety of very useful and impressive solutions to the Laplace equation.
This is followed by the most commonly used method for solving partial
differential equations, namely, separation of variables. The emphasis of the
treatment presented herein is to arrive at the appropriate form of solution
quickly through observation of the boundary conditions and adherence to
the rules governing solutions obtainable by this method.

There are not as many methods available for solving nonlinear differential
equations as there are for solving linear differential equations. Two methods
that are applicable to nonlinear differential equations are used in this book,
namely, similarity solutions and group invariance methods. Similarity solu-
tions are introduced in Chapter 7 and used extensively in Chapters 9 and
10. Group invariance methods are used to reduce the ordinary differential
equations obtained through similarity techniques to their simplest possible
form. In this way, any numerical solutions that are required involve only the
simplest of numerical techniques.

14.1 Fourier Series

If a function, call it f(x), is periodic over some distance 2L, then the Fourier
theorem states that it may be represented by a series of trigonometric func-
tions of various amplitudes. That is, f(x) may be represented by the following
series:

flx)=ay+ 2 a, cos% + E b, sin n—:x (14.1a)
n= n=1

The coefficients in this expression are obtained by utilizing the orthogonal
properties of the trigonometric functions. The orthogonal properties are as
follows:

L
f Cos@sinn—:xdx =0forallmand n
-L

- mmx nmx Oform=n
f_LCOSTCOSde = {Lform=n

L ommx . nnx Oform=n
f_Lsm L sdex={Lform=n.
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Hence, multiplying both sides of Equation 14.1a first by cos (mnx/L) and
then by sin (mnx/L), and integrating over the interval —~L < x < L in each case,
the following values for a,, 4,, and b, are obtained:

n/

4y = ﬁ f LL f(x)dx (14.1b)
a, = % fL £(x) cos%dx (14.10)
b, = % f : £(x) sin%dx. (14.1d)

In the above equations, we have taken the integration to be over the inter-
val ~L < x < L. There is no loss of generality in this choice of interval because
we know that the function f(x) is periodic between x = x; and x = x,, where
these two locations are a distance 2L apart, as illustrated in Figure 14.1a.
Then, we can choose x; = —L and x, = L.

Equation 14.1b shows that the quantity a,, by virtue of its definition, corre-
sponds to the mean value of f(x) over the interval —L < x < L. For the particular
function illustrated in Figure 14.1a, the value of a, will be zero, but this will not
necessarily be the case in general. Equations 14.1c and 14.1d are valid for n > 1.

(@) Sx)

R
888

e e

FIGURE 14.1
Square wave of (a) general form, (b) even form, and (c) odd form.
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In many cases involving fluid flow problems, it is possible to represent the
desired quantity using only the even or the odd component of the Fourier
series. Figures 14.1b and 14.1c show, respectively, the even and odd represen-
tations of a square wave where we are interested in only the shaded part of
the function. By choosing the origin of the coordinate system as indicated,
we may represent the square wave by a cosine series alone or by a sine series
alone. In such cases, the symmetric or antisymmetric nature of the function
f(x) may be invoked to simplify the series represented by Equation 14.1a.

Consider first the symmetric case illustrated in Figure 14.1b by a typical
even function—the square wave. If the function f(x) is even in x, it follows
that f(x) = f(—x) over the interval —L < x < L. Hence, the integration indicated
in Equation 14.1b may be evaluated over the half range 0 < x < L, and the
answer doubled. Likewise, the integration indicated in Equation 14.1c may
be evaluated over the same half range, and the answer doubled. In Equation
14.1d, the integrand involves the product of an even function f(x) and an odd
function sin (nnx/L), so that b, will be zero for all values of n. That is, for an
even function in x, the Fourier series for f(x) over the interval —L < x < L may
be represented by the following series:

©

fx)=a,+ 2 a, cos% (14.2a)
where
1 L
a3 =7 fo f(x)dx (14.2b)
L
a, = %fo f(x) cosn—zxdx. (14.2¢)

Equation 14.2c is valid for n > 1. For the particular function illustrated in
Figure 14.1b, the following specific values apply:

+H forO<x<L/2

f(x)=
-H forlL/2<x=<L.

Substituting these values into Equations 14.2b and 14.2c produces the fol-
lowing results:

a,=0 and a,,=£sinﬂ for n=1.
nw 2
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The coefficients 4, in this case cycle through values that are positive, then
zero, and then negative as n increases. Thus, Equation 14.2a for this specific
case becomes

flx)= ——sin— cos——. (14.3)
Ly nn

Although Equation 14.3 is valid over the entire interval -L < x < L, we typi-
cally use it only over the subinterval -L/2 < x < L/2 to represent a boundary
value, as indicated in Figure 14.1b.

Next, consider the asymmetric case illustrated in Figure 14.1c by a typical
odd function—the square wave again, but with a shift in the origin of the
coordinate system. If the function f(x) is odd in x, it follows that f(x) = — f(—x)
over the interval —L < x < L. Hence, the integral indicated in Equation 14.1b
will be zero. Likewise, the integral indicated by Equation 14.1c will be zero
because the integrand is odd in x. Finally, the integral in Equation 14.1d may
be evaluated over the half range 0 < x < L, and the answer doubled because
the integrand is even in x, being the product of the odd function f(x) and the
odd function sin (nnx/L). That is, for an odd function in x, the Fourier series
for f(x) over the interval —L < x < L. may be represented by the following series:

f(x) = 2 b, sin% (14.4a)
where b =2 F(x)sin 2 dx (14.4b)
5 J; v, .

Equation 14.4b is valid for n > 1. For the particular function illustrated in
Figure 14.1c, the following specific values apply:

fly=HforO0O<x<L.

Substituting this value into Equation 14.4b produces the following result:
b, = 4H for n=1.

n
ni

Thus, Equation 14.4a for this specific case becomes

o 4H | nmx
f(x) = 2 sintE (14.5)
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Although Equation 14.5 is valid over the entire interval —L < x < L, we typi-
cally use it only over the subinterval 0 < x < L to represent a boundary value,
as indicated in Figure 14.1c.

The Fourier series corresponding to a number of frequently encountered
shapes are presented in Appendix D for reference purposes.

14.2 Complex Variables

The theory of complex variables is quite extensive, and we use only some of
the mainstream results of the theory in Chapter 4. In Chapter 14, we review
and summarize some of the material that is employed in this book.

14.2.1 Analytic Functions

A function F(z) of the complex variable z = x + iy is said to be analytic if the
derivative dF/dz exists at a point z, and in some neighborhood of z,, and if
the value of dF/dz is independent of the direction in which it is calculated. A
singular point of the function F(z) is any point at which F(z) is not analytic. If
F(z) is analytic in some neighborhood of the point z,, but not at z, itself, then
2, is called an isolated singular point of F(z).

If F(z) is analytic, then dF/dz will exist and may be calculated in any
direction. Thus, if we evaluate dF/dz along the line y = constant, we get
dF/dz = JF/ox, and if we evaluate it along the line x = constant, we get
dF/dz = 0F/d(iy) = —i 0F/dy. That is,

Hence, if F(z) = ¢(x,y) + iy(x,y), where both ¢p(x,y) and y(x,y) are real func-
tions, then we conclude from the line above that

dF 90 0% _ ;00 0w

dz d9x  ox dy Iy

Then, equating the real and imaginary parts on both sides of the last equal-
ity shows that

9% _ v

14.6
ox oy (14.62)
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% __ (14.6b)

These equations are called the Cauchy—Riemann equations, and they
are necessary, but not sufficient, conditions for a function to be analytic.
Eliminating first ¢, and then y, from the Cauchy—Riemann equations shows
that both ¢ and y are harmonic functions. That is, for an analytic function,
both the real and imaginary parts of the function must satisfy the Laplace
equation:

2400 (14.60)

2 2
L (14.6d)
0x ay
Many functions are analytic but assume more than one value at any point
on the complex plane, z = Re®, as 6 increases in multiples of 2x. This difficulty
or ambiguity may be overcome by replacing the single complex plane, which
is valid for all values of 6, by a series of Riemann sheets, which are connected
to each other along a branch cut that runs (usually along the negative real
axis) between two branch points (frequently z = 0 and z = ), which are sin-
gular points of the function. Then, the principal value of the function may be
considered to be valid over the range -t < 0 < m, and each additional Riemann
sheet will apply to a different range of 6, such as n < 6 < 3r, etc. In this way, the
square root function will have two Riemann sheets, the cube root function
will have three Riemann sheets, etc.

14.2.2 Integral Representations

There are several important integral theorems associated with complex
variable theory. In this book, we do not use them directly, but we do use
some important results that require these integral theorems for their proof.
Therefore, the integral theorems are recorded here, but without an extensive
proof.

One of the most fundamental integral theorems is the Cauchy integral theo-
rem or the Cauchy—Goursat theorem. This theorem states that if F(z) is a func-
tion that is analytic at all points inside and on a closed contour C, then

f F(z)dz = 0. (14.7)
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This theorem is established by using P = ¢ and Q = y in the following form
of Green’s theorem:

f(de+Qdy) ﬂ’ oo dedy

Then, using a second application of Green’s theorem with P =y and Q = ¢
leads to the result quoted in Equation 14.7.

Another important integral result is the Cauchy integral formula. This for-
mula states that if F(z) is analytic at all points inside and on a closed contour
C, and if z, is any point inside C, then

1 F(2)
F(zy) = o fc . dz (14.8a)
and
dnn (zo) = ot &dz forn=1. (14.8b)

23121 C(Z Z)n+1

The proof of this formula uses the Cauchy integral theorem, Equation 14.7,
with the contour as illustrated in Figure 14.2. The contour of integration con-
sists of an outer contour C and an inner contour C, with these two contours
joined to form a single contour through two straight lines that join C and C,,.
The inner contour C, encloses the reference point, z,, at which the function

Co Zq

FIGURE 14.2
Contour of integration for evaluating a function at the point z, using the Cauchy integral
formula.
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F(z) is being evaluated. The integrations along the two straight line segments
cancel one another, and the result presented in Equation 14.8a is achieved.

14.2.3 Series Representations

There are two types of series representations for a function that are of use:
the Taylor series and the Laurent series. The Taylor series says that if F(z) is
analytic at all points within a circle r < r, whose center is at z,, then F(z) may
be represented by the following series:

(z-z))* d°F

F@) = Fla)+ (- 2) L o)+

where the radius of convergence is the distance from the point z; to the near-
est singularity. The general form of this series, as given above, is known as
the Taylor series, and the special case z, = 0 is known as Maclaurin’s series. The
proof for the case of complex variables is the same as that for real variables.

If a circle of radius 7 is considered, where r > r; as defined above, there will
be at least one point within the circle for which the given function is not ana-
lytic. That is, the function will have at least one singular point, and the Taylor
series of Equation 14.9 will not be valid. In such cases, we must represent the
given function by a Laurent series, which is defined as follows.

If F(z) is analytic at all points within the annular region r; < r < r, whose
center is at the point z,, then F(z) may be represented by the following series:

F= Yoz sy (14.10a)
n= prm (Z - ZO)
where
_lr PO g,
" 2mi fcz (€2 d§ n=0,1,2,.. (14.10b)
and

1 E)

= dg n=-1,2,.... 14.10c
2i)e 6=z (1409

n

As shown in Figure 14.3, the contour C; corresponds to the inner radius
r =r,, and the contour C, corresponds to the outer radius r = r, of the annular
region. The quantity  is a variable of integration. The series is convergent



514 Fundamental Mechanics of Fluids

G

FIGURE 14.3
Contour of integration for establishing the Laurent series about the point z,.

if the smallest inner radius r, and the largest outer radius r, are such that
there is no singularity in the region r; < r < r,. The part of the series that
contains the b, coefficients is known as the principal part. The general form
of the series is known as the Laurent series, and in the special case in which
r, can be extended to zero (i.e., there is no principal part), the series becomes
the Taylor series.

The starting point for the development of the Laurent series is the follow-
ing identity:

l1=(1-a)+a(l-o)+o*(1-a)+--+a" (1-a)+a”.
Dividing both sides of this equation by (1 — «) produces the following result:

L4«
=l+a+a’+-+a""+ )
l-a l-a

If an appropriate choice of the quantity « in this identity is used in the
Cauchy integral formula, Equation 14.8a, the results quoted above are
obtained. The Laurent series is a very useful and important representation
of a function F(z), and Equation 14.10a is encountered in Chapter 4.

14.2.4 Residues and Residue Theorem

The residue of a function at a point z, is defined as the coefficient b, in its
Laurent series about the point z,. That is, the residue of the function F(z) at
the point z, is the coefficient of the term 1/(z — z,) in the Laurent series of the
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function as written about the point z,. The significance of this definition lies
in the residue theorem, which is a powerful theorem that can be used in the
evaluation of integrals in the complex plane. The residue theorem may be
stated as follows.

If F(z) is analytic within and on a closed curve C, except for a finite number
of singular points z,, z,,..., z,, then

f F(z)dz = 2mi(R, + R, + -+ R,) (14.11)
C

where R, is the residue of F(z) at z;, R, is the residue at z,, and R, is the residue
at z,. In order to evaluate the residue of a function at some point, it is useful
to know the type of singularity that exists at that point.

Branch points: These are the singular points at the end of each branch

cut of a multivalued function. For example, the function 1/4/1- Z2
has branch points at z = + 1. The residue theorem does not apply at
branch points.

Essential singular points: If the principal part of the Laurent series of the
expansion of a function about some point contains an infinite num-
ber of terms, that point is an essential singular point. For example,
the function log z has an essential singularity at z = 0.

Pole of order m: If the principal part of the Laurent series of expansion of
a function about some point contains only terms up to (z — z,)", that
point is a pole of order m. That is, if F(z,) is a pole of order m, then
(z — zy)"F(z,) will be analytic.

Simple pole: If the principal part of the Laurent series of the expansion
of a function about some point contains only a term inversely pro-
portional to z — z,, that point is a simple pole. That is, if F(z,) is a
simple pole, then (z — zy)F(z,) will be analytic.

The following methods may be used to calculate the residue of a function
F(z) at a singular point z:

1. Expand F(z) in a series about z; and so obtain the coefficient of the
term 1/(z — z,). This fundamental method uses the definition of the
residue and is valid for all types of singularities.

2. If the point z, is a simple pole, the residue may be calculated by tak-
ing the following limit:

R = !j_I)zl(Z - 2y)F(2). (14.12a)
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3. If the point z; is a pole of order m, the residue may be calculated by
taking the following limit:

m-1
— lim 1 d
2=z (m - 1)! dz™!

[(z - z))"E(2)]. (14.12b)

4. If F(z) may be put in the form F(z) = p(2)/q(z) where q(z,) = 0 but
dg/dz(z,) # 0, and where p(z) # 0, the residue may be calculated by
taking the following limit:

o
— 14.12
=z dg/dz (14.12¢)

The residue theorem is one of the most useful theorems in the analysis of
potential flow problems, and it is used to great advantage in Chapter 4 to
calculate the fluid forces and moments that act on bodies.

14.2.5 Conformal Transformations

A conformal transformation is a mapping from the z plane to the { plane
of the form z = f({), where f is an analytic function of {. Conformal transfor-
mations preserve angles between small arcs except at points where df/d¢ =
0. Such points are called critical points of the transformation, and smooth
curves through such points in the { plane may give angular corners in the
z plane.

Useful conformal transformations are provided by the elementary func-
tions, the Joukowski transformation, and the Schwarz—Christoffel trans-
formation. The elementary functions and the Joukowski transformation
are covered in detail in Chapter 4. The Schwarz—-Christoffel transformation
maps the interior of a closed polygon in the z plane onto the upper half of the
¢ plane, whereas the boundary of the polygon maps onto the real axis of the
¢ plane. The transformation is of the following form:

jz = KC-a)"™ '€ -bf™ ' (C-c)™" . (14.13)

Here, the vertices A, B, C, etc., in the z plane subtend the interior angles o, f3,
y, etc,, as indicated in Figure 14.4, and 4, b, ¢, etc,, are the points on the real axis
of the { plane corresponding to the vertices A, B, C, etc. Since the polygon in
the z plane is closed, the angles o, f, y, etc., must satisfy the following relation:

a+PB+y+-=(n-2)m.
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b

z plane { plane

S e

\

FIGURE 14.4
Corresponding regions in the original plane (z plane) and the mapped plane (¢ plane) for the
Schwarz—Christoffel transformation.

Here, 1 is the number of vertices in the polygon. The constant K determines
the scale of the polygon and its orientation, whereas the constant of integra-
tion determines the location of the origin in the z plane. Of the constants 4, b,
¢, etc., any two may be chosen arbitrarily, for example, (-1,1) or (0,1), and any
remaining ones will be determined by the shape of the polygon. This trans-
formation is used extensively in Chapter 4.

14.3 Separation of Variable Solutions

The three types of second-order partial differential equations with two inde-
pendent variables, in their simplest or canonical form, are the following:

2 2
Fo P
ox*  dy
ou  du
& _p
it oy’

u  , 0'u
= —

e ot

The first equation above is the Laplace equation, which is of the elliptic
type, and it has no real characteristics. The second equation is the diffusion
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equation or heat conduction equation, which is of the parabolic type, and it has
one real characteristic. The third equation is the wave equation, which is of
the hyperbolic type, and it has two real characteristics.

Separation of variables is the most commonly used method for solving
partial differential equations of all three types. It is used extensively in this
book, particularly to obtain solutions to the Laplace equation in two dimen-
sions. It is useful in cases where the solution obtained by this method is
capable of satisfying the boundary conditions.

Suppose we have a dependent variable ¢ that is a function of two indepen-
dent variables x and y. If ¢ must satisfy the Laplace equation, then we must
solve the following differential equation:

We look for a “separable” solution to this equation in which each compo-
nent is a function of one independent variable only:

$x, y) = XY (y).

Substituting this assumed form of solution into the differential equation,
then dividing both sides by the product XY, gives

The essential argument is that if we change the value of x, but not the value
of y, the left side of this equation will change but the right side will not. The
only way that this equation can remain valid under these circumstances is
for each side of the equation to be equal to the same constant. Calling this
constant a2, the two resulting ordinary differential equations are

2 2
dX+(x2X=O and d’y

-a’Y =0.
dx? dy?

The first ordinary differential equation yields trigonometric solutions,
whereas the second yields exponential solutions. We note that if we had
chosen the separation constant to be —a? instead of +a?, the trigonometric
solution would have been in the y direction, and the exponential solution
would have been in the x direction. The important point to note is that the
coefficient of x and the coefficient of y in both the trigonometric and the
exponential solutions are both o, the square root of the separation constant.
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The solutions to the two ordinary differential equations above are
X =A cos ax + B sin ox (14.14a)
Y = Ce + De®. (14.14b)

The relationship that exists between exponential functions and hyperbolic
functions permits us to present the solution given in Equation 14.14b in the
following alternative form:

Y = C* sinh ay + D* cosh ay. (14.14¢)

The exponential version, given by Equation 14.14b, is the preferred form
of solution if the fluid extends to infinity in either the positive or negative
direction. However, if the fluid extends over only a finite domain, the hyper-
bolic version of the solution is preferred. This is because of the fact that one
of the two components of Equation 14.14c, namely, sinh ay, vanishes when its
argument is zero. Then, if we have a boundary condition that requires the
dependent variable to be zero for y = 0, we should use the hyperbolic func-
tion solution and take D* = 0.

By invoking the identity for a compound angle of the sinh function, we can
identify yet another useful form of the solution for Y. The identity is

sinh(P — Q) = sinh P cosh Q — cosh P sinh Q.

Hence, if P = h = constant and Q = y = the coordinate, this identity may be
written as follows:

K'sinh(h —y) = K'sinh h cosh y — K cosh i sinh y = K; cosh y + K, siny

where K; = K sinh & and K, = =K cosh h. Hence, if our solution must be zero
on y = 0, we can use the right side of the above equation with K; = 0, and if it
must be zero on y = h instead, we can use the left side of the above equality.
That is, an equivalent form of Equation 14.14c is

Y = C** sinh a(h — y) + D** cosh a(h —). (14.144d)

Generally speaking, we can determine which coordinate has the trigono-
metric solution and which has the exponential (or hyperbolic) solution by
looking at the boundary conditions. It is frequently possible to also deter-
mine the value of the separation constant by looking at the boundary condi-
tions. In this way, we are normally able to determine the functional forms
of the solution in both the x and y directions so that all that remains is to
determine the constant of proportionality. A typical example is discussed
below.
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The problem of determining the speed of a travelling surface wave on a
liquid is discussed in Section 6.3 where it is shown that the problem to be
solved for the velocity potential ¢(x,y,t) is the following:

2 2
7o, 30 _,

x> ay’

a—q)(x, 0,t)= —s@cosz—n(x —ct)
ay A A

9y = 0.
ay

In addition to the two boundary conditions listed above, there is a dynamic
boundary condition to be satisfied on the free surface, but this condition has
no effect on the solution we will obtain, so it will be omitted here in the
interests of clarity. That boundary condition determines the value of speed
of the wave c.

The first boundary condition tells us three things: it tells us that the solu-
tion should be trigonometric in x, that only the cosine function is required,
and that the value of the separation constant a in this case is 2x/A. Then,
preserving the same time dependence since it does not appear explicitly in
the partial differential equation, the solution in the x direction will be of the
functional form

2%
cos— (x —ct).
x( )

It therefore follows that the solution in y must be exponential or hyper-
bolic, and as required by the second boundary condition, the derivative of
the solution must vanish on y = —h. Therefore, using the same separation
constant as for the x direction, the functional form of the solution in the y
direction is chosen to be

cosh 2% (h+y).

That is, by merely looking at the boundary conditions, we conclude that
the appropriate functional form of the solution will be

o(x,y,t)=A cosz%(x —ct) coshz%(h +1).
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We now establish the value of the constant A by matching this solution to
the first boundary condition, not just invoking its functional form. This gives
the result

CE

sinhz—nh
A

A=—

Thus, the complete solution to the problem posed is

o(x,y,t)=- COS%(X —ct) CoshZ%(h + ).

ce
sinh %
A

This is the same value as we found in Section 6.6, that is, Equation 6.7a. If
a boundary condition is given in the form of a graph, such as a square wave
or a triangular wave, we can convert the graph into trigonometric functions
by means of a Fourier series.

If the partial differential equation is of the parabolic type, that is, where
one of the possible highest derivatives is missing, the equation to be solved
will be of the following type:

This equation was encountered in Chapter 7 where u(y,t) is the x compo-
nent of the velocity vector. If we carry out the same procedure as we did for
the Laplace equation, we will get the following result:

u(y,t) = (Acos % y + Bcos % y)e
v v

The separation constant has been taken to be —a?. If we choose the separa-
tion constant to be +a?, we will get solutions that grow exponentially with
time. Since we normally deal with situations where the dependent variable
decays with time, we reject the positive separation constant. Then, it follows
that the solution in the y direction will be trigonometric. Note that the sepa-
ration constant appears explicitly as a coefficient of ¢, but only as a square
root in the coefficient of y in the solution. As with the Laplace equation, the
boundary conditions will reveal the required value of the separation con-
stant, and if a graphical value is specified, we can represent it as a Fourier
series.
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If the partial differential equation is of the hyperbolic type, the equation to
be solved will be of the following type:

Using separation of variables shows that the two components of the solu-
tion must both be of the same functional form, namely, either trigonometric
or exponential. This is not the most suitable form for matching the solutions
to typical boundary conditions in fluid mechanics. However, an alternative
method exists that produces a general form of solution, namely, the method
of characteristics. The wave equation is discussed in Section 11.1, where it is
established that the appropriate solution will be of the following form:

u(x, t) = flx — ct) + g(x + ct).

In the above, f and g are two differential functions that are established by
the boundary conditions or the initial conditions. The first term on the right-
hand side of this solution represents a wave travelling in the positive x direc-
tion with velocity ¢, and the second term represents a wave travelling in the
negative x direction with velocity c. This form of solution is well suited for
matching the type of boundary or initial conditions that are encountered in
typical fluid mechanics applications.

14.4 Similarity Solutions

Similarity solutions are solutions that reduce the number of independent
variables by one. That is, if a similarity solution is applied to a partial differ-
ential equation with two independent variables, it will reduce the problem
to an ordinary differential equation. Unlike separation of variables, the simi-
larity solution will result in just one ordinary differential equation, not two,
and it may be applied to nonlinear problems as well as linear problems. The
method was introduced in Section 74, and it is used extensively in Chapters
9 and 10.

Similarity solutions may be applied to partial differential equations of
the parabolic type, such as the heat conduction equation or the diffusion
equation, as well as a few instances in the analysis of fluid flows. They work
where there are no physical length scales involved, so that the solutions
are all “similar” to each other. In Section 74, the Stokes’ first problem was
discussed, and it was established as requiring the solution to the following
problem for the velocity u(y,t):
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ot (14.15a)

where u(0, t) = UH()
and u(y, t) = finite.

In the above, H(f) is the Heaviside function, or unit step function, which
has the following properties:

0 fort<0
H(t) =
1 fort > 0.

Physically, Stokes’ first problem considers a semi-infinite body of fluid, cor-
responding to y > 0, which is originally at rest. The fluid is in contact with a
horizontal surface that is located along the x axis at y = 0, as shown in Figure
14.5a. The horizontal surface, which may be considered to be a movable belt,

() (b)
y

Fluid Fluid

~_ x 0
Movable surface (belt)

(c)

SE

|

FIGURE 14.5
(a) Definition sketch for an impulsively moving surface, (b) expected form of the solution
curves, and (c) typical curve that represents u/U = constant.
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is impulsively set into sliding motion with constant horizontal velocity U at
time t = O+ The anticipated motion of the fluid is shown in Figure 14.5b for a
variety of times after initiation of the motion.

In Figure 14.5b, we show a vertical line corresponding to the velocity ratio
u/U =\, which is a fixed number in the range 0 < A < 1. The dashed line inter-
sects each of the indicated velocity profiles at some time ¢; and some distance
from the belt ;. Then, it follows that the velocity ratio 1/U is constant when

y=y,andt ="t
y=1y,and t=t,
y=ysand t=1,
y=y,andt=t,

and so on.

If we plot these data points of y versus ¢, we will generate a curve like the one
shown in Figure 14.5¢c. That is, along the line shown in Figure 14.5¢, the velocity
ratio u/U will be constant. If we represent the curve in Figure 14.5¢ by the rela-
tion y ~ #*, then we can say that u/U is constant when y/#" is constant. That is

" some function of l.
u "

Hence, we should look for a solution to Equation 14.15a in which the veloc-
ity ratio u/U is a function of a single independent variable, call it n, where
n ~ y/t". That is, we should look for a solution of the following functional form:

u Yy
—= wheren=a-~-.
U f(m) n=a-,

In the above relation, a constant « has been added to permit us to make
sure that the variable 1 is dimensionless, as it should be, since the left side of
the equation is dimensionless. The variable 1 is called the similarity variable.

Generating the derivatives required by Equation 14.15a, we get

ou ay n
—=-Un =-Un—
at tn+1 f t f
ou a

g 7f

ay t

o*u o?
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Here, the primes denote differentiation with respect to n. In the first equa-
tion above, we have eliminated the independent variable y using the defini-
tion of the new variable n. Substituting these results into Equation 14.15a,
we get

2

n a
“Un—f =vU—f .
ntf v thf

Thus, if we choose the value n = 1%, this equation reduces to the following
form:

f o+ ZUnaz f =0wheren= octl%. (14.15b)

The solution to Equation 14.15b is pursued further in Section 7.4, but for
our purposes here, we can stop and say that we have been able to reduce a
partial differential equation having two independent variables to an ordi-
nary differential equation. Stokes’ first problem may also be treated by linear
methods, such as a Laplace transform.

Now, we will look at a nonlinear problem where linear methods are not
appropriate. We consider the boundary layer flow over a flat surface, which
is discussed in Section 9.3. This flow problem, posed in terms of the stream
function y(x,y), is as follows:

K Py o Y Y
dy oxdy ax ay® Y’

M 0)y=(x 0)=
0 ay (x,00=0

?;(x,y)ellasyeoo.

This is a third-order, nonlinear, partial differential equation in two inde-
pendent variables. There are no geometric length scales, so that the possi-
bility of a similarity solution exists. If we were to superimpose the velocity
profiles from different values of x, we would be led to the same conclusion
here as we were in the case of the Stokes’ first problem. Namely, the veloc-
ity profiles at different values of x exhibit the same behavior as they did at
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different values of t in the Stokes’ first problem. That is, the problem behaves
in a similar manner with the independent variable x in this case replac-
ing the independent variable ¢ in the Stokes’ first problem. Thus, we would
expect a similarity solution to exist in which

u
u- g(n) where n = axly7

where g is some undetermined function. Then, since u = dy/dy, we have

?Tw - Ug(v). (14.16a)

Then, by integration, it follows that the stream function will be of the fol-
lowing form:

) (14.16b)

where f(n) = %Bg (n)and n = axll/ﬁ'

In this last equation, a constant p has been included in the definition of the
function f, as was done with the definition of 1) in the Stokes’ first problem, in
order to make the dimensions the same on both sides of the equation. It may
be readily verified that Equation 14.16b is a solution to Equation 14.16a by dif-
ferentiating it with respect to y and verifying that dy/dy = Ug is recovered.
If we substitute the expression in Equation 14.16b into the partial differential
equation, it will be readily verified that the assumed form of y does indeed
reduce the partial differential equation to an ordinary differential equation.
The solution is developed further in Section 9.3.

Having reduced a second-order linear partial differential equation into
a second-order ordinary differential equation, and a third-order nonlinear
partial differential equation to a third-order ordinary differential equa-
tion, the question arises as to the possibility of generalizing the concepts
that were used to solve other problems. This approach was considered by
Falkner and Skan who generalized the solution presented in Equation 14.16b.
As discussed in Section 9.4, the problem posed by the boundary layer equa-
tions, for flows other than constant velocity flows approaching the surface,
requires solving the following problem:

o oty g Ity udu ER)

dy oxdy  ox ay> dx oy’
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M 2,00- Y (x,0-0
0x ay

?;(x, y)—=U(x)as y — oo.

The difference between this problem and the previous problem is that we
no longer restrict the outer velocity U(x) to be a constant. In order to solve
this problem, Falkner and Skan decided to generalize the form of the solu-
tion as presented in Equation 14.16b by generalizing the function of x that
appears in the definitions of the stream function y and in the similarity vari-
able 1. Thus, they decided to explore the possibility of the existence of solu-
tions of the following form:

v = Ux)gx)fn) (14.17a)
where 1 = % (14.17D)

and then u(x, y) = Ux)f' ().

As we show in Section 9.4, substitution of this assumed form of solution
into the partial differential equation leads to an ordinary differential equa-
tion for some specific types of outer flow velocities U(x) and some specific
functions &(x). The detailed solution is presented in Section 9.4, and all we
have tried to do here is to present a rational basis for the assumed form of the
solution specified above.

In summary, a similarity solution may exist to a particular problem if the
following conditions apply and the solution may be obtained if the proce-
dures indicated are followed:

e The partial differential equation is of the parabolic type, that is,
one of the highest derivatives does not appear in the differential
equation.

® There are no geometric length scales associated with the problem.
Examples include the Stokes’ first problem (Section 7.4) and bound-
ary layer flows (Chapter 9). If a horizontal surface at y = h is intro-
duced in the Stokes’ first problem, the geometric length scale will
preclude a similarity solution that is capable of satisfying the bound-
ary conditions. However, in such cases, a separation of variable solu-
tion will be able to match the boundary conditions through the use
of a Fourier series representation of the boundary condition.
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¢ For problems where the dependent variable is the quantity that is
“similar,” such as the velocity in the Stokes’ first problem, look for a
solution of the following form:

% = Bt" f(n) (14.18a)

y

wheren=a-~-. (14.18b)

Substitute this assumed form of solution into the partial differ-
ential equation and see if some choice of the parameters m and n
reduced the problem to an ordinary differential equation. If so, the
ability of the resulting ordinary differential equation to satisfy the
required boundary conditions may then be investigated.

e If the dependent variable is not the quantity that is “similar,” such
as in boundary layer theory, look for a solution of the form speci-
fied by Equations 14.17a and 14.17b. If substitution into the partial
differential equation leads to an ordinary differential equation, for
some choice of the quantities U(x) and §(x), then a similarity solution
exists. The ability of the solution to the resulting ordinary differen-
tial equation to satisfy the required boundary conditions may then
be investigated.

14.5 Group Invariance Methods

In this book, we use only the idea of coordinate invariance as applied to the
solution of ordinary differential equations. This material is useful in solving
the nonlinear ordinary differential equations, which are created by similar-
ity solutions to nonlinear partial differential equations, such as the boundary
layer equations. The method can be used to reduce the order of an ordinary
differential equation, including nonlinear equations, from 7 to (n — 1).

The method can be readily illustrated by considering the simple case of a
spring-mass vibration system, as illustrated in Figure 14.6. For such a system,
the differential equation to be solved for the displacement x as a function of
the time ¢ is

—+x=0. (14.19a)
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FIGURE 14.6
Free vibrations of a spring mass system with no damping.

The variables here have been made dimensionless by using the frequency
parameter ® to make the time dimensionless, where w? = k/m, and the dis-
placement x can be made dimensionless by using the initial displacement
from the rest position. If we now introduce the velocity v = dx/dt, we get

The last equation follows from the original differential equation. If we now
divide these two first-order differential equations, the time f is eliminated,
giving

do

dx

X
.

That is, the original second-order equation has been reduced to two first-
order equations, which are

do__x (14.19b)
dx v
% -0 (14.19¢)

If we integrate Equation 14.19b and obtain an expression for v, we can then
substitute the result into Equation 14.19c and integrate again to get x. The
solution to Equation 14.19b is called the phase plane solution, that is, the plot
of v versus x is known as the phase plane.
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The reason that we were able to reduce the problem to two first-order dif-
ferential equations this way is that the independent variable t did not appear
explicitly in the given differential equation—only as a derivative. If a more
general version of Equation 14.19a is to be solved, the procedure would be the
same. For example, suppose that the given differential equation is nonlinear
and of the following form:

d’x dx " .
? + A E + [33( = O (1420)

Then, if we take the velocity v as the new dependent variable and x as
the new independent variable as before, we get the following two first-order
equations:

% =-av" - px"
dx
—=u.
dt

Dividing these two equations eliminates t as before, giving the result

do  oo" +px"

dx v
Another way of expressing the condition that made the method work is to
say that the given differential equation is invariant to the following coordi-
nate transformation:
X=x
T =t + constant.
In terms of the new variables X and T, Equation 14.20 becomes

d2x ax "
+o ——
dr? dT

+BX" = 0.

Another way of looking at what happened above is to note that we found
an invariant coordinate, which in this case was x. Then, by taking the invari-
ant coordinate x to be the new independent variable, and taking v = dx/df to
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be the new dependent variable, the second-order equation for x(f) became a
tirst-order equation for v(x).

This method would also work if you found any invariant coordinate to
use as a new independent variable. If (x, t) is a coordinate that arises from
invariance of the differential equation, then we should choose ¢ to be the
new independent variable and G(£) = t" dx/dt to be the new dependent vari-
able. Including the coefficient " in the definition of the new dependent vari-
able G(§) is a generalization similar to what we used to get the Falkner—Skan
similarity solutions by generalizing the method used to obtain the Blasius
solution for flow past a flat surface. The inclusion of the factor #” gives us a
free parameter 1, which can be used to clear the transformed differential
equation of both x and ¢ to yield a differential equation for G(§).

To illustrate this situation, we consider the following differential equation
that is nonlinear and that contains the independent variable explicitly:

d*x .,
—+tx° =0. 14.21
dr (14.21)

This equation cannot be solved by taking v = dx/dt as the new dependent
variable and x as the new independent variable. If v = dx/dt, it follows from
the differential equation that

d
e
dt

an dr_ v

dat
Dividing these two equations gives

@ __w

dx v

Note that the variable t has not been eliminated because it appears explic-
itly in the differential equation. However, we can show that if we take G(§) =
tidx/dt as the new dependent variable, instead of v = dx/d¢t, and & = #3x as
the new independent variable, instead of x, we can reduce the second-order
differential equation to a first-order differential equation. The way we arrive
at these variable changes is as follows.

First, we must establish a coordinate that leaves the differential equation
unchanged. To do this, let

x=aXand t=bT
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Then, Equation 14.21 becomes

a d?X

p* dT?

+(a2b)TX2 —0.

If the differential equation is to be the same in both coordinate systems,
we must make

biz =a’b a= b%
That is,
x T
X £
So that
xt® = XT3

That is, we have established that the quantity xt* is an invariant coordi-
nate for Equation 14.21, so we will take this as our new independent variable.
Then, taking the equivalent of the “velocity” as our new dependent variable,
we are looking for a solution to Equation 14.21, which is of the following form:

GE)=t" % is our new dependent variable (14.22a)

and ¢ = 3x is our new independent variable. (14.22b)

Now, we substitute these expressions into Equation 14.21 and try to elimi-

nate both x and £ to yield an ordinary differential equation for the new vari-
able G(¢). From Equation 14.22a

dx

—=t"G
dt
d’>x d dG
.—=—t”G = -nt™ "G+ 7"
dr* (G) = -n AT
th_%dﬁ_i(t x)§= 3t2x t3dx dG

ar Tarde ars e a de
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From Equation 14.22a, we have dx/dt = +"G, and from Equation 14.22b, we
have the result x = t-3¢. Hence, our expression for dG/dt becomes

% = {3t‘1§ + t3‘”G} (;g.

Substituting this result into the equation for d?x/dt? above produces the result

dzx —-(n+1) -n -1 3-n dG
—=-nt G+t {3t E+t G}d—g

- G 4 30 9O joang 4G
dg dg

Substituting this expression into the given equation, Equation 14.21, we get

—nt~"IG 4 3 g j(; + t(3'2")GCC11§ +17E% =0.

Then, in order to eliminate ¢, we must equate the powers of ¢ in this equa-
tion. That is,

-n+1)=@3-2n)=-5.

These equations are satisfied by the value n = 4. The resulting ordinary
differential equation is

dG dG
4G +3E—+G—+E*=0
+ §d§+ dE+§

4G _4G-€ (14.23a)
dE G+3t

Once the solution for G is obtained, the solution for x will be given by the
solution to the following ordinary differential equation:

dv _ .
— =t"G. 14.23b
& ( )

That is, the original second-order ordinary differential equation, Equation
14.21, has been reduced to two first-order ordinary differential equations,
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Equations 14.23a and 14.23b. Many methods exist for solving first-order ordi-
nary differential equations, including simple numerical methods if required.

In summary, when solving nonlinear ordinary differential equations, use
the following methods—in order of priority:

1. Look for direct integration—for example
X"+ xx"=0= x"+ (x)?/2 = constant.

2. Look for a simple group invariance, that is, if the independent vari-
able does not appear explicitly, then choose v = dx/dt as the new
dependent variable and x as the new independent variable.

3. For less simple cases, look for an invariant coordinate, &(x,), and

choose G(£) = t"dx/dt as the new dependent variable and &(x,f) as the
new independent variable.

PROBLEMS

14.1 In Appendix D, Item D.2, a Fourier series representation is given
for a square wave that is odd in x. Use the methods discussed in
Section 14.1 to verify the expression that is quoted.

14.2 It is required to represent the shaded area in the figure for this
problem by a Fourier series. The upper limit of the shaded area is
a parabola that is defined by the following equation over the range
-L<x<L:

2

X
f(x) = H?

This gives f (0) = 0 and f (L) = H. Obtain the Fourier series for this
particular curve.

14.3 It is required to represent the shaded area in the figure for this
problem by a Fourier series. The upper limit of the shaded area is
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a parabola that is defined by the following equation over the range
-L<x<L:

x2
fx)=H

Here, f(0) = 0, but (L) = 0, which is different from Problem 14.2
above. Obtain the Fourier series for this particular curve.

Slx)

¥
a :
S

144 In Appendix D, Item D.2, a Fourier series representation is given for
a square wave that is odd in x. Due to the discontinuity in slope at
the two ends, x = 0 and x = L, this is one of the most difficult curves
to represent using a Fourier series. Use the given series and any
appropriate software, such as Excel, to plot the series representa-
tion using a truncated number of terms as follows:

H

(@ The first 10 terms corresponding ton =1 to n = 10
(b) The first 20 terms corresponding ton =1 ton =20
(o) The first 30 terms corresponding ton =1ton =30

Plot f(x)/H versus x/L and use a sufficient number of x/L values
to produce smooth curves in each case. You will notice oscillations
that persist and move toward the two corners as n increases. This
is known as Gibb’s phenomenon.

14.5 Use the series representation of each of the following complex
functions to identify the value of the residue at each singularity:

@ F@)- coiz
z

(b) F(2) = cos?z
z

© F@=

(z-a)
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14.6 Use the formula given by Equation 14.12b to determine the value
of the residue of each of the following functions at their singular

points:

@ F)=%
z

®) F(z)= O
z
eZ

© Fa= <

14.7 Use the formula given by Equation 14.12¢ to determine the value
of the residue of each of the following functions at their singular

points:
(@ Fz)="1"
sin z
z
(b) F (Z)=m
eZ
© F&= 2y

14.8 Considering o to be a real number, determine the residues of the
following function at each of its singular points:

40

F(Z)=m.

149 It is required to obtain solutions to the Laplace equation for two
situations as outlined below.

(@) The first situation has the boundary conditions indicated in the
figure for this problem, so that the problem to be solved for
T(xy) is as follows:

P17

2

YR =0
Y

T(x,0) = Asin%; T(x,H)=0; T(0,y)=0; T(Ly)=0.
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y

H Tx,H) =0

IS o
I I
= =
S =2
13 13

0  T(x0)=Asin(mx/L) L x

Obtain the solution in three simple steps as follows:

Step 1: Start by writing down the x component of the solution
that satisfies the boundary conditions on x = 0 and on x =
L and that has the correct functional form and the correct
amplitude along the surface y = 0.

Step 2: Next, write down the y component of the solution that
satisfies the boundary condition on y = H and that has the
value unity on the surface y = 0.

Step 3: Multiply these two components of the solution together
and verify that the resulting expression satisfies all of the
required conditions.

(b) The second situation has the boundary conditions as indicated
in the figure for this problem, so that the problem to be solved
becomes

T(x,0)=0; T(x,H)= Bsin%x; T(0,y)=0; T(L,y)=0.

H T(x,H) = Bsin(nx/L)

T(0,) =0
T(Ly) = 0

0 T(x,0) =0 L x
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Follow a similar procedure to the three steps outlined in Part (a).

Step 1: Start by writing down the x component of the solution
that satisfies the boundary conditions on x = 0 and on x =
L and that has the correct functional form and the correct
amplitude along the surface y = H.

Step 2: Next, write down the y component of the solution that sat-
isfies the boundary condition on y = 0 and that has the value
unity on the surface y = H.

Step 3: Multiply these two components of the solution together
and verify that the resulting expression satisfies all of the
required conditions.

14.10 It is required to obtain solutions to the Laplace equation for two
situations as outlined below.
(@) The first situation has the boundary conditions indicated in the
following figure, so that the problem to be solved for T(x,y) is as
follows:

T o°T
S
aw® oy’

T(x,0)=0; T(x,H)=0; T(o,y>=Csin%; T(L,y)=0.

y
I TH) =0
)
2 o
£ I
S =
I =)
— 39
2
e
I3
0 T(x0) =0 L X

Obtain the solution in three simple steps as follows:

Step 1: Start by writing down the y component of the solution
that satisfies the boundary conditions on y =0 and on y =
H and that has the correct functional form and the correct
amplitude along the surface x = 0.

Step 2: Next, write down the x component of the solution that
satisfies the boundary condition on x = L and that has the
value unity on the surface x = 0.
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Step 3: Multiply these two components of the solution together
and verify that the resulting expression satisfies all of the
required conditions.

(b) The second situation has the boundary conditions as indicated

in the figure for this problem, so that the problem to be solved
becomes

T(x,0)=0; T(x,H)=0; T(O,5)=0; T(L,y)=Dsin’.

y
I TwH) =0
T
o &
I £
3 2
g i
2
=
=
0 T(x,0) =0 L X

Follow a similar procedure to the three steps outlined in Part (a).

Step 1: Start by writing down the y component of the solution
that satisfies the boundary conditions on y = 0 and on y =
H and that has the correct functional form and the correct
amplitude along the surface x = L.

Step 2: Next, write down the x component of the solution that
satisfies the boundary condition on x = 0 and that has the
value unity on the surface x = L.

Step 3: Multiply these two components of the solution together
and verify that the resulting expression satisfies all of the
required conditions.

14.11 Using the results of Problems 14.9 and 14.10, write down the solu-
tion to the following problem:

T(x,0) = Asin%x
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T(x,H) = Bsin%x

. Ty
T(0,y) = Csin —Z
©0,y) sin -,

. Ty
T(L,y) = Dsin —~.
(L,y) T

14.12 Use separation of variables to obtain the solution for u(y,t)
in the range 0 < y < 2h and t > 0, which satisfies the following

conditions:
ou a’u
&2
a oy’
u(y,0) = Usin

h
u(y,t) > 0ast — oo

14.13 In Problem 9.2, it was indicated that the boundary layer equa-
tions for flow over a flat surface may be reduced to the following
second-order nonlinear ordinary differential equation after apply-
ing a similarity method of solution and a simple group invariance
method:

&eF 1 dr
&°F 1 dF
a2 F df

It was also stated in Problem 9.2, without proof, that a certain
transformation of coordinates reduces this second-order equation
to a first-order equation.

(@) Find a coordinate, call it &, that leaves the foregoing differential
equation invariant.

(b) Using group invariance ideas, define a new dependent variable
as follows:

dF
GE)=f"—.
©=1"y
Show that this transformation reduces the given second-order

equation to a first-order equation, and find the first-order equation
for G(€).
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Further Reading—Part V

The following references are representative of the extensive amount of mate-
rial that exists relating to the methods of analysis that are commonly used in
fluid mechanics problems.

Ablowitz, M. J. and A. S. Fokas: Complex Variables, Cambridge University Press,
Cambridge, UK, 2003.

Bender, C. M. and S. A. Orszag: Advanced Mathematical Methods for Scientists and
Engineers, Springer-Verlag, New York, 1999.

Bluman, G. W. and S. C. Anco: Symmetry and Integration Methods for Differential
Equations, Springer-Verlag, New York, 2002.

Cousteix, J. and J. Mauss: Asymptotic Analysis and Boundary Layers, Springer-Verlag,
Berlin, Heidelberg, 2007.

Hydon, P. E.: Symmetry Methods for Differential Equations, Cambridge University Press,
Cambridge, UK, 2000.

Schlichting, H., K. Gersten and K. Gersten: Boundary-Layer Theory, Springer-Verlag,
Berlin, Heidelberg, 2000.

Sobey, 1. J.: Introduction to Interactive Boundary Layer Theory, Oxford University Press,
Oxford, UK, 2000.

Van Dyke, M.: Perturbation Methods in Fluid Dynamics, Academic Press, New York,
1964.






Appendix A: Vector Analysis

The vector relations given in this appendix are particularly useful in the
study of fluid mechanics. The derivation of these relationships may be found
in most books that cover the topic of vector analysis.

A.1 Vector Identities

In the following formulas, ¢ is any scalar and a, b, and ¢ are any vectors:

x ¢=0

Ga)=¢ a+ar ¢

x(pa)= odxa+d( xa)
( xa)=0

(@~ Ja=5 (ara)-ax( xa)

x( xa)= ( -a)- ‘a

x(@xb)=a( ‘b)-b( -a)-(a- )b+(b: )a
‘(axb)=b-( xa)-a-( xDb).

A.2 Integral Theorems

In the following two theorems, which relate surface integrals to volume inte-
grals, V is any volume and S is the surface that encloses V, with the unit nor-
mal on S being denoted by n. ¢ is any scalar and a is any vector.

Gauss’ theorem (also known as the divergence theorem):

a-ndS= -adV
Jomas-],

543
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Green's theorem:

99 1o . 2
f¢and5—![ o o+ ZpldV

Stokes’ theorem:

9Sa-dl=f( xa)-ndA.

This theorem relates a line integral to an equivalent surface integral. The
surface A is arbitrary, but it must terminate on the line I.

A.3 Orthogonal Curvilinear Coordinates

Let xy, x,, x; be a set of orthogonal curvilinear coordinates with e, e,, e; as
the corresponding unit base vectors.

Position vector:
r=xe, + ye,+ ze,

where e,, e, and e, are fixed in space.

4

Base vectors:

o
ox;

ar
0x;

Metric-scale factors:

Line element:

(dr-dr) = k5 (dx,)? + K3 (dx,)* + h3(dx,).
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Cartesian coordinates (rectangular coordinates):

Cylindrical coordinates:
x =R X, =0 X3 =2

h =1 h, =R hy =1
x=Rcos® y=Rsin® z=z

Spherical coordinates:
X, =1 X, =0 X3 =W

h=1 hy=r hy =rsin®

x=rsinBcosw  y=rsinBsinw  z=rcosH.

Vector operations:

545

In the following, let ¢ be any scalar and let a = a,e; + a,e, + a;e; be any

vector.
1. Gradient:

_ 1o, Lo, 1

b o T, a2y o,
2. Divergence:
1 d d 0
= —(h,h —(hh —(hh .
Iy, axl( haa,) + axz( sa,) + 8x3( 1hhas)
3. Curl:
he, e, e,
1 d d d
xa= —

Chhy | ax, 0x, 9x;

ha, Iha, ha,
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4. Laplacian:

SN SN NV IV S U
mhyhy ox; hy dx;  dx, hy 9x,  9xy  hy dx,
» 1 9 1 d
= ——( - — —(a))-—h
2 hy ox, a)+ hyh,  dx, h 8x3( ) ( 43)
9 hy 9
-— —(h h
ox, hh, axl( K0p) = (may) €
1 9 a  hy
=% . 2 i _ h
hon ' i ax iy ox 1( 242) ( 1)
o b 0 F
-— —(hya;) - ——(h
ox, hh, ox, (hya3) ox, (hya,) €
1 9 d hy
=9 0. v h
I, 8x3( a)+ h, ox, h I, ox 2( 303) = ( 20;)
a h
- h h .
ox, hhy @ ( 187) = ( 303) €
5. Lagrangian derivative
1 aa, aa, da; 4,
: =— — — — = h h
(@ )a I, 0 ox, 25 1 3 o, I, ( L0;) = ( o)
a 1 da da da
+i o, (hal) (h ;) e1+g aliﬂzzi a3£
a
—i (h az) = (h a,) + ]71 Tl(h 20,) = (h a) e
0a, 0a, 0a, a
— — — — - h h
+ I a ox, +a, ox,  Pax, b, ox 3( 1) = ( 3003)
a d ad
+h72 —(hyay) = —(ay) e,
, 0%, X3

Appendix A
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Figure A.1 shows the geometric relationship between Cartesian coordi-
nates and the two most common curvilinear coordinate systems.

(@) (b)

(R,6,z) (n,8,0)
0

FIGURE A.1
Relationship between Cartesian coordinates and (a) cylindrical coordinates and (b) spherical
coordinates.






Appendix B: Tensors

In this appendix, some of the basic properties of tensors are reviewed.
Although much of the material is general, the discussion is restricted to
Cartesian tensors since curvilinear tensors are not used in this book.

B.1 Notation and Definition
B.1.1 Notation

The following rules of notation will be followed throughout:

1. If a given index appears only once in each term of a tensor equation,
itis a free index and the equation holds for all possible values of that
index.

2. If an index appears twice in any given term, it is understood that a
summation is to be made over all possible values of that index.

3. No index may appear more than twice in any term.

B.1.2 Definition

A tensor of rank r is a quantity having n" components in n-dimensional space.
The components of a tensor quantity expressed in two different coordinate
systems are related as follows:

z/ka = Cisttcku“'Cmsttumv
where the quantities C,,, are the direction cosines between the axes of the
two coordinate systems.

A tensor of rank 2 is sometimes called a dyadic, a tensor of rank 1 is a vector,
and a tensor of rank 0 is a scalar.

549
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B.2 Tensor Algebra
B.2.1 Addition
Two tensors of equal rank may be added to yield a third tensor of the same

rank as follows:

C,']'...k = A,‘j...k + B,']'“.k-

B.2.2 Multiplication

If tensor A has rank 2 and tensor B has rank b, the multiplication of these two
tensors yields a third one of rank c.
C

A, B,..

kst = Lijok
B.2.3 Contraction

If any two indices of a tensor of rank 7 > 2 are set equal, a tensor of rank r -2
is obtained. For example, if C; is defined by

C,=AB,

then by setting i = j, the tensor C;, which is of rank 2, becomes a tensor of

rank O (i.e., a scalar):

ij/

Ci=AB,.

Thus, contraction is equivalent to taking the scalar product of two vectors
in vector algebra.

B.2.4 Symmetry
If the tensor A has the property that

Ai-uj---kn-l = Ai-~-k--~j---l

then the tensor A is said to be symmetric in the indices j and k. As a con-
sequence of the relation above, the tensor has only n(n + 1)/2 independent
components.

If the tensor A has the property that

Ai...j-.-k.-.l = _Ai~-~k-~-j---l
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then the tensor A is said to be antisymmetric in the indices j and k. Such ten-
sors have only n(n — 1)/2 independent components.

L]
B.3 Tensor Operations

B.3.1 Gradient

The gradient of a tensor of rank r is defined by

OR;.

Tig=—""
iji--kl
0x,

and yields a tensor of rank (v + 1).

B.3.2 Divergence

The divergence of a tensor of rank r results in a tensor of rank (r - 1):

T _ aRimjklmm

B.3.3 Curl

If R is a tensor of rank 7, the curl operation will produce an antisymmetric
tensor of rank (r + 1). In general, the operation is defined by

aRi,.,]z..k _ (")R,-N.[mk
ox, oy

Ti---]‘---kl =

In three dimensions, the curl of a tensor of rank 1 (i.e., a vector) may be
written in the form

R,
T, = —Eijk a

where ¢, is a constant pseudoscalar defined by

ij
€1p3 = €319 = Ep31 = 1
€13 = €391 = €13 = —1

g = 0 otherwise.
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B.4 Isotropic Tensors
B.4.1 Definition

An isotropic tensor is one whose components are invariant with respect to
all possible rotations of the coordinate system. That is, there are no preferred
directions, and the quantity represented by the tensor is a function of posi-
tion only and not of orientation.

B.4.2 Isotropic Tensors of Rank 0

All tensors of rank 0 (i.e., scalars) are isotropic.

B.4.3 Isotropic Tensors of Rank 1

There are no isotropic tensors of rank 1. That is, vectors are not isotropic
since they have preferred directions.

B.4.4 Isotropic Tensors of Rank 2

The only isotropic tensors of rank 2 are of the form ad;, where a is a scalar

and §;; is the Kronecker delta, which has the property that

0 wheni=j

1 wheni=j.

B.4.5 Isotropic Tensors of Rank 3

The isotropic tensors of rank 3 are of the form ae;;, where a is a scalar and ¢
is a pseudoscalar defined under Section B.3.

B.4.6 Isotropic Tensors of Rank 4

The most general isotropic tensor of rank 4 is of the form

8,8, + B(8;,8;, + 8;,8;) + v(8;,8;, — 5;,8,)

where a, B, and y are scalars.
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B.5 Integral Theorems

The following two theorems were given in vector form in Appendix A, and
they are reproduced here in tensor form.

B.5.1 Gauss’ Theorem (Divergence Theorem)

fain,» as= [ %qy.

v 0x;

B.5.2 Stokes’ Theorem

0a;
?ai dl. =f —gy —n dA.
A X,






Appendix C: Governing Equations

This appendix gives the continuity, Navier-Stokes, and energy equations
together with the components of stress in the three most commonly used
coordinate systems: Cartesian, cylindrical, and spherical coordinates. The
equations are valid for calorically perfect incompressible, Newtonian fluids

in which p and k are constants.

C.1 Cartesian Coordinates

Coordinates r = (x, , 2)

Velocity u = (4, v, w)

o o _

+—+ =
ox dy 0z
ou ou ou ou ap
p—+U—+V —+W— =-—+
ot 0x ay 0z ox
Jv v dv v ap
P —+U—+0 —+W——— =———
ot ox ay 0z ay
ow ow ow ow ap
p —+U—+V——+W— =-——+
ot ox ay 0z 0z

or 0T T

pC

=k T+2 —
ax

2
ou dv
o o
day ox

» UV +W
ot ox ay

2
u

o
0z

U+ pf,
20+ of,

w + pf,

2
Jv Jw
— +

ay 0z

2
ou ow
i
Jdz ox

2
v Jw
i

dz  dy
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where
, 9 9F 9
=2t ot o
ox*  Jdy° oz
’ ou ou dJu
Ty = a Txy=ryx= @"'87
5 du Jdv  Jdw
Tyy = @ Tz =Ty = g*’@
T, =2 dw Ty =T, = ow + ou
= 0z oo ox 9z
——

C.2 Cylindrical Coordinates
Coordinates r = (R, 6, 2)
Velocity u = (ug, g, 1,)

1 18u9 ou
2 =0
R R ( ) + R a0 oz

0 auR ‘g Jug Uy duig —u—‘2’+u2 ity
R R 96 R 0z

ap 2 Ug 2 duy

= - + Up ——— — +
oR R R2 RZ 90 pr

d d d d
o Dy By Uy Oy ity Oy
ot R R 00 R 0z

1 ap 2 Uy 2 dug

=—— 4 Uy — —0 +
R 90 TRt R 0p TP

ou, ou, Uy ou, ou, ap )
+u +——=+u =-—+ U, +
P oot TMRGR TR 90 T oz 0z =+ 0




Appendix C 557

oT T uy oT T

C, —+ug—+ +uU, —
P ot TPROR TR a0 T a2
g 1 0 o
-k 2T+2 R, ~ ﬁ+u,{ +
oR R 06 0z
2 2 2
ouy 1 ou, du, Jdug 1 dug A

+ + + ——+
dz R 00 oR 0z R 96 R R

where
, 19 9 1 9> 9
oo Non Yoo T
R oR oR R* 06 0z
ou 0 u 1 ou
T =2 —= Tpe =Ter = R— % +——X
KR aR RO R R R R 99
1 duy,  ug oy 1 du,
T = — 9 4 K Tg, =T,p= —0+——=%
% R a0 R 0= 9z R 08
u ou u
T,=2 —° TR =Tg, = 2 —R
zz aZ zR Rz aR aZ
|

C.3 Spherical Coordinates
Coordinates r = (7, 6, ®)

Velocity u = (u,, gy, u,)

1
—i(rzur)+

r?or

[

rsin® ow

d
—(u_sin 0) +
rsineae( o )



558 Appendix C

ou, ou, Uy du,  u, ou,  up +u;
Pt T T 00 T rsind oo r
d 2 20 2 2 9
--F 2%‘7 r—*zﬂ—*z o cot®—— o +pf,
ar r r> d0 r°sinf Jdw
o Ol Oy Uy Gty U, Gy Uy Uy o
ot ar r 90 rsinB dw r r
19 2 Ju U 2 cot0 Ju,
=_77p+ 2 0 2 - - 2 .82 _72 . +pf6
r 00 r® 0w r°sin“0O r° sinB® Jdw
0 d 0 d
0 U, . Uy +E U, + u.u) U, U, Uy, + Uglh,, cotO
ot ar r 90 rsinB dw r r

1o, L 2

r

. uw
rsin® dw r2sin® dw

2cos0 du, U

+ =5 +pf,
r2sin?0 do  r*sin’ 0

aT oT  u, oT u, 0T
pCp — U, —F——+ ————
ot ar r 00 rsinB dw

r

ar r 09 r

2
=k *T+2 oy + laﬂ+u—

u, U
. 4+ coth
rsin® dw r r

1 ou,

(0]

2

2
1 ou, d U 1 ou, Ja u,
— +r— — 4+ - +r— =2
r 00 or r rsin® Jw o r

1 %+sinei u,
rsin® ow r 00 sinB




Appendix C
where
2_ 19
r2 or
T(’iw

, 0 1 a9 . 9 1 02
ot oy St
ar r“sin® 00 a0 r°sin” 0 dw

d
VV=2 ur
ar
1ouy, u,
—_— 20U M
o0 r 00 r
B 1 ou, u, u,cotd
rsin® Jw r r

Ty=T ri Yo +lau’

oo ar r r 90
_sin® 3 u, 1 duy

o0 r 00 sin® rsin® Jw

. - ri U, 1 ou,
e ar r rsin® dw

559






Appendix D: Fourier Series

This appendix lists some of the more commonly used Fourier series repre-
sentations that are particularly useful in the study of fluid mechanics. In
each case, the function f (x) is periodic over the distance 2L and may be used
to represent the shaded area of the corresponding diagram.

1. Square wave (even)

2. Square wave (odd)

) = Hinzn{l—(—l)”}siniT
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3. Triangular wave (even)

f(x)=H2 4 {1-(-1)"}cos%

n2m?

S)

H

o
ViV

4. Triangular wave (odd)

\ 8 I nmx
x)=H E in— sin——
f(x) 2 sin > sin L

n2m?

Slx)

HA

5. Sawtooth wave (forward): f (x) = x H/L

()= Hi 2 psin
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6. Sawtooth wave (backward): f (x) = H (1- x/L)

o 2 nmx
x)=H Y —sin——
) Zm .

flx)

H

‘ x
S

7. Parabola (even): f (x) = H{1- (x/L)?}

flx)= gH—HZ 4 (-1)" cosn—;jx

n2m?

8. Parabola (odd): f (x) = 4H(x/L){1 - (]x|/L)}

-85,

niwx

{1-(-1)”}5111T

16
n’n’

Sx)







Appendix E: Thermodynamics

Thermodynamics is a complete subject in itself. However, in the study of
fluid mechanics, only a few fundamental relationships from thermodynam-
ics are needed, and these arise mainly in the study of compressible flow. The
following summary contains all the results from thermodynamics that are
required in this book.

E.1 Zeroth Law

The zeroth law of thermodynamics states that there exists a variable of state,
the temperature T, and that two systems that are in thermal contact are in
equilibrium only if their temperatures are equal.

E.2 First Law

The first law of thermodynamics states that there exists a variable of state,
the internal energy e. If an amount of work 8w is done on a thermodynamic
system and an amount of heat 64 is added to it, the equilibrium states before
and after the process are related by

de = dw + 4.

That is, the change in the internal energy equals the work done on the sys-
tem plus the heat added to it during the process or event.

E.3 Equations of State

There are two commonly used thermal equations of state, one being for ideal
gases and the other for real gases.
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E.3.1 Thermally Perfect Gas

The equation of state for a thermally perfect gas is
p=pRT

where R is the gas constant for that particular gas. Frequently, this equation
is used to define a perfect gas. That is, gases that obey the above equation of
state are defined as perfect gases.

E.3.2 Van der Waals Equation

An approximate equation of state for real gases is given by the Van der Waals
equation, which is

p=pRT - -
1-Bp RT
where
¢ _Zpr
B8
and
o
ﬁ—2=27pc.

Here, p. and T, are, respectively, the critical pressure and temperature of
the gas.

E.4 Enthalpy
The enthalpy / of a gas is defined by the following:

P

h=e+==e+pv

where v is the volume of the gas per unit mass.
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E.5 Specific Heats

There are two specific heats in common usage, that at constant volume and
that at constant pressure.

E.5.1 Constant Volume

The specific heat at constant volume C, is defined as follows:

c oo 1 _oe_on on o
dT =~ oT dT  9p aT |

Fromthe definingidentity, the otherrelations follow withoutapproximation.

E.5.2 Constant Pressure

The specific heat at constant pressure C, is defined by

dg _de se ok

C = = + + .
r dTp oT avp an oT

E.5.3 Perfect Gas

Using the above definitions of the specific heats and considering a perfect
gas, it follows that

C,-C,=R.

Under these circumstances, it can be shown that e and & are functions of
the temperature T only and may be expressed in the form

e(T) = f C,dT + constant

WT) = f C,dT + constant.

If C, and C, are constants, independent of T, the gas is called calorically
perfect, and it follows that

e =C,T + constant

h= CpT + constant.
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I
E.6 Adiabatic, Reversible Processes
The following relations are valid for adiabatic, reversible processes:

1/(v-1)

h_ L

Po Ty

P 1 Y/(Y—l)_ o Y
Po T, Po

where y = C,/C, and p,, T, p, are constants.

E.7 Entropy

There exists a variable of state, the entropy s. If heat is added to a system, the
change in entropy between the initial and final equilibrium states will be
given by

Pd
A

where the integral is evaluated for a reversible process.

E.8 Second Law

The second law of thermodynamics states that for any spontaneous process,
the entropy change is positive or zero. That is,

B

d
Sp— S, zfA?q.

For a calorically perfect gas, it follows that

p

T
§=5,=C, logT—O—Rlog%

T P
=C,log— +Rlog—>.
Ty p
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E.9 Canonical Equations of State

The heat-addition term in the first law may be eliminated in favor of the

entropy, yielding an equation that involves variables of state only. From this,
the following identities may be established:

de de

- =T =
ds ) Jv . P
o =T o =0.
ds ap

E.10 Reciprocity Relations

By considering s to be a function of p and T, the following reciprocity rela-
tions follow:

ds 1 oh s 1 oh

= -0
or T oT ap T dp

From these reciprocity relations, the following equation is obtained, which
relates the caloric and thermal equations of state:

oh v
—=0-T—.
ap oT

Similarly, by considering s to be a function of v and T, the following reci-
procity relations are obtained:

ds 1 oge Jds 1 ae+
oT T T w T o' F

From these, the relation between the caloric and thermal equations of state
is found to be

)

v P oT
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